SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

Form 10-Q

(Mark One)

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended June 30, 2003

or

o TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d)
OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

Commission file number 1-12154

Waste Management, Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

73-1309529 (I.R.S. Employer Identification No.)

1001 Fannin

Suite 4000 Houston, Texas 77002 (Address of principal executive offices)

(713) 512-6200

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No o

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Securities Exchange Act of 1934.) Yes 🗵 No o

The number of shares of Common Stock, \$0.01 par value, of the registrant outstanding at August 1, 2003 was 592,213,162 (excluding treasury shares of 38,069,299).

TABLE OF CONTENTS

PART I

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (In Millions, Except Per Share Amounts) (Unaudited)

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In Millions) (Unaudited)

WASTE MANAGEMENT, INC. CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (In Millions, Except Shares in Thousands) (Unaudited)

WASTE MANAGEMENT, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

PART II

SIGNATURES

INDEX TO EXHIBITS

Employment Agreement - Steve Ragiel

Ten-Year Letter of Credit & Term Loan Agreement

Five-Year Letter of Credit & Term Loan Agreement

Seven-Year Letter of Credit & Term Loan Agreement

Directors' Deferred Compensation Plan

Computation of Ratio of Earnings to Fixed Charges

Certification of CEO Pursuant to Rule 15d-14(a)

Certification of CFO Pursuant to Rule 15d-14(a)

Certification of CEO Pursuant to Section 906

Certification of CFO Pursuant to Section 906

PART I

Item 1. **Financial Statements**

WASTE MANAGEMENT, INC.

CONDENSED CONSOLIDATED BALANCE SHEETS

(In Millions, Except Share and Par Value Amounts)

ASSETS

	June 30, 2003	December 31, 2002
	(Unaudited)	
Current assets:		
Cash and cash equivalents	\$ 753	\$ 264
Accounts receivable, net of allowance for doubtful accounts of \$60	1,447	1,379
Notes and other receivables	231	265
Parts and supplies	77	80
Deferred income taxes	562	551
Prepaid expenses and other assets	116	161
Total current assets	3,186	2,700
Property and equipment, net of accumulated depreciation and amortization		
of \$9,116 and \$8,498, respectively	10,812	10,612
Goodwill	5,212	5,079
Other intangible assets, net	119	105
Other assets	1,127	1,135
Total assets	\$20,456	\$19,631

LIABILITIES AND STOC	CKHOLDERS' EQUITY	
Current liabilities:		
Accounts payable	\$ 552	\$ 526
Accrued liabilities	2,060	2,002
Deferred revenues	415	414
Current portion of long-term debt	108	231
Total current liabilities	3,135	3,173
Long-term debt, less current portion	8,350	8,062
Deferred income taxes	1,625	1,548

Landfill and environmental remediation liabilities	1,146	884
Other liabilities	569	637
Total liabilities	14,825	14,304
Minority interest in subsidiaries	47	19
Commitments and contingencies		
Stockholders' equity:		
Common stock, \$0.01 par value; 1,500,000,000 shares authorized;		
630,282,461 shares issued	6	6
Additional paid-in capital	4,510	4,513
Retained earnings	2,110	1,873
Accumulated other comprehensive loss	(83)	(179)
Treasury stock at cost, 38,524,621 and 35,682,000 shares, respectively	(959)	(905)
Total stockholders' equity	5,584	5,308
Total liabilities and stockholders' equity	\$20,456	\$19,631
	_	

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(In Millions, Except Per Share Amounts) (Unaudited)

	Three Months Ended June 30,		Six Months Ended June 30,	
	2003	2002	2003	2002
Operating revenues	\$2,915	\$2,825	\$5,631	\$5,434
Costs and expenses:				
Operating (exclusive of depreciation and amortization shown below)	1,887	1,732	3,680	3,297
Selling, general and administrative	301	316	624	703
Depreciation and amortization	325	313	627	607
Restructuring	23	_	43	37
Asset impairments and unusual items	(6)	_	(7)	(6)
	2 520	2 261	4.067	4 620
	2,530	2,361	4,967	4,638
Income from operations	385	464	664	796
meome from operations		——		
Other income (expense):				
Interest expense	(110)	(116)	(219)	(234)
Interest income	4	5	6	9
Minority interest	(2)	(2)	(3)	(3)
Other income, net	5	1	9	3
	(103)	(112)	(207)	(225)
Income before income taxes	282	352	457	571
Provision for income taxes	106	135	174	218
Income before cumulative effect of changes in accounting principles	176	217	283	353
Cumulative effect of changes in accounting principles Cumulative effect of changes in accounting principles, net of income tax benefit	170	217	203	333
of \$31 and \$0 for the six months ended June 30, 2003 and 2002, respectively	_	_	(46)	2
of the time of the six monais chaca same so, 2000 and 2002, respectively			——————————————————————————————————————	
Net income	\$ 176	\$ 217	\$ 237	\$ 355
	_	_	_	_
Basic earnings per common share:				
Income before cumulative effect of changes in accounting principles	\$ 0.30	\$ 0.35	\$ 0.48	\$ 0.57
Cumulative effect of changes in accounting principles	_	· —	(80.0)	_
ŭ .			<u> </u>	
Net income	\$ 0.30	\$ 0.35	\$ 0.40	\$ 0.57
	_	_	_	_
Diluted earnings per common share:				
Income before cumulative effect of changes in accounting principles	\$ 0.30	\$ 0.35	\$ 0.48	\$ 0.57
Cumulative effect of changes in accounting principles	_	_	(80.0)	_
Net income	\$ 0.30	\$ 0.35	\$ 0.40	\$ 0.57
	_	_	_	_
Pro forma income and earnings per common share assuming changes in accounting principles described in Note 1 are applied retroactively:				
Income before cumulative effect of changes in accounting principles	\$ 176	\$ 210	\$ 283	329
Basic earnings per common share before cumulative effect of changes in				
accounting principles	\$ 0.30	\$ 0.34	\$ 0.48	0.53
Diluted earnings per common share before cumulative effect of changes				
in accounting principles	\$ 0.30	\$ 0.34	\$ 0.48	0.53

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(In Millions) (Unaudited)

Six Months Ended June 30,

	June	30,
	2003	2002
Cash flows from operating activities:		
Net income	\$ 237	\$ 355
Adjustments to reconcile net income to net cash provided		
by operating activities:		
Cumulative effect of changes in accounting principles	46	(2)
Provision for bad debts	24	17
Depreciation and amortization	627	607
Deferred income tax provision	69	110
Minority interest	3	3
Net gain on disposal of assets	(5)	(3)
Effect of asset impairments and unusual items	(7)	(6)
Change in operating assets and liabilities, net of effects of		
acquisitions and divestitures:		
Receivables	(24)	(39)
Prepaid expenses and other current assets	17	(7)
Other assets	100	(29)
Accounts payable and accrued liabilities	(37)	(152)
Deferred revenues and other liabilities	23	30
Net cash provided by operating activities	1,073	884
	<u> </u>	
Cash flows from investing activities:		
Acquisitions of businesses, net of cash acquired	(172)	(64)
Capital expenditures	(501)	(552)
Proceeds from divestitures of businesses, net of cash divested,	()	()
and other sales of assets	34	71
Net receipts from restricted funds	109	103
Other	18	(7)
Net cash used in investing activities	(512)	(449)
ver cash asea in investing acaviacs		(1.5)
Cash flows from financing activities:		
New borrowings	67	498
Debt repayments	(68)	(468)
Common stock repurchases	(71)	(500)
Exercise of common stock options and warrants	9	23
Other	(9)	
Other		
Net cash used in financing activities	(72)	(447)
Net cash used in midneing activities	(72)	(447)
offers of such as a such a house on such and such assistant		
Effect of exchange rate changes on cash and cash equivalents	_	2
ncrease (decrease) in cash and cash equivalents	489	(10)
Cash and cash equivalents at beginning of period	264	730
Cash and cash equivalents at end of period	\$ 753	\$ 720

CONDENSED CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY

(In Millions, Except Shares in Thousands) (Unaudited)

	Common Stock		Common Stock Additional Paid-in Retained		Accumulated Other Comprehensive	Treasury Stock	
	Shares	Amount	Capital	Earnings	Loss	Shares	Amount
Balance, December 31, 2002	630,282	\$ 6	\$4,513	\$1,873	\$ (179)	(35,682)	\$(905)
Net income	_	_	_	237	_	_	_
Common stock issued upon exercise of stock options and warrants and grants of restricted stock, net of tax benefit of							
\$1	_	_	(2)	_	_	542	12
Common stock repurchases, net of settlements	_	_	_	_	_	(3,563)	(71)
Unrealized loss resulting from changes in fair values of derivative instruments, net of tax benefit of \$2	_	_	_	_	(3)	_	_
Minimum pension liability adjustment, net of taxes of \$1	_	_	_	_	1	_	_
Translation adjustment of foreign currency					98		
statements Other		_	(1)	_	90	178	5
Oulei		_					
Balance, June 30, 2003	630,282	\$ 6	\$4,510	\$2,110	\$ (83)	(38,525)	\$(959)

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

The condensed financial statements presented herein represent the consolidation of Waste Management, Inc., a Delaware corporation, and its subsidiaries. Waste Management, Inc. is a holding company that conducts all of its operations through its subsidiaries. When the terms the "Company," "we," "us" or "our" are used in this document, those terms refer to Waste Management, Inc. and all of its consolidated subsidiaries. When we use the term "WMI," we are referring only to the parent holding company, and are not including any of the subsidiaries.

The condensed consolidated financial statements as of and for the three and six month periods ended June 30, 2003 are unaudited. In the opinion of management, these financial statements include all adjustments, which, except as described elsewhere herein, are of a normal recurring nature, necessary for a fair presentation of the financial position, results of operations, and cash flows for the periods presented. The results for interim periods are not necessarily indicative of results for the entire year. The financial statements presented herein should be read in connection with the financial statements included in our Annual Report on Form 10-K for the year ended December 31, 2002.

The preparation of financial statements in conformity with accounting principles generally accepted in the United States requires our management to make certain estimates and assumptions. These estimates and assumptions affect the reported amounts of assets and liabilities and disclosure of the contingent assets and liabilities at the date of the financial statements. These estimates and assumptions will also affect the reported amounts of certain revenues and expenses during the reporting period. Actual results could differ materially based on any changes in the estimates and assumptions that we use in the preparation of our financial statements. Additionally, the estimates and assumptions used in determining landfill airspace amortization rates per ton, final capping, closure and post-closure liabilities as well as environmental remediation liabilities require significant operations, engineering and accounting input. We review these estimates and assumptions no less than annually. In many circumstances, the ultimate outcome of these estimates and assumptions may not be known for decades into the future. Actual results could differ materially from these estimates and assumptions due to changes in environmental related regulations or future operational plans, and the inherent imprecision associated with estimating matters so far into the future. See the "Management's Discussion and Analysis" section of this report for more information regarding our critical accounting estimates and assumptions.

In the first quarter of 2003, we adopted certain changes in accounting principles that impact the comparability of the financial information presented herein. See Note 1 for information regarding these changes.

Reclassifications — On January 1, 2003, we adopted Statement of Financial Accounting Standards ("SFAS") No. 145, Rescission of FASB Statements No. 4, 44, and 64, Amendment of FASB Statement No. 13, and Technical Corrections. SFAS No. 145 requires that gains and losses from extinguishment of debt be classified as extraordinary items only if they meet the criteria in Accounting Principles Board Opinion No. 30. Upon the adoption of SFAS No. 145, we reclassified first quarter 2002 losses on the extinguishment of debt of \$1 million that had previously been reported as extraordinary. The impact of this reclassification on the results of operations for the six months ended June 30, 2002 was a \$2 million increase in interest expense and a \$1 million decrease in the tax provision. The adoption of SFAS No. 145 did not impact the presentation of the results of operations for the three months ended June 30, 2002.

In March 2002, we adopted a new organizational structure to align collection, transport, recycling and disposal resources into market areas, and reduced the number of layers of management and consolidated certain administrative and support functions. During our 2003 planning processes we determined that certain costs we previously reported as selling, general and administrative expenses were more appropriately classified as operating expenses after the adoption of the new organizational structure. These costs include certain maintenance and repairs, property taxes, insurance and claims, rent, utilities, permits and fees. We began

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

reporting these costs as operating expenses in the first quarter of 2003. In order to conform the prior period presentation of these costs with the current period presentation, we have reclassified \$43 million of costs previously reported as selling, general and administrative expenses for the three and six months ended June 30, 2002 to operating expenses in the accompanying condensed consolidated statement of operations. Costs identified for reclassification are those costs incurred beginning in the second quarter of 2002, as this was the first full accounting period that these organizational changes were effective. Similar reclassifications will be made to the statement of operations for the third and fourth quarters of 2002. Additional disclosures associated with this restructuring are included in Note 9.

Certain reclassifications have also been made in the prior period condensed consolidated statement of cash flows in order to conform to the current period presentation.

1. 2003 Accounting Changes

On January 1, 2003, we implemented changes in our methods of accounting for major repairs and maintenance, loss contracts and asset retirement obligations, as described below. If the accounting changes we implemented on January 1, 2003 had been effective in 2002, the impact on income before cumulative effect of changes in accounting principles and earnings per common share for the three and six months ended June 30, 2002 (in millions, except per share amounts) would have been as follows:

	Three Months Ended June 30, 2002	Six Months Ended June 30, 2002
Reported income before cumulative effect of changes in accounting principles	\$ 217	\$ 353
Repairs and maintenance, net of tax	1	1
Loss contracts, net of tax	(1)	(10)
Adoption of SFAS No. 143, net of tax	(7)	(15)
Pro forma income	\$ 210	\$ 329
		_
Basic earnings per common share:		
Reported income before cumulative effect of changes in accounting principles	\$ 0.35	\$ 0.57
Repairs and maintenance, net of tax	_	_
Loss contracts, net of tax	_	(0.02)
Adoption of SFAS No. 143, net of tax	(0.01)	(0.02)
Pro forma income	\$ 0.34	\$ 0.53
		_
Diluted earnings per common share:		
Reported income before cumulative effect of changes in accounting principles	\$ 0.35	\$ 0.57
Repairs and maintenance, net of tax	<u> </u>	<u> </u>
Loss contracts, net of tax	_	(0.02)
Adoption of SFAS No. 143, net of tax	(0.01)	(0.02)
Pro forma income	\$ 0.34	\$ 0.53

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Repairs and Maintenance

Through December 31, 2002, we accrued in advance for major repairs and maintenance expenditures and we deferred costs associated with annual plant outages at our waste-to-energy facilities and independent power production plants. Effective January 1, 2003, we changed our policy from this method to one that expenses these costs as they are incurred. In the first quarter of 2003, we recorded approximately \$25 million, net of taxes, or \$0.04 per diluted share, as a credit to cumulative effect of changes in accounting principles. We believe our current method of accounting is preferable because it (i) provides operating results that more clearly reflect the timing and amount of required expenditures, (ii) more clearly reflects our assets and liabilities, and (iii) reduces the need to make additional estimates and assumptions. The effect of this change in accounting principle is not material to our results of operations for the three and six months ended June 30, 2003.

Loss Contracts

Through December 31, 2002, if our customer contracts were projected to have direct costs greater than revenues over the life of the contract, we accrued for those future losses. Effective January 1, 2003, we changed our policy from this method to one that expenses these losses as they are incurred. In the first quarter of 2003, we recorded approximately \$30 million, net of taxes, or \$0.05 per diluted share, as a credit to cumulative effect of changes in accounting principles. We believe our current method of accounting is preferable because it (i) provides operating results that more clearly reflect the timing and amount of contract losses generated, (ii) more clearly reflects our liabilities, and (iii) reduces the need to make additional estimates and assumptions. The effect of this change in accounting principle is not material to our results of operations for the three and six months ended June 30, 2003.

Adoption of SFAS No. 143 — Accounting for Asset Retirement Obligations

In connection with the adoption of SFAS No. 143, *Accounting for Asset Retirement Obligations*, we recorded approximately \$101 million, including tax benefit, or \$0.17 per diluted share, in the first quarter of 2003 as a charge to cumulative effect of changes in accounting principles. Substantially all of this charge was related to changes in accounting for landfill final capping, closure and post-closure costs. The application of SFAS No. 143 reduced income before cumulative effect of changes in accounting principles for the three months ended June 30, 2003 by approximately \$8 million, net of tax benefit, or \$0.01 per diluted share, and reduced income before cumulative effect of changes in accounting principles for the six months ended June 30, 2003 by approximately \$16 million, net of tax benefit, or \$0.03 per diluted share.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Effective January 1, 2003, our method of accounting for landfill closure and post-closure, as well as landfill final capping, changed as a result of our adoption of SFAS No. 143. SFAS No. 143 does not change the basic landfill accounting that we and others in the waste industry have followed historically. Through December 31, 2002, the waste industry generally recognized expenses associated with (i) amortization of capitalized and future landfill asset costs and (ii) future closure and post-closure obligations on a units-of-consumption basis as airspace was consumed over the life of the related landfill. This practice, referred to as life-cycle accounting within the waste industry, continues to be followed. The table below compares our historical practices to the method prescribed by SFAS No. 143.

Description	Historical Practice	Current Practice (Effective January 1, 2003)		
Definitions:				
Final Capping	Capital asset related to installation of flexible membrane and geosynthetic clay liners, drainage and compacted soil layers and topsoil constructed over areas of landfill where total airspace capacity has been consumed	Reflected as an asset retirement obligation, on a discounted basis, rather than a capital asset		
Closure	Includes last final capping event, final portion of methane gas collection system to be constructed, demobilization, and the routine maintenance costs incurred after site ceases to accept waste, but prior to being certified closed	No change, except that last final capping event of each landfill will be treated as a part of final capping		
Post-closure	Includes routine monitoring and maintenance of a landfill after it has closed, ceased to accept waste and been certified as closed by the applicable state regulatory agency	No change		
Discount Rate:	Risk-free rate (5.0% at December 31, 2002)	Credit-adjusted, risk-free rate (7.25% at January 1, 2003)		
	8			

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

escription	Historical Practice	Current Practice (Effective January 1, 2003)		
Cost Estimates:	Costs were estimated based on performance, principally by third parties, with a small portion performed by the Company	No change, except that the cost of any activities performed internally must be increased to represent an estimate of the amount a third party would charge to perform such activity		
Inflation:	Cost was inflated to period of performance (2.0% at December 31, 2002)	Inflation rate changed to 2.5% effective January 1, 2003		
Closure and post-closure	Accrued over the life of the landfill; the discounted cash flow associated with such liabilities was recorded to accrued liabilities, with a corresponding charge to cost of operations as airspace is consumed	Accrued over the life of the landfill; the discounted cash flow associated with such liabilities is recorded to accrued liabilities, with a corresponding increase in landfill assets as airspace is consumed		
Statement of Operations Expense:				
Liability accrual	Expense charged to cost of operations at same amount accrued to liability	Not applicable		
Landfill asset amortization	Not applicable for landfill closure and post closure obligations. For final capping, the capitalized and expected future costs (on an undiscounted basis) were amortized as airspace was consumed over the life of the landfill	Landfill asset is amortized to depreciation and amortization expense as airspace is consumed over the life of the specific final capping event or life of landfill for closure and post-closure		
Accretion	Expense, charged to cost of operations, was accrued at risk-free rate over the life of the landfill as airspace was consumed	Expense, charged to cost of operations, is accreted at credit-adjusted, risk-free rate (7.25%) under the effective interest method		

We amortize landfill retirement costs arising from closure and post-closure obligations, which are capitalized as part of the landfill asset, using our historical landfill accounting practices. However, we amortize landfill retirement costs arising from final capping obligations, which are also capitalized as part of the landfill asset, on a units-of-consumption method as airspace is consumed over the estimated capacity associated with each final capping event. This change from our historical method requires the Company to make additional estimates related to the capacity of each final capping event and to track landfill consumption by event. These estimates are reviewed at least annually.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The following table summarizes the balance sheet impact of the adoption of SFAS No. 143 (in millions):

	Balance at December 31, 2002	Change	Balance at January 1, 2003
Landfill assets	\$ 8,607	\$ 257	\$ 8,864
Accumulated landfill airspace amortization	(3,539)	(161)	(3,700)
Net landfill assets	\$ 5,068	\$ 96	\$ 5,164
Current landfill and environmental remediation liabilities	\$ 114	\$ 67	\$ 181
Long-term landfill and environmental remediation liabilities	884	199	1,083
Total landfill and environmental remediation liabilities	\$ 998	\$ 266	\$ 1,264

2. Landfill and Environmental Remediation Liabilities

Landfill

We have material financial commitments for final capping, closure and post-closure obligations with respect to our landfills. We develop our estimates of these obligations using input from our operations, engineers and accountants. Our estimates are based on our interpretation of current requirements and proposed regulatory changes and are intended to approximate fair value. Absent quoted market prices, the estimate of fair value should be based on the best available information, including the results of present value techniques. In general, we contract with third parties to fulfill most of our obligations for final capping, closure and post-closure. Accordingly, the fair value of these obligations is based upon quoted and actual prices paid for similar work. However, we intend to perform some of these activities using internal resources. Where internal resources are expected to be used to fulfill an asset retirement obligation, we have added a profit margin to the estimated cost of such services to better reflect the fair value of the obligation. When we then perform these services internally, the added profit margin is recognized as a component of operating income in the period earned. An estimate of fair value should include the price that marketplace participants are able to receive for bearing the uncertainties in cash flows. However, when using discounted cash flow techniques, reliable estimates of market premiums may not be obtainable. In the waste industry, there is no market for selling the responsibility for final capping, closure and post-closure obligations independent of selling the landfill in its entirety. Accordingly, we do not believe that it is possible to develop a methodology to reliably estimate a market risk premium and have therefore excluded any such market risk premium from our determination of expected cash flows for landfill asset retirement obligations.

Once we have determined the final capping, closure and post-closure costs, we then inflate those costs to the expected time of payment and discount those expected future costs back to present value. We have inflated these costs in current dollars until the expected time of payment using an inflation rate of 2.5% at June 30, 2003, and have discounted these costs to present value using a credit-adjusted, risk-free discount rate of 7.25% at June 30, 2003. The credit-adjusted, risk-free rate is based on the risk-free interest rate on obligations of similar maturity adjusted for our own credit rating. Changes in our credit-adjusted, risk-free rate do not change recorded liabilities, but subsequently recognized obligations are measured using the revised credit-adjusted, risk-free rate.

Management reviews the estimates of our obligations at least annually. Significant changes in inflation rates or the amount or timing of future final capping, closure and post-closure cost estimates typically result in both (i) a current adjustment to the recorded liability (and corresponding adjustment to the landfill asset), based on the landfill's capacity consumed to date, and (ii) a change in liability and asset amounts to be

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

recorded prospectively over the remaining capacity of the landfill. Any changes related to the capitalized and future cost of the landfill assets are then recognized in accordance with our amortization policy, which would generally result in amortization expense being recognized prospectively over the remaining capacity of the final capping event or the landfill, as appropriate.

We record the estimated fair value of final capping, closure and post-closure liabilities for our landfills based on the capacity consumed through the current period. The liability and corresponding asset are recorded on a per-ton basis. The estimated fair value of each final capping event will be fully accrued when the tons associated with such capping event have been disposed in the landfill. Additionally, the estimated fair value of total future final capping, closure and post-closure costs will be fully accrued for each landfill at the time the site discontinues accepting waste and is closed. Closure and post-closure accruals consider estimates for methane gas control, leachate management and ground-water monitoring and other operational and maintenance costs to be incurred after the site discontinues accepting waste, which is generally expected to be for a period of up to thirty years after final site closure. Daily maintenance activities, which include many of these costs, are expensed as incurred during the operating life of the landfill. Daily maintenance activities include leachate disposal; surface water, groundwater, and methane gas monitoring and maintenance; other pollution control activities; mowing and fertilizing the landfill final cap; fence and road maintenance; and third party inspection and reporting costs. For purchased disposal sites, we assess and record the estimated fair value of final capping, closure and post-closure liabilities at the time we assume such responsibilities. Such liabilities are based on the percentage of airspace consumed related to such obligations as of the date we assumed the responsibility. Thereafter, we account for the landfill and related final capping, closure and post-closure obligations consistent with the policy described above.

Interest accretion on final capping, closure and post-closure liabilities is recorded using the effective interest method and is recorded as final capping, closure and post-closure expense, which is included in operating costs on the income statement.

In the United States, the final capping, closure and post-closure requirements are established by the EPA's Subtitles C and D regulations, as implemented and applied on a state-by-state basis. The costs to comply with these requirements could change materially as a result of legislation or regulation.

Environmental Remediation

We routinely review and evaluate sites that require remediation, including sites listed on the EPA's National Priorities List, or NPL. We consider whether we were an owner, operator, transporter, or generator at the site, the amount and type of waste hauled to the site and the number of years we were connected with the site. We also review the same information with respect to other named and unnamed PRPs, or potentially responsible parties. We then determine our estimated cost for the likely remedy, which we base on (i) management's judgment and experience in remediating sites; (ii) information available from regulatory agencies as to costs of remediation; (iii) the number, financial resources and relative degree of responsibility of other PRPs who may be liable for remediation of a specific site; and (iv) the typical allocation of costs among PRPs. These estimates are sometimes a range of "reasonably possible" outcomes. "Reasonably possible" outcomes are those outcomes that are considered more than remote and less than likely. In cases where our estimates are a range, we use the amount within the range that constitutes our best estimate. If no amount within the range appears to be a better estimate than any other, we use the amounts that are the low ends of such ranges in accordance with SFAS No. 5, *Accounting for Contingencies*, and its interpretations. Were we to use the high ends of such ranges, our aggregate potential liability would be approximately \$180 million higher on a discounted basis than the estimate recorded in the consolidated financial statements as of June 30, 2003.

As of June 30, 2003, we had been notified that we are a PRP in connection with 74 locations listed on the NPL. Through various acquisitions, we have come to own 17 of these sites that were initially developed by others. We are working with the government to characterize or remediate identified site problems and have

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

either agreed with other legally liable parties on an arrangement for sharing the costs of remediation or are pursuing resolution of an allocation formula. We generally expect to receive any amounts due from these parties at, or near, the time that we make remedial expenditures. Claims have been made against us at another 57 sites we do not own where we have been an operator, transporter or generator of waste. These claims are at different procedural stages under the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, which is also known as Superfund. At some of these sites, our liability is well defined as a consequence of a governmental decision and an agreement among liable parties as to the allocation of costs. At others where no remedy has been selected or the liable parties have been unable to agree on an appropriate allocation, our future costs are uncertain. Any of these matters could have a material adverse effect on our consolidated financial statements.

Estimating our degree of responsibility for remediation of a particular site is inherently difficult and determining the method and ultimate cost of remediation requires that a number of assumptions be made. Our ultimate responsibility may differ materially from current estimates. It is possible that technological, regulatory or enforcement developments, the results of environmental studies, the inability to identify other PRPs, the inability of other PRPs to contribute to the settlements of such liabilities, or other factors could require us to record additional liabilities that could be material. Additionally, our ongoing review of our remediation liabilities could result in revisions that could cause upward or downward adjustments to income from operations. These adjustments could be material in any given period.

Where we believe that both the amount of a particular environmental remediation liability and the timing of the payments are reliably determinable, we inflate the cost in current dollars (2.5% at June 30, 2003 and 2.0% at December 31, 2002) until the expected time of payment and discount the cost to present value using a risk-free discount rate with a term approximating the weighted average period until settlement of the underlying obligation (4.0% at June 30, 2003 and 5.0% at December 31, 2002). We review the discount rate, which is based on the rates for United States Treasury bonds, and the inflation rate on an annual basis. For remedial liabilities that have been discounted, we include interest accretion, based on the effective interest method, in operating costs and expenses.

Liabilities for landfill and environmental remediation costs are presented in the table below (in millions):

		June 30, 2003			December 31, 2002	
	Landfill(a)	Environmental Remediation	Total	Landfill(a)	Environmental Remediation	Total
Current (in accrued liabilities)	\$ 124	\$ 61	\$ 185	\$ 49	\$ 65	\$ 114
Long-term	854	292	1,146	606	278	884
	\$ 978	\$ 353	\$ 1,331	\$ 655	\$ 343	\$ 998

a) As of June 30, 2003, landfill liabilities include our final capping, closure and post-closure obligations pursuant to SFAS No. 143. Final capping obligations to be discharged during the operating lives of landfills were not included in landfill liabilities as of December 31, 2002.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The changes to landfill and environmental remediation liabilities for the six months ended June 30, 2003 and June 30, 2002 are as follows (in millions):

	Landfill	Environmental Remediation
December 31, 2002	\$655	\$343
Cumulative effect of change in accounting principle	266	
Obligations incurred and capitalized	26	
Obligations settled	(20)	(17)
Interest accretion	30	4
Revisions in estimates	1	11
Acquisitions, divestitures and other adjustments	20	12
June 30, 2003	\$978 ——	\$353
December 31, 2001	\$625	\$321
Expense	22	10
Spending	(10)	(20)
Acquisitions, divestitures and other adjustments		
June 30, 2002	\$656	\$328
		_

At several of our landfills, we provide financial assurance by depositing cash into escrow accounts or trust funds that are legally restricted for purposes of settling closure and post-closure obligations. These funds are generally invested in cash or cash-equivalent instruments. The fair value of these escrow accounts and trust funds was approximately \$105 million at June 30, 2003, and is included as other long-term assets in our condensed consolidated balance sheet.

3. Debt and Interest Rate Derivatives

Debt

Debt consisted of the following (in millions):

	June 30, 2003	December 31, 2002
Revolving credit facilities	\$	
Senior notes and debentures, maturing through 2032, interest of 6.375% to 8.75%	6,212	6,164
Tax-exempt bonds maturing through 2031, fixed and variable interest rates ranging from 1% to 10% (weighted average interest rate of 3.1% at June 30,		
2003)	1,376	1,262
Project bonds, principal payable in periodic installments, maturing through 2027, fixed and variable interest rates ranging from 1% to 9% (weighted average		
interest rate of 5.1% at June 30, 2003)	628	634
5.75% convertible subordinated notes due 2005	32	32
Capital leases and other notes payable, maturing through 2022, interest rates up to		
12%	210	201
	8,458	8,293
Less current portion	108 (a),(b),(c)	231 (a)
	\$8,350	\$8,062
	_	

As of December 31, 2002, we had \$434 million of 6.375% senior notes due December 1, 2003 and \$450 million of 7.1% senior notes due August 1, 2026 that were subject to redemption on August 1, 2003 at the option of the holders. We classified \$770 million of these borrowings as long-term at December 31, 2002. As of June 30, 2003, we have \$434 million of 6.375% senior notes due December 1, 2003, \$150 million of 8.0% senior notes due April 30, 2004 and \$200 million of 6.5% senior

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

notes due May 15, 2004. We classified all of these borrowings as long-term at June 30, 2003. The classification of these obligations as long-term was based upon our available capacity under our two long-term revolving credit facilities and our intent to refinance the borrowings with other long-term financings. In the event other sources of long-term financing are not available, we intend to use our revolving credit facilities.

- b) We have \$543 million of tax-exempt bonds at June 30, 2003 that mature through 2031 that are remarketed weekly by a remarketing agent. If the remarketing agent is unable to remarket the bonds, then the remarketing agent can put the bonds to the Company. We have obtained letters of credit, issued under our revolving credit facilities, to guarantee repayment of the bonds in this event. We classified these borrowings as long-term at June 30, 2003 because the borrowings are supported by letters of credit issued under our two revolving credit facilities, which are both long-term.
- c) The Company has \$450 million of 7.1% senior notes due August 1, 2026. These notes were subject to repayment on August 1, 2003 at the option of the holder. Approximately \$1 million of the notes were put to the Company for redemption. The right to put the remaining \$449 million of the senior notes expired July 15, 2003. Therefore, \$449 million of the 7.1% senior notes are classified as long-term in the accompanying condensed consolidated balance sheet.

As of June 30, 2003, we had a three-year, \$650 million syndicated revolving credit facility and a five-year, \$1.75 billion syndicated revolving credit facility. The three-year revolver matures in June 2005 and the five-year revolver matures in June 2006. At June 30, 2003, no borrowings were outstanding under our revolving credit facilities and we had unused and available credit capacity under these facilities of approximately \$870 million. The unused and available capacity under these facilities was approximately \$770 million at December 31, 2002.

As part of our operations, and in connection with issuances of tax-exempt bonds, we use letters of credit to support our bonding and funding needs. In order to increase our letter of credit availability, on June 30, 2003 we entered into a five-year, \$15 million letter of credit and term loan agreement, a seven-year, \$175 million letter of credit and term loan agreement, and a ten-year, \$105 million letter of credit and term loan agreement, which expire in June 2008, 2010, and 2013, respectively (collectively, the "LC and term loan agreements"). At June 30, 2003, letters of credit were issued and outstanding for the full \$295 million of credit capacity under these agreements.

As of June 30, 2003, we had letters of credit in the aggregate amount of approximately \$1.93 billion (of which approximately \$1.53 billion are issued under the revolving credit facilities, \$295 million are issued under the LC and term loan agreements and the remainder are issued under other various lines of credit). These letters of credit generally have terms allowing for automatic renewal after one year. In the event of an unreimbursed draw on a letter of credit, we have the ability to convert that amount into a term loan for the remaining term under its respective agreement or facility.

As of June 30, 2003, we are required to maintain the following financial covenants under our revolving credit facilities: (i) an interest coverage ratio; (ii) total debt to EBITDA; and (iii) minimum net worth, all as defined in the credit facilities solely for the purpose of determining compliance with the covenants. The interest coverage ratio requires that at the end of any fiscal quarter we will not permit the ratio of (A) our consolidated net income plus interest expense and income taxes ("EBIT") for the four fiscal quarters then ending to (B) consolidated total interest expense for such period, to be less than 3 to 1. The total debt to EBITDA covenant requires that at the end of any fiscal quarter, we will not permit the ratio of (A) all indebtedness and certain contingent liabilities such as financial guarantees to (B) EBIT plus depreciation and amortization expense ("EBITDA") for the four fiscal quarters then ending to exceed 3.25 to 1. Our minimum net worth covenant restricts us from allowing the sum of stockholders' equity to be less than \$3.5 billion plus 75% of our cumulative consolidated net income for each fiscal quarter, beginning with the first fiscal quarter ended March 31, 2001. We are in compliance with all covenants under our revolving credit facilities and all other debt instruments.

Our tax-exempt and project bonds are primarily related to industrial revenue bonds issued for the construction of various facilities. Proceeds from these financing arrangements are treated as non-cash financing activities for purposes of the statement of cash flows as the proceeds are directly deposited into trust

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

funds, and we do not have the ability to use the funds in regular operating activities. Accordingly, our 2003 issuances of approximately \$116 million of tax-exempt bonds maturing through 2028 were treated as non-cash financing activities.

Our debt balances are generally unsecured, except for approximately \$612 million of the project bonds outstanding at June 30, 2003 that are issued by certain of our subsidiaries within our Wheelabrator Group and secured by the related subsidiary's assets and revenue.

Interest rate swaps

We manage the interest rate risk of our debt portfolio principally by using interest rate derivatives to achieve a desired position of fixed and floating rate debt, which was approximately 67% fixed and 33% floating at June 30, 2003. Interest rate swap agreements outstanding as of December 31, 2002 and June 30, 2003 are set forth in the table below (dollars in millions):

As of	Notional Amount		Receive		Pay	Maturity Date	Fair Value Asset/(Liability)(a)
December 31, 2002	\$ 19	Floating	1.38%	Fixed	7.27%	Through December 31, 2012	\$ (3)(b)
December 31,	Ψ 15	1 louting	1.3070	1 IACU	7.2770	Timough December 51, 2012	Ψ(5)(6)
2002	\$2,000	Fixed	6.38%-7.65%	Floating	2.97%-4.91%	Through July 15, 2028	\$34 (c)
June 30, 2003	\$ 18	Floating	1.12%	Fixed	7.27%	Through December 31, 2012	\$(3)(b)
June 30, 2003	\$2,250	Fixed	6.38-7.65%	Floating	3.71-5.51%	Through December 15, 2017	\$(8)(c),(d)

- a) The fair value of interest rate derivatives is included in our balance sheets as components of other long-term assets and other long-term liabilities. Fair values of these interest rate derivatives are based on third party pricing models.
- b) The terms of this interest rate derivative contract do not qualify for hedge accounting. Therefore, the contract is accounted for at fair value with changes in fair value recognized immediately in interest expense.
- c) These interest rate derivatives qualify for hedge accounting. Therefore, changes in fair value of these interest rate swap contracts are deferred and recognized as an adjustment to interest expense over the remaining life of the hedged instrument.
- d) The fair value of these interest rate derivatives is a net fair value liability of \$8 million that is comprised of \$12 million of other long-term assets and \$20 million of other long-term liabilities.

During the first and second quarters of 2003, we terminated several interest rate swap agreements with a notional amount of \$2.35 billion prior to the scheduled maturities and received cash of \$117 million (which was comprised of \$109 million for the fair value of the swaps terminated and \$8 million of interest receivable) from the counterparties to the interest rate swaps. We had designated these swap agreements as fair value hedges, and as such the unamortized adjustment to long-term debt for the change in fair value of the swaps remains classified in long-term debt. The proceeds received from the termination of the interest rate swap agreements have been classified as a change in other assets or other liabilities within operating activities in the accompanying condensed consolidated statement of cash flows.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Fair value hedge accounting for interest rate swap contracts increased the carrying value of debt instruments by approximately \$285 million as of June 30, 2003 and \$239 million as of December 31, 2002. The following table summarizes the accumulated fair value adjustments from interest rate swap agreements by underlying debt instrument category (in millions):

Increase in carrying value of debt due to hedge accounting for interest rate swaps	June 30, 2003	December 31, 2002
Senior notes and debentures:		
Active swap agreements	\$ (8)	\$ 34
Terminated swap agreements	292 (a)	203
	284	237
Tax-exempt and project bonds:		
Terminated swap agreements	1 (a)	2
	\$285	\$239
	_	

a) Of these amounts, \$51 million (on a pre-tax basis) is scheduled to be reclassified as a credit to interest expense over the next twelve months.

Interest rate swap agreements reduced net interest expense by \$23 million and \$47 million for the three and six months ended June 30, 2003, respectively, and \$23 million and \$43 million for the three and six months ended June 30, 2002, respectively. The significant terms of the interest rate contracts and the underlying debt instruments are identical and therefore no ineffectiveness has been realized.

Interest rate locks

During 2002 and 2001, we entered into cash flow hedges to secure the underlying interest rates in anticipation of our senior note issuances. These hedging agreements resulted in a deferred loss, net of taxes, of \$38 million at June 30, 2003, which is included in accumulated other comprehensive income. Of this amount, \$5 million (on a pre-tax basis) is scheduled to be reclassified into interest expense over the next twelve months.

4. Income Taxes

The current tax obligations associated with the provision for income taxes recorded in the statements of operations are reflected in the accompanying condensed consolidated balance sheets as a component of accrued liabilities, and the deferred tax obligations are reflected in deferred income taxes. The difference in federal income taxes computed at the federal statutory rate and reported income taxes for the three and six months ended June 30, 2003 and June 30, 2002 is primarily due to state and local income taxes, offset in part by non-conventional fuel tax credits.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

5. Comprehensive Income

Comprehensive income represents all changes in our equity except for changes resulting from investments by, and distributions to, stockholders. Comprehensive income for the three and six months ended June 30, 2003 and June 30, 2002 was as follows (in millions):

	Three Months Ended June 30,		Six Months Ended June 30,	
	2003	2002	2003	2002
Net income	\$176	\$217	\$237	\$355
Other comprehensive income:				
Minimum pension liability adjustment, net of taxes	_	_	1	_
Unrealized loss resulting from changes in fair values of				
derivative instruments, net of tax benefit	_	(24)	(3)	(23)
Gain on derivative instruments reclassified into earnings, net of taxes				(1)
Unrealized loss on marketable securities, net of tax				(1)
benefit	_	(2)	_	(5)
Translation adjustment of foreign currency statements	58	36	98	37
Other comprehensive income	58	10	96	8
		_		
Comprehensive income	\$234	\$227	\$333	\$363
	_	_	_	_

The components of accumulated other comprehensive loss were as follows:

	June 30, 2003	December 31, 2002
Minimum pension liability adjustment, net of taxes	\$ —	\$ (1)
Accumulated unrealized loss on derivative instruments, net of tax benefit	(42)	(39)
Cumulative translation adjustment of foreign currency statements	(41)	(139)
	\$(83)	\$(179)
	_	

6. Earnings Per Share and Stock Based Compensation

The following reconciles net income as presented on the condensed consolidated statements of operations with diluted net income for purposes of calculating diluted earnings per common share (in millions):

	Three Months Ended June 30,		Six Months June 3	
	2003	2002	2003	2002
Net income	\$176	\$217	\$237	\$355
Interest on convertible securities, net of taxes	_	_	_	1
Diluted net income	\$176	\$217	\$237	\$356
	_	_	_	_

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The following reconciles the number of common shares outstanding at June 30 of each year presented to the weighted average number of common shares outstanding and to the weighted average number of common and potentially dilutive common shares outstanding for the purpose of calculating basic and diluted earnings per common share (shares in millions):

	Three Months Ended June 30,		Six Month June	
	2003	2002	2003	2002
Number of common shares outstanding at end of period Effect of using weighted average common shares	591.8	611.4	591.8	611.4
outstanding	(0.3)	3.9	0.8	9.2
Weighted average basic common shares outstanding	591.5	615.3	592.6	620.6
Dilutive effect of common stock options and warrants and other contingently issuable shares	3.3	3.9	2.6	4.1
Dilutive effect of convertible subordinated notes	_	_	_	1.7
Weighted average diluted common shares outstanding	594.8	619.2	595.2	626.4
		_	_	

For the three and six months ended June 30, 2003 and the three months ended June 30, 2002, we excluded the effect of our convertible subordinated notes from the diluted earnings per share calculation because the inclusion of this item would have been antidilutive.

At June 30, 2003, there were approximately 52.7 million shares of common stock potentially issuable with respect to stock options, warrants and convertible debt. We excluded approximately 32.3 million of these shares from the diluted earnings per share computation because their exercise price was greater than the average per share market price of our stock for the three and six months ended June 30, 2003. Including the impact of these potentially issuable shares in the current period calculations would have been antidilutive at June 30, 2003, but may dilute earnings per share in future periods.

We account for our stock-based compensation using the intrinsic value method prescribed by APB Opinion No. 25, *Accounting for Stock Issued to Employees*, as amended. Pursuant to APB No. 25, we recognize no compensation cost for our stock option grants because the number of shares potentially issuable and the exercise price, which is equal to the fair market value of the underlying stock on the date of grant, is fixed.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The following schedule reflects the pro forma impact on net income and earnings per common share of accounting for our stock option grants using SFAS No. 123, *Accounting for Stock-Based Compensation*, which would result in the recognition of compensation expense for the fair value of stock option grants as computed using the Black-Scholes option-pricing model (in millions, except per share amounts).

	Three Months Ended June 30,			nths Ended ne 30,
	2003	2002	2003	2002
Reported net income	\$ 176	\$ 217	\$ 237	\$ 355
Less: compensation expense per SFAS No. 123, net of tax benefit	18		33	40
Pro forma net income	\$ 158	\$ 195	\$ 204	\$ 315
	_	_	_	_
Basic earnings per common share:				
Reported net income	\$ 0.30	\$ 0.35	\$ 0.40	\$ 0.57
Less: compensation expense per SFAS No. 123, net of tax benefit	0.03	0.03	0.06	0.06
Pro forma net income	\$ 0.27	\$ 0.32	\$ 0.34	\$ 0.51
Diluted earnings per common share:				
Reported net income	\$ 0.30	\$ 0.35	\$ 0.40	\$ 0.57
Less: compensation expense per SFAS No. 123, net of tax benefit	0.03	0.04	0.06	0.07
Pro forma net income	\$ 0.27	\$ 0.31	\$ 0.34	\$ 0.50

7. Common Stock Repurchases

In February 2002 we announced that our Board of Directors had approved a stock repurchase program for up to \$1 billion in annual repurchases each year through 2004, to be implemented at management's discretion. Under the program, our purchases may be made in either open market or privately negotiated transactions primarily using cash flows from operations.

The following is a summary of our 2003 activity under the stock repurchase program (in millions, except shares in thousands and price per share in dollars).

	Agre	ement		Common Stock	Total Purchase	Settlement (Received)/	Net Common Stock
Transaction Type	Initiating Date	Settlement Date	Shares	Price per Share	Price	Paid	Repurchases
Private Accelerated Purchase(a)	December 2002	February 2003	1,731	\$24.52	\$ 42	\$ (3)(b)	\$ 39
Private Accelerated Purchase(a) Open Market Purchases(d)	March 2003 N/A	May 2003 N/A	2,400 1,163	\$20.00 \$19.70 - \$20.77	\$ 48 \$ 23	\$ 3 (c) N/A	\$ 51 \$ 23

- a) We accounted for the initial payments as a purchase of treasury stock and classified the future settlements with the counterparty as an equity instrument because we have the option under these agreements to settle our obligations, if any, in shares of our common stock.
- b) The weighted average daily market price of our stock during the valuation period times the number of shares we purchased was approximately \$3 million less than the approximately \$42 million we initially paid. Pursuant to the agreement, the counterparty paid us the difference of approximately \$3 million at the end of the valuation period to settle the agreement.
- c) The weighted average daily market price of our stock during the valuation period times the number of shares we purchased was approximately \$3 million more than the approximately \$48 million we initially paid. Pursuant to the agreement, we paid the counterparty the difference of approximately \$3 million at the end of the valuation period to settle the agreement.
- d) We engaged in open market purchases during the first quarter of 2003 when trading was allowed pursuant to law and our insider trading policy. We did not engage in open market purchases during the second quarter of 2003.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

8. Commitments and Contingencies

Financial instruments — We have obtained letters of credit, performance bonds and insurance policies, and have established trust funds and issued financial guarantees to support tax-exempt bonds, contracts, performance of landfill closure and post-closure requirements, and other obligations. We also use insurance policies issued by our wholly-owned insurance company, the sole business of which is to issue us policies, to secure such performance obligations. In those instances where our use of captive insurance is not allowed, we generally have available alternative bonding mechanisms. Because virtually no claims have been made against these financial instruments in the past, and considering our current financial position, management does not expect that these instruments will have a material adverse effect on our consolidated financial statements. We have not experienced any unmanageable difficulty in obtaining performance bonds or letters of credit for our current operations. During the second quarter of 2003, we entered into the LC and term loan agreements to provide us with an additional source of capacity from which we may obtain letters of credit. See Note 3 for additional information related to these agreements. In an ongoing effort to mitigate risks of future cost increases and reductions in available capacity, we continue to evaluate various options to access cost-effective sources of financial assurance.

For the 14 months ended January 1, 2000, we insured certain risks, including auto, general liability and workers' compensation, with Reliance National Insurance Company, whose parent filed for bankruptcy in June 2001. In October 2001, the parent and certain of its subsidiaries, including Reliance National Insurance Company, were placed in liquidation. We believe that because of various state insurance guarantee funds and potential recoveries from the liquidation, it is unlikely that events relating to Reliance will have a material adverse impact on our financial statements.

Guarantees — We have entered into the following guarantee agreements associated with our operations.

- Waste Management Holdings, Inc. ("WM Holdings"), one of WMI's wholly-owned subsidiaries, has fully and unconditionally guaranteed WMI's senior indebtedness that matures through 2032. WMI has fully and unconditionally guaranteed the senior indebtedness of WM Holdings that matures through 2026 and WM Holdings' 5.75% convertible subordinated notes due 2005. Performance under these guarantee agreements would be required if either party defaulted on their respective obligations. No additional liability has been recorded for these guarantees because the underlying obligations are reflected in our consolidated balance sheets. See Note 12 for further information.
- WMI has guaranteed the tax-exempt bonds of its subsidiaries. If a subsidiary fails to meet its obligations associated with tax-exempt bonds as they come due, WMI will be required to perform under the related guarantee agreement. No additional liability has been recorded for these guarantees because the underlying obligations are reflected in our consolidated balance sheets. See Note 3 for information related to the balances and maturities of our tax-exempt bonds.
- As of June 30, 2003, we have guaranteed approximately \$49 million of debt primarily for entities that we account for under the equity method of accounting. The related debt, which matures through 2021, is not recorded on our consolidated balance sheets, and we have not recorded any liability for these guarantees. We have ongoing projects with the entities and have deemed that our performance under these guarantees is not likely.
- Certain of our subsidiaries have guaranteed the market value of certain homeowners' properties that are adjacent to our landfills. These guarantee agreements extend over the life of the landfill. Under these agreements, we would be responsible for the difference between the sale value and the market value of the homeowners' properties, if any. We do not believe it is possible to determine the contingent obligation associated with these guarantees, but we do not believe it would have a material effect on our financial position or results of operations.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

- The Company has indemnified the purchasers of businesses or divested assets for the occurrence of specified events under certain of our divestiture agreements. We do not believe that it is possible to determine the contingent obligations associated with these indemnities.
- WMI guarantees the lease obligations of certain of its subsidiaries. If a subsidiary fails to meet its contractual lease obligations as they come due, WMI has an unconditional obligation to perform on its behalf. No additional liability has been recorded for these guarantees because the subsidiaries' obligations are properly accounted for as operating or capital leases, as appropriate.

We currently believe that it is not reasonably likely that we will be required to perform under these guarantee agreements or that any performance requirement would have a material impact on our consolidated financial statements.

Environmental matters — Our business is intrinsically connected with the protection of the environment. As such, a significant portion of our operating costs and capital expenditures could be characterized as costs of environmental protection. Such costs may increase in the future as a result of legislation or regulation. However, we believe that we generally tend to benefit when environmental regulation increases, because such regulations increase the demand for our services, and we have the resources and experience to manage environmental risk. For more information regarding environmental matters, see Note 2.

Litigation — In December 1999, an individual brought an action against the Company, five former officers of WM Holdings, and WM Holdings' former independent auditor, Arthur Andersen LLP, in Illinois state court on behalf of a proposed class of individuals who purchased WM Holdings common stock before November 3, 1994, and who held that stock through February 24, 1998. The action is for alleged acts of common law fraud, negligence and breach of fiduciary duty. In May 2001, the court granted in part and denied in part the defendants' motion to dismiss. This action is currently in the discovery stage and the extent of possible damages, if any, has not yet been determined.

We are one of the defendants in a class action lawsuit arising from events related to our earnings announcements in July and August of 1999. On November 7, 2001, we announced that we had reached a settlement agreement with the plaintiff in this case, resolving all claims against us as well as claims against our current and former officers and directors. The agreement provides for a payment of \$457 million to members of the class and for us to consent, for settlement purposes, to the certification of a class of purchasers or acquirers of our securities from June 11, 1998 through November 9, 1999. The payment provided for by the settlement agreement, including interest accrued at the Federal Funds rate, is included in the accompanying condensed consolidated balance sheets as a component of accrued liabilities. A hearing was held April 29, 2002 at which the settlement was approved. A former participant in WM Holdings' ERISA plans filed an appeal of both the final ruling and related motions on March 27, 2003. That appeal has now been conditionally remanded back to the trial court where a settlement among plaintiffs' counsel and the appellant was approved by the court. This settlement requires no additional commitment from the Company. If no appeal is taken from this most recent approval from the trial court, the case will be final and payment will be due in September. This former participant and another individual also filed a separate case in Washington, D.C. against us and others, attempting to increase the recovery of a class of ERISA plan participants based on allegations related to both the events alleged in, and the settlements relating to, the class action against WM Holdings that was settled in 1998 and the complaint in this action. This D.C. case has been stayed pending finalization of the class action settlement.

Also on November 7, 2001, we announced that we would receive \$20 million (less fees of approximately \$5 million awarded to counsel for the derivative plaintiffs) as a result of a settlement reached between the derivative plaintiffs and Arthur Andersen in a stockholder derivative suit filed on July 3, 2001 in Texas state court against Arthur Andersen, as our former independent auditor. The derivative plaintiffs alleged, among other things, that Arthur Andersen engaged in professional malpractice in connection with certain services

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

that it performed for us. Arthur Andersen informed us that neither the complaint nor the settlement affected its independence in 2001 or prior years, when Arthur Andersen was serving as our independent auditor. The settlement was approved in May 2002. Arthur Andersen's payment will become due at the same time our payment to the plaintiffs in settlement of the class action lawsuit described above becomes due. Because of the appeal filed in that case in March 2003, we cannot predict when the class action payment will be due, and thus cannot predict when the Arthur Andersen receivable will become due. Although Arthur Andersen has ceased operations, we presently have no reason to believe that the settlement amount will not be paid and we will not collect the receivable we have recorded. Any prolonged delay in the payment ultimately becoming due could make such payment more uncertain, depending on Arthur Andersen's financial status at such time. If such uncertainties develop in the future, it is reasonably possible that we will be required to write down the receivable up to the full amount due from Arthur Andersen.

Three groups of stockholders have filed separate lawsuits in state courts in Texas and federal court in Illinois against us and certain of our former officers. The petitions allege that the plaintiffs are substantial stockholders of the Company's common stock who intended to sell their stock in 1999, or to otherwise protect themselves against loss, but that the individual defendants made false and misleading statements regarding the Company's prospects that, along with public statements, induced the plaintiffs to retain their stock or not to take other measures. The plaintiffs assert that the value of their retained stock declined dramatically. The plaintiffs assert claims for fraud, negligent misrepresentation, and conspiracy. The Texas state court granted our motion for summary judgment in the first of these cases in March 2002, which the plaintiffs have appealed. The other two cases are in early stages, and the extent of damages, if any, cannot yet be determined.

Our business is intrinsically connected with the protection of the environment, and there is the potential for unintended or unpermitted noncompliance with environmental laws or regulations. From time to time, we pay fines or penalties in environmental proceedings relating primarily to waste treatment, storage or disposal facilities. As of June 30, 2003, there were 11 proceedings involving our subsidiaries where the sanctions involved in each could potentially exceed \$100,000. The matters involve allegations that subsidiaries (i) operated a hazardous waste incinerator in such a way that its air emissions exceeded permit limits, (ii) operated a waste-to-energy facility that, as a result of intermittent and isolated equipment malfunctions, exceeded emission limits and failed to meet monitoring requirements, (iii) are responsible for remediation of landfill gas and chemical compounds required pursuant to a Unilateral Administrative Order associated with an NPL site, (iv) are responsible for late performance of work required under a Unilateral Administrative Order, (v) in two separate cases improperly operated a solid waste landfill and caused excess odors, (vi) improperly operated a solid waste landfill by failing to maintain required records, properly place and cover waste and adhere to proper leachate levels, (vii) violated the state's clean water act, (viii) did not comply with air regulations requiring control of emissions at a closed landfill, and (ix) improperly operated a solid waste landfill by failing to maintain required leachate levels and erosion control and failing to properly operate and monitor gas wells and adequately control odors and stormwater. On April 11, 2003, Alliance Sanitary Landfill, Inc., a whollyowned subsidiary, entered into a Consent Order with the Pennsylvania Department of Environmental Protection, pursuant to which we voluntarily suspended operations at the landfill to address certain gas collection system, leachate collection system and capping deficiencies. The landfill reopened on May 9, 2003, and we are negotiating the resolution of any fines arising from the Consent Order. We believe that the resolution will likely exceed \$100,000. We do not believe that the fines or other penalties in any of these matters will, individually or in the aggregate, have a material adverse effect on our financial condition or results of operations.

It is not always possible to predict the impact that lawsuits, proceedings, investigations and inquiries may have on us, nor is it possible to predict whether additional suits or claims may arise out of the matters described above in the future. We and each of our subsidiaries intend to defend ourselves vigorously in all the above matters. However, it is possible that the outcome of any of these matters may ultimately have a material adverse impact on our financial condition or results of operations in one or more future periods.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

We and certain of our subsidiaries are also currently involved in other routine civil litigation and governmental proceedings relating to the conduct of our business. We do not believe that any of these routine matters will have a material adverse impact on our consolidated financial statements.

Tax matters — We are currently under audit by the Internal Revenue Service and from time to time are audited by other taxing authorities. We fully cooperate with all audits, but defend our positions vigorously. Our audits are in various stages of completion. Results of audit assessments by taxing authorities could have a material effect on our quarterly or annual cash flows. However, we do not believe that any of these matters will have a material adverse impact on our results of operations.

9. Restructurings

In 2002 we reorganized the structure of the Company into market areas to better align our collection, transport, recycling and disposal resources. As part of the restructuring, we reduced the number of field layers of management and eliminated approximately 1,900 field-level administrative and operational positions. In the first quarter of 2002, we recorded \$37 million of pre-tax charges for costs associated with the implementation of the new structure. An additional \$1 million was recorded in the third quarter of 2002. These charges included \$36 million for employee severance and benefit costs and \$2 million related to abandoned operating lease agreements. As of June 30, 2003, payments of \$36 million had been recorded against the restructuring liability that was previously established and \$2 million of the related accrual remained.

In February 2003 we reduced the number of market areas that make up our geographic operating Groups to 66 from a total of 91 at December 31, 2002, and reduced certain overhead positions to further streamline our organization. Our market areas all report to one of our seven Groups that divide our operations geographically into the Eastern, Midwest, Southern, Western and Canadian operations and functionally into Recycling and Wheelabrator. We manage and evaluate our operations through these seven operating Groups, which represent our reporting segments as further described in Note 11. Management believes that this structure will result in a more effective utilization of resources and enable us to serve our customers more efficiently. In connection with the restructuring, there was a workforce reduction of about 700 employees and 270 contract workers.

The operational efficiencies provided by these organizational changes and a focus on fully utilizing the capabilities of our information technology resources enabled us to further reduce our workforce in June 2003. This workforce reduction resulted in the elimination of an additional 600 employee positions and 200 contract worker positions.

In the first quarter of 2003, we recorded \$20 million of pre-tax charges for costs associated with our February 2003 restructuring and workforce reduction, all of which was associated with employee severance and benefit costs. No material costs for this restructuring and workforce reduction have been incurred in the second quarter of 2003. In the second quarter of 2003, we recorded \$23 million of pre-tax charges for costs associated with the June 2003 workforce reduction.

We anticipate incurring a total of approximately \$44 million in employee severance and benefit costs for the 2003 restructuring and workforce reduction efforts, with \$43 million incurred during the six month period ended June 30, 2003 and the remaining \$1 million to be incurred throughout the remainder of the year. The

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

following table summarizes the total costs recorded for the three and six months ended June 30, 2003 for each of our operating Groups (in millions).

	Three Months Ended June 30, 2003	Six Months Ended June 30, 2003
Canadian	\$ <i>—</i>	\$ 1
Eastern	6	10
Midwest	3	5
Southern	3	7
Western	4	8
Wheelabrator	_	_
Recycling	_	1
Corporate	7	11
	_	_
Total	\$ 23	\$43
	_	_

We paid approximately \$9 million of the employee severance and benefit costs incurred as a result of the February 2003 restructuring and workforce reduction during the three month period ended June 30, 2003 and have paid \$13 million of these costs during the six month period ended June 30, 2003. As of June 30, 2003, \$7 million of the related accrual remained for employee severance and benefit costs. The length of time we are obligated to make severance payments to employees associated with the February 2003 reorganization and workforce reduction varies, with the longest period of obligation ending in the first quarter of 2005.

As of June 30, 2003, we had not paid any of the employee severance and benefit costs incurred as a result of the June 2003 workforce reduction. The length of time we are obligated to make severance payments to employees associated with the June 2003 workforce reduction varies, with the longest period of obligation ending in the third quarter of 2005.

10. Purchase Acquisitions

During the six month period ended June 30, 2003, we paid \$172 million, net of cash acquired, for the acquisitions of certain businesses. We paid approximately \$85 million in the first quarter of 2003 for acquisitions, which was primarily associated with our acquisition of the Peltz Group, the largest privately-held recycler in the United States. Its assets were contributed to Recycle America Alliance. See Note 11 for further discussion. Approximately \$87 million of the acquisitions occurred during the second quarter of 2003, the most significant of which was the acquisition of certain collection assets from Allied Waste Industries.

Additionally, in the second quarter of 2003 we acquired certain operations from Bio-Energy Partners, a general partnership in which we have a 50% ownership interest, for \$18 million. Bio-Energy Partners owns and operates facilities that produce electrical power from landfill gas that is ultimately sold to public utilities and other commercial users. Concurrent with this transaction, the Company received net cash proceeds from Bio-Energy Partners of \$30 million in exchange for assuming a like amount of indebtedness of the partnership. We continue to account for our remaining interests in Bio-Energy Partners as an equity investment.

All 2003 acquisitions were accounted for under the purchase method of accounting, as required by SFAS No. 141, *Accounting for Business Combinations*, with the purchase price being allocated to the net assets acquired based on their respective fair values. As a result of these acquisitions, we recorded approximately \$225 million in additional assets, including approximately \$99 million of goodwill and approximately \$32 million of other intangible assets. No single acquisition has been material to our consolidated financial position or our results of operations for the periods presented herein, and we do not anticipate that these acquisitions, when considered individually or in the aggregate, will have a material impact on our results of operations in future periods.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

11. Segment and Related Information

We manage and evaluate our operations primarily through our Eastern, Midwest, Southern, Western, Canadian, Wheelabrator and Recycling Groups. These seven operating Groups are presented below as our reportable segments. These reportable segments, when combined with certain other operations not managed through the seven operating Groups, comprise our North American Solid Waste, or NASW, operations. NASW, our core business, provides integrated waste management services consisting of collection, disposal (solid waste landfill and hazardous waste landfill), transfer, waste-to-energy facilities and independent power production plants that are managed by Wheelabrator, recycling and other miscellaneous services to commercial, industrial, municipal and residential customers throughout the United States, Puerto Rico and Canada. The operations not managed through our seven operating Groups are presented herein as "Other NASW." We also had international waste management services and non-solid waste services, all of which were divested by March 31, 2002. However, we continue to incur minimal administrative expenses in connection with these divestitures. These operations are included in the table below as "Other."

Summarized financial information concerning our reportable segments for the three and six months ended June 30 is shown in the following table (in millions). In prior interim period disclosures to the condensed consolidated financial statements, NASW was presented as one reportable segment. For comparability purposes, prior period information has been restated to conform to the current year presentation.

Three Months Ended:	Gross Operating Revenues	Intercompany Operating Revenues(d)	Net Operating Revenues(e)	Income from Operations(f)
June 30, 2003				
Canadian	\$ 150	\$ (16)	\$ 134	\$ 17
Eastern	981	(204)	777	91
Midwest	569	(101)	468	84
Southern	758	(116)	642	141
Western	625	(84)	541	83
Wheelabrator	203	(14)	189	71
Recycling(a)	139	(3)	136	(1)
Other NASW(b)	48	(20)	28	(9)
· /				
Total NASW	3,473	(558)	2,915	477
Other	, <u> </u>		· —	_
Corporate(c)	_	_	_	(92)
•				
Total	\$3,473	\$(558)	\$2,915	\$385
				_
June 30, 2002				
Canadian	\$ 137	\$ (13)	\$ 124	\$ 18
Eastern	958	(192)	766	132
Midwest	577	(83)	494	97
Southern	755	(115)	640	149
Vestern	624	(89)	535	96
Wheelabrator	193	(13)	180	62
Recycling(a)	74	(2)	72	4
Other NASW(b)	21	(7)	14	(6)
,				
Total NASW	3,339	(514)	2,825	552
Other				_
Corporate(c)	_	_	_	(88)
1 (-)				
Total	\$3,339	\$(514)	\$2,825	\$464
20111	ψ9,555	ψ(514)	Ψ2,025	Ψ+0+

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Six Months Ended:	Gross Operating Revenues	Intercompany Operating Revenues(d)	Net Operating Revenues(e)	Income from Operations(f)
June 30, 2003				
Canadian	\$ 267	\$ (28)	\$ 239	\$ 24
Eastern	1,843	(369)	1,474	149
Midwest	1,071	(185)	886	142
Southern	1,497	(226)	1,271	281
Western	1,227	(163)	1,064	167
Wheelabrator	402	(30)	372	113
Recycling(a)	272	(6)	266	(1)
Other NASW(b)	94	(35)	59	(6)
· /				
Total NASW	6,673	(1,042)	5,631	869
Other	_	_		(2)
Corporate(c)	_	_	_	(203)
Total	\$6,673	\$(1,042)	\$5,631	\$ 664
1000	\$ 0,0.7 S	\$(1,0 i=)	\$5,051	\$ 55.
June 30, 2002				
Canadian	\$ 251	\$ (24)	\$ 227	\$ 30
Eastern	1,818	(351)	1,467	239
Midwest	1,078	(151)	927	166
Southern	1,480	` '		287
Western		(221)	1,259	171
Wheelabrator	1,213 375	(170) (28)	1,043 347	100
	132	` '	128	
Recycling(a)	42	(4)	28	(5)
Other NASW(b)	42	(14)	20	(12)
E . LNIA CV.		(062)		
Total NASW	6,389	(963)	5,426	976
Other	9	(1)	8	(2)
Corporate(c)	_	_	_	(178)
				
Total	\$6,398	\$ (964)	\$5,434	\$ 796

- a) Our Recycling Group is comprised of Recycle America Alliance, L.L.C. Recycle America Alliance includes certain recycling assets transferred from our geographic operating groups as well as assets contributed by the Peltz Group, our minority interest partner in Recycle America Alliance.
- b) Other NASW includes operations provided throughout our operating Groups for methane gas recovery and certain third party sub- contract and administration revenues managed by our national accounts department. Also included are certain quarter-end adjustments related to the reportable segments but not included in the measure of segment profit or loss used to assess their performance for the periods disclosed.
- c) Corporate functions include the treasury, legal, information technology, tax, insurance, management of closed landfills and related insurance recoveries, centralized service center and other typical administrative functions. Certain of the associated costs for support services are allocated to the seven operating Groups.
- d) Intercompany operating revenues reflect each segment's total intercompany sales including intercompany sales within a segment and between segments. Transactions within and between segments are generally made on a basis intended to reflect the market value of the service.
- e) Our operating revenues tend to be somewhat lower in the winter months, primarily due to the lower volume of construction and demolition waste. The volumes of industrial and residential waste in certain regions also tend to decrease during the winter months. Our first and fourth quarter results of operations typically reflect these seasonal trends. In addition, particularly harsh weather conditions may result in the temporary suspension of certain of our operations.
- f) For those items included in the determination of income from operations, the accounting policies of our segments are generally the same as those described in the summary of significant accounting policies included in our December 31, 2002 Form 10-K except as discussed in Note 1 included herein.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The table below shows the total revenues contributed by the principal lines of business within NASW (in millions).

	Three Mon June		Six Months Ended June 30,		
	2003	2002	2003	2002	
NASW:					
Collection	\$1,953	\$1,916	\$ 3,822	\$3,739	
Landfill	690	694	1,279	1,289	
Transfer	407	375	746	691	
Wheelabrator	203	193	402	375	
Recycling and other(a)	220	161	424	295	
Intercompany(b)	(558)	(514)	(1,042)	(963)	
Operating revenues	\$2,915	\$2,825	\$ 5,631	\$5,426	

- a) In addition to the revenue generated by our Recycling Group, we have included revenues generated within our five geographic operating Groups derived from recycling, methane gas operations, sweeping services and Port-o-let® services in the recycling and other line of business.
- b) Intercompany revenues between lines of business are eliminated within the condensed consolidated financial statements included herein.

12. Condensed Consolidating Financial Statements

WM Holdings, which is 100% owned by WMI, has fully and unconditionally guaranteed all of WMI's senior indebtedness, as well as WMI's 4% convertible subordinated notes that matured and were repaid in February 2002. WMI has fully and unconditionally guaranteed the 5.75% convertible subordinated debentures due 2005 issued by WM Holdings. However, none of WMI's other subsidiaries guaranteed any of WMI's or WM Holdings' debt. As a result of these guarantee arrangements, we are required to present the following condensed consolidating financial information (in millions).

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

CONDENSED CONSOLIDATING BALANCE SHEETS

June 30, 2003

(Unaudited)

ASSETS

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Current assets:					
Cash and cash equivalents	\$ 809	\$ —	\$ (56)	\$ —	\$ 753
Other current assets	_	_	2,433	_	2,433
	809	_	2,377	_	3,186
Property and equipment, net	_	_	10,812	_	10,812
Intercompany and investment in subsidiaries	9,358	5,246	(6,705)	(7,899)	_
Other assets	33	137	6,288	_	6,458
		—			
Total assets	\$10,200	\$5,383	\$12,772	\$(7,899)	\$20,456
	_	_	_	_	_
	LIABILITIES AND ST	OCKHOLDERS' EQUITY	<i>I</i>		
Current liabilities:					
Current portion of long-term debt	\$ —	\$ 7	\$ 101	\$ —	\$ 108
Accounts payable and other accrued liabilities	72	46	2,909	_	3,027
	72	53	3,010	_	3,135
Long-term debt, less current portion	4,526	1,973	1,851	_	8,350
Other liabilities	18	2	3,320	_	3,340
Total liabilities	4,616	2,028	8,181	_	14,825
Minority interest in subsidiaries	· —	_	47	_	47
Stockholders' equity	5,584	3,355	4,544	(7,899)	5,584
Total liabilities and stockholders' equity	\$10,200	\$5,383	\$12,772	\$(7,899)	\$20,456

December 31, 2002

ASSETS

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Current assets:					
Cash and cash equivalents	\$ 316	\$ —	\$ (52)	\$ —	\$ 264
Other current assets	_	4	2,432	_	2,436
	316	4	2,380	_	2,700
Property and equipment, net	_	_	10,612	_	10,612
Intercompany and investment in subsidiaries	9,484	5,694	(7,277)	(7,901)	_
Other assets	57	123	6,139	_	6,319
Total assets	\$9,857	\$5,821	\$11,854	\$(7,901)	\$19,631
	_	_	_		_
	LIABILITIES AND ST	OCKHOLDERS' EQUITY	7		
Current liabilities:					
Current portion of long-term debt	\$ —	\$ 112	\$ 119	\$ —	\$ 231
Accounts payable and other accrued liabilities	73	32	2,837	_	2,942
	73	144	2,956	_	3,173
Long-term debt, less current portion	4,476	1,863	1,723	_	8,062
Other liabilities	_	_	3,069	_	3,069
Total liabilities	4,549	2,007	7,748	_	14,304
Minority interest in subsidiaries	· —	_	19	_	19
Stockholders' equity	5,308	3,814	4,087	(7,901)	5,308
Total liabilities and stockholders' equity	\$9,857	\$5,821	\$11,854	\$(7,901)	\$19,631
	_	_			

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS

Three Months Ended June 30, 2003

(Unaudited)

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Operating revenues	\$ —	\$ —	\$2,915	\$ —	\$2,915
Costs and expenses	_	_	2,530		2,530
Income from operations	_	_	385	_	385
Other income (expense):					
Interest income (expense), net	(59)	(33)	(14)	_	(106)
Equity in subsidiaries, net of taxes	213	234	_	(447)	_
Minority interest	_	_	(2)	_	(2)
Other, net	_	1	4	_	5
	154	202	(12)	(447)	(103)
Income before income taxes	154	202	373	(447)	282
Provision for (benefit from) income taxes	(22)	(11)	139	_	106
Net income	\$176	\$213	\$ 234	\$(447)	\$ 176
	_				

Three Months Ended June 30, 2002

(Unaudited)

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Operating revenues	\$ —	\$ —	\$2,825	\$ —	\$2,825
Costs and expenses	_	_	2,361	_	2,361
Income from operations	_	_	464	_	464
Other income (expense):					
Interest income (expense), net	(58)	(39)	(14)	_	(111)
Equity in subsidiaries, net of taxes	254	278	_	(532)	_
Minority interest	_	_	(2)	_	(2)
Other, net	_	1			1
	196	240	(16)	(532)	(112)
	_				
Income before income taxes	196	240	448	(532)	352
Provision for (benefit from) income taxes	(21)	(14)	170	_	135
Net income	\$217	\$254	\$ 278	\$(532)	\$ 217

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS

Six Months Ended June 30, 2003

(Unaudited)

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Operating revenues	\$ —	\$ —	\$5,631	\$ —	\$5,631
Costs and expenses	_	_	4,967	_	4,967
Income from operations			664		664
Other income (expense):					
Interest income (expense), net	(119)	(66)	(28)	_	(213)
Equity in subsidiaries, net of taxes	312	353	_	(665)	_
Minority interest	_	_	(3)	_	(3)
Other, net	_	2	7	_	9
	193	289	(24)	(665)	(207)
Income before income taxes	193	289	640	(665)	457
Provision for (benefit from) income taxes	(44)	(23)	241	_	174
Income before cumulative effect of changes in accounting					
principles	237	312	399	(665)	283
Cumulative effect of changes in accounting principles		_	(46)	_	(46)
Net income	\$ 237	\$312	\$ 353	\$(665)	\$ 237
	_	_	_	_	_

Six Months Ended June 30, 2002

(Unaudited)

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Operating revenues	\$ —	\$ —	\$5,434	\$ —	\$5,434
Costs and expenses	_	_	4,638	_	4,638
Income from operations	_	_	796	_	796
Other income (expense):					
Interest income (expense), net	(114)	(79)	(32)	_	(225)
Equity in subsidiaries, net of taxes	427	476	_	(903)	_
Minority interest	_	_	(3)	_	(3)
Other, net	_	1	2	_	3
				—	
	313	398	(33)	(903)	(225)
				—	
Income before income taxes	313	398	763	(903)	571
Provision for (benefit from) income taxes	(42)	(29)	289	_	218
Income before cumulative effect of change in accounting					
principle	355	427	474	(903)	353
Cumulative effect of change in accounting principle	_	_	2	_	2
	_				
Net income	\$ 355	\$427	\$ 476	\$(903)	\$ 355
	_	_	_	_	_

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS

Six Months Ended June 30, 2003

(Unaudited)

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Cash flows from operating activities:					
Net income	\$ 237	\$ 312	\$ 353	\$(665)	\$ 237
Equity in earnings of subsidiaries, net of taxes	(312)	(353)	_	665	_
Other adjustments and charges	9	3	824	_	836
Net cash provided by (used in) operating activities	(66)	(38)	1,177	_	1,073
Cash flows from investing activities:					
Acquisitions of businesses, net of cash acquired	_	_	(172)	_	(172)
Capital expenditures	_	_	(501)	_	(501)
Proceeds from divestitures of businesses, net of cash					
divested, and other sales of assets	_	_	34	_	34
Net receipts from restricted funds and other	_	_	109	_	109
Other	_	_	18	_	18
Net cash used in investing activities	_	_	(512)	_	(512)
Cash flows from financing activities:					
New borrowings	_	_	67	_	67
Debt repayments	_	_	(68)	_	(68)
Common stock repurchases	(71)	_	_	_	(71)
Exercise of common stock options and warrants	9	_	_	_	9
Other	(3)	_	(6)	_	(9)
(Increase) decrease in intercompany and investments,					
net	624	38	(662)	_	_
Net cash provided by (used in) financing activities	559 ——	38	(669)	_	(72)
Increase (decrease) in cash and cash equivalents	493	_	(4)	_	489
Cash and cash equivalents at beginning of period	316	_	(52)	_	264
Cash and cash equivalents at end of period	\$ 809	\$ <u> </u>	\$ (56)	\$ <u> </u>	\$ 753

Six Months Ended June 30, 2003

(Unaudited)

	WMI	WM Holdings	Non-Guarantor Subsidiaries	Eliminations	Consolidated
Cash flows from operating activities:					
Net income	\$ 355	\$ 427	\$ 476	\$(903)	\$ 355
Equity in earnings of subsidiaries, net of taxes	(427)	(476)	_	903	_
Other adjustments and charges	(12)	(10)	551	_	529
Net cash provided by (used in) operating activities	(84)	(59)	1,027	_	884
Cash flows from investing activities:					
Acquisitions of businesses, net of cash acquired	_	_	(64)	_	(64)
Capital expenditures	_	_	(552)	_	(552)
Proceeds from divestitures of businesses, net of cash					
divested, and other sales of assets	_	_	71	_	71
Net receipts from restricted funds	_	_	103	_	103
Other	_	_	(7)	_	(7)
Net cash used in investing activities	_	_	(449)	_	(449)
Cash flows from financing activities:					
New borrowings	498	_	_	_	498

Debt repayments	(427)	(24)	(17)	_	(468)
Common stock repurchases	(500)		_	_	(500)
Exercise of common stock options and warrants	23	_	_	_	23
(Increase) decrease in intercompany and investments,					
net	525	83	(608)	_	_
Net cash provided by (used in) financing activities	119	59	(625)	_	(447)
Effect of exchange rate changes on cash and cash					
equivalents	_	_	2	_	2
Increase (decrease) in cash and cash equivalents	35	_	(45)	_	(10)
Cash and cash equivalents at beginning of period	757	_	(27)	_	730
Cash and cash equivalents at end of period	\$ 792	\$ —	\$ (72)	\$ —	\$ 720

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

13. New Accounting Pronouncement

FIN 46

In January 2003, the Financial Accounting Standards Board issued FASB Interpretation No. 46, *Consolidation of Variable Interest Entities* ("FIN 46"). FIN 46 requires variable interest entities to be consolidated by their primary beneficiaries. A primary beneficiary is the party that absorbs a majority of the entity's expected losses or residual benefits. FIN 46 is currently effective for all variable interest entities created or modified after January 31, 2003 and is effective for all variable interest entities, regardless of when created, in quarterly periods beginning after June 15, 2003. Management has determined that a limited number of the Company's arrangements are with variable interest entities and may require consolidation and/or disclosure as of July 1, 2003. However, we have yet to definitively conclude whether consolidation or disclosure will be required upon full adoption of FIN 46 or to measure the impact that this accounting change will have on our results of operations or financial position. Upon full adoption, we will disclose the nature, purpose, size and activities of the variable interest entities in which we are either the primary beneficiary or have significant variable interests. We will also disclose our maximum exposure to loss as a result of our involvement with these entities.

14. Subsequent Event

On August 5, 2003, we announced that our Board of Directors approved a quarterly dividend program beginning in 2004. The dividend will be \$0.75 per share annually and we expect to announce the record and payment dates of the first quarter dividend in January 2004, which we expect will be payable in March 2004.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

When we make statements containing projections about our accounting and finances, about our plans, programs and objectives for the future, about our future economic performance or statements containing any other projections or estimates about our assumptions relating to these types of statements, we are making forward-looking statements. The statements usually relate to future events and anticipated revenues, earnings or other aspects of our operations or operating results. We make these statements in an effort to keep stockholders and the public informed about our business, and have based them on our current expectations about future events. You should view such statements with caution. These statements are not guarantees of future performance or events. All phases of our business are subject to uncertainties, risks and other influences, many over which we have no control. Any of these factors, either alone or taken together, could have a material adverse effect on us and could change whether any forward-looking statement ultimately turns out to be true. Additionally, we assume no obligation to update any forward-looking statements as a result of future events or developments.

Outlined below are some of the risks that we face and that could affect our business and financial statements for the remainder of 2003 and beyond. However, they are not the only risks that we face. There may be additional risks that we do not presently know or that we currently believe are immaterial which could also impair our business.

- possible changes in our estimates of site remediation requirements, final closure and post-closure obligations, compliance and regulatory developments;
- the possible impact of regulations on our business, including the cost to comply with regulatory requirements and the potential liabilities associated with disposal operations, as well as our ability to obtain and maintain permits needed to operate our facilities;
- the effect of limitations or bans on disposal or transportation of out-of-state waste or certain categories of waste;
- possible charges against earnings as a result of shut-down operations, uncompleted acquisitions, development or expansion projects or other events;
- the effects that trends toward requiring recycling, waste reduction at the source and prohibiting the disposal of certain types of wastes could have on volumes of waste going to landfills and waste-to-energy facilities;
- the effect the weather has on our quarter to quarter results, as well as the effect of extremely harsh weather on our operations;
- the effect that price fluctuations on commodity prices may have on our operating revenues;
- the outcome of litigation or threatened litigation;
- the effect competition in our industry could have on our profitability;
- possible diversions of management's attention and increases in operating expenses due to efforts by labor unions to organize our employees;
- $\bullet \ possible \ increases \ in \ operating \ expenses \ due \ to \ fuel \ price \ increases \ or \ fuel \ supply \ shortages;$
- the effects of general economic conditions, including the ability of insurers to fully or timely meet their contractual commitments and of surety companies to continue to issue surety bonds;
- the need for additional capital if cash flows are less than we expect or capital expenditures are more than we expect, and the possibility that we cannot obtain additional capital on acceptable terms if needed;
- ullet possible errors or problems upon implementation of new information technology systems; and
- possible disruptions due to the implementation of our workforce reductions.

General

Our principal executive offices are located at 1001 Fannin Street, Suite 4000, Houston, Texas 77002. Our telephone number at that address is (713) 512-6200. Our website address is http://www.wm.com. Our annual reports on Form 10-K, quarterly reports on Form 10-Q and current reports on Form 8-K are all available, free of charge, on our website as soon as practicable after we file the reports with the SEC. Our stock is traded on the New York Stock Exchange under the symbol "WMI."

We are the waste industry's leading provider of integrated waste services in North America. Through our subsidiaries we provide collection, transfer, recycling and resource recovery, and disposal services. We are also a leading developer, operator and owner of waste-to-energy facilities in the United States. Our customers include commercial, industrial, municipal and residential customers, other waste management companies and governmental entities.

Our collection services involve picking up and transporting waste from where it was generated to a transfer station or disposal site. Transfer stations are facilities located near residential and commercial collection routes where collection trucks take the solid waste that has been collected. The solid waste is then transferred via transfer trucks, rail or barge to disposal sites. These disposal sites include landfills and waste-to-energy facilities. As of June 30, 2003, we owned or operated 284 solid waste landfills, five hazardous waste landfills, 16 waste-to-energy facilities and eight independent power production plants ("IPPs"). The operation of a solid waste landfill includes excavation, construction of liners and final caps, continuous spreading and compacting of waste, and covering of waste with earth or other inert materials. These operations are carefully planned to maintain sanitary conditions, to ensure the best possible use of the airspace and to prepare the site so it can ultimately be used for other purposes. We also operate secure hazardous waste landfills in the United States. Our hazardous waste landfills are sited, constructed and operated in a manner designed to provide long-term containment of hazardous waste. At waste-to-energy facilities, solid waste is burned at high temperatures in specially designed boilers, producing heat that is converted into high-pressure steam. We use that steam to generate electricity for sale to electric utilities under long-term contracts. Our IPPs also convert various waste and conventional fuels into electricity. The IPPs combust wood waste, anthracite coal waste (culm), tires, landfill gas and natural gas. In addition to electricity production, the IPPs also produce steam, which is sold to industrial and commercial users.

We actively pursue landfill gas-to-energy projects and are currently supplying methane gas from several of our solid waste landfills. The processed gas is delivered to engine generators to produce electricity that is sold to public utilities, municipal utilities or power cooperatives, delivered via pipeline to industrial customers as a direct substitute for fossil fuels in industrial processes or processed and delivered to natural gas suppliers.

In addition to disposing of waste, we offer recycling services, which involve the removal of reusable materials from the waste stream for processing and resale or other disposition for use in various applications. These services are mainly offered through our Recycling Group, which is composed of our majority-owned recycling entity, Recycle America Alliance, L.L.C. During the first quarter of 2003, we transferred the majority of the recycling assets and businesses of our subsidiaries to Recycle America Alliance to optimize the capacity and improve the profitability of our recycling operations.

We also rent and service portable restroom facilities under the name Port-o-let®, provide street and parking lot sweeping services and provide in-plant services, in which we outsource our employees to provide full service waste management to customers at their plants. Our vertically integrated waste management operations allow us to provide these customers with full management of their waste, including choosing the right-sized containers, finding recycling opportunities, minimizing their waste and transporting and disposing of their waste.

Our North American Solid Waste operations are comprised of our seven Groups that divide our operations geographically into the Eastern, Midwest, Southern, Western and Canadian operations and functionally into Recycling and Wheelabrator. Our Other NASW services include our national accounts and methane gas recovery operations that provide services throughout the Groups. Each of the Groups within NASW is reported as a separate segment, as is Other NASW. We also had international waste management

services and non-solid waste services, all of which were divested by March 31, 2002. The impact of these divested operations is included in our presentations in this report as "Other."

Critical Accounting Estimates and Assumptions

In preparing our financial statements, we make several estimates and assumptions that affect our assets and liabilities and revenues and expenses. We must make these estimates and assumptions because certain of the information that is used in the preparation of our financial statements is dependent on future events, cannot be calculated with a high degree of precision from available data or is simply not capable of being readily calculated based on generally accepted methodologies. In some cases, these estimates are particularly difficult to determine and we must exercise significant judgment. The most difficult, subjective and complex estimates and the assumptions that deal with the greatest amount of uncertainty are related to our accounting for landfills, environmental liabilities and asset impairments, as described in Item 7 of our Annual Report on Form 10-K for the year ended December 31, 2002.

Effective January 1, 2003, we adopted SFAS No. 143, *Accounting for Asset Retirement Obligations*. Although the adoption of SFAS No. 143 has impacted our landfill accounting policies, the estimates and assumptions that we used as of December 31, 2002 and that are inherent in the application of these policies continue to be a relevant component of our accounting for landfill retirement obligations.

As a result of adopting SFAS No. 143, additional landfill accounting related estimates and assumptions are required in determining the obligations associated with retiring landfill assets. The cost estimates for final capping, closure and post-closure activities at landfills for which we have responsibility are estimated based on our interpretations of current requirements and proposed or anticipated regulatory changes. We also estimate additional costs, pursuant to the requirements of SFAS No. 143, based on the amount a third party would charge us to perform such activities even when we expect to perform these activities internally. Additionally, we estimate the airspace to be consumed related to each final capping event and the timing of each final capping event and post-closure activities. Because landfill final capping, closure and post-closure obligations are measured at estimated fair value using present value techniques, changes in the estimated future timing of landfill final capping and closure activities would have an effect on these liabilities, related assets and results of operations.

Results of Operations for the Three and Six Months Ended June 30, 2003 and 2002

The following table presents, for the periods indicated, the period to period change in dollars (in millions) and percentages for the respective consolidated statement of operations line items.

	Period to Period Change For the Three Months Ended June 30, 2003 and 2002		Period to Period For the Six Mont June 30, 2003 and 20	hs Ended
Statement of Operations:				
Operating revenues	\$ 90	3.2 %	\$ 197	3.6 %
Costs and expenses:				
Operating (exclusive of depreciation and				
amortization shown below)	155	8.9	383	11.6
Selling, general and administrative	(15)	(4.7)	(79)	(11.2)
Depreciation and amortization	12	3.8	20	3.3
Restructuring	23	N/A	6	16.2
Asset impairments and unusual items	(6)	N/A	(1)	(16.7)
	169	7.2	329	7.1
Income from operations	(79)	(17.0)	(132)	(16.6)
Other income (expense):				
Interest expense	6	5.2	15	6.4
Interest and other income, net	3	50.0	3	25.0
Minority interest	_	_	_	_
	—			
	9	8.0	18	8.0
Income before income taxes	(70)	(19.9)	(114)	(20.0)
Provision for income taxes	(29)	(21.5)	(44)	(20.2)
Income before cumulative effect of changes in				
accounting principles	\$ (41)	(18.9)%	\$ (70)	(19.8)%

The following table presents, for the periods indicated, the percentage relationship that the respective consolidated statement of operations line items bear to operating revenues:

	Three Months Ended June 30,		Six Months June 3	
	2003	2002	2003	2002
Statement of Operations:				
Operating revenues	100.0 %	100.0 %	100.0 %	100.0 %
Costs and expenses:				
Operating (exclusive of depreciation				
and amortization shown below)	64.7	61.3	65.3	60.7
Selling, general and administrative	10.3	11.2	11.1	12.9
Depreciation and amortization	11.2	11.1	11.1	11.2
Restructuring	8.0	_	0.8	0.7
Asset impairments and unusual items	(0.2)	_	(0.1)	(0.1)
	86.8	83.6	88.2	85.4
Income from operations	13.2	16.4	11.8	14.6
Other income (expense):				
Interest expense	(3.7)	(4.1)	(3.9)	(4.3)
Interest and other income, net	0.3	0.3	0.3	0.2
Minority interest	(0.1)	(0.1)	(0.1)	_
	(3.5)	(3.9)	(3.7)	(4.1)
Income before income taxes	9.7	12.5	8.1	10.5
Provision for income taxes	3.7	4.8	3.1	4.0
Income before cumulative effect of				
changes in accounting principles	6.0 %	7.7 %	5.0 %	6.5 %
01 1				

Operating Revenues

Our operating revenues for the three months ended June 30, 2003 were \$2.9 billion, compared to \$2.8 billion in 2002. For the six months ended June 30, 2003, our operating revenues were \$5.6 billion, as compared to \$5.4 billion in 2002. As shown below (in millions), NASW is our principal operation, which is comprised of seven operating Groups within North America, along with our Other NASW services. Our "Other" operations consisted of international waste management services and non-solid waste services, all of which were divested as of March 31, 2002.

		Three Months Ended June 30,		ns Ended 30,
	2003	2002	2003	2002
Canadian	\$ 150	\$ 137	\$ 267	\$ 251
Eastern	981	958	1,843	1,818
Midwest	569	577	1,071	1,078
Southern	758	755	1,497	1,480
Western	625	624	1,227	1,213
Wheelabrator	203	193	402	375
Recycling	139	74	272	132
Other NASW	48	21	94	42
Intercompany	(558)	(514)	(1,042)	(963)
Total NASW	2,915	2,825	5,631	5,426
Other	_	<u> </u>	_	8
Net operating revenues	\$2,915	\$2,825	\$ 5,631	\$5,434
	_	_	_	_

Our NASW operating revenues generally come from fees charged for our collection, disposal and transfer station services. Some of the fees we charge to our customers for collection services are billed in advance; a liability for future service is recorded when we bill the customer and operating revenues are recognized as services are actually provided. Revenues from our disposal operations consist of tipping fees charged to third parties based on the volume and type of waste being disposed of at our disposal facilities and are normally billed monthly or semi-monthly. Fees charged at transfer stations are based on the volume of waste deposited, taking into account our cost of loading, transporting, and disposing of the solid waste at a disposal site. Recycling revenue, which is generated by our Recycling Group as well as our five geographic operating Groups, generally consists of the sale of recyclable commodities to third parties and tipping fees. Intercompany revenues between our operations have been eliminated in the consolidated financial statements.

The mix of NASW operating revenues from our different services is reflected in the table below (in millions).

		nths Ended e 30,	Six Month June	
	2003	2002	2003	2002
Collection	\$1,953	\$1,916	\$ 3,822	\$3,739
Landfill	690	694	1,279	1,289
Transfer	407	375	746	691
Wheelabrator	203	193	402	375
Recycling and other	220	161	424	295
Intercompany	(558)	(514)	(1,042)	(963)
Total NASW	\$2,915	\$2,825	\$ 5,631	\$5,426
	_	_	_	_

We experienced positive internal revenue growth for the three and six months ended June 30, 2003, as compared with the corresponding prior year periods. The following table provides details associated with the period to period change in NASW revenues (dollars in millions) along with an explanation of the significant components of the current period changes.

	Change Three Mont	Period to Period Change for the Three Months Ended June 30, 2003 and 2002		Period for the is Ended 3 and 2002
Price:				
Base business(a)	\$ 17	0.6 %	\$ 24	0.4 %
Commodity(b)	5	0.2	27	0.5
Electricity(c)	2	0.1	1	_
Fuel(d)	11	0.4	25	0.5
	_			
Total price	35	1.3	77	1.4
Volume(e)	(11)	(0.4)	3	0.1
	_	_		_
Internal growth	24	0.9	80	1.5
Acquisitions(f)	70	2.5	136	2.5
Divestitures	(17)	(0.6)	(29)	(0.5)
Foreign currency translation(g)	13	0.4	18	0.3
	_			
	\$ 90	3.2 %	\$205	3.8 %
	_		_	

- a) The increases in our base business revenues during the three and six months ended June 30, 2003 are attributable to our collection, transfer and waste-to-energy businesses with a significant increase in our residential collection operations. However, for our landfill business we experienced decreases in price-related revenue due to our special waste and municipal solid waste landfill operations, even after considering a significant price-related revenue increase at our landfill operations in Pennsylvania for a new \$4 per ton landfill tax imposed by the state. The decrease in price-related revenue from special waste landfill operations is due principally to an increase in lower priced event work in the southern and western portions of the United States. The lower price-related revenue from municipal solid waste landfill operations was driven in part by increased competition and the mix of volumes received at certain landfills, particularly in the western portion of the United States.
- b) Increases in commodity prices resulted in additional revenues of approximately \$5 million for the three months ended June 30, 2003, with the majority of this increase attributable to the prices of old newsprint ("ONP"). Commodity prices played an even greater role in our revenue increase for the six months ended June 30, 2003, increasing revenue by approximately \$27 million. Specifically, (i) the prices of old corrugated cardboard ("OCC") increased to an average of \$64 per ton for the six months ended June 30, 2003 as compared to an average of \$58 per ton for the corresponding prior year period and (ii) the prices of ONP increased to an average of \$68 per ton for the six months ended June 30, 2003 as compared to an average of \$53 per ton for the corresponding prior year period.
- c) We experienced positive price increases due to increased electricity rates at our IPP facilities.
- d) We experienced positive price increases due to increased fuel surcharges.
- e) The volume-related decline in revenue during the three months ended June 30, 2003 is primarily attributable to the loss of a contract with the City of Chicago that generated \$19 million of revenue during the corresponding prior year period. Losing this contract negatively impacted revenues by \$27 million for the six-month period ended June 30, 2003. However, favorably impacting our revenue during these periods were volume-related revenue increases in special waste activity within our landfill line of business in the southern and western portions of the United States. Additionally, our waste-to-energy facilities contributed to the overall volume-related revenue increase for the six month period ended June 30, 2003.
- f) Business acquisitions positively impacted our revenues for both the three and six months ended June 30, 2003. The most significant acquisition consummated during this period was associated with the formation of Recycle America Alliance, which resulted in our acquisition of the Peltz Group, the largest privately-held recycler in the United States, in January 2003.
- g) Fluctuations in the relative value of the Canadian dollar favorably impacted revenues for both the three and six months ended June 30, 2003.

Operating Costs and Expenses (Exclusive of Depreciation and Amortization Shown Below)

Our operating costs and expenses include direct and indirect labor and related taxes and benefits, risk management costs, fuel, facility operating costs, maintenance and repairs of equipment and facilities, tipping fees paid to third party disposal facilities and transfer stations, and accretion of and expense revisions relating

to future landfill final capping, closure and post-closure costs and environmental remediation. Certain direct landfill development expenditures are capitalized and amortized over the estimated useful life of a site as capacity is consumed, and include acquisition, engineering, grading, construction, capitalized interest, and permitting costs.

Operating costs and expenses increased \$155 million, or 8.9%, for the three months ended June 30, 2003 as compared to the corresponding prior year period. As a percentage of operating revenues, operating costs and expenses increased 3.4 percentage points, to 64.7%, in the same year-over-year comparison. For the six months ended June 30, 2003, operating costs and expenses were \$383 million higher than the prior year period, representing an increase of 11.6%. As a percentage of operating revenue, operating costs and expenses were 65.3% for the six months ended June 30, 2003, which is a 4.6 percentage point increase from the prior year period.

The increase in operating costs and expenses for the three months ended June 30, 2003 as compared to the previous year was generally due to:

- Increased cost of goods sold of approximately \$54 million, largely related to our acquisition of the Peltz Group in January 2003
- Increased disposal costs of approximately \$31 million, partially attributable to the redirection of waste as a result of landfill constraints in the eastern portion of the United States. The increase was also attributable to increased taxes of approximately \$9 million on disposal of waste in Pennsylvania, although approximately 90% of the tax increases have been passed on to our customers as price increases;
- Increases in subcontractor costs of approximately \$11 million, due to the redirection of waste as a result of landfill constraints in the eastern portion of the United States, increases in subcontracted transportation costs due to the current period increase in special waste activity within our landfill line of business and increased use of subcontractors for our national accounts in areas where we do not provide services;
- Salary and wage increases of approximately \$15 million, for annual merit raises and increased overtime expenses, partially offset by headcount reductions;
- Increased risk management costs of approximately \$10 million, associated with our insurance and claims expenditures;
- Increased fuel costs of approximately \$0.18 per gallon, or approximately \$10 million, for the quarter; and
- The effect of the strengthening Canadian dollar on foreign currency translation, which resulted in an \$11 million increase in operating expenses in our Canadian operations.

Increased costs from business acquisitions accounted for approximately \$60 million of the increases discussed above.

Operating costs and expenses were also higher for the six months ended June 30, 2003, as compared to the same period of the prior year, due to:

- The Company's reorganization of its operations in March 2002. During our 2003 planning processes we determined that certain employee costs and facility-related expenses were more appropriately classified as operating expenses after the adoption of the new organizational structure and have reclassified these costs beginning in the second quarter of 2002, as this was the first full accounting period that these organizational changes were effective. During the first quarter of 2003, approximately \$50 million of such costs have been classified as operating expenses that would have been reported as selling, general and administrative costs prior to the reorganization;
- Increased cost of goods sold of approximately \$110 million, largely related to our acquisition of the Peltz Group in January 2003;

- Increased disposal costs of approximately \$62 million, partially attributable to the redirection of waste as a result of landfill constraints in the eastern portion of the United States. The increase was also attributable to increased taxes of approximately \$17 million on disposal of waste in Pennsylvania, although approximately 87% of the tax increases have been passed on to our customers as price increases;
- Increases in subcontractor costs of approximately \$19 million, due to the redirection of waste as a result of landfill constraints in the eastern portion of the United States, increases in subcontracted transportation costs due to the current period increase in special waste activity within our landfill line of business and increased use of subcontractors for our national accounts in areas where we do not provide services;
- Salary and wage increases of approximately \$32 million, for annual merit raises and increased overtime expenses, partially offset by headcount reductions;
- An increase in fuel costs of approximately \$35 million, an average of \$0.30 per gallon;
- The effect of strengthening of the Canadian dollar on foreign currency translation, which resulted in a \$15 million increase in operating expenses for our Canadian operations;
- Increased maintenance costs of approximately \$15 million, due mostly to changes in the timing and scope of certain maintenance projects at our waste-to-energy facilities;
- Increased landfill and environmental costs of approximately \$18 million, due to the increase in the inflation rate applicable to future costs of environmental remedial reserves from 2.0% to 2.5%, and due to increases in costs principally associated with accretion expense on landfill asset retirement obligations, related to our adoption of SFAS No. 143 in 2003; and
- Increased risk management expenses of approximately \$24 million, due primarily to the increased year-over-year average cost per claim;

Business acquisitions accounted for approximately \$115 million of the increases discussed above.

Selling, General and Administrative

Our selling, general and administrative expenses include management salaries and benefits; clerical and administrative costs; marketing costs; professional services; provision for doubtful accounts; and administrative-related costs including telecommunications, travel and insurance costs; as well as costs related to our sales force and customer service.

Selling, general and administrative expenses decreased \$15 million, or 4.7%, for the three months ended June 30, 2003, as compared to the prior year period. As a percentage of operating revenues, our selling, general and administrative expenses decreased from 11.2% for the three months ended June 30, 2002 to 10.3% for the three months ended June 30, 2003. This decrease was due principally to the favorable settlement of a legal dispute for \$11 million as well as a reduction of nearly \$3 million in costs due to management's focus on reducing spending related to professional fees, supplies, travel and entertainment, and other administrative costs. This decrease was offset partially by increased bad debt expense of over \$4 million attributable largely to the weaker economy. The salary and wage expense reductions resulting from restructurings were largely offset by increases in such expenses associated with annual merit increases and businesses acquired.

For the six months ended June 30, 2003, selling, general and administrative expenses decreased by \$79 million, or 11.2%, as compared to the prior year period. As a percentage of revenue, the decrease was 1.8 percentage points from 12.9% for the six months ended June 30, 2002 to 11.1% for the six months ended June 30, 2003. In addition to the second quarter reductions, we experienced a decrease of approximately \$50 million for the six months ended June 30, 2003 when compared with the same period of the prior year as a result of the re-characterization of certain costs as discussed above in *Operating Costs and Expenses*. Those costs related principally to employee costs and facility-related expenses, including property taxes, utilities, and risk management expenses, that are reflected as operating costs since April 1, 2002. The decrease was also a result of our March 2002 reorganization and February 2003 restructuring, which allowed us to reduce expenses

for salaries and other labor-related costs. Further, selling, general and administrative costs declined in the six months ended June 30, 2003 compared to the prior year period by \$18 million due to management's focus on reducing spending related to professional fees, supplies, travel and entertainment, and other administrative costs, but was offset by \$8 million due to increased bad debt expenses.

Depreciation and Amortization

Depreciation and amortization includes (i) amortization of intangible assets with a definite life, other than goodwill, on a straight-line basis over the definitive terms of the related agreements (generally from 3 to 7 years); (ii) depreciation of property and equipment on a straight-line basis from 3 to 50 years; (iii) amortization of landfill costs, including those incurred and all estimated future costs for closure and post-closure, on a units-of-consumption method as landfill airspace is consumed over the estimated remaining capacity of a site; and (iv) as a result of our adoption of SFAS No. 143, amortization of landfill asset retirement costs arising from final capping obligations on a units-of-consumption method as airspace is consumed over the estimated capacity associated with each final capping event.

Depreciation and amortization expense increased by \$12 million and \$20 million during the three and six months ended June 30, 2003, respectively, compared to the corresponding prior year periods. This increase is primarily attributable to the accounting changes that resulted from our adoption of SFAS No. 143. Depreciation and amortization expense as a percentage of operating revenues was 11.2% and 11.1% for the three and six months ended June 30, 2003, respectively, and 11.1% and 11.2% for the three and six months ended June 30, 2002, respectively.

Restructuring

In 2002, we reorganized the structure of the Company into market areas to align collection, transport, recycling and disposal resources. As part of the restructuring, we reduced the number of field layers of management and eliminated approximately 1,900 field-level administrative and operational positions. In the first quarter of 2002, we recorded \$37 million of pre-tax charges for costs associated with the implementation of the new structure. These charges included \$34 million for employee severance and benefit costs and \$3 million related to abandoned operating lease agreements. No additional costs were recorded during the second quarter of 2002.

In February 2003, we reduced the number of market areas that make up our geographic operating Groups to 66 from a total of 91 at December 31, 2002, and reduced certain overhead positions to further streamline our organization. Management believes that this structure results in a more effective utilization of resources and enables us to serve our customers more efficiently. In connection with these restructuring efforts, there was a workforce reduction of about 700 employees and 270 contract workers. In the first quarter of 2003, we recorded \$20 million of pre-tax charges for costs associated with the implementation of the new structure, all of which was associated with employee severance and benefit costs. No material costs for this restructuring and workforce reduction have been incurred in the second quarter of 2003.

The operational efficiencies provided by these organizational changes and a focus on fully utilizing the capabilities of our information technology resources enabled us to further reduce our workforce in June 2003. This workforce reduction resulted in the elimination of 600 employee positions and 200 contract worker positions. In the second quarter of 2003, we recorded \$23 million of pre-tax charges for employee severance and benefit costs associated with this workforce reduction.

Asset Impairments and Unusual Items

During the six months ended June 30, 2003, we recorded a net credit of \$7 million to asset impairments and unusual items primarily as a result of gains recognized on divestitures of certain small operations that were offset, in part, by asset impairment charges. Approximately \$6 million of this net activity occurred during the second quarter of 2003.

During the first quarter of 2002 we recorded a net credit of \$6 million to asset impairments and unusual items primarily due to a reversal of a loss contract reserve and adjustments for revisions of estimated losses on assets held-for-sale that were partially offset by asset impairment charges. We incurred no such costs during the second quarter of 2002.

Interest Expense

Interest expense decreased by \$6 million and \$15 million for the three and six months ended June 30, 2003, respectively, when compared to the corresponding prior year periods. The decrease in interest expense is partially attributable to our increased utilization of tax-exempt financing, which has resulted in a decline in our weighted average interest rate. The remaining decrease in interest expense between 2003 and 2002 is attributable to the overall positive impact of our interest rate derivative transactions, including the amortization of the fair value adjustments to debt for the early termination of many interest rate swap agreements. Interest rate derivative contracts reduced interest expense by \$23 million and \$47 million for the three and six months ended June 30, 2003, respectively, and \$23 million and \$43 million for the three and six months ended June 30, 2002, respectively.

Provision for Income Taxes

We recorded a provision for income taxes of \$106 and \$174 million for the three and six months ended June 30, 2003, respectively, and \$135 million and \$218 million for the corresponding periods in 2002. The difference in federal income taxes computed at the federal statutory rate and reported income taxes for the three and six months ended June 30, 2003 and June 30, 2002 is primarily due to state and local income taxes, offset in part by non-conventional fuel tax credits.

Cumulative Effect of Changes in Accounting Principle

In the first quarter of 2003, we recorded a charge of \$46 million, net of taxes, to cumulative effect of changes in accounting principles for the adoption of certain accounting changes described below.

- Through December 31, 2002, we accrued in advance for major repairs and maintenance expenditures and deferred costs associated with annual plant outages at our waste-to-energy facilities and independent power production plants. Effective January 1, 2003, we changed our policy from this method to one that expenses these costs as they are incurred. We recorded approximately \$25 million, net of taxes, or \$0.04 per diluted share, as a credit to cumulative effect of changes in accounting principles.
- Through December 31, 2002, we accrued for future losses under customer contracts that over the contract life were projected to have direct costs greater than revenues. Effective January 1, 2003, we changed our policy from this method to one that expenses these losses as they are incurred. We recorded approximately \$30 million, net of taxes, or \$0.05 per diluted share, as a credit to cumulative effect of changes in accounting principles.
- In connection with the adoption of SFAS No. 143, we recorded approximately \$101 million, including tax benefit, or \$0.17 per diluted share, in the first quarter of 2003 as a charge to cumulative effect of changes in accounting principles. Substantially all of this charge was related to the impact of changes in accounting for landfill final capping, closure and post-closure costs.

In the first quarter of 2002, we recorded a credit of \$2 million, net of tax, to cumulative effect of change in accounting principle to write-off the aggregate amount of negative goodwill as a result of adopting SFAS No. 141, *Accounting for Business Combinations*.

Income From Operations by Reportable Segment

As discussed in Note 11 to the condensed consolidated financial statements, we manage and evaluate our operations primarily through our Eastern, Midwest, Southern, Western, Canadian, Wheelabrator and Recycling Groups. The operations not managed through our seven operating Groups are presented herein as "Other NASW." We also had international waste management services and non-solid waste services, all of

which were divested by March 31, 2002. However, we do continue to incur minimal administrative expenses in connection with these divested operations. These operations are included in the table below as "Other."

As discussed in Note 9 to the condensed consolidated financial statements, our restructuring and workforce reductions have impacted the operating results of our reportable segments in both 2002 and 2003. The following table summarizes income from operations by reportable segment for the three and six months ended June 30, 2003 and June 30, 2002 and provides explanations of other significant factors contributing to the identified variances.

	Three Months Ended June 30,			Six Months Ended June 30,		
	2003	2002	Change	2003	2002	Change
Canadian	\$ 17	\$ 18	\$ (1)	\$ 24	\$ 30	\$ (6)
Eastern	91	132	(41)(a)	149	239	(90)(a)
Midwest	84	97	(13)(b)	142	166	(24)(b)
Southern	141	149	(8)	281	287	(6)
Western	83	96	(13)(c)	167	171	(4)
Wheelabrator	71	62	9	113	100	13
Recycling	(1)	4	(5)	(1)	(5)	4
Other NASW	(9)	(6)	(3)	(6)	(12)	6
Total NASW	477	552	(75)	869	976	(107)
Other	_	_		(2)	(2)	` <u>_</u>
Corporate	(92)	(88)	(4)	(203)	(178)	(25)(d)
•	<u> </u>			<u>`</u>		
Total	\$385	\$464	\$(79)	\$ 664	\$ 796	\$(132)
	_	_		_	_	

- a) The Eastern Group was significantly impacted by harsh winter weather during the first quarter of 2003. Increased operating costs, including higher disposal costs, fuel costs, landfill operating costs and repairs and maintenance also contributed to the Group's lower operating income in both the first and second quarters of 2003.
- b) Harsh winter weather conditions in the first quarter of 2003, higher fuel and disposal costs and lower base business pricing negatively affected the Midwest Group's results of operations during 2003 when compared to the corresponding prior year period.
- c) Increased operating costs including higher disposal costs, repairs and maintenance and fuel costs negatively affected the Western Group's income from operations during the three months ended June 30, 2003.
- d) The increase in costs in the Corporate Group is due to higher environmental remediation expenses for closed sites, which in part relates to an increase in the inflation rate used in connection with inflating future cash flows to estimate the present value of remedial liabilities, and the transfer of printing and postage costs to the centralized service center related to certain customer billing activities that were previously included in the results of operations of the Eastern, Midwest, Southern and Western Groups.

Liquidity and Capital Resources

The following is a summary of our cash balances and cash flows for the six months ended June 30, 2003 and 2002 (in millions):

	Six Month June	
	2003	2002
Cash and cash equivalents at the end of the period	\$ 753	\$ 720
	_	
Cash provided by operating activities	\$1,073	\$ 884
Cash used in investing activities	\$ (512)	\$(449)
Cash used in financing activities	\$ (72)	\$(447)
	_	

We generated cash flows from operations of approximately \$1.1 billion for the six months ended June 30, 2003, which includes approximately \$109 million received for the fair value of interest rate swap agreements terminated prior to their scheduled maturities. We used \$512 million for investing activities, consisting of \$501 million for capital expenditures and \$172 million for the acquisition of businesses, net of cash acquired, which was offset by net receipts of \$109 million from restricted funds and proceeds from divestitures of businesses, net of cash divested, and other asset sales of \$34 million. Additionally, we used \$72 million for financing activities, which consisted primarily of the repurchase of approximately 3.6 million shares of our common stock for \$71 million.

We generated cash flows from operations of \$884 million for the six months ended June 30, 2002. During that period, we spent \$449 million for investing activities including capital expenditures of \$552 million and acquisitions of businesses, net of cash acquired, of \$64 million, which were offset by net receipts of \$103 million from restricted funds and proceeds from sales of assets and cash generated from other investing activities of \$64 million. In addition, we used \$447 million for financing activities, which included \$500 million spent for our stock buy back program, offset by \$30 million of net debt borrowings and \$23 million from exercises of common stock options and warrants.

We operate in a capital intensive business and continuing access to various financing sources is vital to our operations. In the past, we have been successful in obtaining financing from a variety of sources on terms we consider attractive. Based on several key factors we believe are considered by credit rating agencies and financial markets to be important in determining our future access to financing, we expect to continue to maintain access to capital sources in the future. These factors include:

- the essential nature of the services we provide and our large and diverse customer base;
- · our ability to generate strong and consistent cash flows despite the economic environment;
- our liquidity profile;
- · our asset base; and
- our commitment to maintaining a moderate financial profile and disciplined capital allocation.

In addition to our working capital needs for ongoing operations, we have capital requirements for (i) capital expenditures for construction and expansion of landfill sites, as well as new trucks and equipment for collection and other operations, (ii) refurbishments and improvements at waste-to-energy facilities and (iii) business acquisitions. We currently expect to spend approximately \$700 million for capital expenditures during the final two quarters of 2003.

In February 2002 we announced that our Board of Directors had approved a stock repurchase program for up to \$1 billion in annual repurchases for each year through 2004, to be implemented at management's

discretion. We expect to utilize cash flows from operations for purchases that will be made in open market or privately negotiated transactions.

The following is a summary of 2003 activity for our stock repurchase program (in millions, except shares in thousands and price per share in dollars).

	Agree	ement		Common Stock	Total Purchase	Settlement (Received)/	Net Common Stock
Transaction Type	Initiating Date	Settlement Date	Shares	Price per Share	Price	Paid	Repurchases
Private Accelerated Purchase(a)	December 2002	February 2003	1,731	\$ 24.52	\$ 42	\$ (3)(b)	\$ 39
Private Accelerated Purchase(a) Open Market Purchases(d)	March 2003 N/A	May 2003 N/A	2,400 1,163	\$ 20.00 \$19.70 - \$20.77	\$ 48 \$ 23	\$ 3 (c) N/A	\$ 51 \$ 23

- a) We accounted for the initial payments as a purchase of treasury stock and classified the future settlements with the counterparty as an equity instrument because we have the option under these agreements to settle our obligations, if any, in shares of our common stock.
- b) The weighted average daily market price of our stock during the valuation period times the number of shares we purchased was approximately \$3 million less than the approximately \$42 million we initially paid. Pursuant to the agreement, the counterparty paid us the difference of approximately \$3 million at the end of the valuation period to settle the agreement.
- c) The weighted average daily market price of our stock during the valuation period times the number of shares we purchased was approximately \$3 million more than the approximately \$48 million we initially paid. Pursuant to the agreement, we paid the counterparty the difference of approximately \$3 million at the end of the valuation period to settle the agreement.
- d) We engaged in open market purchases during the first quarter of 2003 when trading was allowed pursuant to law and our insider trading policy. We did not engage in open market purchases during the second quarter of 2003.

We expect a net cash outflow, after considering insurance, tax deductions and related settlement costs, of approximately \$230 to \$240 million for settlement of our securities class action lawsuit. The settlement fund began to accrue interest, at the Federal Funds rate, during the second quarter of 2002, although payment will not become due until the judgment is final. That judgment will not be final until an appeal of the approval of the settlement filed in March 2003, is decided. We currently expect the payment to be made before the end of 2003.

Our strategy is to primarily utilize cash flows from operations to meet our capital needs and contractual obligations. However, we also have bank borrowings available to meet our capital needs and contractual obligations and, when appropriate, will obtain financing by issuing debt or common stock.

As of June 30, 2003, we had a three-year, \$650 million syndicated revolving credit facility and a five-year, \$1.75 billion syndicated revolving credit facility. The three-year revolver matures in June 2005 and the five-year revolver matures in June 2006. At June 30, 2003, no borrowings were outstanding under our revolving credit facilities and we had unused and available credit capacity under these facilities of approximately \$870 million. The unused and available capacity under the facilities was approximately \$770 million at December 31, 2002.

As part of our operations, and in connection with issuances of tax-exempt bonds, we use letters of credit to support our bonding and funding needs. In order to increase our letter of credit availability, on June 30, 2003 we entered into a five-year, \$15 million letter of credit and term loan agreement, a seven-year, \$175 million letter of credit and term loan agreement, and a ten-year, \$105 million letter of credit and term loan agreement, which expire in June 2008, 2010, and 2013, respectively (collectively, the "LC and term loan agreements"). At June 30, 2003, letters of credit were issued and outstanding for the full \$295 million of credit capacity under these agreements.

As of June 30, 2003, we had letters of credit in the aggregate amount of approximately \$1.93 billion (of which approximately \$1.53 billion are issued under the revolving credit facilities, \$295 million are issued under the LC and term loan agreements and the remainder are issued under other various lines of credit). These letters of credit generally have terms allowing for automatic renewal after one year. In the event of an

unreimbursed draw on a letter of credit, we have the ability to convert that amount into a term loan for the remaining term under its respective agreement or facility.

As of June 30, 2003, we are required to maintain the following financial covenants under our revolving credit facilities: (i) an interest coverage ratio; (ii) total debt to EBITDA; and (iii) minimum net worth, all as defined in the credit facilities solely for the purpose of determining compliance with the covenants. The interest coverage ratio requires that at the end of any fiscal quarter we will not permit the ratio of (A) our consolidated net income plus interest expense and income taxes ("EBIT") for the four fiscal quarters then ending to (B) consolidated total interest expense for such period, to be less than 3 to 1. The total debt to EBITDA covenant requires that at the end of any fiscal quarter, we will not permit the ratio of (A) all indebtedness and certain contingent liabilities such as financial guarantees to (B) EBIT plus depreciation and amortization expense ("EBITDA") for the four fiscal quarters then ending to exceed 3.25 to 1. Our minimum net worth covenant restricts us from allowing the sum of stockholders' equity to be less than \$3.5 billion plus 75% of our cumulative consolidated net income for each fiscal quarter, beginning with the first fiscal quarter ended March 31, 2001. We are in compliance with all covenants under our revolving credit facilities and all other debt instruments.

Additionally, we have issued industrial revenue bonds primarily for the construction of various facilities. Proceeds from these financing arrangements are directly deposited into trust funds and we do not have the ability to utilize the funds in regular operating activities. Accordingly, we report these amounts as an investing activity when the cash is released from the trust funds and a financing activity when the industrial revenue bonds are repaid. At June 30, 2003, approximately \$425 million of funds were held in trust to meet future capital expenditures at various facilities. These fund balances are included as other long-term assets in the accompanying condensed consolidated balance sheets.

As of June 30, 2003, we have \$434 million of 6.375% senior notes due December 1, 2003, \$150 million of 8.0% senior notes due April 30, 2004 and \$200 million of 6.5% senior notes due May 15, 2004. We classified all of these borrowings as long-term at June 30, 2003. The classification of these obligations as long-term was based upon our available capacity under our two long-term revolving credit facilities and our intent to refinance the borrowings with other long-term financings. However, in the event other sources of long-term financing are not available, we intend to use our revolving credit facilities.

We manage the interest rate risk of our debt portfolio principally by using interest rate derivatives to achieve a desired position of fixed and floating rate debt, which was approximately 67% fixed and 33% floating at June 30, 2003. In addition, we periodically enter into derivative transactions to secure the then-current market interest rate in anticipation of senior debt issuances.

Accounting for Stock Options

We account for our stock-based compensation using the intrinsic value method prescribed by APB Opinion No. 25, *Accounting for Stock Issued to Employees*, as amended. Pursuant to APB No. 25, we recognize no compensation cost for our stock option grants because the number of shares potentially issuable and the exercise price, which is equal to the fair market value of the underlying stock on the date of grant, is fixed.

Were the Company to recognize compensation expense based on the fair value of stock options granted, as provided for under SFAS No. 123, *Accounting for Stock Based Compensation*, we would have recognized compensation expense of \$18 million, or \$0.03 per diluted share, for the three months ended June 30, 2003, and \$22 million, or \$0.04 per diluted share, for the three months ended June 30, 2002. The compensation expense for the six month periods ended June 30, 2003 and June 30, 2002 would have been \$33 million, or \$0.06 per diluted share, and \$40 million, or \$0.07 per diluted share, respectively.

Seasonal Trends and Inflation

Our operating revenues tend to be somewhat lower in the winter months, primarily due to the lower volume of construction and demolition waste. The volumes of industrial and residential waste in certain

regions where we operate also tend to decrease during the winter months. Our first and fourth quarter results of operations typically reflect this seasonality. We also use the slower winter months for scheduled maintenance at our waste-to-energy facilities, so repair and maintenance expense is generally higher in our first quarter than in other quarters during the year. In addition, particularly harsh weather conditions may result in the temporary suspension of certain of our operations.

We believe that inflation has not had, and is not expected to have, any material adverse effect on our results of operations in the near future. However, management's estimates associated with inflation have had, and will continue to have, an impact on our accounting for landfill and environmental liabilities.

Recent Development

On August 5, 2003, we announced that our Board of Directors approved a quarterly dividend program beginning in 2004. The dividend will be \$0.75 per share annually and we expect to announce the record and payment dates of the first quarter dividend in January 2004, which we expect will be payable in March 2004.

New Accounting Pronouncement

FIN 46

In January 2003, the Financial Accounting Standards Board issued FASB Interpretation No. 46, *Consolidation of Variable Interest Entities* ("FIN 46"). FIN 46 requires variable interest entities to be consolidated by their primary beneficiaries. A primary beneficiary is the party that absorbs a majority of the entity's expected losses or residual benefits. FIN 46 is currently effective for all variable interest entities created or modified after January 31, 2003 and is effective for all variable interest entities, regardless of when created, in quarterly periods beginning after June 15, 2003. Management has determined that a limited number of the Company's arrangements are with variable interest entities and may require consolidation and/or disclosure as of July 1, 2003. However, we have yet to definitively conclude whether consolidation or disclosure will be required upon full adoption of FIN 46 or to measure the impact that this accounting change will have on our results of operations or financial position. Upon full adoption, we will disclose the nature, purpose, size and activities of the variable interest entities in which we are either the primary beneficiary or have significant variable interests. We will also disclose our maximum exposure to loss as a result of our involvement with these entities.

Item 4. Controls and Procedures.

Within the 90 days prior to the date of this Form 10-Q, an evaluation was performed under the supervision and with the participation of our management, including the Chief Executive Officer, Chief Financial Officer and Chief Accounting Officer, of the effectiveness of the design and operation of our disclosure controls and procedures. Based on that evaluation, our management, including the Chief Executive Officer, Chief Financial Officer and Chief Accounting Officer, concluded that our disclosure controls and procedures were effective. There have been no significant changes in our internal controls and procedures or in other factors that could significantly affect internal controls subsequent to the date the Company carried out its evaluation.

PART II

Item 1. Legal Proceedings

Information regarding our legal proceedings can be found under the "Litigation" section of Note 8, *Commitments and Contingencies*, to the condensed consolidated financial statements.

Item 4. Submission of Matters to a Vote of Security Holders

At our 2003 Annual Meeting of Stockholders held on May 16, 2003, a proposal to elect the nominees listed in the following table as directors of the Company was submitted to a vote of our stockholders. The following table also shows the results of voting as to each nominee:

	For	Withheld
Pastora San Juan Cafferty	504,017,386	23,717,879
Frank M. Clark, Jr.	504,112,884	23,622,381
Robert S. Miller	304,990,847	222,744,418
A. Maurice Myers	502,151,389	25,583,876
John C. Pope	503,879,507	23,855,758
W. Robert Reum	504,041,189	23,694,076
Steven G. Rothmeier	504,002,513	23,732,752
Carl W. Vogt	346,865,620	180,869,646
Ralph V. Whitworth	503,396,734	24,338,531

At the same meeting, the following proposals were also adopted by our stockholders. The voting was as follows:

	For	Against	Withheld
Ratification of appointment of Ernst & Young LLP as our independent auditors for the fiscal year			
ending December 31, 2003	516,514,272	6,405,937	4,815,056
Amendment to the Company's Employee Stock			
Purchase Plan	510,187,933	12,140,036	5,405,820

The following proposal was submitted to our stockholders, but was not approved. The vote was as follows:

	For	Against	Withheld
Require disclosure of management's plan to oppose privatization	25,119,108	383,873,987	64,625,267

Item 5. Other Information

None.

Item 6. Exhibits and Reports on Form 8-K

(a) Exhibits:

Exhibit Number	_	Description
10.1	_	Employment Agreement between Recycle America Alliance, L.L.C. and Steve Ragiel, dated March 30, 2003.
10.2	_	Ten-Year Letter of Credit and Term Loan Agreement among the Registrant, Waste Management Holdings, Inc., and Bank of America, N.A., as Administrative Agent and Letter of Credit Issuer and the Lenders party thereto, dated as of June 30, 2003.
10.3	_	Five-Year Letter of Credit and Term Loan Agreement among the Registrant, Waste Management Holdings, Inc., and Bank of America, N.A., as Administrative Agent and Letter of Credit Issuer and the Lenders party thereto, dated as of June 30, 2003.
10.4	_	Seven-Year Letter of Credit and Term Loan Agreement among the Registrant, Waste Management Holdings, Inc., and Bank of America, N.A., as Administrative Agent and Letter of Credit Issuer and the Lenders party thereto, dated as of June 30, 2003.
10.5	_	2003 Waste Management, Inc. Directors' Deferred Compensation Plan.
12	_	Computation of Ratio of Earnings to Fixed Charges.
31.1	_	Certification Pursuant to Rule 15d — 14(a) under the Securities Exchange Act of 1934, as amended, of A. Maurice Myers, President and Chief Executive Officer.
31.2	_	Certification Pursuant to Rule $15d - 14(a)$ under the Securities Exchange Act of 1934 , as amended, of David P. Steiner, Executive Vice President and Chief Financial Officer.

Exhibit Number		Description
32.1 32.2	- - -	Certification Pursuant to 18 U.S.C. §1350 of A. Maurice Myers, President and Chief Executive Officer. Certification Pursuant to 18 U.S.C. §1350 of David P. Steiner, Executive Vice President and Chief Financial Officer.
(b) Reports on Form 8-K:		
None.		
		50

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

WASTE MANAGEMENT, INC.

By:

/s/ DAVID P. STEINER

David P. Steiner

Executive Vice President and
Chief Financial Officer
(Principal Financial Officer)

WASTE MANAGEMENT, INC.

By: /s/ ROBERT G. SIMPSON

Robert G. Simpson Senior Vice President and Chief Accounting Officer (Principal Accounting Officer)

Date: August 5, 2003

INDEX TO EXHIBITS

Exhibit Number	_	Description
10.1	_	Employment Agreement between Recycle America Alliance, L.L.C. and Steve Ragiel, dated March 30, 2003.
10.2	_	Ten-Year Letter of Credit and Term Loan Agreement among the Registrant, Waste Management Holdings, Inc., and Bank of America, N.A., as Administrative Agent and Letter of Credit Issuer and the Lenders party thereto, dated as of June 30, 2003.
10.3	_	Five-Year Letter of Credit and Term Loan Agreement among the Registrant, Waste Management Holdings, Inc., and Bank of America, N.A., as Administrative Agent and Letter of Credit Issuer and the Lenders party thereto, dated as of June 30, 2003.
10.4	_	Seven-Year Letter of Credit and Term Loan Agreement among the Registrant, Waste Management Holdings, Inc., and Bank of America, N.A., as Administrative Agent and Letter of Credit Issuer and the Lenders party thereto, dated as of June 30, 2003.
10.5	_	2003 Waste Management, Inc. Directors' Deferred Compensation Plan.
12	_	Computation of Ratio of Earnings to Fixed Charges.
31.1	_	Certification Pursuant to Rule 15d — 14(a) under the Securities Exchange Act of 1934, as amended, of A. Maurice Myers, President and Chief Executive Officer.
31.2	_	Certification Pursuant to Rule 15d — 14(a) under the Securities Exchange Act of 1934, as amended, of David P. Steiner, Executive Vice President and Chief Financial Officer.
32.1	_	Certification Pursuant to 18 U.S.C. §1350 of A. Maurice Myers, President and Chief Executive Officer.
32.2	_	Certification Pursuant to 18 U.S.C. §1350 of David P. Steiner, Executive Vice President and Chief Financial Officer.

EMPLOYMENT AGREEMENT

This EMPLOYMENT AGREEMENT (the "Agreement") is made and entered into on this 30th day of March, 2003 by and between Recycle America Alliance, L.L.C. and Steve Ragiel (the "Executive"). Recycle America Alliance, L.L.C. is an indirect subsidiary of Waste Management, Inc. ("WMI"). Recycle America Alliance, L.L.C. and all of its affiliates shall be referred to herein as "the Company."

1. Employment and Termination of Previous Employment Agreement.

The Company shall employ Executive, and Executive shall be employed by the Company upon the terms and subject to the conditions set forth in this Agreement.

Executive acknowledges and represents that the certain Employment Agreement between he and WMI dated on or about October 14, 1998 is terminated, and that any and all obligations of WMI created thereunder, whether express or implied, are null and void and of no further force or effect, and that the only rights, obligations, and duties between the Company and Executive are those expressly set forth in this Agreement.

2. Term of Employment.

The period of Executive's employment under this Agreement shall commence on January 1, 2003 ("Employment Date"), and shall continue for a period of two (2) years thereafter, and shall automatically be renewed for successive one (1) year periods thereafter, unless Executive's employment is terminated in accordance with Section 5 below. The period during which Executive is employed hereunder shall be referred to as the "Employment Period."

3. Duties and Responsibilities.

- (a) Executive shall serve as President of Recycle America Alliance, L.L.C. In such capacity, Executive shall perform such duties and have the power, authority, and functions as set forth in the Operating Agreement of the Company, in the policies and procedures of WMI (which shall be applicable to the Company), as otherwise set forth by the Board of Managers of the Company, and as may be assigned to Executive from time to time by the Chief Executive Officer, Chief Administrative Officer or Chief Operating Officer of WMI (in any such case, hereinafter referred to as the "Chief Executive Officer").
- **(b)** Executive shall devote substantially all of his working time, attention and energies to the business of the Company, and its affiliated entities. Executive may make and manage his personal investments (provided such investments in other activities do not violate the provisions of Section 8 of this Agreement), be involved in charitable and professional activities, and, with the prior written consent of the Chief Executive Officer, serve on boards of other for profit entities, provided such activities do not materially interfere with the performance of his duties hereunder.

4. Compensation and Benefits.

- (a) Base Salary. During the Employment Period, the Company shall pay Executive a base salary at the annual rate of Two Hundred Seventy-Five Thousand and 00/100ths Dollars (\$275,000.00) per year, or such higher rate as may be determined from time to time by the Chief Executive Officer ("Base Salary"). Such Base Salary shall be paid in accordance with the Company's standard payroll practice for its executive officers. Once increased, Base Salary shall not be reduced.
- **(b) Annual Bonus.** During the Employment Period, Executive will be entitled to participate in an annual incentive compensation plan of the Company. The Executive's target annual bonus will be seventy-five percent (75%) of his Base Salary in effect for such year (the "Target Bonus"), and his actual annual bonus may range from 0% to 150% (two times Target Bonus), and will be determined based upon (i) the achievement of certain WMI and Company performance goals, as may be established and approved by from time to time by the Compensation Committee of the WMI Board of Directors, and (ii) the achievement of personal performance goals as may be established by the Chief Executive Officer.
- (c) Stock Options. Executive shall be eligible to be considered for stock option grants under Waste Management, Inc.'s annual stock option award program as administered by, and at the discretion of, the Compensation Committee of the WMI Board of Directors.
- (d) Benefit Plans and Vacation. Subject to the terms of such plans, Executive shall be eligible to participate in or receive benefits under any long-term incentive plan, pension plan, profit sharing plan, salary deferral plan, medical and dental benefits plan, life insurance plan, short-term and long-term disability plans, or any other health, welfare or fringe benefit plan, generally made available to similarly-situated WMI executive employees. Neither WMI nor the Company shall be obligated to institute, maintain, or refrain from changing, amending, or discontinuing any benefit plan, or perquisite, so long as such changes are similarly applicable to similarly-situated employees generally.

During the Employment Period, Executive shall be entitled to vacation each year in accordance with the Company's policies in effect from time to time, but in no event less than four (4) weeks paid vacation per calendar year.

- **(e) Expense Reimbursement.** The Company shall promptly reimburse Executive for the ordinary and necessary business expenses incurred by Executive in the performance of the duties hereunder in accordance with the Company's customary practices applicable to its executive officers.
- **(f) Other Perquisites.** Executive shall be entitled to all perquisites provided to Senior Vice Presidents of WMI as approved by the Chief Executive Officer or the Compensation Committee of the WMI Board of Directors, and as they may exist from time to time.

5. Termination of Employment.

Executive's employment hereunder may be terminated during the Employment Period

under the following circumstances:

- (a) **Death.** Executive's employment hereunder shall terminate upon Executive's death.
- **(b) Total Disability.** The Company may terminate Executive's employment hereunder upon Executive becoming "Totally Disabled." For purposes of this Agreement, Executive shall be considered "Totally Disabled" if Executive has been physically or mentally incapacitated so as to render Executive incapable of performing the essential functions of Executive's position with or without reasonable accommodation. Executive's receipt of disability benefits under the Company's long-term disability plan or receipt of Social Security disability benefits shall be deemed conclusive evidence of Total Disability for purpose of this Agreement; provided, however, that in the absence of Executive's receipt of such long-term disability benefits or Social Security benefits, the Board of Managers may, in its reasonable discretion (but based upon appropriate medical evidence), determine that Executive is Totally Disabled.
- **(c) Termination by the Company for Cause.** The Company may terminate Executive's employment hereunder for "Cause" at any time after providing a Notice of Termination for Cause to Executive.
 - (i) For purposes of this Agreement, the term "Cause" means any of the following: (A) willful or deliberate and continual refusal to perform Executive's employment duties reasonably requested by the Company or the Chief Executive Officer after receipt of written notice to Executive of such failure to perform, specifying such failure (other than as a result of Executive's sickness, illness or injury) and Executive fails to cure such nonperformance within ten (10) days of receipt of said written notice; (B) breach of any statutory or common law duty of loyalty to the Company; (C) has been convicted of, or pleaded *nolo contendre* to, any felony; (D) willfully or intentionally caused material injury to the Company, its property, or its assets; (E) disclosed to unauthorized person(s) proprietary or confidential information of the Company; or (F) breach of any of the covenants set forth in Section 8 hereof.
 - (ii) For purposes of this Agreement, the phrase "Notice of Termination for Cause" shall mean a written notice that shall indicate the specific termination provision in Section 5(c)(i) relied upon, and shall set forth in reasonable detail the facts and circumstances which provide the basis for termination for Cause. Further, a Notification of Termination for Cause shall be required to include a copy of a resolution duly adopted by at least a majority of the entire membership of the Company's Board of Managers at a meeting of the Board of Managers which was called for the purpose of considering such employment termination, and at which Executive and his representative had the right to attend and address the Board, finding that, in the good faith belief of the Board, Executive engaged in conduct set forth in Section 5(c)(i) herein and specifying the particulars thereof in reasonable detail. The date of termination for Cause shall be the date indicated in the Notice of Termination for Cause.

- **(d) Voluntary Termination by Executive.** Executive may terminate his employment hereunder with or without Good Reason at any time upon written notice to the Company.
 - A termination for "Good Reason" means a resignation of employment by Executive by written notice ("Notice of Termination for Good Reason") (i) given to Chief Executive Officer within ninety (90) days after the occurrence of the Good Reason event, unless such circumstances are substantially corrected prior to the date of termination specified in the Notice of Termination for Good Reason. For purposes of this Agreement, "Good Reason" shall mean the occurrence or failure to cause the occurrence, as the case may be, without Executive's express written consent, of any of the following circumstances: (A) the Company substantially changes Executive's core duties or removes Executive's responsibility for those core duties, so as to effectively cause Executive to no longer be performing the duties of his position (except in each case in connection with the termination of Executive's employment for Cause or Total Disability or as a result of Executive's death, or temporarily as a result of Executive's illness or other absence), provided that the change in the geographic area of Executive's responsibility or the change in reporting structure shall not constitute Good Reason under any circumstances; further provided that if the Company becomes a fifty percent or more subsidiary of any other entity, or merges with, transfers assets to, or the business is otherwise transferred to a third party (which may be owned in whole or in part, directly or indirectly by WMI), Executive shall be deemed to have a substantial change in the core duties of his position unless he is the equivalent of a Senior Vice-President of WMI or such other successor entity and retains the core duties and responsibilities of his existing position; (B) other than as contemplated by (a) above, removal or the non-reelection of the Executive from the officer position with the Company specified herein, or removal of the Executive from any of his then officer positions; (C) any material breach by the Company of any provision of this Agreement, including without limitation Section 10 hereof; or (D) failure of any successor to the Company or to WMI (whether direct or indirect and whether by merger, acquisition, consolidation or otherwise) to assume in a writing delivered to Executive upon the assignee becoming such, the obligations of the Company hereunder.
 - (ii) A "Notice of Termination for Good Reason" shall mean a notice that shall indicate the specific termination provision relied upon and shall set forth in reasonable detail the facts and circumstances claimed to provide a basis for Termination for Good Reason. The Notice of Termination for Good Reason shall provide for a date of termination not less than ten (10) nor more than sixty (60) days after the date such Notice of Termination for Good Reason is given, provided that in the case of the events set forth in Sections 5(d)(i)(A) or (B), the date may be five (5) business days after the giving of such notice.
- **(e) Termination by the Company without Cause.** The Company may terminate Executive's employment hereunder without Cause at any time upon written notice to Executive.

(f) Effect of Termination. Upon any termination of employment for any reason, Executive shall immediately resign from all Board memberships and other positions with the Company, or any of its parent corporations or subsidiaries held by him at such time.

6. Compensation Following Termination of Employment.

In the event that Executive's employment hereunder is terminated, Executive shall be entitled to the following compensation and benefits upon such termination:

- (a) **Termination by Reason of Death.** In the event that Executive's employment is terminated by reason of Executive's death, the Company shall pay the following amounts to Executive's beneficiary or estate:
 - (i) Any accrued but unpaid Base Salary for services rendered to the date of death, any accrued but unpaid expenses required to be reimbursed under this Agreement, any accrued but unused vacation, any earned but unpaid bonuses for any prior period, and, to the extent not otherwise paid, a pro-rata bonus or incentive compensation payment to the extent payments are awarded to senior executives of WMI and paid at the same time as senior executives are paid.
 - (ii) Any benefits to which Executive may be entitled pursuant to the plans, policies and arrangements (including those referred to in Section 4(d) hereof), as determined and paid in accordance with the terms of such plans, policies and arrangements.
 - (iii) An amount equal to the Base Salary (at the rate in effect as of the date of Executive's death) which would have been payable to Executive if Executive had continued in employment for two additional years. Said payments will be paid to Executive's estate or beneficiary at the same time and in the same manner as such compensation would have been paid if Executive had remained in active employment.
 - (iv) As of the date of termination by reason of Executive's death, stock options previously awarded to Executive as of the date of death shall be fully vested, and Executive's estate or beneficiary shall have up to one (1) year from the date of death to exercise all such previously-awarded options, provided that in no event will any option be exercisable beyond its term. No stock options contemplated by this Agreement, but not yet awarded to Executive as of the time of his death, shall be granted.
- **(b) Termination by Reason of Total Disability.** In the event that Executive's employment is terminated by reason of Executive's Total Disability as determined in accordance with Section 5(b), the Company shall pay the following amounts to Executive:
 - (i) Any accrued but unpaid Base Salary for services rendered to the date of termination, any accrued but unpaid expenses required to be reimbursed under

this Agreement, any accrued but unused vacation, and any earned but unpaid bonuses for any prior period. Executive shall also be eligible for a prorata bonus or incentive compensation payment to the extent such awards are made to senior executives of WMI for the year in which Executive is terminated, and to the extent not otherwise paid to the Executive.

- (ii) Any benefits to which Executive may be entitled pursuant to the plans, policies and arrangements (including those referred to in Section 4(d) hereof) shall be determined and paid in accordance with the terms of such plans, policies and arrangements.
- (iii) An amount equal to the Base Salary (at the rate in effect as of the date of Executive's Total Disability) which would have been payable to Executive if Executive had continued in active employment for two years following termination of employment, less any payments under any long-term disability plan or arrangement paid for by the Company. Payment shall be made at the same time and in the same manner as such compensation would have been paid if Executive had remained in active employment until the end of such period.
- (iv) As of the date of termination by reason of Executive's Total Disability, stock options previously awarded to Executive as of the date of termination shall be fully vested, and Executive or his legal guardian shall have up to one (1) year from the date of death to exercise all such previously-awarded options, provided that in no event will any option be exercisable beyond its term. No stock options contemplated by this Agreement, but not yet awarded to Executive as of the time of his employment termination, shall be granted.
- **(c) Termination for Cause.** In the event that Executive's employment is terminated by the Company for Cause, the Company shall pay the following amounts to Executive:
 - (i) Any accrued but unpaid Base Salary for services rendered to the date of termination, any accrued but unpaid expenses required to be reimbursed under this Agreement, any accrued but unused vacation, and any earned but unpaid bonuses for any prior period.
 - (ii) Any benefits to which Executive may be entitled pursuant to the plans, policies and arrangements (including those referred to in Section 4(d) hereof up to the date of termination) shall be determined and paid in accordance with the terms of such plans, policies and arrangements.
 - (iii) All options, whether vested or not vested prior to the date of such termination of employment, shall be automatically cancelled on the date of employment termination. However, it is expressly understood and agreed that Executive would have no obligation to repay or otherwise reimburse the Company for funds received as a result of Executive's having exercised any previously-vested stock options prior to his employment termination.

- **(d) Voluntary Termination by Executive.** In the event that Executive voluntarily terminates employment other than for Good Reason, the Company shall pay the following amounts to Executive:
 - (i) Any accrued but unpaid Base Salary for services rendered to the date of termination, any accrued but unpaid expenses required to be reimbursed under this Agreement, any accrued but unused vacation, and any earned but unpaid bonuses for any prior period.
 - (ii) Any benefits to which Executive may be entitled pursuant to the plans, policies and arrangements (including those referred to in Section 4(d) hereof up to the date of termination) shall be determined and paid in accordance with the terms of such plans, policies and arrangements.
 - (iii) Any stock options that have not vested prior to the date of such termination of employment shall be automatically cancelled as of that date, and Executive shall have ninety (90) days following the date of termination of employment to exercise any previously vested options; provided that in no event will any option be exercisable beyond its term. No stock options contemplated by this Agreement, but not yet awarded to Executive as of the time of his employment termination, shall be granted.
- **(e) Termination by the Company Without Cause; Termination by Executive for Good Reason.** In the event that Executive's employment is terminated by the Company for reasons other than death, Total Disability or Cause, or Executive terminates his employment for Good Reason, the Company shall pay the following amounts to Executive:
 - (i) Any accrued but unpaid Base Salary for services rendered to the date of termination, any accrued but unpaid expenses required to be reimbursed under this Agreement, any accrued but unused vacation, and any earned but unpaid bonuses for any prior period.
 - (ii) Any benefits to which Executive may be entitled pursuant to the plans, policies and arrangements referred to in Section 4(d) hereof shall be determined and paid in accordance with the terms of such plans, policies and arrangements.
 - (iii) An amount equal to two times the sum of Executive's Base Salary plus his Target Annual Bonus (in each case as then in effect), of which one-half shall be paid in a lump sum within ten (10) days after such termination and one-half shall be paid during the two (2) year period beginning on the date of Executive's termination and shall be paid at the same time and in the same manner as Base Salary would have been paid if Executive had remained in active employment until the end of such period.
 - (iv) The Company will continue for Executive and Executive's spouse and dependents, all health benefit plans, whether group or individual, disability, and other similar benefit plans, in which Executive was participating on the date of

termination, until the earliest to occur of (A) two years after the date of termination; (B) Executive's death (provided that benefits provided to Executive's spouse and dependents shall not terminate upon Executive's death); or (C) with respect to any particular plan, the date Executive becomes eligible to participate in a comparable benefit provided by a subsequent employer. The first eighteen (18) months of any medical and dental insurance coverage provided by the Company pursuant to the foregoing sentence will be provided pursuant to the provisions of COBRA, 29 U.S.C. §1161; provided, however, that Executive shall only be required to pay the normal employee premium required of similarly-situated active employees of the Company, and the Company shall pay the difference in the cost of the coverage directly to the medical and/or dental plans or their administrator. Dental and health insurance coverage for any remaining period required to be provided by the Company to Executive hereunder shall be provided by Company pursuant to private insurance purchased by the Company for Executive for which the Executive shall only be required to pay an amount equal to the premiums that Executive would be required to pay under the Company's dental and health insurance plans then in effect.

- (v) Executive shall continue to vest in all stock option awards or restricted stock awards over the two (2) year period commencing on the date of such termination. Executive shall have two (2) years and six (6) months after the date of termination to exercise all options to the extent then vested, provided that in no event may any option be exercisable beyond its term.
- **(f) No Other Benefits or Compensation.** Except as may be provided under this Agreement, under the terms of any incentive compensation, employee benefit, or fringe benefit plan applicable to Executive at the time of Executive's termination or resignation of employment, Executive shall have no right to receive any other compensation, or to participate in any other plan, arrangement or benefit, with respect to future periods after such termination or resignation.
- **(g) No Mitigation; No Set-Off.** In the event of any termination of employment hereunder, Executive shall be under no obligation to seek other employment and there shall be no offset against any amounts due Executive under this Agreement on account of any remuneration attributable to any subsequent employment that Executive may obtain. The amounts payable hereunder shall not be subject to setoff, counterclaim, recoupment, defense or other right which the Company may have against the Executive or others, except upon obtaining by the Company of a final non-appealable judgment against Executive.
 - 7. Resignation by Executive for Good Reason and Compensation Payable Following Change in Control.
- (a) Resignation for Good Reason following Change in Control. In the event a "Change in Control" occurs and Executive terminates his employment for Good Reason thereafter, or the Company terminates Executive's employment other than for Cause, or such termination for Good Reason or without Cause occurs in contemplation of such Change in Control (any termination within six (6) months prior to such Change in Control being presumed

to be in contemplation unless rebutted by clear and demonstrable evidence to the contrary), the Company shall pay the following amounts to Executive:

- (i) The payments and benefits provided for in Section 6(e), except that (A) the amount and period with respect to which severance is calculated pursuant to Section 6(e)(iii) will be three (3) years and the amount shall be paid in a lump-sum and (B) the benefit continuation period in Section 6(e)(iv) shall be for three years.
- (ii) In lieu of Section 6(e)(v), Executive will be 100% vested in all benefits, awards, and grants (including stock option grants and stock awards, all of such stock options exercisable for three (3) years following Termination, provided that in no event will any option be exercisable beyond its term) accrued but unpaid as of the date of termination under any non-qualified pension plan, supplemental and/or incentive compensation or bonus plans, in which Executive was a participant as of the date of termination. Executive shall also receive a bonus or incentive compensation payment (the "bonus payment"), payable at 100% of the maximum bonus available to Executive, pro-rated as of the effective date of the termination. The bonus payment shall be payable within five (5) days after the effective date of Executive's termination. Except as may be provided under this Section 7 or under the terms of any incentive compensation, employee benefit, or fringe benefit plan applicable to Executive at the time of Executive's termination of employment, Executive shall have no right to receive any other compensation, or to participate in any other plan, arrangement or benefit, with respect to future periods after such resignation or termination.

(b) Certain Additional Payments by the Company.

- (i) In the event that the Executive shall become entitled to payments and/or benefits provided by this Agreement or any other amounts in the "nature of compensation" (whether pursuant to the terms of this Agreement or any other plan, arrangement or agreement with the Company, any person whose actions result in a change of ownership or effective control covered by Section 280G(b)(2) of the Code or any person affiliated with the Company or such person) as a result of such change in ownership or effective control (collectively the "Company Payments"), and such Company Payments will be subject to the tax (the "Excise Tax") imposed by Section 4999 of the Code (and any similar tax that may hereafter be imposed by any taxing authority) the Company shall pay to the Executive at the time specified in subsection (iv) below an additional amount (the "Gross-up Payment") such that the net amount retained by the Executive, after deduction of any Excise Tax on the Company Payments and any U.S. federal, state, and for local income or payroll tax upon the Gross-up Payment provided for by this Section 7(b), but before deduction for any U.S. federal, state, and local income or payroll tax on the Company Payments, shall be equal to the Company Payments.
- (ii) For purposes of determining whether any of the Company Payments and Gross-up Payments (collectively the "Total Payments") will be subject to the Excise Tax

and the amount of such Excise Tax, (x) the Total Payments shall be treated as "parachute payments" within the meaning of Section 280G(b)(2) of the Code, and all "parachute payments" in excess of the "base amount" (as defined under Code Section 280G(b)(3) of the Code) shall be treated as subject to the Excise Tax, unless and except to the extent that, in the opinion of the Company's independent certified public accountants appointed prior to any change in ownership (as defined under Code Section 280G(b)(2)) or tax counsel selected by such accountants (the "Accountants") such Total Payments (in whole or in part) either do not constitute "parachute payments," represent reasonable compensation for services actually rendered within the meaning of Section 280G(b)(4) of the Code in excess of the "base amount" or are otherwise not subject to the Excise Tax, and (y) the value of any non-cash benefits or any deferred payment or benefit shall be determined by the Accountants in accordance with the principles of Section 280G of the Code.

(iii) For purposes of determining the amount of the Gross-up Payment, the Executive shall be deemed to pay U.S. federal income taxes at the highest marginal rate of U.S. federal income taxation in the calendar year in which the Gross-up Payment is to be made and state and local income taxes at the highest marginal rate of taxation in the state and locality of the Executive's residence for the calendar year in which the Company Payment is to be made, net of the maximum reduction in U.S. federal income taxes which could be obtained from deduction of such state and local taxes if paid in such year. In the event that the Excise Tax is subsequently determined by the Accountants to be less than the amount taken into account hereunder at the time the Gross-up Payment is made, the Executive shall repay to the Company, at the time that the amount of such reduction in Excise Tax is finally determined, the portion of the prior Gross-up Payment attributable to such reduction (plus the portion of the Gross-up Payment attributable to the Excise Tax and U.S. federal, state and local income tax imposed on the portion of the Gross-up Payment being repaid by the Executive if such repayment results in a reduction in Excise Tax or a U.S. federal, state and local income tax deduction), plus interest on the amount of such repayment at the rate provided in Section 1274(b)(2)(B) of the Code. Notwithstanding the foregoing, in the event any portion of the Gross-up Payment to be refunded to the Company has been paid to any U.S. federal, state and local tax authority, repayment thereof (and related amounts) shall not be required until actual refund or credit of such portion has been made to the Executive, and interest payable to the Company shall not exceed the interest received or credited to the Executive by such tax authority for the period it held such portion. The Executive and the Company shall mutually agree upon the course of action to be pursued (and the method of allocating the expense thereof) if the Executive's claim for refund or credit is denied.

In the event that the Excise Tax is later determined by the Accountant or the Internal Revenue Service to exceed the amount taken into account hereunder at the time the Gross-up Payment is made (including by reason of any payment the existence or amount of which cannot be determined at the time of the Gross-up Payment), the Company shall make an additional Gross-up Payment in respect of

such excess (plus any interest or penalties payable with respect to such excess) at the time that the amount of such excess is finally determined.

- (iv) The Gross-up Payment or portion thereof provided for in subsection (iii) above shall be paid not later than the thirtieth (30th) day following an event occurring which subjects the Executive to the Excise Tax; provided, however, that if the amount of such Gross-up Payment or portion thereof cannot be finally determined on or before such day, the Company shall pay to the Executive on such day an estimate, as determined in good faith by the Accountant, of the minimum amount of such payments and shall pay the remainder of such payments (together with interest at the rate provided in Section 1274(b)(2)(B) of the Code), subject to further payments pursuant to subsection (iii) hereof, as soon as the amount thereof can reasonably be determined, but in no event later than the ninetieth day after the occurrence of the event subjecting the Executive to the Excise Tax. In the event that the amount of the estimated payments exceeds the amount subsequently determined to have been due, such excess shall constitute a loan by the Company to the Executive, payable on the fifth day after demand by the Company (together with interest at the rate provided in Section 1274(b)(2) (B) of the Code).
- (v) In the event of any controversy with the Internal Revenue Service (or other taxing authority) with regard to the Excise Tax, the Executive shall permit the Company to control issues related to the Excise Tax (at its expense), provided that such issues do not potentially materially adversely affect the Executive, but the Executive shall control any other issues. In the event the issues are interrelated, the Executive and the Company shall in good faith cooperate so as not to jeopardize resolution of either issue, but if the parties cannot agree the Executive shall make the final determination with regard to the issues. In the event of any conference with any taxing authority as to the Excise Tax or associated income taxes, the Executive shall permit the representative of the Company to accompany the Executive, and the Executive and the Executive's representative shall cooperate with the Company and its representative.
- (vi) The Company shall be responsible for all charges of the Accountant.
- (vii) The Company and the Executive shall promptly deliver to each other copies of any written communications, and summaries of any verbal communications, with any taxing authority regarding the Excise Tax covered by this Section 7(b).
- (c) Change in Control. For purposes of this Agreement, "Change in Control" means the occurrence of any of the following events:
- (i) any Person is or becomes the Beneficial Owner, directly or indirectly, of securities of WMI (not including in the securities beneficially owned by such person any securities acquired directly from WMI or its Affiliates) representing twenty-five percent (25%) or more of the combined voting power of WMI's then outstanding voting securities;

- the following individuals cease for any reason to constitute a majority of the number of directors then serving: individuals who, on the Employment Date, constitute the Board of WMI and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest, including but not limited to a consent solicitation, relating to the election of directors of WMI) whose appointment or election by the Board or nomination for election by WMI's stockholders was approved or recommended by a vote of the at least two-thirds (2/3rds) of the directors then still in office who either were directors on the Employment Date or whose appointment, election or nomination for election was previously so approved or recommended;
- (iii) there is a consummated merger or consolidation of the Company pursuant to which no securities of the Company are owned by WMI or its direct or indirect subsidiaries, or a merger or consolidation of WMI or any direct or indirect subsidiary of WMI (other than the Company) with any other corporation, other than (A) a merger or consolidation which would result in the voting securities of WMI outstanding immediately prior thereto continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving or parent entity) more than fifty percent (50%) of the combined voting power of the voting securities of WMI or such surviving or parent entity outstanding immediately after such merger or consolidation or (B) a merger or consolidation effected to implement a recapitalization of WMI (or similar transaction) in which no Person, directly or indirectly, acquired twenty-five percent (25%) or more of the combined voting power of WMI's then outstanding securities (not including in the securities beneficially owned by such person any securities acquired directly from WMI or its Affiliates); or
- (iv) the stock holders of the Company or WMI approve a plan of complete liquidation of the Company or WMI or there is consummated an agreement for the sale or disposition by the Company or WMI of all or substantially all of the Company's or WMI's assets (or any transaction having a similar effect), other than a sale or disposition by the Company or WMI of all or substantially all of the Company's or WMI's assets to an entity, at least fifty percent (50%) of the combined voting power of the voting securities of which are owned by stockholders of the Company or WMI in substantially the same proportions as their ownership of the Company or WMI immediately prior to such sale.

For purposes of this Section 7(c), the following terms shall have the following meanings:

- (i) "Affiliate" shall mean an affiliate of WMI, as defined in Rule 12b-2 promulgated under Section 12 of the Securities Exchange Act of 1934, as amended from time to time (the "Exchange Act");
- (ii) "Beneficial Owner" shall have the meaning set forth in Rule 13d-3 under the Exchange Act;
- (iii) "Person" shall have the meaning set forth in Section 3(a)(9) of the

Exchange Act, as modified and used in Sections 13(d) and 14(d) thereof, except that such term shall not include (1) WMI, (2) a trustee or other fiduciary holding securities under an employee benefit plan of WMI, (3) an underwriter temporarily holding securities pursuant to an offering of such securities or (4) a corporation owned, directly or indirectly, by the stockholders of WMI in substantially the same proportions as their ownership of shares of Common Stock of WMI.

8. Covenants

- **(a) This Agreement**. The terms of this Agreement constitute Confidential Information, which Executive shall not disclose to anyone other than Executive's spouse, attorneys, advisors, or as required by law. Disclosure of these terms is a material breach of this Agreement and could subject Executive to disciplinary action, including without limitation, termination of employment for Cause.
- **(b) Company Property.** All written materials, records, data, and other documents prepared or possessed by Executive during Executive's employment with the Company are the Company's property. All information, ideas, concepts, improvements, discoveries, and inventions that are conceived, made, developed, or acquired by Executive individually or in conjunction with others during Executive's employment (whether during business hours and whether on the Company's premises or otherwise) which relate to the Company's business, products, or services are the Company's sole and exclusive property. All memoranda, notes, records, files, correspondence, drawings, manuals, models, specifications, computer programs, maps, and all other documents, data, or materials of any type embodying such information, ideas, concepts, improvements, discoveries, and inventions are the Company's property. At the termination of Executive's employment with the Company for any reason, Executive shall return all of the Company's documents, data, or other Company property to the Company.
- **(c) Confidential Information; Non-Disclosure.** Executive acknowledges that the business of the Company is highly competitive and that the Company has agreed to provide and immediately will provide Executive with access to "Confidential Information" relating to the business of the Company, WMI, and their respective affiliates.

For purposes of this Agreement, "Confidential Information" means and includes the Company's and WMI's confidential and/or proprietary information and/or trade secrets that have been developed or used and/or will be developed and that cannot be obtained readily by third parties from outside sources. Confidential Information includes, by way of example and without limitation, the following information regarding customers, employees, contractors, and the industry not generally known to the public; strategies, methods, books, records, and documents; technical information concerning products, equipment, services, and processes; procurement procedures and pricing techniques; the names of and other information concerning customers, investors, and business affiliates (such as contact name, service provided, pricing for that customer, type and amount of services used, credit and financial data, and/or other information relating to the Company's relationship with that customer); pricing strategies and price curves; positions, plans, and strategies for expansion or acquisitions; budgets; customer lists; research; weather data; financial and sales data; trading methodologies and terms; evaluations, opinions, and interpretations of information and data; marketing and merchandising techniques;

prospective customers' names and marks; grids and maps; electronic databases; models; specifications; computer programs; internal business records; contracts benefiting or obligating the Company; bids or proposals submitted to any third party; technologies and methods; training methods and training processes; organizational structure; personnel information, including salaries of personnel; payment amounts or rates paid to consultants or other service providers; and other such confidential or proprietary information. Information need not qualify as a trade secret to be protected as Confidential Information under this Agreement, and the authorized and controlled disclosure of Confidential Information to authorized parties by Company in the pursuit of its business will not cause the information to lose its protected status under this Agreement. Executive acknowledges that this Confidential Information constitutes a valuable, special, and unique asset used by the Company, WMI, and their respective affiliates in their businesses to obtain a competitive advantage over their competitors. Executive further acknowledges that protection of such Confidential Information against unauthorized disclosure and use is of critical importance to the Company, WMI, and their respective affiliates in maintaining their competitive position.

Executive also will have access to, or knowledge of, Confidential Information of third parties, such as actual and potential customers, suppliers, partners, joint venturers, investors, financing sources, and the like, of the Company, WMI, and their affiliates.

The Company also agrees to provide Executive with one or more of the following: access to Confidential Information, specialized training regarding the Company's and WMI's methodologies and business strategies, and/or support in the development of goodwill such as introductions, information and reimbursement of customer development expenses consistent with Company policy. The foregoing is not contingent on continued employment, but is contingent upon Executive's use of the Confidential Information access, specialized training, and goodwill support provided by Company for the exclusive benefit of the Company and upon Executive's full compliance with the restrictions on Executive's conduct provided for in this Agreement.

In addition to the requirements set forth in Section 5(c)(i), Executive agrees that Executive will not after Executive's employment with the Company, make any unauthorized disclosure of any then Confidential Information or specialized training of the Company, WMI, or their respective affiliates, or make any use thereof, except in the carrying out of his employment responsibilities hereunder. Executive also agrees to preserve and protect the confidentiality of third party Confidential Information to the same extent, and on the same basis, as the Company's Confidential Information.

(d) Unfair Competition Restrictions. Upon Executive's Employment Date, the Company agrees to and shall provide Executive with immediate access to Confidential Information. Ancillary to the rights provided to Executive following employment termination, the Company's provision of Confidential Information, specialized training, and/or goodwill support to Executive, and Executive's agreements, regarding the use of same, and in order to protect the value of the above-referenced stock options, training, goodwill support and/or the Confidential Information described above, the Company and Executive agree to the following provisions against unfair competition. Executive agrees that for a period of two (2) years following the termination of employment for any reason ("Restricted Term"), Executive will not, directly or indirectly, for Executive or for others, anywhere in the United States (including all

parishes in Louisiana) (the "Restricted Area") do the following, unless expressly authorized to do so in writing by the Chief Executive Officer:

Engage in, or assist any person, entity, or business engaged in, the selling or providing of products or services that would displace the products or services that (i) the Company or WMI are currently in the business of providing and were in the business of providing, or were planning to be in the business of providing, at the time Executive was employed with the Company, and (ii) that Executive had involvement in or received Confidential Information about in the course of employment; the foregoing is expressly understood to include, without limitation, the business of the collection, transfer, recycling and resource recovery, or disposal of solid waste, including the operation of waste-to-energy facilities and alternative energy facilities, and trading or hedging of positions (physical and financial) in any fiber or fiber-related materials, including, but not limited to, those marketed or traded by the Company

It is further agreed that during the Restricted Term, Executive cannot engage in any of the enumerated prohibited activities in the Restricted Area by means of telephone, telecommunications, satellite communications, correspondence, or other contact from outside the Restricted Area. Executive further understands that the foregoing restrictions may limit his ability to engage in certain businesses during the Restricted Term, but acknowledges that these restrictions are necessary to protect the Confidential Information the Company and WMI have provided to Executive.

A failure to comply with the foregoing restrictions will create a presumption that Executive is engaging in unfair competition. Executive agrees that this Section defining unfair competition with the Company or with WMI does not prevent Executive from using and offering the skills that Executive possessed prior to receiving access to Confidential Information, confidential training, and knowledge from the Company and WMI. This Agreement creates an advance approval process, and nothing herein is intended, or will be construed as, a general restriction against the pursuit of lawful employment in violation of any controlling state or federal laws. Executive shall be permitted to engage in activities that would otherwise be prohibited by this covenant if such activities are determined in the sole discretion of the Chief Executive Officer to be no material threat to the legitimate business interests of the Company or WMI.

(e) Non-Solicitation of Customers. For a period of two (2) years following the termination of employment for any reason, Executive will not call on, service, or solicit competing business from customers of the Company, WMI, or their respective affiliates whom Executive, within the previous twelve (12) months, (i) had or made contact with, or (ii) had access to information and files about, or induce or encourage any such customer or other source of ongoing business to stop doing business with the Company or WMI.

- **(f) Non-Solicitation of Employees.** During Executive's employment, and for a period of two (2) years following the termination of employment for any reason, Executive will not, either directly or indirectly, call on, solicit, encourage, or induce any other employee or officer of the Company, WMI, or their respective affiliates whom Executive had contact with, knowledge of, or association within the course of employment with the Company to terminate his or her employment, and will not assist any other person or entity in such a solicitation.
- **(g) Non-Disparagement.** Executive covenants and agrees that Executive shall not engage in any pattern of conduct that involves the making or publishing of written or oral statements or remarks (including, without limitation, the repetition or distribution of derogatory rumors, allegations, negative reports or comments) which are disparaging, deleterious or damaging to the integrity, reputation or good will of the Company or WMI, their respective management, or of management of corporations affiliated with the Company or WMI.

9. Enforcement of Covenants.

- (a) Termination of Employment and Forfeiture of Compensation. Executive agrees that any breach by Executive of any of the covenants set forth in Section 8 hereof during Executive's employment by the Company, shall be grounds for immediate dismissal of Executive for Cause pursuant to Section 5(c)(i), which shall be in addition to and not exclusive of any and all other rights and remedies the Company may have against Executive.
- **(b) Right to Injunction**. Executive acknowledges that a breach of the covenants set forth in Section 8 hereof will cause irreparable damage to the Company and/or WMI with respect to which the remedy at law for damages will be inadequate. Therefore, in the event of breach or anticipatory breach of the covenants set forth in this Section 8 by Executive, Executive and the Company agree that the Company and/or WMI shall be entitled to seek the following particular forms of relief, in addition to remedies otherwise available to it at law or equity: (A) injunctions, both preliminary and permanent, enjoining or restraining such breach or anticipatory breach and Executive hereby consents to the issuance thereof forthwith and without bond by any court of competent jurisdiction; and (B) recovery of all reasonable sums as determined by a court of competent jurisdiction expended and costs, including reasonable attorney's fees, incurred by the Company and/or WMI to enforce the covenants set forth in this section.
- (c) Separability of Covenants. The covenants contained in Section 8 hereof constitute a series of separate but ancillary covenants, one for each applicable State in the United States and the District of Columbia, and one for each applicable foreign country. If in any judicial proceeding, a court shall hold that any of the covenants set forth in Section 8 exceed the time, geographic, or occupational limitations permitted by applicable laws, Executive and the Company agree that such provisions shall and are hereby reformed to the maximum time, geographic, or occupational limitations permitted by such laws. Further, in the event a court shall hold unenforceable any of the separate covenants deemed included herein, then such unenforceable covenant or covenants shall be deemed eliminated from the provisions of this Agreement for the purpose of such proceeding to the extent necessary to permit the remaining separate covenants to be enforced in such proceeding. Executive and the Company further agree that the covenants in Section 8 shall each be construed as a separate agreement independent of any other provisions of this Agreement, and the existence of any claim or cause of action by

Executive against the Company or WMI whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by the Company of any of the covenants of Section 8.

10. Indemnification.

The Company shall indemnify and hold harmless Executive to the fullest extent permitted by Delaware law for any action or inaction of Executive while serving as an officer or director of the Company or, at the Company's request, as an officer or director of any other entity or as a fiduciary of any benefit plan. This provision includes the obligation and undertaking of the Executive to reimburse the Company for any fees advanced by the Company on behalf of the Executive should it later be determined that Executive was not entitled to have such fees advanced by the Company under Delaware law. The Company shall cover the Executive under directors and officers liability insurance both during and, while potential liability exists, after the Employment Period in the same amount and to the same extent as the Company covers its other officers and directors.

11. Disputes and Payment of Attorney's Fees.

If at any time during the term of this Agreement or afterwards there should arise any dispute as to the validity, interpretation or application of any term or condition of this Agreement, the Company agrees, upon written demand by Executive (and Executive shall be entitled upon application to any court of competent jurisdiction, to the entry of a mandatory injunction, without the necessity of posting any bond with respect thereto, compelling the Company) to promptly provide sums sufficient to pay on a current basis (either directly or by reimbursing Executive) Executive's costs and reasonable attorney's fees (including expenses of investigation and disbursements for the fees and expenses of experts, etc.) incurred by Executive in connection with any such dispute or any litigation, provided that Executive shall repay any such amounts paid or advanced if Executive is not the prevailing party with respect to at least one material claim or issue in such dispute or litigation. The provisions of this Section 11, without implication as to any other section hereof, shall survive the expiration or termination of this Agreement and of Executive's employment hereunder.

12. Withholding of Taxes.

The Company may withhold from any compensation and benefits payable under this Agreement all applicable federal, state, local, or other taxes.

13. Source of Payments.

All payments provided under this Agreement, other than payments made pursuant to a plan which provides otherwise, shall be paid from the general funds of the Company, and no special or separate fund shall be established, and no other segregation of assets made, to assure payment. Executive shall have no right, title or interest whatever in or to any investments which the Company may make to aid the Company in meeting its obligations hereunder. To the extent that any person acquires a right to receive payments from the Company hereunder, such right shall be no greater than the right of an unsecured creditor of the Company.

14. Assignment.

Except as otherwise provided in this Agreement, this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective heirs, representatives, successors and assigns. This Agreement shall not be assignable by Executive (but any payments due hereunder which would be payable at a time after Executive's death shall be paid to Executive's designated beneficiary or, if none, his estate) and shall be assignable by the Company only to any financially solvent corporation or other entity resulting from the reorganization, merger or consolidation of the Company with any other corporation or entity or any corporation or entity to or with which the Company's business or substantially all of its business or assets may be sold, exchanged or transferred, and it must be so assigned by the Company to, and accepted as binding upon it by, such other corporation or entity in connection with any such reorganization, merger, consolidation, sale, exchange or transfer in a writing delivered to Executive in a form reasonably acceptable to Executive (the provisions of this sentence also being applicable to any successive such transaction).

15. Entire Agreement; Amendment.

This Agreement shall supersede any and all existing oral or written agreements, representations, or warranties between Executive and the Company, WMI, or any of their respective parent, subsidiaries or affiliated entities relating to the terms of Executive's employment by the Company or WMI. It may not be amended except by a written agreement signed by both parties.

16. Governing Law.

This Agreement shall be governed by and construed in accordance with the laws of the State of Texas applicable to agreements made and to be performed in that State, without regard to its conflict of laws provisions.

17. Requirement of Timely Payments.

If any amounts which are required, or determined to be paid or payable, or reimbursable, to Executive under this Agreement (or any other plan, agreement, policy or arrangement with the Company) are not so paid promptly at the times provided herein or therein, such amounts shall accrue interest, compounded daily, at an 8% annual percentage rate, from the date such amounts were required or determined to have been paid or payable, reimbursed or reimbursable to Executive, until such amounts and any interest accrued thereon are finally and fully paid, provided, however, that in no event shall the amount of interest contracted for, charged or received hereunder, exceed the maximum non-usurious amount of interest allowed by applicable law.

18. Notices.

Any notice, consent, request or other communication made or given in connection with this Agreement shall be in writing and shall be deemed to have been duly given when delivered

or mailed by registered or certified mail, return receipt requested, or by facsimile or by hand delivery, to those listed below at their following respective addresses or at such other address as each may specify by notice to the others:

To the Company: Recycle America Alliance, L.L.C.

1001 Fannin, Suite 4000 Houston, Texas 77002

Attention: Corporate Secretary

Copy to WMI: Waste Management, Inc.

1001 Fannin, Suite 4000 Houston, Texas 77002

Attention: Corporate Secretary

To Executive: At the address for Executive set forth below.

19. Miscellaneous.

- (a) Waiver. The failure of a party to insist upon strict adherence to any term of this Agreement on any occasion shall not be considered a waiver thereof or deprive that party of the right thereafter to insist upon strict adherence to that term or any other term of this Agreement.
- **(b) Separability.** Subject to Section 9 hereof, if any term or provision of this Agreement is declared illegal or unenforceable by any court of competent jurisdiction and cannot be modified to be enforceable, such term or provision shall immediately become null and void, leaving the remainder of this Agreement in full force and effect.
 - (c) Headings. Section headings are used herein for convenience of reference only and shall not affect the meaning of any provision of this Agreement.
 - (d) Rules of Construction. Whenever the context so requires, the use of the singular shall be deemed to include the plural and vice versa.
- **(e) Counterparts.** This Agreement may be executed in any number of counterparts, each of which so executed shall be deemed to be an original, and such counterparts will together constitute but one Agreement.

IN WITNESS WHEREOF, this Agreement is EXECUTED and EFFECTIVE as of the day set forth above.

STEVE RAGIEL ("Executive")	
/s/ Steve Ragiel	
Steve Ragiel	
2740 Centenary St	(address)
Houston, Texas 77005	
RECYCLE AMERICA ALLIANCE, L.L.C. (The "Company")	
By: /s/ A. Maurice Myers	
A. Maurice Myers On Behalf of the Board of Managers	

\$105,000,000

TEN YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT

Dated as of June 30, 2003

among

WASTE MANAGEMENT, INC.

WASTE MANAGEMENT HOLDINGS, INC.

BANK OF AMERICA, N.A.,

as Administrative Agent

BANK OF AMERICA, N.A.,

as Letter of Credit Issuer

and

The Lenders Party Hereto

BANC OF AMERICA SECURITIES LLC

Sole Lead Arranger and Sole Book Manager



		Page
ARTICLE I	DEFINITIONS AND ACCOUNTING TERMS	1
1.01 Defined T	Terms	1
1.02 Other Inte	erpretive Provisions	17
1.03 Accounting	ng Terms	17
1.04 Rounding		18
1.05 Reference	es to Agreements and Laws	18
1.06 Credit An		18
ARTICLE II	THE CREDIT-LINKED DEPOSITS AND CREDIT EXTENSIONS	18
2.01 Credits		18
2.02 Cash Coll	ateral	27
2.03 Reduction	n of Total Credit-Linked Deposits; Conversion to Funded Term Loans	28
2.04 Interest or	n Borrowings Other Than Term Loans	30
2.05 Fees and	Other Charges	31
2.06 Computat	ion of Interest and Fees	32
2.07 Evidence	of Debt	32
2.08 Payments	Generally	33
2.09 Sharing of	f Payments	33
ARTICLE III	TAXES AND YIELD PROTECTION	34
3.01 Taxes		34
3.02 Increased	Cost and Reduced Return; Capital Adequacy or Risk-Based Capital	35
3.03 Illegality		36
3.04 Funding I	Losses	36
3.05 Matters A	pplicable to All Requests for Compensation	37
3.06 Survival		37
3.07 Replacem	ent of Lenders	37
ARTICLE IV	CONDITIONS PRECEDENT TO CREDIT EXTENSIONS	38
4.01 Condition	s of Credit-Linked Deposits and Initial Credit Extension	38
4.02 Condition	s to all Credit Extensions by the Issuer	40
4 03 Condition	s to Term Loans	<i>4</i> 1

(continued)

	Page
ARTICLE V REPRESENTATIONS AND WARRANTIES	41
5.01 Organization; Power and Authority	41
5.02 Authorization, etc.	42
5.03 Disclosure	42
5.04 Organization and Ownership of Shares of Subsidiaries; Affiliates	43
5.05 Financial Statements	43
5.06 Compliance with Laws, Other Instruments, etc.	44
5.07 Governmental Authorizations, etc.	44
5.08 Litigation; Observance of Agreements, Statutes and Orders	44
5.09 Taxes	45
5.10 Title to Property; Leases	45
5.11 Licenses, Permits, etc.	45
5.12 Compliance with ERISA	46
5.13 Private Offering by Company	46
5.14 Use of Proceeds; Margin Regulations	47
5.15 Existing Indebtedness	47
5.16 Foreign Assets Control Regulations, etc.	47
5.17 Status under Certain Statutes	48
5.18 Environmental Matters	48
5.19 Compliance With Issuance Limits	48
5.20 Tax Shelter Regulations	48
ARTICLE VI AFFIRMATIVE COVENANTS	49
6.01 Financial and Business Information	49
6.02 Officer's Certificate	51
6.03 Inspection	52
6.04 Compliance with Laws	52
6.05 Insurance	52
6.06 Maintenance of Properties	52
6.07 Payment of Taxes and Claims	52
6.08 Corporate Existence, Etc.	53

(continued)

	Page
6.09 Books and Records	53
6.10 Purpose of Credits	53
ARTICLE VII NEGATIVE COVENANTS	53
7.01 Transactions with Affiliates	53
7.02 Merger, Consolidation, Sale of Assets, etc.	54
7.03 Liens	56
7.04 Limitation on Consolidated Debt	58
7.05 Limitation on Priority Debt	58
7.06 Nature of Business	58
7.07 Designation of Unrestricted Subsidiaries	58
7.08 Margin Stock	59
ARTICLE VIII EVENTS OF DEFAULT AND REMEDIES	59
8.01 Events of Default	59
8.02 Remedies Upon Event of Default	61
8.03 Application of Funds	62
ARTICLE IX ADMINISTRATIVE AGENT	63
9.01 Appointment and Authorization of Administrative Agent	63
9.02 Delegation of Duties	63
9.03 Liability of Administrative Agent	64
9.04 Reliance by Administrative Agent	64
9.05 Notice of Default	64
9.06 Credit Decision; Disclosure of Information by Administrative Agent	65
9.07 Indemnification of Administrative Agent	65
9.08 Administrative Agent in its Individual Capacity	67
9.09 Successor Administrative Agent	67
ARTICLE X MISCELLANEOUS	67
10.01 Amendments, Etc.	67
10.02 Notices and Other Communications; Facsimile Copies	69
10.03 No Waiver; Cumulative Remedies	70
10.04 Attorney Costs, Expenses and Taxes	70

(continued)

		Page
10.05 Indemnifi	ication by the Company	71
10.06 Payments	Set Aside	71
10.07 Successor	rs and Assigns	72
10.08 Confident	tiality	74
10.09 [Intention		75
10.10 Represent	tations and Warranties of Each Lender	75
10.11 Counterpa	arts	77
10.12 Integratio	n	77
10.13 Survival	of Representations and Warranties	77
10.14 Severabil	ity	78
10.15 Tax Form	IS .	78
10.16 Governin	g Law	80
10.17 Waiver of	f Right to Trial by Jury	80
10.18 Time of the	he Essence	80
10.19 Credit Ra	ting Downgrade Event or Deposit Account Interest Non-Payment Event	80
ARTICLE XI	HOLDINGS GUARANTY	83
11.01 Guaranty		83
11.02 Guaranty		84
11.03 Effectives	ness; Enforcement	84
11.04 Waiver		84
11.05 Expenses		85
11.06 Concernia	ng Joint and Several Liability of Holdings	85
11.07 Waiver		87
11.08 Subrogati	on; Subordination	87
11.09 Release o	f Guaranty	87
ARTICLE XII	ENTIRE AGREEMENT	88
12.01 Entire Ag	greement	88
SIGNATURES		S-1

SCHEDULES

	A	Applicable Rates; Interest on Credit-Linked Deposits
	1.01	Existing Credits
	2.01	Credit-Linked Deposits and Pro Rata Shares
	5.04	Organization and Ownership Shares of Subsidiaries; Affiliates
	5.05	Financial Statements
	5.08	Litigation; Observance of Agreements, Statutes and Orders
	5.11	Licenses, Permits, etc.
	5.15	Existing Indebtedness
	5.17	Status under Certain Statutes
	7.03	Liens
	10.02	Administrative Agent's Office; Certain Addresses for Notices
EXHIB	BITS	
		Form of
	A	Form of Request for Credit Extension
	A B	
		Request for Credit Extension
	В	Request for Credit Extension Assignment and Assumption
	B C-1	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings
	B C-1 C-2	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative Agent and the
	B C-1 C-2 C-3	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative Agent and the Issuer
	B C-1 C-2 C-3	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative Agent and the Issuer Opinion of Special Counsel to the Lenders
	B C-1 C-2 C-3 C-4 D	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative Agent and the Issuer Opinion of Special Counsel to the Lenders Letter of Credit Linked Note
	B C-1 C-2 C-3 C-4 D	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative Agent and the Issuer Opinion of Special Counsel to the Lenders Letter of Credit Linked Note Request for Term Loans
	B C-1 C-2 C-3 C-4 D E	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative Agent and the Issuer Opinion of Special Counsel to the Lenders Letter of Credit Linked Note Request for Term Loans Term Note

TEN YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT

This TEN YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT ("Agreement") is entered into as of June 30, 2003, among WASTE MANAGEMENT, INC., a Delaware corporation (the "Company"), WASTE MANAGEMENT HOLDINGS, INC., a Delaware corporation and wholly-owned Subsidiary of the Company ("Holdings") (executing this Agreement for the limited purposes set forth in Article XI hereof), BANK OF AMERICA, N.A., as issuer of letters of credit (the "Issuer"), the financial institutions having a Credit-Linked Deposit set forth opposite their names on Schedule 2.01 hereto under the heading "Credit-Linked Deposit" (collectively, the "Lenders" and individually, a "Lender"), and BANK OF AMERICA, N.A., as Administrative Agent.

The Company has requested that the Issuer and the Lenders provide a letter of credit facility, and the Issuer and the Lenders are willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I DEFINITIONS AND ACCOUNTING TERMS

1.01 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth below:

"Adjusted Consolidated EBITDA" means for any period, the total of the following calculated without duplication on a consolidated basis for such period: (i) Consolidated EBITDA, plus (ii) on a pro forma basis, the EBITDA of each Person acquired, or, as applicable, the EBITDA attributable to the assets acquired from any Person, by the Company or any Restricted Subsidiary in the subject period, for any portion of such subject period occurring prior to the date of the acquisition by the Company or any Restricted Subsidiary of such Person or related assets; provided, further, that any calculation pursuant to clause (ii) above of pro forma EBITDA for any subject period which gives effect to asset acquisitions and/or stock purchases during the subject period on a pro forma basis as provided above shall in any event be determined by the Company in accordance with sound financial practice in accordance with GAAP and on the basis, to the extent available, of appropriate financial statements and tax returns for the applicable period, and shall be set forth in a certificate of a Senior Financial Officer of the Company accompanied by calculations in reasonable detail showing the manner of determination thereof, which certificate shall be furnished to the Administrative Agent, the Issuer and each Lender not later than the certificate required to be furnished by the Company in respect of the fiscal period in which such date of determination occurs pursuant to Section 6.02 of this Agreement.

"Administrative Agent" means Bank of America in its capacity as administrative agent under any of the Credit Documents, or any successor administrative agent.

- "Administrative Agent's Office" means the Administrative Agent's address and, as appropriate, account as set forth on Schedule 10.02, or such other address or account as the Administrative Agent may from time to time specify by notice to the Company, the Issuer and the Lenders.
 - "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.
- "Advance" means, with respect to each Lender, such Lender's funding of its participation in any Borrowing (other than Term Loans) in accordance with its Pro Rata Share.
- "Affiliate" means, at any time, (a) with respect to the Company and each of its Restricted Subsidiaries, any Person (other than the Company and each of its Restricted Subsidiaries) that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, the Company, (b) with respect to any Person (other than the Company and each of its Restricted Subsidiaries), any other Person that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, such first Person, and (c) any Person beneficially owning or holding, directly or indirectly, 10% or more of any class of voting or equity interests of the Company or any Subsidiary of the Company or any corporation of which the Company and its Subsidiaries beneficially own or hold, in the aggregate, directly or indirectly, 10% or more of any class of voting or equity interests. As used in this definition, "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. Unless the context otherwise clearly requires, any reference to an "Affiliate" is a reference to an Affiliate of the Company.
 - "Agent Indemnified Liabilities" has the meaning specified in Section 9.07.
- "Agent-Related Persons" means the Administrative Agent or the Issuer, together with its Affiliates (including, in the case of Bank of America in its capacity as the Administrative Agent, BAS), and the officers, directors, employees, agents and attorneys-in-fact of such Persons and Affiliates.
 - "Agreement" means this Ten Year Letter of Credit and Term Loan Agreement.
 - "Applicable Rate" has the meaning set forth on Schedule A annexed hereto.
 - "Arranger" means BAS, in its capacity as sole lead arranger and sole book manager.
 - "Assignment and Assumption" means an Assignment and Assumption substantially in the form of Exhibit B.
 - "Attorney Costs" means and includes all fees, expenses and disbursements of any law firm or other external counsel.

"Audited Financial Statements" means the audited consolidated balance sheet of the Company and its Subsidiaries for the fiscal year ended December 31, 2002, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year of the Company and its Subsidiaries, including the notes thereto.

"Authorized Representative" means any employee of a Credit Party involved principally in the financial administration, controllership or treasury function of such Credit Party who is authorized or designated as an Authorized Representative, as evidenced by the certificate delivered pursuant to Section 4.01(a)(iv) or from time to time by a certificate delivered by a Senior Financial Officer of such Credit Party to the Administrative Agent in form and substance satisfactory to it. Any document delivered hereunder that is signed by an Authorized Representative of a Credit Party shall be conclusively presumed to have been authorized by all necessary corporate action on the part of such Credit Party and such Authorized Representative shall be conclusively presumed to have acted on behalf of such Credit Party.

"Auto-Reinstating Credit" means any Credit containing any provisions providing for automatic reinstatement of the stated amount after any drawing thereunder.

"Auto-Renewal Credit" has the meaning specified in Section 2.01(b)(iii).

"Bank of America" means Bank of America, N.A. and its successors.

"BAS" means Banc of America Securities LLC.

"Base Rate" means for any day a fluctuating rate per annum equal to the higher of (a) the Federal Funds Rate plus 1/2 of 1% and (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate." Such rate is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such rate announced by Bank of America shall take effect at the opening of business on the day specified in the public announcement of such change.

"Beneficial Ownership" has the meaning set forth in Rule 13d-3 promulgated by the SEC under the Securities Exchange Act.

"Borrowing" means, without duplication, an extension of credit resulting from a drawing under any Credit which has not been reimbursed on the applicable Reimbursement Date, including the making of Advances, or the making of a Term Loan, as the context may require.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, New York or Dallas, or (if the Administrative Agent's Office is not located in New York or Dallas) the state where the Administrative Agent's Office is located, and, if such day relates to the Credit-Linked Deposit Account or any LIBO Rate Term Loan or LIBO Rate Advance, means any such day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Capital Leases" or "Capitalized Leases" means leases under which the Company or any of its Restricted Subsidiaries is the lessee or obligor, the discounted future rental payments under which are required to be capitalized on the balance sheet of the lessee or obligor in accordance with GAAP.

"Cash Collateral" has the meaning specified in Section 2.02.

"Cash Collateralize" has the meaning specified in Section 2.02.

"Closing Date" means the first date all the conditions precedent in Section 4.01 are satisfied or waived in accordance with Section 4.01 (or, in the case of Section 4.01(b), waived by the Person entitled to receive the applicable payment).

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"Company" has the meaning specified in the introductory paragraph hereto.

"Consolidated Debt" means, as of any date of determination, the total of all Indebtedness of the Company and its Restricted Subsidiaries determined on a consolidated basis in accordance with GAAP.

"Consolidated EBITDA" means, for any period, Consolidated Net Income or Consolidated Net Loss, as the case may be, for such period, plus the sum of (a) interest expense; (b) income tax expense; (c) depletion and depreciation expense; and (d) amortization expense; in each case, which were deductible in determining Consolidated Net Income or Consolidated Net Loss for such period, determined on a consolidated basis for the Company and its Restricted Subsidiaries in accordance with GAAP.

"Consolidated Net Income or Consolidated Net Loss" means, respectively, for any period, the net income or loss for such period determined on a consolidated basis for the Company and its Restricted Subsidiaries in accordance with GAAP.

"Consolidated Net Worth" means, as of any date of determination, the consolidated stockholders' equity of the Company and its Restricted Subsidiaries (excluding any amounts attributable to mandatorily redeemable Preferred Stock) determined as of such date.

"Consolidated Tangible Assets" means, as of any date of determination, Consolidated Total Assets less the sum of (i) Intangible Assets and (ii) all amounts representing write-up in the book value of any assets of the Company and its Restricted Subsidiaries resulting from a revaluation thereof.

"Consolidated Total Assets" means all assets of the Company and its Restricted Subsidiaries determined on a consolidated basis in accordance with GAAP.

"Credit" means any standby letter of credit issued or outstanding hereunder (including a so-called "direct pay" standby letter of credit) that is a Permitted Credit and shall include the Existing Credits.

"Credit Documents" means this Agreement, the Fee Letter, each Request for Credit Extension, each Issuer Document, each Request for Term Loan, each Letter of Credit Linked Note and each Term Note and each other document, instrument or agreement from time to time executed by the Company or any of its Subsidiaries, Holdings, or any secretary, assistant secretary, Responsible Officer or Authorized Representative thereof and delivered in connection with this Agreement.

"Credit Expiration Date" means the day that is seven (7) Business Days prior to the Maturity Date (or, if such day is not a Business Day, the next preceding Business Day).

"Credit Extension" means, (a) with respect to any Credit, the issuance thereof or extension of the expiry date thereof, or the renewal, reinstatement or increase of the amount thereof and (b) the making of any Term Loans.

"Credit-Linked Deposit" shall have the meaning specified in Section 2.01(c)(ii).

"Credit-Linked Deposit Account" shall mean the deposit account established by the Administrative Agent in its name and under its sole and exclusive control maintained at the office of Bank of America at its principal office in Chicago, Illinois, designated as the "Bank of America N.A. as Administrative Agent Waste Management Inc Ten Year LOC Facility Lender Credit Linked Notes Dep Acct Treasury" that shall be used solely for the purposes set forth in Sections 2.01(c)(ii) and (iii) and in Section 2.03(b)(iii), or such successor account established pursuant to Section 10.19 of this Agreement.

"Credit-Linked Lenders" means the Lenders under this Agreement and the lenders under the Seven Year Agreement and the Five Year Agreement.

"Credit Obligations" means, without duplication, as at any date of determination, the aggregate undrawn face amount of all outstanding Credits plus the aggregate amount of all Unreimbursed Amounts in respect of any Credits, and, without duplication, all outstanding Borrowings arising from any Credits, plus the aggregate amount of the Term Loans. For all purposes of this Agreement, if on any date of determination a Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Credit Parties" means, collectively, the Company and Holdings.

"Credit Rating Downgrade Event" has the meaning specified in Section 10.19(a)(i).

"Debtor Relief Laws" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the

United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means an event or condition the occurrence or existence of which would, with the lapse of time or the giving of notice or both, become an Event of Default.

"Default Rate" means (a) with respect to obligations other than Advances and Term Loans, the Base Rate plus 2.00% per annum, and (b) with respect to Advances and Term Loans, the LIBO Rate plus 3.00% per annum, in each case to the fullest extent permitted by applicable Laws.

"Deposit Account Interest Non-Payment Event" has the meaning specified in Section 10.19(a)(ii).

"Disclosure Documents" means the Company's financial statements referred to in Section 5.05 and filings made by the Company or Holdings with the SEC that were publicly available prior to the date of this Agreement and which were provided to the Issuer and the Lenders.

"Dollar" and "\$" means lawful money of the United States.

"EBITDA" means, (without duplication) with respect to any Person, division or business and with respect to any period, the sum of (i) (a) net income for such period from continuing operations after extraordinary items (excluding gains or losses from disposition of assets); (b) consolidated interest charges for such period; (c) the amount of taxes, based on or measured by income, used or included in the determination of such net income; (d) the amount of depreciation and amortization expense deducted in determining such net income; and (e) any extraordinary losses or non-recurring expenses not requiring the expenditure of cash decreasing net income for such period, minus (ii) any extraordinary gains or non-recurring items increasing net income for such period.

"Eligible Assignee" means any Institutional Investor.

"Environmental Laws" means any and all federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including but not limited to those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder from time to time in effect.

"ERISA Affiliate" means any trade or business (whether or not incorporated) that is treated as a single employer together with the Company under Section 414 of the Code.

"Escalating Credit" means a Credit which provides for a face amount that increases from time to time in accordance with its terms.

"Event of Default" has the meaning specified in Section 8.01.

"Existing Credits" means the standby letters of credit listed on Schedule 1.01.

"Facility Fee" has the meaning specified in Section 2.05(a).

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" has the meaning specified in Section 2.05(b).

"Five Year Agreement" means the Five Year Letter of Credit and Term Loan Agreement dated as of even date herewith (as amended, restated, supplemented or otherwise modified from time to time) among the Company, Holdings, each lender party thereto and Bank of America as administrative agent and issuer of letters of credit; provided, however, if all Credit Obligations (as such term is defined in the Five Year Agreement) are paid in full and the Five Year Agreement is terminated, then this definition of "Five Year Agreement" shall no longer be applicable to this Agreement.

"Foreign Lender" has the meaning specified in Section 10.15(a).

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Fee" has the meaning specified in Section 2.05(b).

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means

(a) the government of

- (i) the United States of America or any state or other political subdivision thereof, or
- (ii) any jurisdiction in which the Company or any Subsidiary conducts all or any part of its business, or which asserts jurisdiction over any properties of the Company or any Subsidiary, or
- (b) any entity exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, any such government.

"Guarantee" by any Person means any obligation, contingent or otherwise, of such Person directly or indirectly guaranteeing any Indebtedness of any other Person and, without limiting the generality of the foregoing, any obligation, direct or indirect, contingent or otherwise, of such Person (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness (whether arising by virtue of partnership arrangements, by agreement to keep-well, to purchase assets, goods, securities or services, to take-or-pay, or to maintain financial statement conditions or otherwise) or (b) entered into for the purpose of assuring in any other manner the obligee of such Indebtedness of the payment thereof or to protect such obligee against loss in respect thereof (in whole or in part); provided that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business. The term "Guarantee" used as a verb has a corresponding meaning.

"Guaranteed Obligations" has the meaning specified in Section 11.01.

"Hazardous Material" means any hazardous waste as defined by 42 U.S.C. §6903(5), any hazardous substance as defined by 42 U.S.C. §9601(14), any pollutant or contaminant as defined by 42 U.S.C. §9601(33) or any toxic substances, oil or other chemicals or substances regulated by the Environmental Laws, excluding household hazardous waste.

"Holdings" means Waste Management Holdings, Inc., a Delaware corporation and wholly-owned Subsidiary of the Company.

"Holdings Guaranty" means the meaning specified in Section 11.01.

"Indebtedness" means, collectively, without duplication, whether classified as Indebtedness, an investment or otherwise on the obligor's balance sheet, (a) all indebtedness for borrowed money, (b) all obligations for the deferred purchase price of property or services (other than trade payables incurred in the ordinary course of business which either (i) are not overdue by more than ninety (90) days, or (ii) are being disputed in good faith and for which adequate reserves have been established in accordance with GAAP), (c) all obligations evidenced by notes, bonds, debentures or other similar debt instruments, (d) all obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (e) all obligations, liabilities and indebtedness under Capitalized Leases, (f) all obligations, liabilities or indebtedness under drawn surety or performance bonds, or any other drawn bonding arrangement, (g) Guarantees with respect to all Indebtedness of others referred to in clauses (a) through (f) above, and (h) all

Indebtedness of others referred to in clauses (a) through (f) above secured or supported by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured or supported by) any Lien on the property or assets of the Company or any Subsidiary, even though the owner of the property has not assumed or become liable, contractually or otherwise, for the payment of such Indebtedness; provided that if a Permitted Receivables Transaction is outstanding and is accounted for as a sale of accounts receivable under GAAP, Indebtedness shall also include the additional Indebtedness, determined on a consolidated basis, which would have been outstanding had such Permitted Receivables Transaction been accounted for as a borrowing.

"Indemnified Liabilities" has the meaning specified in Section 10.05.

"Indemnitees" has the meaning specified in Section 10.05.

"INHAM Exemption" has the meaning specified in Section 10.10(b)(v).

"Institutional Investor" means (a) any original Lender or any Affiliate of a Lender, (b) any Lender holding more than \$5,000,000 of the Credit-Linked Deposits and/or Credit Obligations then outstanding, and (c) any bank, trust company, savings and loans association or other financial institution, any pension plan, any investment company, any insurance company, any broker or dealer, any Person (other than a natural Person) engaged in making, purchasing, holding or otherwise investing in commercial loans in the ordinary course of its business, or any other similar financial institution or entity, regardless of legal form.

"Intangible Assets" means goodwill, the purchase price of acquired assets in excess of the fair value thereof, trademarks, trade names, service marks, customer lists, brand names, copyrights, patents and licenses, and rights with respect to the foregoing.

"Interest Period" means, (a) as to the Credit-Linked Deposits, the period commencing on the date such Credit-Linked Deposit is made and ending on the date three months thereafter; (b) as to Advances and Term Loans, the period commencing on the date any such Advance or Term Loan is disbursed (except that in respect of Term Loans made from the proceeds of the Credit-Linked Deposits pursuant to Sections 2.03(b)(i)(A) or (B), the Interest Period shall be deemed to commence on the same day as the Interest Period commencement date for the Credit-Linked Deposits, and in respect of Term Loans made through the conversion of Advances pursuant to Section 2.03(b)(i)(C), the Interest Period shall be deemed to commence on the same day as the Interest Period commencement date for the Credit-Linked Deposits, if any, on the date of such conversion) and ending on the earlier of (x) the date three months thereafter and (y) the last day of the Interest Period with respect to the Credit-Linked Deposits; and (c) after the initial Interest Periods described in (a) and (b) above, as to the Credit-Linked Deposits, Advances and Term Loans, the period commencing on the last day of the immediately preceding Interest Period and ending on the date three months thereafter; provided that:

(i) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;

- (ii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and
 - (iii) no Interest Period shall extend beyond the Maturity Date;

provided further that after the occurrence and during the continuance of an Event of Default the Required Lenders, acting through the Administrative Agent, may designate one or more Interest Periods for the Advances and/or the Term Loans which are shorter than that set forth above, such designation to be effective after the last day of the applicable existing Interest Period.

"IRS" means the United States Internal Revenue Service.

"ISP" means, with respect to any Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance).

"Issuance Limits" has the meaning specified in Section 2.01(a)(i).

"Issuer" means Bank of America in its capacity as issuer of Credits hereunder, or any successors, and shall include any branch of the Issuer through which the Issuer may cause Credits to be issued hereunder.

"Issuer Documents" means with respect to any Credit issued by the Issuer, such Credit, any applications for issuance or amendment of such Credit, and any other document, agreement and instrument entered into by the Issuer and the Company or in favor of the Issuer and relating to any such Credit, including any of the Issuer's standard form documents for issuances and amendments of letters of credit and guarantees or other similar undertakings.

"Issuer-Related Persons" means as to the Issuer, together with its Affiliates, and the officers, directors, employees, agents and attorneys-in-fact of the Issuer and its Affiliates.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"Lender" has the meaning specified in the introductory paragraph hereto.

"Lending Office" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Company and the Administrative Agent.

"Letter of Credit Linked Note" has the meaning specified in Section 2.07(b).

"LIBO Rate" means for any Interest Period:

- (a) the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate that appears on the page of the Telerate screen (or any successor thereto) that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (b) if the rate referenced in the preceding clause (a) does not appear on such page or service or such page or service shall not be available, the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate on such other page or other service that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (c) if the rates referenced in the preceding clauses (a) and (b) are not available, the rate per annum determined by the Administrative Agent as the rate of interest at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Total Credit-Linked Deposit or the average amount of the Advances of the Lenders or LIBO Rate Term Loans, as applicable, being made or continued and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 4:00 p.m. (London time) two Business Days prior to the first day of such Interest Period.

"LIBO Rate Term Loans" means Term Loans bearing interest based on the LIBO Rate.

"Lien" means, with respect to any Person, any mortgage, lien, pledge, charge, security interest or other encumbrance, or any interest or title of any vendor, lessor, lender or other secured party to or of such Person under any conditional sale or other title retention agreement or Capital Lease, upon or with respect to any property or asset of such Person (including in the case of stock, stockholder agreements, voting trust agreements and all similar arrangements).

"Long-Term Debt" has the meaning specified in Section 10.19(a).

"Material" or "Materially" means material or materially, as applicable, in relation to the business, operations, affairs, financial condition, assets, properties, or prospects of the Company and its Restricted Subsidiaries taken as a whole.

"Material Adverse Effect" means a material adverse effect on (a) the business, operations, affairs, financial condition, prospects, assets or properties of the Company and its Restricted Subsidiaries taken as a whole, (b) the ability of the Company to perform its obligations under this Agreement, the Letter of Credit Linked Notes or the Term Notes, or (c) the

validity or enforceability of this Agreement, the Letter of Credit Linked Notes or the Term Notes.

"Material Debt" means any Indebtedness (other than any Indebtedness of the Company or Holdings in respect of the Credit Obligations) of any one or more of the Company, its Restricted Subsidiaries and Holdings in an aggregate principal amount for all such Persons exceeding \$50,000,000.

"Maturity Date" means (a) June 30, 2013, or (b) such earlier date upon which the obligations of the Issuer and the Lenders to make Credit Extensions may be terminated in accordance with the terms hereof.

"Memorandum" has the meaning specified in Section 5.03.

"Money Market Account Agreement" means the Money Market Account Agreement between the Administrative Agent and Bank of America, as depository, substantially in the form of Exhibit G attached hereto, concerning the Credit-Linked Deposit Account.

"Moody's" means Moody's Investors Service, Inc., or any successor thereto that is a nationally-recognized rating agency.

"Multiemployer Plan" means any Plan that is a "multiemployer plan" (as such term is defined in Section 4001(a)(3) of ERISA).

"New Account" has the meaning specified in Section 10.19(b)(i).

"New Depository Bank" has the meaning specified in Section 10.19(b)(i).

"Nonrenewal Notice Date" has the meaning specified in Section 2.01(b)(iii).

"Officer's Certificate" means a certificate of a Senior Financial Officer or of any other officer of the Company whose responsibilities extend to the subject matter of such certificate.

"Other Taxes" has the meaning specified in Section 3.01(b).

"Outstanding Amount" means, with respect to any Credit Obligations on any date, the amount of such Credit Obligations on such date after giving effect to any Credit Extension occurring on such date and any other changes in the aggregate amount of the Credit Obligations as of such date, including as a result of any reimbursements of outstanding unpaid drawings under any Credits or any reductions in the maximum amount available for drawing under Credits taking effect on such date. Without limiting the provisions of Section 1.07, for avoidance of doubt, the portion of the Credit Obligations in respect of the undrawn face amount of any Escalating Credits shall be deemed to be the maximum aggregate amount available to be drawn under such Escalating Credits (after giving effect to all increases).

"Participant" has the meaning specified in Section 10.07(d).

"Participation" has the meaning specified in Section 2.01(c)(ii).

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA or any successor thereto.

"Permitted Credit" means a Credit used directly or indirectly to cover (a) a default in the performance of any non-financial or commercial obligations of the Company or any Subsidiary under specific contracts, including a Credit issued in favor of a bank or other surety who in connection therewith issues a guarantee or similar undertaking, performance bond, surety bond or other similar instrument that covers a default in the performance of any non-financial or commercial obligations under specific contract, or (b) the payment of any financial contractual obligation of the Company or any Subsidiary.

"Permitted Liens" has the meaning specified in Section 7.03.

"Permitted Receivables Transaction" means any sale or sales of, and/or securitization of, any accounts receivable of the Company and/or any of its Restricted Subsidiaries (the "Receivables") pursuant to which (a) the Company and its Restricted Subsidiaries realize aggregate net proceeds of not more than \$750,000,000 at any one time outstanding, including, without limitation, any revolving purchase(s) of Receivables where the maximum aggregate uncollected purchase price (exclusive of any deferred purchase price) for such Receivables at any time outstanding does not exceed \$750,000,000, and (b) which Receivables shall not be discounted more than 25%.

"Person" means any individual, trustee, corporation, general partnership, limited partnership, limited liability company, joint stock company, trust, unincorporated organization, bank, business association, firm, joint venture, Governmental Authority or other legal entity.

"Plan" means an "employee benefit plan" (as defined in Section 3(3) of ERISA) that is or, within the preceding five years, has been established or maintained, or to which contributions are or, within the preceding five years, have been made or required to be made, by the Company or any ERISA Affiliate or with respect to which the Company or any ERISA Affiliate may have any liability.

"Preferred Stock" means any class of capital stock of a corporation that is preferred over any other class of capital stock of such corporation as to the payment of dividends or the payment of any amount upon liquidation or dissolution of such corporation.

"Prepayment Premium" shall mean in connection with any prepayment of Advances, Term Loans or reduction of the available amount of the Total Credit-Linked Deposit an amount equal to the applicable percentage of the principal amount of the Advances, Term Loans or Total Credit-Linked Deposit so prepaid or reduced as follows: (i) on or after the second anniversary of the Closing Date and prior to the third anniversary of the Closing Date, a premium of 3%, (ii) on or after the fourth anniversary of the Closing Date and prior to the fifth anniversary of the Closing Date, a premium of 1%.

"Priority Debt" means (without duplication) the sum of (a) unsecured Indebtedness of the Company's Restricted Subsidiaries other than (i) Indebtedness of the Company's Restricted

Subsidiaries, including Holdings, listed on Schedule 5.15 and any extension, renewal or refinancing by such parties of such Indebtedness, provided that the terms of any such extension, renewal or refinancing are substantially the same as the terms and conditions in effect on the Closing Date, or more favorable to such parties, and the principal amount of such Indebtedness is not increased, (ii) Indebtedness owing to the Company or any other Restricted Subsidiary and (iii) any unsecured Indebtedness of Holdings consisting of Guarantees of the Company's Indebtedness existing as of the Closing Date and, so long as the Holdings Guaranty remains in full force and effect, Indebtedness consisting of Guarantees of the Company's Indebtedness incurred thereafter, plus (b) Indebtedness with respect to drawn landfill closure bonds, plus (c) Indebtedness with respect to Permitted Receivables Transactions, plus (d) Indebtedness of the Company or any of its Restricted Subsidiaries secured by a Lien not permitted by clauses (a) through (l) of Section 7.03.

"Pro Rata Share" means, with respect to each Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the Credit-Linked Deposit of such Lender at such time and the denominator of which is the amount of the Total Credit-Linked Deposit; provided that if the obligations of the Issuer to make Credit Extensions have been terminated pursuant to Section 8.02(a), the numerator of such fraction shall be such Lender's Outstanding Amount of all Credit Obligations at such time, and the denominator shall be the Outstanding Amount of all Credit Obligations of all Lenders at such time (with the aggregate amount of any Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Lender for purposes of this definition). The initial Pro Rata Share of each Lender is set forth opposite the name of such Lender on Schedule 2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"property" or "properties" means, unless otherwise specifically limited, real or personal property of any kind, tangible or intangible, existing or inchoate.

"PTE" has the meaning specified in Section 10.10(b)(i).

"QPAM Exemption" has the meaning specified in Section 10.10(b)(iv).

"Register" has the meaning specified in Section 10.07(c).

"Reimbursement Date" has the meaning specified in Section 2.01(c)(i).

"Replacement Lender" has the meaning specified in Section 3.07.

"Request for Credit Extension" means a request for the issuance or amendment of a Credit by the Issuer pursuant to a Request for Credit Extension substantially in the form of Exhibit A, with such changes thereto as the Administrative Agent or the Issuer may from time to time approve.

"Request for Term Loans" means a request for the making of Term Loans from the proceeds of Credit-Linked Deposits, substantially in the form of Exhibit E.

"Required Credit-Linked Lenders" means, Credit-Linked Lenders (other than the Company and its Affiliates) having Combined Pro Rata Shares aggregating more than 50%. For purposes hereof, the "Combined Pro Rata Share" of a Credit-Linked Lender means, with respect to such Credit-Linked Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the sum of (A) the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under this Agreement at such time plus (B) the sum of the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under the Seven Year Agreement at such time (as such terms are defined in the Seven Year Agreement), plus (C) the sum of the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under the Five Year Agreement at such time (as such terms are defined in the Five Year Agreement), and the denominator of which is the amount of the sum of (X) the Total Credit-Linked Deposit plus the Outstanding Amount of all Credit-Linked Lenders under this Agreement at such time (with the aggregate amount of any Credit-Linked Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Credit-Linked Lenders under the Seven Year Agreement at such time (as such terms are defined in the Seven Year Agreement) (with the aggregate amount of any Credit-Linked Lender's risk participation and funded participation in Credit Obligations of all Credit-Linked Lender for purposes of this definition) plus (Z) the Total Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of all Credit-Linked Lenders under the Five Year Agreement at such time (as such terms are defined in the Five Year Agreement) (with the aggregate amount of any Credit-Linked Lenders under the Five Year Agreement) (with the aggregate amount of any Credit-Linked Lender's

"Required Lenders" means, as of any date of determination, Lenders (other than the Company and its Affiliates) having Pro Rata Shares aggregating more than 50%.

"Required Notice" has the meaning specified in Section 10.19(b).

"Responsible Officer" means any Senior Financial Officer and any other officer of a Credit Party with responsibility for the administration of the relevant portion of this Agreement. Any document delivered hereunder that is signed by a Responsible Officer of a Credit Party shall be conclusively presumed to have been authorized by all necessary corporate action on the part of such Credit Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Credit Party.

"Restricted Subsidiary" means any Subsidiary of the Company which is not designated as an Unrestricted Subsidiary in accordance with Section 7.07.

"S&P" means Standard & Poor's Rating Group, a division of The McGraw-Hill Companies, or any successor thereto that is a nationally-recognized rating agency.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

- "Securities Account" has the meaning specified in Section 10.19(b)(ii).
- "Securities Act" means the Securities Act of 1933.
- "Securities Exchange Act" means Securities Exchange Act of 1934.
- "Securities Intermediary" has the meaning specified in Section 10.19(b)(ii).
- "Security" has the meaning specified in Section 2(1) of the Securities Act.
- "Senior Financial Officer" means the chief financial officer, principal accounting officer, treasurer or controller of a Credit Party.
- "Senior Responsible Officer" means any Senior Financial Officer, or any of the President, any Executive Vice President, any Senior Vice President or any Principal Vice President of a Credit Party.

"Seven Year Agreement" means the Seven Year Letter of Credit and Term Loan Agreement dated as of even date herewith (as amended, restated, supplemented or otherwise modified from time to time) among the Company, Holdings, each lender party thereto and Bank of America as administrative agent and issuer of letters of credit; provided, however, if all Credit Obligations (as such term is defined in the Seven Year Agreement) are paid in full and the Seven Year Agreement is terminated, then this definition of "Seven Year Agreement" shall no longer be applicable to this Agreement.

"Source" has the meaning specified in Section 10.10(b).

"Subsidiary" means, as to any Person, any corporation, association or other business entity in which such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries owns sufficient equity or voting interests to enable it or them (as a group) ordinarily, in the absence of contingencies, to elect a majority of the directors (or Persons performing similar functions) of such entity, and any partnership or joint venture if more than a 50% interest in the profits or capital thereof is owned by such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries (unless such partnership or joint venture can and does ordinarily take major business actions without the prior approval of such Person or one or more of its Subsidiaries). Unless the context otherwise clearly requires, any reference to a "Subsidiary" is a reference to a Subsidiary of the Company.

"Taxes" has the meaning specified in Section 3.01(a).

"Term Loan Issuance Date" has the meaning specified in Section 2.03(b)(iii).

"Term Loans" has the meaning specified in Section 2.03(b)(i).

"Term Note" has the meaning specified in Section 2.07(b).

"Total Credit-Linked Deposit" shall mean, at any time, the sum of all Lenders' Credit-Linked Deposits, as the same may be reduced from time to time.

- "Trust" has the meaning specified in Section 10.19(b)(ii).
- "United States" and "U.S." each means the United States of America.
- "Unreimbursed Amount" has the meaning specified in Section 2.01(c)(i).
- "Unrestricted Subsidiary" means any Subsidiary which is designated as such in writing by the Company to each Lender pursuant to Section 7.07.
- "Unused Amount" has the meaning specified in Section 2.03(b)(i).
- **1.02 Other Interpretive Provisions**. With reference to this Agreement and each other Credit Document, unless otherwise specified herein or in such other Credit Document:
 - (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (b) (i) The words "herein," "hereot," and "hereunder" and words of similar import when used in any Credit Document shall refer to such Credit Document as a whole and not to any particular provision thereof.
 - (ii) Article, Section, Exhibit and Schedule references are to the Credit Document in which such references appear.
 - (iii) The term "including" is by way of example and not limitation.
- (iv) The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.
- (c) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including."
- (d) Section headings herein and in the other Credit Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Credit Document.
- **1.03 Accounting Terms.** (a) All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Financial Statements, except as otherwise specifically prescribed herein.
- (b) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Credit Document, and either the Company or the Required Credit-Linked Lenders shall so request, the Administrative Agent, the Lenders and the Company

shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Credit-Linked Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Company shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

1.04 Rounding. Any financial ratios required to be maintained by the Company pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

1.05 References to Agreements and Laws. Unless otherwise expressly provided herein, (a) references to agreements (including the Credit Documents) and other contractual instruments shall be deemed to include all subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Credit Document; and (b) references to any Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Law.

1.06 Credit Amounts. Unless otherwise specified, all references herein to the amount of a Credit at any time shall be deemed to mean the maximum face amount of such Credit after giving effect to all increases thereof contemplated by such Credit or the Issuer Documents related thereto, whether or not such maximum face amount is in effect at such time.

ARTICLE II THE CREDIT-LINKED DEPOSITS AND CREDIT EXTENSIONS

2.01 Credits.

- (a) The Credits.
- (i) Subject to the terms and conditions set forth herein, (A) the Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.01, from time to time on any Business Day during the period from the Closing Date until the Credit Expiration Date, to issue Credits denominated in Dollars for the account of the Company or any Subsidiary, and (B) the Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.01, (1) to amend, extend, reinstate or renew Credits previously issued by it, in accordance with subsection (b) below, and (2) to honor drawings under the Credits; and (C) the Lenders severally agree to participate in Credits issued or outstanding hereunder for the account of the Company or any Subsidiary as provided in Section 2.01(c), and all reimbursement obligations and rights hereunder in respect thereof and the Issuer Documents with respect thereto and the Issuer hereby grants to the Lenders such participation interests in such Credits and all reimbursement obligations and rights hereunder in respect thereof and such related Issuer

Documents; provided that the Company shall not request the issuance of any Credit, the Issuer shall not make any Credit Extension with respect to any Credit, and no Lender shall participate in any Credit, to the extent that, as of the date of such Credit Extension:

- (A) the sum, without duplication, of the applicable Outstanding Amount of all (i) Credits and (ii) Unreimbursed Amounts for which Advances have not yet been made would exceed the Total Credit-Linked Deposit;
- (B) any Lender's Pro Rata Share of the sum, without duplication, of the applicable Outstanding Amount of all (i) Credits and (ii) Unreimbursed Amounts for which Advances have not yet been made would exceed such Lender's Credit-Linked Deposit; or
- (C) the expiry date of any such requested Credit would occur after the Credit Expiration Date (or, in the case of an Auto-Renewal Credit, no Nonrenewal Notice Date would occur on or prior to the Credit Expiration Date);

and provided further that in determining the availability hereunder with respect to any Escalating Credits issued or outstanding hereunder, the Total Credit-Linked Deposit will be deemed to be utilized in respect of such Escalating Credits in the aggregate amount equal to the maximum aggregate amount available to be drawn under all such Escalating Credits (after giving effect to all increases).

Within the foregoing limits (the "Issuance Limits"), and subject to the terms and conditions hereof, the Company's ability to obtain Credits shall be fully revolving, and accordingly the Company may, during the foregoing period, obtain Credits to replace Credits that have expired, been terminated or cancelled, or that have been drawn upon and reimbursed. All Existing Credits up to the respective amounts specified in Schedule 1.01 shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof. Notwithstanding anything to the contrary set forth herein, if any portion of the Total Credit-Linked Deposit is utilized to reimburse the Issuer for any amounts drawn under any Credit (as contemplated by Section 2.01(c) hereof) or to make Term Loans (as contemplated by Section 2.03(b) hereof), the Total Credit-Linked Deposit shall be permanently reduced by the corresponding amount utilized to reimburse the Issuer or to make Term Loans and shall not be reinstated.

- (ii) The Issuer shall not be under any obligation to issue any Credit if:
 - (A) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain the Issuer from issuing such Credit, or any Law applicable to the Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over the Issuer shall prohibit, or request that the Issuer refrain from, the issuance of letters of credit generally or such Credit in particular or shall impose upon the Issuer with respect to such Credit any restriction, reserve or capital requirement (in each case, for which the Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall

impose upon the Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which the Issuer in good faith deems material to it;

- (B) the Administrative Agent or the Issuer reasonably determines that such Credit is not a Permitted Credit;
- (C) the issuance of such Credit would violate any Laws or any policies of the Issuer; or
- (D) except as otherwise agreed by the Administrative Agent and the Issuer, such Credit is in a face amount less than \$1,000,000, or is to be denominated in a currency other than Dollars.
- (iii) The Issuer shall not be under any obligation to amend any Credit if (A) the Issuer would have no obligation at such time to issue such Credit in its amended form under the terms hereof, or (B) the beneficiary of such Credit does not accept the proposed amendment to such Credit.
- (iv) The Issuer shall not issue or amend any Credit if the Issuer has received written notice from any Lender, the Administrative Agent or any Credit Party, on or prior to the Business Day prior to the requested date of issuance or amendment of such Credit, that one or more applicable conditions contained in Article IV shall not then be satisfied.
 - (b) Procedures for Issuance and Amendment of Credits; Auto-Renewal Credits.
- (i) Each Credit shall be issued or amended, as the case may be, upon the request of the Company delivered to the Issuer (with a copy concurrently delivered to the Administrative Agent) in the form of a Request for Credit Extension, appropriately completed and signed by a Responsible Officer or Authorized Representative of the Company. Any such Request for Credit Extension must be received by the Issuer and the Administrative Agent not later than 1:00 p.m., Dallas time, at least five Business Days prior to the proposed issuance date or date of amendment, as the case may be; or such other date or time as the Issuer and the Administrative Agent may agree in a particular instance in their sole discretion. In the case of a request for an initial issuance of a Credit, such request shall be in the form of a Request for Credit Extension and shall specify in form and detail satisfactory to the Issuer and the Administrative Agent: (A) the proposed issuance date of the requested Credit (which shall be a Business Day); (B) the face amount thereof; (C) the expiry date thereof; (D) the purpose and nature of the Credit; (E) the Person whose obligations are supported thereby (in the case of a Credit supporting the obligations of a Subsidiary); (F) the name and address of the beneficiary thereof; (G) any required text to be contained in the Credit; (H) the delivery instructions with respect to the Credit; and (J) whether such Credit will be an Escalating Credit, and if so, the maximum face amount of such Credit after giving effect to all increases. In the case of a request for an amendment of any outstanding Credit, such request shall be in the form of a Request for Credit Extension and shall specify in form and detail satisfactory to the Issuer and the Administrative Agent: (1) the Credit to be amended; (2) the proposed date of amendment thereof (which shall be a Business Day); (3) the nature of the proposed amendment; and (4) delivery instructions with respect to the amendment. Additionally, the Company shall furnish to

the Issuer and the Administrative Agent such other documents and information pertaining to such requested Credit issuance or amendment, including any Issuer Documents as the Issuer or the Administrative Agent may reasonably require. Without limiting the generality of the foregoing, the delivery of a Request for Credit Extension is additional to, and not in replacement or substitution of, any other Issuer Document that the Issuer may require. Upon the effectiveness of any issuance or amendment of a Credit that will constitute a Credit hereunder, the Administrative Agent and the Lenders shall be entitled to assume that the Issuer has obtained such Issuer Documents as it shall have requested, executed by the relevant parties thereto to the extent required hereby.

(ii) Promptly after receipt of any Request for Credit Extension by the Issuer, the Issuer will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has received a copy of such Request for Credit Extension from the Company and, if not, the Issuer will provide the Administrative Agent with a copy thereof. Upon receipt by the Issuer and the Company of confirmation from the Administrative Agent that the requested issuance or amendment is permitted in accordance with the terms hereof, then, subject to the terms and conditions hereof, the Issuer shall, on the requested date, issue a Credit for the account of the Company or the applicable Subsidiary, or enter into the applicable amendment, as the case may be, in each case in accordance with the Issuer's usual and customary business practices.

(iii) If the Company so requests in any applicable Request for Credit Extension, the Issuer agrees, subject to the terms and conditions hereof, to issue a Credit that has automatic renewal provisions (each, an "Auto-Renewal Credit"); provided that any such Auto-Renewal Credit must permit the Issuer to prevent any such renewal at least once during the term thereof (commencing with the date of issuance of such Credit) by giving prior notice to the beneficiary thereof not later than a specified date to be agreed upon at the time such Credit is issued, which shall occur and be effective prior to the Credit Expiration Date (the "Nonrenewal Notice Date"). Unless otherwise directed by the Issuer, the Company shall not be required to make a specific request to the Issuer for any renewal of an Auto-Renewal Credit, but the Issuer shall be required to provide prior notice to the Administrative Agent of any pending renewal of an Auto-Renewal Credit at least ten Business Days before the applicable Nonrenewal Notice Date. Once an Auto-Renewal Credit has been issued (or is permitted to be outstanding hereunder in the case of an Existing Credit that is an Auto-Renewal Credit), the Lenders shall be deemed to have authorized (but may not require) the Issuer to permit the renewal of such Credit at any time to a date not later than the Credit Expiration Date; provided, however, that the Issuer shall not permit any renewal of an Auto-Renewal Credit if (A) the Issuer has determined that it would have no obligation at such time to issue such Credit in its renewed form under the terms hereof (by reason of the provisions of Section 2.01(a)(ii), Section 4.02 or otherwise), (B) after giving effect to any such renewal, the earlier of the (x) expiry date of such Auto-Renewal Credit and (y) next occurring Nonrenewal Notice Date of such Auto-Renewal Credit would occur after the Credit Expiration Date, or (C) it has received notice (which may be by telephone or in writing) on or before the day that is two Business Days before the Nonrenewal N

- (iv) If the Issuer shall issue any Credit or amend any Credit (if the effect thereof is to increase the face amount), without obtaining prior consent from the Administrative Agent (as provided in clause (ii) above), or if the Issuer shall permit the extension or renewal of an Auto-Renewal Credit without giving timely prior notice to the Administrative Agent or when such extension or renewal is not permitted hereunder (as provided in clause (iii) above), or if the Issuer shall permit the reinstatement of an Auto-Reinstating Credit when such reinstatement is not permitted hereunder (as provided in clause (vi) below), such Credit (in the case of any such amendment, to the extent of the increased face amount thereof, and in the case of any reinstatement, to the extent of the reinstated amount thereof) (A) shall for all purposes be deemed to have been issued by the Issuer solely for its own account and risk, and (B) shall not be considered a Credit outstanding under this Agreement, and no Lender shall be deemed to have any participation therein, effective as of the date of such issuance, amendment, reinstatement, extension or renewal, as the case may be, unless the Required Lenders expressly consent thereto.
- (v) Promptly after its delivery of any Credit or any amendment to a Credit to an advising bank with respect thereto or to the beneficiary thereof, the Issuer will also deliver to the Company and the Administrative Agent a true and complete copy of such Credit or amendment.
- (vi) If any Credit contains provisions providing for automatic reinstatement of the stated amount after any drawing thereunder, the Administrative Agent and the Lenders hereby authorize and direct the Issuer to permit such automatic reinstatement, whether or not a Default or Event of Default then exists, unless the Issuer has received a notice (which may be by telephone or in writing) on or before the day that is two Business Days before the reinstatement date from the Administrative Agent or the Required Lenders that one or more applicable conditions specified in Sections 2.01(a)(ii) or 4.02 is not then satisfied and directing it to cease permitting such automatic reinstatements.
 - (c) Drawings and Reimbursements; Credit-Linked Deposits.
- (i) Upon receipt from the beneficiary of any Credit of any notice of a drawing under such Credit, the Issuer of such Credit shall promptly notify the Company thereof. If the Company obtains written notice (confirmed by telephone) from the Issuer of a drawing under a Credit prior to 1:00 p.m., Dallas time, on any Business Day, the Company shall reimburse the Issuer on such Business Day. If the Company obtains written notice (confirmed by telephone) from the Issuer of a drawing under a Credit at or after 1:00 p.m., Dallas time, on any Business Day, the Company shall reimburse the Issuer on the Business Day immediately following the Business Day upon which such notice was received by the Company. Such reimbursement shall be made directly to the Issuer in an amount equal to the unreimbursed drawing (the "Unreimbursed Amount"). Without affecting the Company's obligations to so reimburse the Issuer on any such Business Day, which shall be absolute and unconditional, if the Company fails to so reimburse the Issuer by such time referenced above on the applicable due date for reimbursement specified above (the "Reimbursement Date"), the Issuer shall so notify the Administrative Agent (with a copy to the Company), which notice shall be provided on a Business Day, and specify in such notice the amount of the Unreimbursed Amount. Immediately upon receipt of such notice from the Issuer, the Administrative Agent shall promptly notify each

Lender of the Reimbursement Date, the Unreimbursed Amount, the amount of such Lender's Pro Rata Share thereof, and shall state that Advances will be funded by application of such Lender's Pro Rata Share of the Credit-Linked Deposits in the Credit-Linked Deposit Account on the sixth Business Day after such Reimbursement Date to the extent the Unreimbursed Amount (or any portion thereof) remains outstanding on such day.

(ii) The Issuer hereby irrevocably grants to each Lender, and each Lender hereby irrevocably accepts and purchases from the Issuer on the terms and conditions hereinafter stated, for such Lender's own account and risk an undivided interest equal to its Pro Rata Share in the Issuer's obligations and rights with respect to the Credits issued pursuant to this Agreement (as to each Lender, its "Participation"), which purchase price when paid to the Administrative Agent shall not be a deposit obligation of the Administrative Agent or Bank of America. The consideration for the Participation of each Lender shall consist of the payment by such Lender to the Administrative Agent of an amount equal to the Dollar amount set forth opposite such Lender's name in Schedule 2.01 under the heading "Credit-Linked Deposit" (as the same may be reduced from time to time in accordance with this Agreement, such Lender's "Credit-Linked Deposit"), subject to the conditions precedent set forth in Section 4.01 hereof. Each Lender shall pay to the Administrative Agent its Credit-Linked Deposit in full on the Closing Date. The Credit-Linked Deposits of all Lenders shall be held by, and in the name of, the Administrative Agent in the Credit-Linked Deposit Account under the sole dominion and control of the Administrative Agent. Each Lender unconditionally and irrevocably agrees with the Administrative Agent and the Issuer that, if a drawing is paid under any Credit for which the Issuer is not reimbursed in full by the Company in cash within five Business Days after the applicable Reimbursement Date or converted into a Term Loan under Section 2.03(b), such Lender hereby authorizes the Administrative Agent to reimburse to the Issuer the Unreimbursed Amount (or the outstanding portion thereof) related to such drawing on or after the sixth Business Day after such Reimbursement Date, to the extent of such Lender's Pro Rata Share of the Unreimbursed Amount, solely from such Lender's Pro Rata Share of the Credit-Linked Deposits in the Credit-Linked Deposit Account, and each Lender hereby irrevocably authorizes the Administrative Agent to charge the Credit-Linked Deposit Account for such purpose, in satisfaction of such Lender's reimbursement obligation arising with respect to such drawing hereunder. Without limiting the generality of Section 9.01, in charging the Credit-Linked Deposit Account or otherwise exercising any rights of set-off with respect thereto, the Administrative Agent acts as the agent of the Issuer. The amount of each Lender's Pro Rata Share of such Unreimbursed Amount (or portion thereof) which is paid the Issuer as set forth above shall be deemed to be an Advance by such Lender to the Company hereunder. The outstanding principal amount of each Advance, together with interest thereon as provided in Section 2.04, shall be due and payable, in Dollars, on demand. Each Lender shall be subrogated to the rights and remedies of the Issuer against the Company and any Subsidiary liable under such Credit to the extent such Lender has reimbursed the Issuer as set forth in this Section 2.01(c)(ii). The Issuer shall reasonably cooperate in exercising and enforcing such rights and remedies as may be requested by the Required Lenders (and such cooperation shall be subject to any applicable indemnification set forth in Section 9.07 or Section 10.05 of this Agreement). The Issuer shall have no right to withdraw or set-off against monies on deposit in the Credit-Linked Deposit Account other than as set forth in this Section 2.01(c)(ii).

- (iii) The Credit-Linked Deposits shall be held by the Administrative Agent in its name in the Credit-Linked Deposit Account and no Person other than the Administrative Agent shall have a right of withdrawal from the Credit-Linked Deposit Account nor any other right or power with respect to the Credit-Linked Deposits or the Credit-Linked Deposit Account. Notwithstanding anything in this Agreement to the contrary, the sole funding obligation of each Lender in respect of its Participation shall be satisfied upon funding of its Credit-Linked Deposit.
- (iv) Until an Advance is funded pursuant to this Section 2.01(c) to reimburse the Issuer for the Unreimbursed Amount in respect of any Credit, or such Unreimbursed Amount is converted to a Term Loan pursuant to Section 2.03, interest in respect of each Lender's Pro Rata Share of the Borrowing in the amount of such Unreimbursed Amount shall be solely for the account of the Issuer.
- (v) Except as expressly provided herein, each Lender's agreement to fund Advances, by application of such Lender's Pro Rata Share of the Credit-Linked Deposits, to reimburse the Issuer for amounts drawn under Credits issued by the Issuer, as contemplated by this Section 2.01(c), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against the Issuer, the Company, any Subsidiary or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or Event of Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing.
- (vi) Notwithstanding that a Credit issued or outstanding hereunder is in support of any obligations of a Subsidiary, the Company shall be obligated to reimburse the Issuer hereunder for the full amount of any and all drawings under such Credit.
- (vii) In the event of any voluntary payment or prepayment (other than through the making of a Term Loan as provided for in Section 2.03(b)) of an Advance at any time prior to the second anniversary of the Closing Date, the Company shall pay on the date of such prepayment to the Administrative Agent, for the account of the Lenders in accordance with their Pro Rata Shares, a premium in an amount equal to three percent (3.00%) of the amount so prepaid and any prepayment of the Advances thereafter shall be accompanied by the Prepayment Premium, if any.
 - (d) Repayment of Participations.
- (i) At any time after the Issuer has made a payment under any Credit and has received from the Credit-Linked Deposit the proceeds of Advances by the Lenders in respect of such payment in accordance with Section 2.01(c), if the Administrative Agent receives for the account of the Issuer any payment in respect of the related Unreimbursed Amount or interest thereon (whether directly from the Company or otherwise, including proceeds of Cash Collateral applied thereto by the Administrative Agent), the Administrative Agent will distribute to each Lender its Pro Rata Share thereof (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's Advance was outstanding) in Dollars and in the same funds as those received by the Administrative Agent. If a Borrowing has occurred hereunder, the Company shall repay the Unreimbursed Amount and interest thereon as provided in Sections 2.01(c) and 2.04 and shall pay any prepayment premium required to be paid pursuant

to Section 2.01(c)(vii). Notwithstanding the foregoing, if the Issuer shall have received from the Credit-Linked Deposit the proceeds of Advances by the Lenders and thereafter shall receive any direct payment from the Company in respect of a Borrowing, the Issuer shall immediately pay the amount received to the Administrative Agent for distribution to the Lenders in accordance with this Section 2.01(d).

- (ii) If any payment received by the Administrative Agent for the account of the Issuer pursuant to Section 2.01(d)(i) and distributed to the Lenders by the Administrative Agent is required to be returned under any of the circumstances described in Section 10.06, each Lender shall pay to the Administrative Agent for the account of the Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect.
- (e) Obligations Absolute. The obligation of the Company to reimburse the Issuer for each drawing under each Credit and to repay each Borrowing, shall be absolute, unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following:
 - (i) any lack of validity or enforceability of such Credit, this Agreement, or any other Credit Document;
- (ii) the existence of any claim, counterclaim, set-off, defense or other right that the Company or any Subsidiary may have at any time against any beneficiary or any transferee of such Credit (or any Person for whom any such beneficiary or any such transferee may be acting), the Issuer or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by such Credit or any agreement or instrument relating thereto, or any unrelated transaction;
- (iii) any draft, demand, certificate or other document presented under such Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under such Credit;
- (iv) any payment by the Issuer under such Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Credit; or any payment made by the Issuer under such Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Credit, including any arising in connection with any proceeding under any Debtor Relief Law; or
- (v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Company or any Subsidiary.

The Company shall promptly examine a copy of each Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Company's instructions

or other irregularity, the Company will promptly notify the Issuer. The Company shall be conclusively deemed to have waived any such claim against the Issuer and its correspondents unless such notice is given as aforesaid.

- (f) Role of Issuer. Each Lender and the Company agree that, in paying any drawing under a Credit, the Issuer shall not have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by the Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. None of the Issuer, any Issuer-Related Person nor any of the respective correspondents, participants or assignees of the Issuer shall be liable to any Lender for (i) any action taken or omitted in connection herewith at the request or with the approval of the Required Lenders or the Required Credit-Linked Lenders, as applicable; (ii) any action taken or omitted in the absence of gross negligence or willful misconduct; or (iii) the due execution, effectiveness, validity or enforceability of any document or instrument related to any Credit or Request for Credit Extension. The Company hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Company's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuer, any Issuer-Related Person, nor any of the respective correspondents, participants or assignees of the Issuer, shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.01(e); provided, however, that anything in such clauses to the contrary notwithstanding, the Company may have a claim against the Issuer, and the Issuer may be liable to the Company, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Company which the Company proves were caused by the Issuer's willful misconduct or gross negligence or the Issuer's willful failure to pay under any Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Credit. In furtherance and not in limitation of the foregoing, the Issuer may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the Issuer shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.
- (g) Applicability of ISP98. Unless otherwise expressly agreed by the Issuer and the Company when a Credit is issued (including any such agreement applicable to an Existing Credit), the rules of the ISP shall apply to each Credit.
- (h) Issuer Documents. In the event of any conflict between the terms hereof, the terms of any Request for Credit Extension and the terms of any Issuer Document (other than any Credit), the terms hereof shall control.
 - (i) Existing Credits. The Company and the Issuer certify that Schedule 1.01 accurately and completely sets forth the Existing Credits.

- (j) Interest on Credit-Linked Deposit; Repayment of Credit-Linked Deposit.
- (i) The Administrative Agent hereby agrees to cause Bank of America to pay interest to the Administrative Agent, as holder of the Credit-Linked Deposit Account, and the Administrative Agent, as holder of the Credit-Linked Deposit Account, agrees to distribute to each Lender, from such interest payments received from Bank of America, such Lender's Pro Rata Share of interest on the outstanding amount of the Credit-Linked Deposits in the Credit-Linked Deposit Account at the rate per annum set forth on Schedule A annexed hereto. Such interest will be distributed to the Lenders by the Administrative Agent quarterly in arrears on the first Business Day following the end of each Interest Period relating to the Credit-Linked Deposits. The Administrative Agent shall compute all amounts due under this Section 2.01(j)(i) and shall notify the Company and such Lender of each such amount due.
- (ii) Subject to, and to the extent of, the Company's compliance with the cash-collateralization requirements set forth in Section 2.02, the Administrative Agent shall distribute the aggregate amount of any remaining Credit-Linked Deposits to the Lenders on the Maturity Date to the extent of the amount credited to the Credit-Linked Deposit Account. The Credit-Linked Deposit Account in the name of the Administrative Agent under its sole dominion and control, and held by it with Bank of America subject to the terms and conditions of this Agreement. No Lender, whether upon the receivership or liquidation of such Lender or otherwise, shall be entitled to its Credit-Linked Deposit other than as expressly provided in this Agreement.
- (iii) The Company shall have no right, title or interest in or to the Credit-Linked Deposits or the Credit-Linked Deposit Account and no obligations with respect thereto, it being acknowledged and agreed by the parties hereto that the making of the Credit-Linked Deposits by the Lenders, the deposit by the Administrative Agent of the Credit-Linked Deposits in the Credit-Linked Deposits Account, the provisions of this Section 2.01(j) and the application of the Credit-Linked Deposits in the manner contemplated by Section 2.01(c) constitute agreements among the Administrative Agent, the Issuer and each Lender in respect of the funding obligations of each Lender in respect of its Participation in Credits, and do not constitute any loan or extension of credit to the Company.

2.02 Cash Collateral

(a) Upon the request of the Administrative Agent, (i) if, as of the Credit Expiration Date, (A) any Credit for any reason remains outstanding and partially or wholly undrawn or (B) any amount remains available to be drawn under any Credit by reason of the operation of Rule 3.14 of the ISP, or (ii) upon the occurrence of the circumstances described in Section 8.02(c) requiring the Company to Cash Collateralize Credits, the Company shall within two Business Days after receipt of such request Cash Collateralize the then Outstanding Amount of all Credits (in an amount equal to the Outstanding Amount thereof). For purposes hereof, "Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Issuer and the Lenders, as collateral for the Credit Obligations, cash or deposit account balances ("Cash Collateral") pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent (which documents are hereby consented to by the Lenders). Derivatives of such term have corresponding meanings.

(b) The Company hereby grants the Administrative Agent, for the benefit of the Issuer and the Lenders, a security interest in all such Cash Collateral. Cash Collateral shall be maintained in blocked, non-interest bearing deposit accounts at Bank of America and shall be subject to such Lien documentation as the Administrative Agent shall reasonably request. The Company hereby authorizes the filing by the Administrative Agent of any UCC-1 financing statements in connection therewith.

2.03 Reduction of Total Credit-Linked Deposits; Conversion to Funded Term Loans.

(a) Reduction. The Company may, on or after the second anniversary of the Closing Date, upon notice from the Company to the Administrative Agent, permanently reduce the available amount of the Total Credit-Linked Deposit to an amount not less than the sum of the then Outstanding Amount of all Credit Obligations (other than Advances and Term Loans); provided that (i) any such notice shall be received by the Administrative Agent not later than 1:00 p.m., Dallas time, ten Business Days prior to the date of termination or reduction, (ii) any such partial reduction shall be in an aggregate amount of \$10,000,000 or any whole multiple of \$1,000,000 in excess thereof, or such lesser amount as would reduce the Total Credit-Linked Deposit to \$0; and (iii) a Prepayment Premium, if any, shall be payable with respect to any amounts reduced. Such notice shall specify the proposed effective date of such reduction and the amount of the Prepayment Premium, if any. The Administrative Agent will promptly notify the Lenders of any such notice of reduction of the Total Credit-Linked Deposit. Once reduced in accordance with this Section, the Total Credit-Linked Deposit may not be increased. Any reduction of the Total Credit-Linked Deposit shall be applied to each Lender according to its Pro Rata Share. All Facility Fees and Fronting Fees shall accrue until the effective date of any reduction of the Total Credit-Linked Deposit and shall be paid on the effective date of such reduction if the Total Credit-Linked Deposit is \$0 after giving effect to such reduction. Upon any such reduction, the Administrative Agent shall promptly remit to each Lender its Pro Rata Share of the amount of the reduction in the Total Credit-Linked Deposit and Prepayment Premium, if any.

(b) Conversion to Funded Term Loans.

(i) Subject to the terms and conditions set forth herein (including satisfaction of the applicable conditions specified in Sections 4.02 and 4.03), the Company may elect to (A) cause the Lenders to make, and each Lender agrees to make, funded loans ("Term Loans") to the Company in an aggregate principal amount up to the excess of (x) the Total Credit-Linked Deposit over (y) the Outstanding Amount of all Credit Obligations (other than Advances and Term Loans) (the amount of such excess being the "Unused Amount"), or (B) cause the Lenders to make, and each Lender agrees to make, one or more funded Term Loans in an aggregate principal amount up to the amount of a forthcoming drawing on any Credit, or (C) cause the Lenders to make, and each Lender agrees to make, one or more funded Term Loans in the aggregate amount of any Advances then outstanding by converting such Advances to Term Loans; in each case in accordance with the procedures set forth in this Section 2.03(b). Any Term Loans made under this Section 2.03(b) and subsequently repaid or prepaid may not be reborrowed.

- (ii) The Company may irrevocably request the making of one or more Term Loans in an aggregate amount up to the Unused Amount or in an aggregate amount up to the anticipated drawing on a Credit from the proceeds of Credit-Linked Deposits, or in an aggregate amount up to the aggregate amount of Advances then outstanding by converting such Advances to Term Loans, as applicable, by delivering a Request for Term Loans therefor to the Administrative Agent, appropriately completed and signed by a Responsible Officer or Authorized Representative of the Company, not later than 10 Business Days prior to the proposed date of making of the Term Loans; provided that the Term Loans requested pursuant to Section 2.03(b)(i)(A) shall only be made on a day which is the last day of an Interest Period with respect to the Credit-Linked Deposits and the Company shall in any event pay the amounts, if any, payable under Section 3.04; and provided further that the minimum aggregate amount of Term Loans made on any date shall be (A) \$25,000,000, in the case of Term Loans requested pursuant to Section 2.03(b)(i)(B), and (B) \$10,000,000, in the case of Term Loans requested pursuant to Sections 2.03(b)(i)(B) and 2.03(b)(i)(C), or in each case of clauses (A) and (B) of this clause (ii), such lesser amount which, when added to the applicable Outstanding Amount of all (x) Credits and (y) the Unreimbursed Amounts for which Advances have not been made would equal the Total Credit-Linked Deposit.
- (iii) Following receipt of a Request for Term Loans, the Administrative Agent shall promptly notify each Lender of its Pro Rata Share thereof. Upon satisfaction of the applicable conditions set forth in Sections 4.02 and 4.03, the Administrative Agent shall (A) in the case of Term Loans made pursuant to Section 2.03(b)(i)(A), debit the Credit-Linked Deposit Account in an amount equal to the aggregate amount of the proposed Term Loans (but not exceeding the Unused Amount) and shall make such funds available to the Company either by (i) crediting the account of the Company on the books of Bank of America with the amount of such funds or (ii) wire transferring such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Company, (B) in the case of Term Loans made pursuant to Section 2.03(b)(i)(B), debit the Credit-Linked Deposit Account in an amount equal to the aggregate amount of the proposed Term Loans and transfer such funds to the Issuer in reimbursement of the applicable drawing on a Credit, and (C) in the case of Term Loans made pursuant to Section 2.03(b)(i)(C), record in its books and records and on the Register the conversion of the applicable Advances into Term Loans, which credit or transfer or recordation, as applicable, shall be deemed to be the making of the Term Loans by the Lenders hereunder, and such date shall be the "Term Loan Issuance Date" (and the Administrative Agent shall forthwith deliver a notice to each Lender informing such Lender of the applicable Term Loan Issuance Date). Each Lender hereby irrevocably authorizes the Administrative Agent to charge the Credit-Linked Deposit Account for such purpose. No Lender shall have any obligation to make Term Loans to the Company other than from the proceeds of its Credit-Linked Deposits.
 - (iv) Interest Rate on Term Loans.
 - (A) Subject to the provisions of subsection (B) below, each Term Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the LIBO Rate for such Interest Period plus the Applicable Rate.

- (B) If any amount of the Term Loans is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Law. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.
- (C) Interest on each Term Loan shall be due and payable in arrears on the first Business Day following the end of each Interest Period applicable thereto, at such other times as may be specified herein and on the Maturity Date. To the fullest extent permitted by applicable Law, interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.
- (v) Repayment of Term Loans. The Company shall repay to the Administrative Agent for the account of the Lenders on the Maturity Date the aggregate principal amount of the Term Loans outstanding on such date.
- (vi) Optional Prepayment of Term Loans. The Company may, on or after the second anniversary of the Closing Date, at its option, upon notice to the Administrative Agent as provided below, prepay on any Business Day all, or from time to time any part of, the Term Loans (subject to the limitations sets forth below); provided that (i) any such notice shall be received by the Administrative Agent not later than 1:00 p.m., Dallas time, ten Business Days prior to the date of prepayment, (ii) any such partial prepayment shall be in an aggregate amount of \$1,000,000 or any whole multiple of \$1,000,000 in excess thereof, or such lesser amount as would reduce the Term Loans to \$0; and (iii) a Prepayment Premium, if any, shall be payable with respect to any amounts prepaid. Such notice shall specify the proposed effective date of such prepayment, the aggregate principal amount of the Term Notes to be prepaid on such date, the interest to be paid on the prepayment date with respect to such principal amount being prepaid, which shall be the last day of the Interest Period therefor, and the amount of the Prepayment Premium, if any. The Administrative Agent will promptly notify the Lenders of any such notice of prepayment of the Term Loans. Once prepaid in accordance with this Section, such Term Loans may not be reborrowed. Upon any such prepayment, the Administrative Agent shall promptly remit to each Lender its Pro Rata Share of the amount of the Prepayment of the Term Loans and Prepayment Premium, if any.

2.04 Interest on Borrowings Other Than Term Loans.

- (a) Each Borrowing (other than the Term Loans) shall bear interest on the outstanding principal amount thereof (i) from and including the applicable Borrowing date to and excluding the date six Business Days thereafter at an interest rate per annum equal at all times to the Base Rate plus 1.00% per annum, and (ii) from and including the sixth Business Day after the applicable Borrowing date at an interest rate per annum equal at all times to the Default Rate.
- (b) If any other amount payable by the Company under any Credit Document is not paid when due (without regard to any applicable grace periods) whether at stated maturity, by

acceleration or otherwise, each such amount shall thereafter bear interest at an interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Law.

- (c) Accrued and unpaid interest on all such unpaid amounts (including interest on past due interest) shall be due and payable upon demand.
- (d) Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.

2.05 Fees and Other Charges.

- (a) Facility Fee. The Company shall pay to the Administrative Agent for the account of each Lender in accordance with its Pro Rata Share, a fee (the "Facility Fee") equal to the Applicable Rate, expressed on a daily basis, times the actual daily amount of such Lender's Credit-Linked Deposit. The Facility Fee shall accrue effective as of the Closing Date and shall be calculated quarterly in arrears by the Administrative Agent. If there is any change in the Applicable Rate during any quarter, the actual daily Credit-Linked Deposit amount shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect. The Facility Fee shall be due and payable quarterly in arrears on the first Business Day following the end of each Interest Period relating to the Credit-Linked Deposits, commencing with the first such date to occur after the Closing Date, and on the Maturity Date; provided that, in connection with any reduction of the Total Credit-Linked Deposit under Section 2.03, the accrued Facility Fee calculated for the period ending on such date shall also be paid on the date of such reduction, and the following quarterly payment shall be calculated on the basis of the period from such reduction date to such quarterly payment date. The Facility Fee shall accrue at all times, including at any time during which one or more of the conditions in Article IV is not met.
- (b) Fronting Fee. The Company shall pay directly to the Issuer for its own account, a fee (the "Fronting Fee") equal to the rate per annum set forth in the letter agreement dated May 8, 2003 (the "Fee Letter"), among the Company, BAS and the Administrative Agent, times the daily amount of the Total Credit-Linked Deposit, due and payable quarterly in arrears on the first Business Day immediately following the last Business Day of each March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the Maturity Date; provided that, in connection with any reduction of the Total Credit-Linked Deposit under Section 2.03, the accrued Fronting Fee calculated for the period ending on such date shall also be paid on the date of such reduction, and the following quarterly payment shall be calculated on the basis of the period from such reduction date to such quarterly payment date. At any time that an Event of Default exists, the Fronting Fee shall be increased by adding 2% to the amount thereof determined as provided in the Fee Letter. The Fronting Fee shall accrue at all times, including at any time during which one or more of the conditions in Article IV is not met.
- (c) Documentary and Processing Charges Payable to Issuer. The Company shall pay directly to the Issuer for its own account the customary and standard issuance, commission, presentation, amendment and other processing fees, and other standard costs and charges (if any), of the Issuer or any Affiliate of the Issuer, relating to its Credits as from time to time in

effect and agreed upon between the Company and the Issuer. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.

(d) Placement and Administrative Fees. The Company shall pay a placement fee to BAS, and shall pay an administrative fee to the Administrative Agent for the Administrative Agent's own account, in the amounts and at the times specified in the Fee Letter, and if a successor Administrative Agent is appointed pursuant to Section 9.09 hereof, the Company shall pay the administrative fees of such successor Administrative Agent.

2.06 Computation of Interest and Fees.

- (a) All computations of interest when the Base Rate is determined by reference to Bank of America's "prime rate" shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. Computation of all other types of interest and all fees shall be calculated on the basis of a year of 360 days, and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day or 366-day year). Interest shall accrue on each Advance or Term Loan for the day on which the Advance or Term Loan is made, and shall not accrue on an Advance or Term Loan, or any portion thereof, for the day on which the Advance, such Term Loan or such portion is paid, provided that any Advance or Term Loan that is repaid on the same day on which it is made shall bear interest for one day.
- (b) The Administrative Agent shall promptly notify the Company and the Lenders of the LIBO Rate applicable to any Interest Period upon determination of such interest rate. The determination of LIBO Rate by the Administrative Agent shall be conclusive in the absence of manifest error. At any time that Advances or Term Loans accruing interest based on the Base Rate are outstanding, the Administrative Agent shall notify the Company and the Lenders of any change in Bank of America's prime rate used in determining the Base Rate promptly following the public announcement of such change.

2.07 Evidence of Debt.

- (a) The Credit-Linked Deposits, the Credit Extensions made by the Issuer, the participations of each Lender therein and any Advances and Term Loans made by each Lender shall be evidenced by one or more accounts or records maintained by the Issuer, such Lender and by the Administrative Agent in the ordinary course of business. The accounts or records maintained by the Issuer, the Administrative Agent and each Lender shall be conclusive absent manifest error of the amount of the Credit-Linked Deposits, the Credit Extensions made by the Issuer and any Advances and Term Loans made by each Lender and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of the Company hereunder to pay any amount owing with respect to the Credit Obligations. In the event of any conflict between the accounts and records maintained by the Issuer or any Lender and the accounts and records of the Administrative Agent shall control in the absence of manifest error.
- (b) In addition to the accounts and records referred to in subsection (a), (i) the Credit-Linked Deposit of, and the obligation of the Company to pay Advances to, each Lender shall be

evidenced by a Letter of Credit Linked Note in the form of Exhibit D attached hereto (a "Letter of Credit Linked Note"), executed by the Administrative Agent and by the Company, and (ii) the Term Loans of a Lender shall be evidenced by a Term Note in the form of Exhibit F attached hereto (a "Term Note"), executed by the Company.

2.08 Payments Generally.

- (a) All payments to be made by the Company to any Lender, the Administrative Agent, the Issuer or any other Person hereunder shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by the Company hereunder to be made directly to the Issuer shall be made to the Issuer in accordance with its payment instructions in Dollars and in immediately available funds not later than the times and on the dates specified herein. Except as otherwise expressly provided herein and except with respect to such payments to be made directly to the Issuer, all payments by the Company hereunder shall be made to the Administrative Agent, for the account of the respective Lenders or the Issuer to which such payment is owed, at the Administrative Agent's Office in Dollars and in immediately available funds not later than 1:00 p.m., Dallas time, on the date specified herein. The Administrative Agent will promptly distribute to the Issuer its applicable share as provided herein, or to each Lender its Pro Rata Share (or other applicable share as provided herein), of such payment in like funds as received by wire transfer to the Issuer in accordance with its payment instructions or to such Lender at its Lending Office. All payments received by the Administrative Agent after 1:00 p.m., Dallas time, or by the Issuer after the time specified herein or in any Issuer Document, shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.
- (b) If any payment to be made by the Company shall come due on a day other than a Business Day, or any other date specified hereunder would otherwise occur on a day other than a Business Day, then except as otherwise provided herein, such payment shall be made, or other date shall occur, on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.
- (c) The obligations of the Lenders hereunder to fund Participations in Credits are several and not joint. The failure of any Lender to fund its Participation on the Closing Date with its Credit-Linked Deposit shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so purchase its Participation.
- (d) Nothing herein shall be deemed to obligate any Lender to obtain the funds for the Credit-Linked Deposit in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for the Credit-Linked Deposit in any particular place or manner.
- **2.09 Sharing of Payments.** If, other than as expressly provided elsewhere herein, any Lender shall obtain on account of the Term Loans or Advances made by it or participations in Credit Obligations held by it, any payment (whether voluntary, involuntary, through the exercise of any right of set-off, or otherwise) in excess of its Pro Rata Share (or other share contemplated hereunder) thereof, such Lender shall immediately (a) notify the Administrative Agent of such

fact, and (b) purchase from the other Lenders such subparticipations in the Term Loans or Advances made by them or in the participations in Credit Obligations held by them as shall be necessary to cause such purchasing Lender to share the excess payment in respect thereof *pro rata* with each of them; provided, however, that if all or any portion of such excess payment is thereafter recovered from the purchasing Lender under any of the circumstances described in Section 10.06 (including pursuant to any settlement entered into by the purchasing Lender in its discretion), such purchase shall to that extent be rescinded and each other Lender shall repay to the purchasing Lender the purchase price paid therefor, together with an amount equal to such paying Lender's ratable share (according to the proportion of (i) the amount of such paying Lender's required repayment to (ii) the total amount so recovered from the purchasing Lender) of any interest or other amount paid or payable by the purchasing Lender in respect of the total amount so recovered, without further interest thereon. The Company agrees that any Lender so purchasing a subparticipation from another Lender may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of set-off) with respect to such subparticipation as fully as if such Lender were the direct creditor of the Company in the amount of such subparticipation. The Administrative Agent will keep records (which shall be conclusive and binding in the absence of manifest error) of subparticipations purchased under this Section and will in each case notify the Lenders following any such purchases or repayments. Each Lender that purchases a subparticipation pursuant to this Section shall from and after such purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Credit Obligations purchased to the same extent as though the purchasing Lender were the original owne

ARTICLE III TAXES AND YIELD PROTECTION

3.01 Taxes. (a) Subject to Section 10.15(a)(iii), any and all payments by the Company to or for the account of the Administrative Agent, the Issuer or any Lender under any Credit Document shall be made free and clear of and without deduction for any and all present or future taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and all liabilities with respect thereto, excluding, in the case of the Administrative Agent, the Issuer and each Lender, taxes imposed on or measured by its overall net income, and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the Laws of which the Administrative Agent, the Issuer or such Lender, as the case may be, is organized or maintains a lending office (all such non-excluded taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and liabilities being hereinafter referred to as "Taxes"). If the Company shall be required by any Laws to deduct any Taxes from or in respect of any sum payable under any Credit Document to the Administrative Agent, the Issuer or any Lender, (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section), the Administrative Agent, the Issuer or such Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) the Company shall make such deductions, (iii) the Company shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable Laws, and (iv) within 30 days after the date of such payment, the Company shall furnish to the

Administrative Agent (which shall promptly forward the same to the Issuer or such Lender) the original or a certified copy of a receipt evidencing payment thereof.

- (b) In addition, the Company agrees to pay any and all present or future stamp, court or documentary taxes and any other excise or property taxes or charges or similar levies which arise from any payment made under any Credit Document or from the execution, delivery, performance, enforcement or registration of, or otherwise with respect to, any Credit Document (hereinafter referred to as "Other Taxes").
- (c) If the Company shall be required to deduct or pay any Taxes or Other Taxes from or in respect of any sum payable under any Credit Document to the Administrative Agent, the Issuer or any Lender, the Company shall also pay to the Administrative Agent or to such Lender, as the case may be, at the time interest is paid, such additional amount that the Administrative Agent, the Issuer or any Lender specifies as necessary to preserve the after-tax yield (after factoring in all taxes, including taxes imposed on or measured by net income) that the Administrative Agent, the Issuer or any Lender would have received if such Taxes or Other Taxes had not been imposed.
- (d) The Company agrees to indemnify the Administrative Agent, the Issuer and each Lender for (i) the full amount of Taxes and Other Taxes (including any Taxes or Other Taxes imposed or asserted by any jurisdiction on amounts payable under this Section) paid by the Administrative Agent, the Issuer and such Lender, (ii) amounts payable under Section 3.01(c) and (iii) any liability (including additions to tax, penalties, interest and expenses) arising therefrom or with respect thereto, in each case whether or not such Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. Payment under this subsection (d) shall be made within 30 days after the date the Issuer, such Lender or the Administrative Agent makes a demand therefor.
- 3.02 Increased Cost and Reduced Return; Capital Adequacy or Risk-Based Capital. (a) If the Issuer or any Lender determines that as a result of the introduction of or any change in or in the interpretation of any Law, or the Issuer's or such Lender's compliance therewith, there shall be any increase in the cost to the Issuer or such Lender of issuing or participating in Credits or the funding or maintaining Term Loans or Advances, or a reduction in the amount received or receivable by the Issuer or such Lender in connection with the foregoing (excluding for purposes of this subsection (a) any such increased costs or reduction in amount resulting from (i) Taxes or Other Taxes (as to which Section 3.01 shall govern), or (ii) changes in the basis of taxation of overall net income or overall gross income by the United States or any foreign jurisdiction or any political subdivision of either thereof under the Laws of which the Issuer or such Lender is organized or has its Lending Office), then from time to time upon demand of the Issuer or such Lender (with a copy of such demand to the Administrative Agent), the Company shall pay to the Issuer or such Lender such additional amounts as will compensate the Issuer or such Lender for such increased cost or reduction; provided that such compensation will be limited to (A) the period commencing not more than 120 days prior to the date of such demand or (B) any longer period of retroactive effect of any such change or requirement for compliance if such demand is made 120 days or less after such change or requirement for compliance.

- (b) If the Issuer or any Lender determines that the introduction of any Law regarding capital adequacy or risk-based capital or any change therein or in the interpretation thereof, or compliance by the Issuer or such Lender (or its Lending Office) therewith, has the effect of reducing the rate of return on the capital of the Issuer or such Lender or any corporation controlling the Issuer or such Lender as a consequence of the Issuer's or such Lender's obligations (for avoidance of doubt not including outstanding Term Loans and Advances) hereunder (taking into consideration its policies with respect to capital adequacy or risk-based capital and the Issuer's or such Lender's desired return on capital), then from time to time upon demand of the Issuer or such Lender (with a copy of such demand to the Administrative Agent), the Company shall pay to the Issuer or such Lender such additional amounts as will compensate the Issuer or such Lender for such reduction; provided that such compensation will be limited to (A) the period commencing not more than 120 days prior to the date of such demand or (B) any longer period of retroactive effect of any such change or requirement for compliance if such demand is made 120 days or less after such change or requirement for compliance.
- 3.03 Illegality. If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender to make, maintain or fund LIBO Rate Term Loans, or to determine or charge interest rates based upon the LIBO Rate, then, on notice thereof by such Lender to the Company through the Administrative Agent, any obligation of such Lender to make or continue LIBO Rate Term Loans shall be suspended until such Lender notifies the Administrative Agent and the Company that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Company shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay the Term Loans of such Lender, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such LIBO Rate Term Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such LIBO Rate Term Loans. Upon any such prepayment, the Company shall also pay accrued interest on the amount so prepaid.
- **3.04 Funding Losses.** Upon demand of any Lender (with a copy to the Administrative Agent), or demand of the Administrative Agent, from time to time, the Company shall promptly compensate such Lender or the Administrative Agent, as applicable, for and hold such Lender or the Administrative Agent harmless from any loss, cost or expense incurred by it as a result of:
- (a) any payment or prepayment of any LIBO Rate Term Loan or Advance on a day other than the last day of the Interest Period for such Term Loan or Advance (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);
 - (b) any reduction of the Total Credit-Linked Deposits on a day other than the last day of an Interest Period therefor;
 - (c) any failure by the Company to prepay or borrow any Term Loan on the date or in the amount notified by the Company; or
- (d) any assignment of a LIBO Rate Term Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Company pursuant to Section 3.07;

including any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Term Loan or from fees payable to terminate the deposits from which such funds were obtained; provided, however, that such Lender shall use commercially reasonable efforts to mitigate any such loss, cost or expense.

For purposes of calculating amounts payable by the Company to the Lenders under this Section 3.04, each Lender shall be deemed to have funded each LIBO Rate Term Loan made by it at the LIBO Rate used in determining the LIBO Rate for such Term Loan by a matching deposit or other borrowing in the London interbank LIBOR market for a comparable amount and for a comparable period, whether or not such LIBO Rate Term Loan was in fact so funded.

- **3.05 Matters Applicable to All Requests for Compensation.** A certificate of the Administrative Agent, the Issuer or any Lender claiming compensation under this Article III and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. In determining such amount, the Administrative Agent, the Issuer or any Lender may use any reasonable averaging and attribution methods.
- **3.06 Survival.** All of the Company's obligations under this Article III shall survive termination of the Credit-Linked Deposits and repayment of all Credit Obligations and all other amounts due hereunder for a period of one (1) year; provided, however, any and all of the Company's obligations under this Article III related to Taxes shall survive for a period of three (3) years.
- 3.07 Replacement of Lenders. On each occasion that a Lender either makes a demand for compensation pursuant to Section 3.02 in an amount in excess of the amount that the Company would have had to pay pursuant to such Section if such Lender's Credit-Linked Deposit or Term Loan was held by a Lender with respect to which no such amounts would then be payable pursuant to such Section, the Company may, upon at least ten Business Days' prior written notice to the Issuer, each such Lender and the Administrative Agent, in whole permanently replace the Credit-Linked Deposit or Term Loan of such Lender; provided that, the Company shall replace such Credit-Linked Deposit or Term Loan with the Credit-Linked Deposit or Term Loan of (a) any other Lender, or, if no other Lender shall agree to replace such Lender (within 30 days after being offered such option), (b) a substitute lending institution which is an Eligible Assignee (in each case, a "Replacement Lender"). Such Replacement Lender shall, upon the effective date of replacement, purchase the Credit-Linked Deposit or Term Loan of, and the Credit Obligations owed to, such replaced Lender for the aggregate amount thereof and shall thereupon for all purposes become a "Lender" hereunder. Such notice from the Company shall specify an effective date for the replacement of such Lender, which date shall not be earlier than the tenth day after the day such notice is given. On the effective date of any replacement of such Lender pursuant to this Section 3.07 and in accordance with the terms of Section 10.07, the Company shall pay to the Administrative Agent for the account of such Lender (i) any fees or other amounts due to such Lender to the date of such replacement, (ii) accrued interest on the principal amount of outstanding Borrowings and the Credit-Linked Deposit or Term Loan held by such Lender to the date of such replacement and (iii) any other amount or amounts payable to such Lender pursuant to this Section 3.07, such Lender shall cease to be a

"Lender" hereunder, without any other or further act or deed on the part of any other Lender. No such replacement of any such Lender, the purchase of such Lender's Borrowings, Credit-Linked Deposit or Term Loans pursuant to this Section 3.07 shall affect (x) any liability or obligation of the Company or any other Lender to such replaced Lender which accrued on or prior to the date of such termination or (y) such replaced Lender's rights hereunder in respect of any such liability or obligation.

ARTICLE IV CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

- **4.01 Conditions of Credit-Linked Deposits and Initial Credit Extension.** The obligation of each Lender to fund its Credit-Linked Deposit and the obligation of the Issuer to make its initial Credit Extension hereunder are subject to satisfaction or waiver of the following conditions precedent:
- (a) The Administrative Agent's and Lenders' special counsel's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer or Secretary or assistant Secretary, as applicable of the signing Credit Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance reasonably satisfactory to the Administrative Agent, the Issuer, the Lenders and their respective legal counsel:
- (i) executed counterparts of this Agreement, sufficient in number for distribution to the Administrative Agent, the Issuer, each Lender and the Credit Parties;
- (ii) certified copies of the certificate or articles of incorporation (including all amendments thereto) of each Credit Party, certified as of a recent date prior to the Closing Date by the Secretary of State of the respective states of incorporation of such Credit Party;
- (iii) copies of the bylaws (including all amendments thereto) of each Credit Party, certified as of the Closing Date by the Secretary or assistant Secretary of such Credit Party;
- (iv) incumbency certificates of the Secretary or assistant Secretary of each Credit Party evidencing the identity, authority and capacity of each Responsible Officer or Authorized Representative thereof authorized to act as a Responsible Officer or Authorized Representative in connection with this Agreement and the other Credit Documents to which such Credit Party is a party;
- (v) certificates of the Secretary of State or other appropriate official of the respective states of incorporation of each the Credit Party, certifying as of a recent date prior to the Closing Date that such Credit Party is duly organized or formed and validly existing and in good standing in the jurisdiction of its organization or formation as of such date, and certificates of the Secretary of State or other appropriate official of the State of Texas, certifying as of a recent date prior to the Closing Date that such Credit Party is in good standing and qualified to engage in business in the State of Texas;

- (vi) true and correct copies of the resolutions adopted by the respective boards of directors of each Credit Party, duly certified as of the Closing Date by the Secretary or assistant Secretary of such Credit Party, (i) approving the transactions contemplated by (and the form of) this Agreement and the other Credit Documents and all other documents to be entered into in connection with this Agreement to be delivered by such Credit Party hereunder and (ii) authorizing the officers of such Credit Party, and each of them, acting individually or together, to execute and deliver each such document with such changes as such officer or officers shall approve;
- (vii) a certificate signed by a Responsible Officer of the Company certifying that (a) the representations and warranties contained in Article V, or any other Credit Document are true and correct, (b) no Default or Event of Default has occurred and is continuing, and (c) since the date of the Audited Financial Statements, no event or circumstance that either individually or in the aggregate has resulted in or could reasonably be expected to result in a Material Adverse Effect;
- (viii) opinions of counsel to each Credit Party, addressed to the Administrative Agent, the Issuer and each Lender, as to the matters set forth in Exhibit C-1 and such other matters concerning the Credit Parties and the Credit Documents as the Issuer or the Required Credit-Linked Lenders may reasonably request;
- (ix) opinions of counsel to the Administrative Agent and the Issuer, addressed to the Administrative Agent, the Issuer and each Lender, as to the matters set forth in Exhibit C-2 and Exhibit C-3 and such other matters as the Required Credit-Linked Lenders may reasonably request; and
- (x) opinion of special counsel to the Lenders, addressed to each Lender, as to the matters set forth in Exhibit C-4 and such other matters as the Required Credit-Linked Lenders may reasonably request.
 - (b) Any fees required to be paid on or before the Closing Date shall have been paid.
- (c) The Company shall have paid all reasonable Attorney Costs of (i) the Administrative Agent, and (ii) the special counsel to the Lenders, in each case to the extent invoiced prior to or on the Closing Date.
 - (d) The Closing Date shall have occurred on or prior to June 30, 2003.
- (e) On the Closing Date, each Lender's purchase of its Participation shall (i) be permitted by the laws and regulations of each jurisdiction to which such Lender is subject, without recourse to provisions (such as Section 1405(a)(8) of the New York Insurance Law) permitting limited investments by insurance companies without restriction as to the character of the particular investment, (ii) not violate any applicable law or regulation (including, without limitation, Regulation T, U or X of the Board of Governors of the Federal Reserve System) and (iii) not subject such Lender to any tax, penalty or liability under or pursuant to any applicable law or regulation, which law or regulation was not in effect on the date hereof. If requested by each Lender, such Lender shall have received an Officer's Certificate certifying as to such matters of

fact as such Lender may reasonably specify to enable such Lender to determine whether such purchase is so permitted. The wire transfer of funds on the Closing Date by each Lender in the amount of its Credit-Linked Deposit to the Administrative Agent shall constitute its confirmation as to satisfaction of this condition with respect to such Lender.

- (f) A Private Placement number issued by Standard & Poor's CUSIP Service Bureau (in cooperation with the Securities Valuation Office of the National Association of Insurance Commissioners) shall have been obtained for (i) the Letter of Credit Linked Notes and (ii) the Term Notes, which Private Placement number shall not be in the series of numbers issued to Bank of America Corporation and its affiliates, including but not limited to Bank of America.
 - (g) The Issuer shall have received evidence satisfactory to it that each Lender, in respect of its Participation hereunder, has funded its Credit-Linked Deposit.
 - (h) The Company and the Administrative Agent shall have executed and delivered to each Lender such Lender's Letter of Credit Linked Note.
- (i) The Administrative Agent and Bank of America in its capacity as depository shall have executed and delivered the Money Market Account Agreement and special counsel for the Lenders shall have received a copy thereof.
- **4.02 Conditions to all Credit Extensions by the Issuer.** The obligation of the Issuer to make any Credit Extension or honor any Request for Credit Extension is subject to the following conditions precedent:
- (a) The representations and warranties of the Company and each other Credit Party contained in Article V, or any other Credit Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct on and as of the date of such Credit Extension, except to the extent that such representations and warranties relate expressly and solely to an earlier date, and except to the extent of changes resulting from transactions contemplated or permitted by the Credit Documents and changes occurring in the ordinary course of business which either individually or in the aggregate do not result in a Material Adverse Effect.
 - (b) No Default or Event of Default shall exist, or would result from such proposed Credit Extension.
- (c) There has occurred since the date of the Audited Financial Statements, no event or circumstance that either individually or in the aggregate has resulted in or could reasonably be expected to result in a Material Adverse Effect.
- (d) Except in the case of any automatic renewal of any Auto-Renewal Credit or any automatic reinstatement of any Auto-Reinstating Credit, the Administrative Agent and the Issuer shall have received a Request for Credit Extension in accordance with the requirements hereof.

(e) The Administrative Agent shall have received, in form and substance satisfactory to it, such other assurances, certificates, documents or consents related to the foregoing as the Administrative Agent, the Issuer or the Required Lenders reasonably may require.

Each Request for Credit Extension submitted by the Company shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a), (b) and (c) have been satisfied on and as of the date of the applicable Credit Extension.

- **4.03 Conditions to Term Loans.** The obligation of the Lenders to make any Term Loans is subject to the following conditions precedent:
- (a) The Administrative Agent shall have received in accordance with the provisions of Section 2.03(b) hereof an originally executed Request for Term Loans.
- (b) The Company, by a duly authorized Senior Responsible Officer, shall have executed and delivered to the Administrative Agent on the applicable Term Loan Issuance Date, for delivery to the Lenders, Term Notes to evidence such Term Loans, dated as of the Term Loan Issuance Date, in the aggregate principal amount of the Term Loans and with other appropriate insertions.
- (c) The Administrative Agent shall have received an opinion of in-house counsel of the Company, addressed to the Administrative Agent and each Lender, substantially in the form of Exhibit H attached hereto.
- (d) Conditions Satisfied. The Company shall have delivered to the Administrative Agent an Officer's Certificate, dated as of the applicable Term Loan Issuance Date, in form and substance satisfactory to the Administrative Agent, certifying that all conditions precedent set forth in Sections 4.02(a) and (b) and 4.03 have been satisfied on and as of the applicable Term Loan Issuance Date.

Each Request for Term Loans submitted by the Company shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a) and (b) have been satisfied on and as of the date of the applicable Term Loan Issuance Date.

ARTICLE V REPRESENTATIONS AND WARRANTIES

The Company represents and warrants to the Administrative Agent and the Lenders that:

5.01 Organization; Power and Authority.

(a) The Company is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by Law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Company has the corporate power and authority to own or hold under lease the properties it

purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Credit Documents to which it is a party and to perform the provisions hereof and thereof.

(b) Holdings is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by Law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a material adverse effect on Holdings and its Subsidiaries, taken as a whole. Holdings has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Credit Documents to which it is a party and to perform the provisions hereof and thereof.

5.02 Authorization, etc.

- (a) This Agreement and each other Credit Document have been duly authorized by all necessary corporate action on the part of the Company, and this Agreement constitutes, and each other Credit Document to which the Company is a party when executed and delivered will constitute, a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (b) This Agreement and each other Credit Document have been duly authorized by all necessary corporate action on the part of Holdings, and this Agreement constitutes, and each other Credit Document to which Holdings is a party when executed and delivered will constitute, a legal, valid and binding obligation of Holdings enforceable against Holdings in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (c) To the extent Credits are being issued for the account of Subsidiaries, at the time of each such Credit Extension, such Credits do and will inure to the benefit of the Company, and the Company's business derives and will derive substantial benefits from the businesses of such Subsidiaries.
- **5.03 Disclosure.** The Company, through its agent, BAS, has delivered to each Lender a copy of a Private Placement Memorandum, dated May 2003 (the "Memorandum"), relating to the transactions contemplated hereby. As of its date, the Memorandum fairly describes, in all material respects, the general nature of the business and principal properties of the Company and its Subsidiaries. As of the Closing Date, this Agreement, the Memorandum, the documents, certificates or other writings delivered to the Lenders by or on behalf of the Company in connection with the transactions contemplated hereby and the financial statements listed in Schedule 5.05, taken as a whole, do not contain any untrue statement of a material fact or omit to

state any material fact necessary to make the statements therein not misleading in light of the circumstances under which they were made. Since December 31, 2002, there has been no change in the financial condition, operations, business or properties of the Company or any Subsidiary except changes that individually or in the aggregate could not reasonably be expected to have a Material Adverse Effect. There is no fact known to the Company that could reasonably be expected to have a Material Adverse Effect that has not been disclosed herein or in the Memorandum, in the Disclosure Documents, or in the other documents, certificates and other writings delivered to the Lenders by or on behalf of the Company specifically for use in connection with the transactions contemplated hereby.

5.04 Organization and Ownership of Shares of Subsidiaries; Affiliates.

- (a) Schedule 5.04 contains (except as noted therein) complete and correct lists as of the Closing Date of (i) each of the Company's Restricted Subsidiaries, showing, as to each Restricted Subsidiary, the correct name thereof and the jurisdiction of its organization, (ii) each of the Company's Affiliates, and (iii) the Company's directors and Senior Responsible Officers. As of the Closing Date, there are no Unrestricted Subsidiaries.
- (b) As of the Closing Date, all of the outstanding shares of capital stock or similar equity interests of each Restricted Subsidiary owned by the Company and its Subsidiaries have been validly issued, are fully paid and nonassessable and are owned by the Company or another Subsidiary free and clear of any Lien (except for any Permitted Lien or as otherwise disclosed in Schedule 5.04).
- (c) Each Subsidiary identified on Schedule 5.04 is a corporation or other legal entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, and is duly qualified as a foreign corporation or other legal entity and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each such Subsidiary has the corporate or other power and authority to own or hold under lease the properties it purports to own or hold under lease and to transact the business it transacts and proposes to transact.
- (d) As of the Closing Date, no Subsidiary is a party to, or otherwise subject to any legal restriction or any agreement (other than the agreements listed on Schedule 5.04 and customary limitations imposed by corporate law statutes) restricting the ability of such Subsidiary to pay dividends out of profits or make any other similar distributions of profits to the Company or any of its Subsidiaries that owns outstanding shares of capital stock or similar equity interests of such Subsidiary.

5.05 Financial Statements.

The Company has delivered to the Issuer and each Lender the financial statements of the Company and its Subsidiaries listed on Schedule 5.05. All of said financial statements (including in each case the related schedules and notes) fairly present in all material respects the consolidated financial position of the Company and its Subsidiaries as of the respective dates specified in such Schedule and the consolidated results of their operations and cash flows for the

respective periods so specified and have been prepared in accordance with GAAP consistently applied throughout the periods involved except as set forth in the notes thereto (subject, in the case of any interim financial statements, to normal year-end adjustments).

5.06 Compliance with Laws, Other Instruments, etc.

- (a) The execution, delivery and performance by the Company of this Agreement, the Letter of Credit Linked Notes and the Term Notes will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of the Company or any Subsidiary under, any indenture, mortgage, deed of trust, loan, note purchase or credit agreement, or lease, its corporate charter or by-laws, or any other agreement or instrument to which the Company or any Subsidiary is bound or by which the Company or any Subsidiary or any of their respective properties may be bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to the Company or any Subsidiary or (iii) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to the Company or any Subsidiary.
- (b) The execution, delivery and performance by Holdings of this Agreement will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of Holdings under, any indenture, mortgage, deed of trust, loan, note purchase or credit agreement, lease, corporate charter or by-laws, or any agreement or instrument to which Holdings is bound or by which Holdings or any of its properties may be bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to Holdings, or (iii) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to Holdings.

5.07 Governmental Authorizations, etc.

- (a) No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by the Company of this Agreement, the Letter of Credit Linked Notes or the Term Notes.
- (b) No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by Holdings of this Agreement.

5.08 Litigation; Observance of Agreements, Statutes and Orders.

(a) Except as disclosed in Schedule 5.08 or in the Disclosure Documents, there are no actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company or any Subsidiary or any property of the Company or any Subsidiary in any court or before any arbitrator of any kind or before or by any Governmental Authority that, individually or in the aggregate, could reasonably be expected, after offset for claims against insurers, to have a Material Adverse Effect.

- (b) Except as disclosed in Schedule 5.08 or in the Disclosure Documents, neither the Company nor any Subsidiary is in default under any term of any agreement or instrument to which it is a party or by which it is bound, or any order, judgment, decree or ruling of any court, arbitrator or Governmental Authority or is in violation of any applicable law, ordinance, rule or regulation (including without limitation Environmental Laws) of any Governmental Authority, which default or violation, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.
- **5.09 Taxes.** The Company and its Subsidiaries have filed all tax returns that are required to have been filed in any jurisdiction, and have paid all taxes shown to be due and payable on such returns and all other taxes and assessments levied upon them or their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent, except for any taxes and assessments (i) the amount of which is not individually or in the aggregate Material or (ii) the amount, applicability or validity of which is currently being contested in good faith by appropriate proceedings and with respect to which the Company or a Subsidiary, as the case may be, has established adequate reserves in accordance with GAAP. The Company knows of no basis for any other tax or assessment that could reasonably be expected to have a Material Adverse Effect. The charges, accruals and reserves on the books of the Company and its Subsidiaries in respect of federal, state or other taxes for all fiscal periods are adequate in accordance with GAAP. The federal income tax liabilities of the Company and its Subsidiaries have been determined by the IRS and paid for all fiscal years up to and including the fiscal year ended December 31, 1987.
- **5.10 Title to Property; Leases.** The Company and its Restricted Subsidiaries have good and sufficient title to their respective properties that the Company and the Subsidiaries own or purport to own that individually or in the aggregate are Material, including all such properties reflected in the most recent audited balance sheet referred to in Section 5.05 or purported to have been acquired by the Company or any Restricted Subsidiary after said date (except as sold or otherwise disposed of in the ordinary course of business), in each case free and clear of Liens prohibited by this Agreement. All leases that individually or in the aggregate are Material are valid and subsisting and are in full force and effect in all material respects.
 - **5.11 Licenses, Permits, etc.** Except as disclosed in Schedule 5.11,
- (a) the Company and its Restricted Subsidiaries own or possess all licenses, permits, franchises, authorizations, patents, copyrights, service marks, trademarks and trade names, or rights thereto, that individually or in the aggregate are Material, without known material conflict with the rights of others;
- (b) to the best knowledge of the Company, no product of the Company or any of its Restricted Subsidiaries infringes in any Material respect any license, permit, franchise, authorization, patent, copyright, service mark, trademark, trade name or other right owned by any other Person; and
- (c) to the best knowledge of the Company, there is no Material violation by any Person of any right of the Company or any of its Restricted Subsidiaries with respect to any patent,

copyright, service mark, trademark, trade name or other right owned or used by the Company or any of its Restricted Subsidiaries.

5.12 Compliance with ERISA.

- (a) The Company and each ERISA Affiliate have operated and administered each Plan in compliance with all applicable laws except for such instances of noncompliance as have not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any ERISA Affiliate has incurred any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans (as defined in Section 3 of ERISA), and no event, transaction or condition has occurred or exists that could reasonably be expected to result in the incurrence of any such liability by the Company or any ERISA Affiliate, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate, in either case pursuant to Title I or IV of ERISA or to such penalty or excise tax provisions or to Section 401(a)(29) or 412 of the Code, other than such liabilities or Liens as would not be individually or in the aggregate Material.
- (b) The present value of the aggregate accrued benefits under each of the Plans that are subject to Title IV of ERISA (other than Multiemployer Plans), determined as of the valuation date for the most recently ended plan year on the basis of the actuarial assumptions specified for funding purposes in such Plan's most recent actuarial valuation report, did not exceed the aggregate current value of the assets of such Plan allocable to such benefits (i) in the case of any single Plan, by an amount which, if said amount were immediately contributed to the Plan, could have a Material Adverse Effect, and (ii) in the case of all said Plans, by an amount which, if said amount were immediately contributed to the Plans, could have a Material Adverse Effect.
- (c) The Company and its ERISA Affiliates have not incurred withdrawal liabilities (and are not subject to contingent withdrawal liabilities) under Section 4201 or 4204 of ERISA in respect of Multiemployer Plans that individually or in the aggregate are Material.
- (d) The expected obligation for postretirement benefits other than pensions (determined as of the last day of the Company's most recently ended fiscal year in accordance with Financial Accounting Standards Board Statement No. 106, without regard to liabilities attributable to continuation coverage mandated by Section 4980B of the Code and after taking into account any governmental reimbursements) of the Company and its Subsidiaries is not Material.
- (e) The execution and delivery of this Agreement and the issuance and sale of the Letter of Credit Linked Notes and the Term Notes will not involve any transaction that is subject to the prohibitions of Section 406 of ERISA or in connection with which a tax could be imposed pursuant to Section 4975(c)(1)(A)-(D) of the Code. The representations by the Company in the first sentence of this Section 5.12(e) are made in reliance upon and subject to the accuracy of each Lender's representations in Section 10.10(b) as to the sources of the funds used to pay the purchase price of its Participation under this Agreement.
- **5.13 Private Offering by Company.** Neither the Company nor anyone acting on its behalf has offered the Letter of Credit Linked Notes or Term Notes or any similar Securities for sale to, or solicited any offer to buy any of the same from, or otherwise approached or negotiated

in respect thereof with, any Person other than the Lenders and not more than seventeen (17) other Institutional Investors, each of which has been offered the Letter of Credit Linked Notes and Term Notes at a private sale for investment. Neither the Company nor anyone acting on its behalf has taken, or will take, any action that would subject the issuance or sale of the Letter of Credit Linked Notes or Term Notes to the registration requirements of Section 5 of the Securities Act

- **5.14 Use of Proceeds; Margin Regulations.** The Company will apply the proceeds from the Advances and the Term Loans hereunder for general corporate purposes of the Company and its Subsidiaries. No part of the proceeds from the Credit Extensions hereunder will be used, directly or indirectly, for the purpose of buying or carrying any margin stock within the meaning of Regulation U of the FRB (12 CFR 221), or for the purpose of buying or carrying or trading in any Securities under such circumstances as to involve the Company in a violation of Regulation X of said Board (12 CFR 224) or to involve any broker or dealer in a violation of Regulation T of the FRB (12 CFR 220). Margin stock does not constitute more than 5.0% of the value of the consolidated assets of the Company and its Subsidiaries and the Company does not have any present intention that margin stock will constitute more than 5.0% of the value of such assets. As used in this Section, the terms "margin stock" and "purpose of buying or carrying" shall have the meanings assigned to them in said Regulation U.
- **5.15 Existing Indebtedness.** Except as described therein, Schedule 5.15 sets forth a complete and correct list of all outstanding Indebtedness of the Company and its Restricted Subsidiaries as of March 31, 2003, specifying whether such Indebtedness is secured or unsecured. From March 31, 2003 to the Closing Date, there has been no material change in the amounts, interest rates, sinking funds, installment payments or maturities of such Indebtedness of the Company or its Restricted Subsidiaries. As of the Closing Date, neither the Company nor any Restricted Subsidiary is in default and no waiver of default is currently in effect, in the payment of any principal or interest on any such Indebtedness of the Company or such Restricted Subsidiary, and no event or condition exists with respect to any such Indebtedness of the Company or any Restricted Subsidiary, that would permit (or that with notice or the lapse of time, or both, would permit) one or more Persons to cause such Indebtedness to become due and payable before its stated maturity or before its regularly scheduled dates of payment.
- **5.16 Foreign Assets Control Regulations, etc.** Neither any Credit Extension to the Company hereunder nor its use of the proceeds thereof will violate the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto. Without limiting the foregoing, neither the Company nor any of its Subsidiaries or its Affiliates (a) is or will become a Person whose property or interests in property are blocked pursuant to Section 1 of Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)) or (b) engages or will engage in any dealings or transactions, or be otherwise associated, with any such Person. The Company and its Subsidiaries and its Affiliates are in compliance, in all Material respects, with the Uniting And Strengthening America By Providing Appropriate Tools Required To Intercept And Obstruct Terrorism (USA Patriot Act of 2001). No part of the proceeds from any Credit

Extension hereunder will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

- **5.17 Status under Certain Statutes.** Neither the Company nor any Restricted Subsidiary is subject to regulation under the Investment Company Act of 1940, as amended, the Public Utility Holding Company Act of 1935, as amended, or, except as disclosed in Schedule 5.17, the Federal Power Act, as amended.
- **5.18** Environmental Matters. Except as disclosed in the Disclosure Documents, neither the Company nor any Restricted Subsidiary has knowledge of any Material claim or has received any notice of any Material claim, and no proceeding has been instituted raising any Material claim against the Company or any of its Restricted Subsidiaries or any of their respective real properties now or formerly owned, leased or operated by any of them or other assets, alleging any damage to the environment or violation of any Environmental Laws, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect. Except as otherwise disclosed to each Lender in writing,
- (a) neither the Company nor any Restricted Subsidiary has knowledge of any facts which would give rise to any claim, public or private, of violation of Environmental Laws or damage to the environment emanating from, occurring on or in any way related to real properties now or formerly owned, leased or operated by any of them or to other assets or their use, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect;
- (b) neither the Company nor any of its Restricted Subsidiaries has stored any Hazardous Materials on real properties now or formerly owned, leased or operated by any of them in a manner contrary to any Environmental Laws and has not disposed of any Hazardous Materials in a manner contrary to any Environmental Laws, in each case in any manner that could reasonably be expected to result in a Material Adverse Effect; and
- (c) all buildings on all real properties now owned, leased or operated by the Company or any of its Restricted Subsidiaries are in compliance with applicable Environmental Laws, except where failure to comply could not reasonably be expected to result in a Material Adverse Effect.
- **5.19 Compliance With Issuance Limits.** Both before and after giving effect to any Credit Extension, all Credits issued and outstanding hereunder are in full compliance with the Issuance Limits.
- **5.20 Tax Shelter Regulations.** The Company does not intend to treat the Advances, Term Loans and Credits and related transactions as being a "reportable transaction" (within the meaning of Treasury Regulation Section 1.6011-4).

ARTICLE VI AFFIRMATIVE COVENANTS

So long as any Lender shall have any Credit-Linked Deposit, Letter of Credit Linked Note or Term Note hereunder or any Credit hereunder shall remain outstanding:

- **6.01 Financial and Business Information.** The Company shall deliver to the Administrative Agent, the Issuer and to each of the Lenders, in form and detail satisfactory to the Administrative Agent, the Issuer and the Required Credit-Linked Lenders:
- (a) Quarterly Statements within 60 days after the end of each quarterly fiscal period in each fiscal year of the Company (other than the last quarterly fiscal period of each such fiscal year), duplicate copies of,
 - (i) a consolidated balance sheet of the Company and its Subsidiaries as at the end of such quarter, and
- (ii) consolidated statements of income, changes in shareholders' equity and cash flows of the Company and its Subsidiaries, for such quarter and (in the case of the second and third quarters) for the portion of the fiscal year ending with such quarter,

setting forth in each case in comparative form the figures for the corresponding periods in the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP applicable to quarterly financial statements generally, and certified by a Senior Financial Officer as fairly presenting, in all material respects, the financial position of the companies being reported on and their results of operations and cash flows, subject to changes resulting from year-end adjustments, provided that delivery within the time period specified above of copies of the Company's Quarterly Report on Form 10-Q prepared in compliance with the requirements therefor and filed with the SEC, if applicable, shall be deemed to satisfy the requirements of this Section 6.01(a) so long as such Quarterly Report on Form 10-Q includes the financial statements identified in clauses (i) and (ii) above;

- (b) Annual Statements within 100 days after the end of each fiscal year of the Company, duplicate copies of,
 - (i) a consolidated balance sheet of the Company and its Subsidiaries, as at the end of such year, and
 - (ii) consolidated statements of income, shareholder's equity and cash flows of the Company and its Subsidiaries, for such year,

setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP, and accompanied by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that such financial statements present fairly, in all material respects, the financial position of the companies being reported upon and their results of operations and cash flows and have been prepared in conformity with GAAP, and that the examination of such accountants in connection

with such financial statements has been made in accordance with generally accepted auditing standards, and that such audit provides a reasonable basis for such opinion in the circumstances; provided that the delivery within the time period specified above of the Company's Annual Report on Form 10-K for such fiscal year (together with the Company's annual report to shareholders, if any, prepared pursuant to Rule 14a-3 under the Exchange Act) prepared in accordance with the requirements therefor and filed with the SEC, if applicable, shall be deemed to satisfy the requirements of this Section 6.01(b) so long as such Annual Report on Form 10-K includes the financial statements identified in clauses (i) and (ii) above;

- (c) SEC and Other Reports promptly, following the filing or mailing thereof, copies of all material of a financial nature filed with the SEC or sent to the Company's and its Subsidiaries' stockholders generally;
- (d) Outstanding Credits not later than five (5) Business Days after the end of each quarter, the information with respect to the outstanding Credits as is required under Section 2.01(j);
 - (e) [intentionally omitted]
- (f) Notice of Default or Event of Default promptly, and in any event within five (5) Business Days after a Responsible Officer becoming aware of the existence of any Default or Event of Default, a written notice specifying the nature and period of existence thereof and what action the Company is taking or proposes to take with respect thereto;
- (g) ERISA Matters promptly, and in any event within five (5) Business Days after a Responsible Officer becoming aware of any of the following, a written notice setting forth the nature thereof and the action, if any, that the Company or an ERISA Affiliate proposes to take with respect thereto:
- (i) with respect to any Plan, any reportable event, as defined in Section 4043(b) of ERISA and the regulations thereunder, for which notice thereof has not been waived pursuant to such regulations as in effect on the date hereof; or
- (ii) the taking by the PBGC of steps to institute, or the threatening by the PBGC of the institution of, proceedings under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, or the receipt by the Company or any ERISA Affiliate of a notice from a Multiemployer Plan that such action has been taken by the PBGC with respect to such Multiemployer Plan; or
- (iii) any event, transaction or condition that would result in the incurrence of any liability by the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or such penalty or excise tax provisions, if such liability or Lien, taken together with any other such liabilities or Liens then existing, could reasonably be expected to have a Material Adverse Effect;

- (h) Notices from Governmental Authority promptly, and in any event within 30 days of receipt thereof, copies of any notice to the Company or any Subsidiary from any federal or state Governmental Authority relating to any order, ruling, statute or other law or regulation that the Company reasonably believes could have a Material Adverse Effect;
- (i) Additional Reporting Requirement in the event that Unrestricted Subsidiaries account for more than 10% of the Consolidated Total Assets of the Company and its Subsidiaries, or more than 10% of the consolidated revenue of the Company and its Subsidiaries, in each case as of the last day of the fiscal quarter or fiscal year for which the Company is delivering financial information pursuant to Sections 6.01(a) and (b), then each such set of financial information delivered pursuant to Sections 6.01(a) and (b) shall be accompanied by unaudited financial statements for all Unrestricted Subsidiaries of the Company taken as a group, together with consolidating statements reflecting eliminations or adjustments required to reconcile such group statements to the consolidated financial statements of the Company and its Subsidiaries;
- (j) Requested Information with reasonable promptness, such other data and information relating to the business, operations, affairs, financial condition, assets or properties of the Company or any its Subsidiaries or relating to the ability of the Company to perform its obligations hereunder, under the Letter of Credit Linked Notes and under the Term Notes or the ability of Holdings to perform its obligations hereunder as from time to time may be reasonably requested by any Lender, including without limitation and subject to Section 10.07, such information as is required by SEC Rule 144A under the Securities Act to be delivered to a prospective transferee of the Letter of Credit Linked Notes or Term Notes.
- **6.02 Officer's Certificate.** Each set of financial statements delivered to a Lender pursuant to Section 6.01(a) or Section 6.01(b) hereof shall be accompanied by a certificate of a Senior Financial Officer setting forth:
- (a) Covenant Compliance the information (including reasonably detailed calculations) required in order to establish whether the Company was in compliance with the requirements of Sections 7.02(c)(iv), 7.03(m), 7.04 and 7.05, during the quarterly or annual period covered by the statements then being furnished (including with respect to each such Section, where applicable, the calculations of the maximum or minimum amount, ratio or percentage, as the case may be, permissible under the terms of such Sections, and the calculation of the amount, ratio or percentage then in existence); and
- (b) Event of Default a statement that such officer has reviewed the relevant terms hereof and has made, or caused to be made, under his or her supervision, a review of the transactions and conditions of the Company and its Subsidiaries from the beginning of the quarterly or annual period covered by the statements then being furnished to the date of the certificate and that such review shall not have disclosed the existence during such period of any condition or event that constitutes a Default or an Event of Default or, if any such condition or event existed or exists (including, without limitation, any such event or condition resulting from the failure of the Company or any Subsidiary to comply with any Environmental Law), specifying the nature and period of existence thereof and what action the Company shall have taken or proposes to take with respect thereto.

- **6.03 Inspection.** The Company shall permit the representatives of each Lender:
- (a) No Default if no Default or Event of Default then exists, at the expense of such Lender, to visit the principal executive office of the Company and to discuss the affairs, finances and accounts of the Company and its Restricted Subsidiaries with the Company's officers, all at such reasonable times during normal business hours and as often as may be reasonably requested in writing; and
- (b) Default if a Default or Event of Default then exists, at the expense of the Company and upon reasonable prior notice, to visit the principal executive office of the Company, to examine all relevant books of account, records, reports and other papers relating to the Company's and Holdings' performance of their respective obligations under this Agreement, the Letter of Credit Linked Notes and the Term Notes and to make copies and abstracts therefrom, and to discuss the Company's and its Restricted Subsidiaries' affairs, finances and accounts with the Company's officers and independent public accountants (and by this provision the Company authorizes said accountants to discuss the affairs, finances and accounts of the Company and its Restricted Subsidiaries), all at such reasonable times during normal business hours and as often as may be reasonably requested.
- **6.04 Compliance with Laws.** The Company will, and will cause each of its Subsidiaries to, comply with all laws, ordinances or governmental rules or regulations to which each of them is subject, including, without limitation, Environmental Laws, and will obtain and maintain in effect all licenses, certificates, permits, franchises and other governmental authorizations necessary to the ownership of their respective properties or to the conduct of their respective businesses, in each case to the extent necessary to ensure that non-compliance with such laws, ordinances or governmental rules or regulations or failures to obtain or maintain in effect such licenses, certificates, permits, franchises and other governmental authorizations could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **6.05 Insurance.** The Company will, and will cause each of its Restricted Subsidiaries to, maintain, with financially sound and reputable insurers, insurance with respect to their respective properties and businesses against such risks, of such types, on such terms and in such amounts as is customary in the case of entities of established reputations engaged in the same or a similar business and similarly situated.
- **6.06 Maintenance of Properties.** The Company will, and will cause each of its Restricted Subsidiaries to, maintain and keep, or cause to be maintained and keept, their respective properties in good repair, working order and condition (other than ordinary wear and tear), so that the business carried on in connection therewith may be properly conducted at all times, provided that this Section shall not prevent the Company or any Restricted Subsidiary from discontinuing the operation and the maintenance of any of its properties or failing to carry on any such business if the Company has concluded that such discontinuance or failure could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **6.07 Payment of Taxes and Claims.** The Company will, and will cause each of its Subsidiaries to, file all income tax and similar tax returns required to be filed in any jurisdiction and to pay and discharge all taxes shown to be due and payable on such returns and pay all other

taxes, assessments, governmental charges, or levies imposed on them or any of their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent and all claims for which sums have become due and payable that have or might become a Lien on properties or assets of the Company or any Subsidiary, provided that neither the Company nor any Subsidiary need file any such returns or pay any such tax or assessment or claims (i) if the amount, applicability or validity thereof is contested by the Company or such Subsidiary on a timely basis in good faith and in appropriate proceedings, diligently conducted, and the Company or a Subsidiary has established adequate reserves therefor in accordance with GAAP on the books of the Company or such Subsidiary, or (ii) to the extent that the failure to file such returns and the nonpayment of all such taxes, assessments and claims, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

- **6.08 Corporate Existence, Etc.** Subject to Section 7.02, the Company will at all times preserve and keep in full force and effect its corporate existence, and the Company will at all times preserve and keep in full force and effect the corporate existence of each of its Restricted Subsidiaries (unless merged into the Company or a Restricted Subsidiary) and all rights and franchises of the Company and its Restricted Subsidiaries, unless, in the good faith judgment of the Company, the termination of or failure to preserve and keep in full force and effect such corporate existence, right or franchise could not, individually or in the aggregate, have a Material Adverse Effect.
- **6.09 Books and Records.** The Company will, and will cause each Restricted Subsidiary to, (a) maintain proper books of record and account, in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of the Company or such Restricted Subsidiary, as the case may be; and (b) maintain such books of record and account in material conformity with all applicable requirements of any Governmental Authority having regulatory jurisdiction over the Company or such Restricted Subsidiary, as the case may be.
- **6.10 Purpose of Credits.** Each Credit shall be used directly or indirectly to cover (a) a default in the performance of any non-financial or commercial obligations of the Company or any Subsidiary under specific contracts, including a Credit issued in favor of a bank or other surety who in connection therewith issues a guarantee or similar undertaking, performance bond, surety bond or other similar instrument that covers a default in the performance of any non-financial or commercial obligations under specific contracts or the payment of any financial contractual obligation, or (b) the payment of any financial contractual obligation of the Company or any Subsidiary.

ARTICLE VII NEGATIVE COVENANTS

So long as any Lender shall have any Credit-Linked Deposit, any Letter of Credit Linked Note or any Term Note hereunder or any Credit shall remain outstanding:

7.01 Transactions with Affiliates. The Company will not, and will not permit any Restricted Subsidiary to, enter into directly or indirectly any Material transaction or Material

group of related transactions (including without limitation the purchase, lease, sale or exchange of properties of any kind or the rendering of any service) with any Affiliate (other than the Company or another Restricted Subsidiary), except in the ordinary course and pursuant to the reasonable requirements of the Company's or such Restricted Subsidiary's business and upon fair and reasonable terms no less favorable to the Company or such Restricted Subsidiary than would be obtainable in a comparable arm's-length transaction with a Person not an Affiliate.

7.02 Merger, Consolidation, Sale of Assets, etc.

- (a) The Company will not, and will not permit any Restricted Subsidiary to, consolidate with or merge with any other Person unless immediately after giving effect to any consolidation or merger no Default or Event of Default would exist and:
- (i) in the case of a consolidation or merger of a Restricted Subsidiary, either (x) the surviving or continuing corporation is the Company or another Restricted Subsidiary or (y) such consolidation or merger, if considered as the sale of the assets of such Restricted Subsidiary to such other Person, would be permitted by Section 7.02(c); and
- (ii) in the case of a consolidation or merger of the Company, the successor corporation which results from such consolidation or merger (the "surviving corporation"), if not the Company, shall have executed and delivered to the Issuer, the Administrative Agent and each Lender in writing its assumption of the due and punctual payment of all Credit Obligations, and the due and punctual performance and observation of all covenants in this Agreement, the Letter of Credit Linked Notes and the Term Notes, to be performed or observed by the Company hereunder and under the Credit Documents and shall furnish to the Issuer, the Administrative Agent and such Lenders a customary opinion of counsel solely to the effect that the instrument of assumption has been duly authorized, executed and delivered and constitutes the legal, valid and binding contract and agreement of the surviving corporation enforceable in accordance with its terms, except as enforcement of such terms may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the enforcement of creditors' rights generally and by general equitable principles.
- (b) The Company will not sell, lease (as lessor) or otherwise transfer all or substantially all of its assets in a single transaction or series of transactions to any Person unless:
 - (i) immediately after giving effect to such transaction, no Default or Event of Default shall have occurred and be continuing; and
- (ii) the successor corporation to which all or substantially all of the Company's assets have been sold, leased or transferred (the "successor corporation") shall have executed and delivered to the Issuer, the Administrative Agent and each Lender in writing its assumption of the due and punctual payment of all Credit Obligations, and the due and punctual performance and observation of all covenants in this Agreement, the Letter of Credit Linked Notes and the Term Notes, to be performed or observed by the Company hereunder and under the Credit Documents and shall furnish to the Issuer, the Administrative Agent and such Lenders a customary opinion of counsel solely to the effect that the instrument of assumption has been duly authorized, executed and delivered and constitutes the legal, valid and binding contract and

agreement of the successor corporation enforceable in accordance with its terms, except as enforcement of such terms may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the enforcement of creditors' rights generally and by general equitable principles.

No such conveyance, transfer or lease of all or substantially all of the assets of the Company shall have the effect of releasing the Company or any successor corporation that shall theretofore have become such in the manner prescribed in this Section 7.02 from its liability under this Agreement or the Credit Documents.

- (c) The Company will not, and will not permit any Restricted Subsidiary to, sell, lease (as lessor), transfer, abandon or otherwise dispose of (collectively, a "Disposition") any assets in one or a series of transactions (including, without limitation, in any sale-leaseback transaction), other than those assets consisting of accounts receivable sold through any Permitted Receivables Transaction, to any Person other than:
- (i) Dispositions in the ordinary course of business of inventory or of equipment, fixtures, supplies or materials no longer required in the business of the Company or any of its Restricted Subsidiaries or that is obsolete;
 - (ii) Dispositions by the Company to a Restricted Subsidiary or by a Restricted Subsidiary to the Company or to another Restricted Subsidiary;
 - (iii) the sale of assets for cash or other property if all of the following conditions are met:
 - (A) such assets (valued at book value) do not constitute a "substantial part" of the assets of the Company and its Restricted Subsidiaries;
 - (B) in the opinion of a Responsible Officer of the Company, the sale is for fair value and is in the best interests of the Company; and
 - (C) immediately after the consummation of the transaction and after giving effect thereto, no Default or Event of Default would exist; or
- (iv) the sale of assets meeting the conditions set forth in clauses (B) and (C) of clause (iii) above, as long as the cash proceeds from such sale in excess of a substantial part of the assets of the Company and its Restricted Subsidiaries (net of expenses of sale and an amount reasonably estimated by the Company to represent incremental income taxes due as a result of such sale) are (i) applied within 365 days of the date of receipt to the acquisition of productive assets useful and intended to be used in the operation or expansion of the business of the Company or its Restricted Subsidiaries, and/or (ii) used to repay any Indebtedness of the Company or its Restricted Subsidiaries (other than (x) Indebtedness that is in any manner subordinated in right of payment in any respect to Indebtedness arising or incurred under this Agreement, (y) Indebtedness owing to the Company, any of its Subsidiaries or any Affiliate, and (z) Indebtedness in respect of any revolving credit or similar credit facility providing the Company or any of its Restricted Subsidiaries with the right to obtain loans or other extensions

of credit from time to time, except to the extent that in connection with such payment of Indebtedness the availability of credit under such credit facility is permanently reduced by an amount not less than the amount of such proceeds applied to the payment of such Indebtedness).

- (d) For purposes of Section 7.02(c), a sale of assets will be deemed to involve a "substantial part" of the assets of the Company and its Restricted Subsidiaries if the book value of such assets, together with all other assets sold, including mergers effected pursuant to Section 7.02(a)(i)(y) and the issuance, sale, transfer or disposition of stock or other Securities of Restricted Subsidiaries (except to the Company or another Restricted Subsidiary) effected pursuant to Section 7.02(e), during the same fiscal year (except (1) those assets sold pursuant to clauses (i) and (ii) of Section 7.02(c), and (2) those assets consisting of receivables sold through any Permitted Receivables Transaction permitted under Section 7.03(i)) equals at least 15.00% of the Consolidated Total Assets of the Company and its Restricted Subsidiaries determined as of the end of the immediately preceding fiscal year.
- (e) The Company will not permit any Restricted Subsidiary to issue shares of stock or other Securities to any Person other than the Company or another Restricted Subsidiary. The Company will not, and will not permit any Restricted Subsidiary to, issue, sell, transfer or otherwise dispose of its interest in any stock or any other Securities (or any options or warrants to purchase stock or other Securities exchangeable for or convertible into stock or other Securities) of any Restricted Subsidiary (except to the Company or another Restricted Subsidiary) unless such issuance, sale, transfer or disposition (valued at book value) would be permitted under Section 7.02(c).
- **7.03 Liens.** The Company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly create, incur, assume or permit to exist (upon the happening of a contingency or otherwise) any Lien on or with respect to any of its properties or assets, whether now owned or hereafter acquired, except for the following (which are collectively referred to as "Permitted Liens"):
- (a) Mechanics', materialmen's, carriers' or other like liens arising in the ordinary course of business (including construction of facilities) in respect of obligations that are not due or which are being contested in good faith;
- (b) Liens arising by reason of deposits with, or the giving of any form of security to, any governmental agency or any body created or approved by law or governmental regulation, which is required by law or governmental regulation as a condition to the transaction of any business or the exercise of any privilege, franchise or license (including, without limitation, any Lien arising by reason of one or more letters of credit in connection with any international waste management contract to be performed by the Company or any of its Subsidiaries or their respective Affiliates);
- (c) Liens for taxes, assessments or governmental charges or levies not yet delinquent or Liens for taxes, assessments or governmental charges or levies already delinquent but the validity of which is being contested in good faith;

- (d) Liens (including judgment liens) arising in connection with legal proceedings so long as such proceedings are being contested in good faith and, in the case of judgment liens, execution thereon is stayed;
 - (e) Landlords' liens on fixtures located on premises leased by the Company or any Restricted Subsidiary in the ordinary course of business;
 - (f) [intentionally omitted]
- (g) Liens arising by reason of deposits to qualify the Company or any Restricted Subsidiary to conduct business, to maintain self-insurance, or to obtain the benefit of, or comply with, Laws;
- (h) Liens that secure any Indebtedness of a Restricted Subsidiary owing to the Company or another Restricted Subsidiary or by the Company to a Restricted Subsidiary;
- (i) Liens created or deemed to be created under Permitted Receivables Transactions at any time provided such Liens do not extend to any property or assets other than the trade receivables sold pursuant to such Permitted Receivables Transactions, interests in the goods or products (including returned goods and products), if any, relating to the sales giving rise to such trade receivables; any security interests or other Liens and property subject thereto (other than on any leases or related lease payment rights or receivables between the Company and any of its Restricted Subsidiaries, as lessors or sublessors) from time to time purporting to secure the payment by the obligors describing the collateral securing such trade receivables (together with any financing statements signed by such obligors describing the collateral securing such trade receivables Transactions;
 - (j) Liens existing at Closing and reflected on Schedule 7.03 hereto;
- (k) (i) Liens upon property or assets which are created prior to or contemporaneously with, or within 360 days after, (1) in the case of the acquisition of such property or assets, the completion of such acquisition; and (2) in the case of the construction, development or improvement of such property or assets, the later to occur of the completion of such construction, development or improvement or the commencement of operation or use of the property or assets, which Lien secures or provides for the payment, financing or refinancing, directly or indirectly, of all or any part of the acquisition cost of such property or assets or the cost of construction, development or improvement thereof; or (ii) any Lien upon property or assets existing at the time of the acquisition thereof, which Lien secures obligations assumed by the Company or any Restricted Subsidiary; or (iii) any conditional sales agreement or other title retention agreement with respect to any property or assets acquired by the Company or any Restricted Subsidiary; or (iv) any Lien existing on the property or assets or shares of stock of a corporation or firm at the time such corporation or firm is merged into or consolidated with the Company or any Restricted Subsidiary or at the time of a sale, lease or other disposition of the property or assets of such corporation or firm as an entirety or substantially as an entirety to the Company or any Restricted Subsidiary or at the time such corporation becomes a Restricted Subsidiary; or (v) any Lien existing on the property, assets or shares of stock of any successor which shall have become the Company in accordance with the provisions set forth in Section 7.02; provided, in each case, that

any such Lien described in the foregoing clauses (ii), (iii), (iv) or (v) does not attach to or affect property or assets owned by the Company or any Restricted Subsidiary prior to the event referred to in such clauses; provided further (for avoidance of doubt) that any such Lien described in the foregoing clauses (i), (ii) or (iii) may be created in favor of any Governmental Authority.

- (l) Any extension, renewal or refunding (or successive extensions, renewals or refundings) in whole or in part of any Indebtedness secured by any Lien referred to in the foregoing clauses, inclusive, provided that the principal amount of the Indebtedness secured is not increased or the maturity thereof reduced and the Lien securing such Indebtedness shall be limited to the property or assets that, immediately prior to such extension, renewal or refunding, secured such Indebtedness and additions to such property or assets; and
- (m) Liens securing Priority Debt, provided that Priority Debt shall not at any time exceed 15.00% of Consolidated Tangible Assets as of the then most recently ended fiscal quarter or fiscal year of the Company for which financial information is reportable pursuant to Sections 6.01(a) and (b).
- **7.04 Limitation on Consolidated Debt.** The Company will not permit at any time the ratio of (a) Consolidated Debt to (b) Adjusted Consolidated EBITDA for the most recently ended period of four consecutive fiscal quarters of the Company to be greater than 3.75 to 1.00.
- **7.05 Limitation on Priority Debt.** The Company will not permit Priority Debt at any time to exceed 15.00% of Consolidated Tangible Assets as of the end of the most recently ended fiscal quarter or fiscal year of the Company for which financial information is reportable pursuant to Sections 6.01(a) and (b).
- **7.06 Nature of Business.** The Company will not, and will not permit any Restricted Subsidiary to, engage in any business if, as a result, the general nature of the business of the Company and its Restricted Subsidiaries, taken as a whole, which would then be engaged in by the Company and its Restricted Subsidiaries would be substantially changed from the nature of the business engaged in by the Company and its Restricted Subsidiaries, taken as a whole, on the Closing Date.
- **7.07 Designation of Unrestricted Subsidiaries.** The Company may designate any newly-formed or newly-acquired Subsidiary as a Restricted Subsidiary or an Unrestricted Subsidiary; provided that if the Company shall fail to so designate by the end of the fiscal quarter in which such Subsidiary is formed or acquired, such Subsidiary shall be deemed to be a Restricted Subsidiary. In addition, the Company may designate any Restricted Subsidiary as an Unrestricted Subsidiary and any Unrestricted Subsidiary as a Restricted Subsidiary; provided that, (a) if such Subsidiary initially is a Restricted Subsidiary, then such Restricted Subsidiary may be subsequently redesignated as an Unrestricted Subsidiary and such Unrestricted Subsidiary, then such Unrestricted Subsidiary may be subsequently designated as a Restricted Subsidiary and such Restricted Subsidiary may be subsequently redesignated as an Unrestricted Subsidiary, but no further changes in designation may be made; (c) immediately before and after designation of a Restricted Subsidiary as an Unrestricted Subsidiary or of an Unrestricted Subsidiary as a

Restricted Subsidiary there exists no Default or Event of Default; and (d) Holdings may not be designated as an Unrestricted Subsidiary.

7.08 Margin Stock. The Company shall not use the Credits, the Advances or the Term Loans, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock with in the meaning of Regulation U of the FRB.

ARTICLE VIII EVENTS OF DEFAULT AND REMEDIES

- **8.01 Events of Default.** Any of the following shall constitute an Event of Default:
- (a) Non-Payment of Drawing or Principal Amount. The Company fails to pay when and as required to be paid herein or in any other Credit Document (i) within six Business Days after the applicable Reimbursement Date of any drawing under any Credit, the amount of such drawing (unless within such six Business Day period a Request for Term Loans has been received pursuant to Section 2.03(b), and such Term Loans are made in accordance with the terms and conditions of Sections 2.03, 4.02 and 4.03 within 15 Business Days of the receipt by the Administrative Agent of such Request for Term Loans), or (ii) on the Maturity Date, the aggregate principal amount of any Term Loan or Advances outstanding; or
- (b) Non-Payment of Interest and Other Amounts. The Company defaults in the payment of any interest on any Letter of Credit Linked Note or Term Note, any Facility Fee or Fronting Fee due hereunder, or other amount payable hereunder or under any other Credit Document for more than five (5) Business Days after the same becomes due and payable; or
- (c) Specific Covenants. The Company defaults in the performance of or compliance with (i) Section 7.05, which default is not remedied within ten (10) Business Days after a Responsible Officer obtaining actual knowledge of such default, or (ii) Sections 7.02, 7.03, or 7.04; or
- (d) Other Defaults. The Company defaults in the performance of or compliance with any term contained herein (other than those referred to in paragraphs (a), (b) and (c) of this Section 8.01) and such default is not remedied within thirty (30) days after a Responsible Officer obtaining actual knowledge of such default; or
- (e) Representations and Warranties. Any representation or warranty made in writing by or on behalf of the Company or Holdings or by any officer of the Company or Holdings in this Agreement or in any writing furnished in connection with the transactions contemplated hereby proves to have been false or incorrect in any material respect on the date as of which made; or
- (f) Cross-Defaults. (i) A default in the payment of any principal of or premium or make-whole amount or interest on any Material Debt has occurred and continues beyond any period of grace provided with respect thereto, or (ii) any other default in the performance of or compliance with any term of any evidence of any Material Debt or of any mortgage, indenture or other agreement relating thereto exists, and as a consequence of such default such Material Debt has become, or has been declared (or one or more Persons are entitled to declare such Material Debt to be), due and payable before its stated maturity or before its regularly scheduled dates of

payment, or (iii) as a consequence of the occurrence or continuation of any event, (x) the Company or any Restricted Subsidiary has become obligated to purchase or repay any Material Debt before its regular maturity or before its regularly scheduled dates of payment, or (y) one or more Persons have the right to require the Company or any Restricted Subsidiary so to purchase or repay any Material Debt, or (iv) an event or condition shall occur which results in an "Event of Default" under the Seven Year Agreement or the Five Year Agreement; or

- (g) Failure to Pay Debts; Voluntary Proceedings. The Company, any Restricted Subsidiary or Holdings (i) is generally not paying, or admits in writing its inability to pay, its debts as they become due, (ii) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy, for liquidation or to take advantage of any bankruptcy, insolvency, reorganization, moratorium or other similar law of any jurisdiction, (iii) makes an assignment for the benefit of its creditors, (iv) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, (v) is adjudicated as insolvent or to be liquidated, or (vi) takes corporate action for the purpose of any of the foregoing; or
- (h) Involuntary Proceedings. A court or governmental authority of competent jurisdiction enters an order appointing, without consent by the Company, any Restricted Subsidiary or Holdings, as applicable, a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, or constituting an order for relief or approving a petition for relief or reorganization or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy or insolvency law of any jurisdiction, or ordering the dissolution, winding-up or liquidation of the Company, any Restricted Subsidiary or Holdings, or any such petition shall be filed against the Company, any Restricted Subsidiary or Holdings and such petition shall not be dismissed within 60 days; or
- (i) Judgments. A final judgment or judgments for the payment of money aggregating in excess of 10.00% of Consolidated Net Worth are rendered against one or more of the Company, any Restricted Subsidiary or Holdings and which judgments are not, within 60 days after entry thereof, bonded, discharged or stayed pending appeal, or are not discharged within 60 days after the expiration of such stay; or
- (j) Invalidity of Holdings Guaranty. If for any reason (other than payment in full of all obligations under this Agreement, the Letter of Credit Linked Notes and the Term Notes) the Holdings Guaranty shall cease to be in full force and effect (other than as expressly provided in Section 11.09 of this Agreement), or the Company or Holdings (or any other Person on behalf of the Company or Holdings) shall contest the validity or enforceability of such Holdings Guaranty, deny that Holdings has any further liability hereunder, or seek to revoke such Holdings Guaranty; or
- (k) ERISA. If (i) any Plan shall fail to satisfy the minimum funding standards of ERISA or the Code for any plan year or part thereof or a waiver of such standards or extension of any amortization period is sought or granted under Section 412 of the Code, (ii) a notice of intent to terminate any Plan shall have been or is reasonably expected to be filed with the PBGC or the PBGC shall have instituted proceedings under ERISA Section 4042 to terminate or appoint a

trustee to administer any Plan or the PBGC shall have notified the Company or any ERISA Affiliate that a Plan may become a subject of any such proceedings, (iii) the aggregate "amount of unfunded benefit liabilities" (within the meaning of Section 4001(a)(18) of ERISA) under all Plans, determined in accordance with Title IV of ERISA, shall exceed 10.00% of Consolidated Net Worth, (iv) the Company or any ERISA Affiliate shall have incurred or is reasonably expected to incur any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, (v) the Company or any ERISA Affiliate withdraws from any Multiemployer Plan, or (vi) the Company or any Subsidiary establishes or amends any employee welfare benefit plan that provides post-employment welfare benefits in a manner that would increase the liability of the Company or any Subsidiary thereunder; and any such event or events described in clauses (i) through (vi) above, either individually or together with any other such event or events, could reasonably be expected to have a Material Adverse Effect; or

(l) Change in Control. At any time any Person or group of Persons (within the meaning of Section 13 or 14 of the Securities Exchange Act of 1934, as amended) shall have acquired beneficial ownership (within the meaning of Rule 13d-3 promulgated by the Securities and Exchange Commission under said Act) of 25% or more of the outstanding shares of common voting stock of the Company; or during any period of twelve consecutive calendar months, individuals who were directors of the Company on the first day of such period (together with any new directors whose election by such board or whose nomination for election by the shareholders of the Company was approved by a vote of a majority of the directors still in office who were either directors at the beginning of such period or whose election or nomination for election was previously so approved) shall cease to constitute a majority of the board of directors of the Company.

As used in Section 8.01(k), the terms "employee benefit plan" and "employee welfare benefit plan" shall have the respective meanings assigned to such terms in Section 3 of ERISA.

- **8.02 Remedies Upon Event of Default.** If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Credit-Linked Lenders, take any or all of the following actions, which actions shall be taken simultaneously under this Agreement, the Seven Year Agreement and the Five Year Agreement:
- (a) declare any obligation of the Issuer to make Credit Extensions and the obligations of the Lenders to make Term Loans to be terminated, whereupon such commitments and obligation shall be terminated;
- (b) declare all amounts owing or payable hereunder (including the Prepayment Premium, if any) or under any other Credit Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Company;
- (c) require that the Company Cash Collateralize the Credit Obligations (other than Advances and Term Loans) (in an amount equal to the Outstanding Amount thereof, as such amount may vary from time to time); and

(d) exercise on behalf of itself, the Issuer and the Lenders all rights and remedies available to it, the Issuer and the Lenders under the Credit Documents or applicable law;

provided, however, that upon the occurrence of an Event of Default pursuant to Sections 8.01(g) or (h), any obligation of the Issuer or the Lenders to make Credit Extensions shall automatically terminate, all unpaid amounts as aforesaid shall automatically become due and payable, and the obligation of the Company to Cash Collateralize the Credit Obligations (other than Advances and Term Loans) as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent or any Lender.

Thereafter, at the request of the Required Lenders, the Administrative Agent shall (x) withdraw from the Credit-Linked Deposit Account and distribute to the Lenders an amount equal to the excess of the Total Credit-Linked Deposit over the Outstanding Amount of the Credit Obligations (other than Advances and Term Loans), and (y) subject to, and to the extent of, the Company's compliance with any Cash Collateralization requirements as aforesaid, withdraw from the Credit-Linked Deposit Account and distribute to the Lenders the aggregate amount of any remaining Credit-Linked Deposits to the extent of the amount credited to the Credit-Linked Deposit Account; in each case subject to any notice required pursuant to the Money Market Account Agreement.

8.03 Application of Funds. After the exercise of remedies provided for in Section 8.02 (or after the obligations of the Issuer to make Credit Extensions have been automatically terminated and the Credit Obligations have automatically been required to be Cash Collateralized as set forth in the proviso to Section 8.02), any amounts received by the Administrative Agent, the Issuer or the Lenders on account of the Credit Obligations shall be applied by the Administrative Agent in the following order:

First, to payment of fees, indemnities, expenses and other amounts (including interest thereon, Attorney Costs and amounts payable under Article III) payable to the Administrative Agent in its capacity as such;

Second, to payment of fees, indemnities and other amounts (excluding principal and interest on any Borrowings but including interest, if any, on any such fees, indemnities and other amounts) payable to the Issuer and the Lenders (including Attorney Costs and amounts payable under Article III), ratably among them in proportion to the amounts described in this clause Second payable to them;

Third, to payment of that portion of the Credit Obligations constituting accrued and unpaid interest on the Borrowings, to the Issuer and ratably among the Lenders in proportion to the respective amounts described in this clause Third payable to them;

Fourth, to payment of that portion of the Credit Obligations constituting unpaid principal of the Borrowings, to the Issuer and ratably among the Lenders in proportion to the respective amounts described in this clause Fourth held by them;

Fifth, to the Administrative Agent for the account of the Issuer, to Cash Collateralize that portion of Credit Obligations comprised of the aggregate undrawn amount of Credits; and

Last, the balance, if any, after all of the Credit Obligations have been paid in full, to the Company or as otherwise required by Law.

Subject to Section 2.01(c), amounts used to Cash Collateralize the aggregate undrawn amount of Credits pursuant to clause Fifth above shall be applied to satisfy drawings under such Credits as they occur. If any amount remains on deposit as Cash Collateral after all Credits have either been fully drawn or expired, such remaining amount shall be applied to the other Credit Obligations, if any, in the order set forth above.

ARTICLE IX ADMINISTRATIVE AGENT

9.01 Appointment and Authorization of Administrative Agent. (a) Each of the Lenders and the Issuer hereby irrevocably appoints, designates and authorizes the Administrative Agent to take such action on its behalf under the provisions of this Agreement and each other Credit Document and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement or any other Credit Document, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Credit Document, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall the Administrative Agent have or be deemed to have any fiduciary relationship with the Issuer, any Lender or participant, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Credit Document or otherwise exist against the Administrative Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Credit Documents with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable Law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.

(b) The Issuer shall act on behalf of the Lenders with respect to any Credits issued by it and the Issuer Documents associated therewith, and the Issuer shall have all of the benefits and immunities (i) provided to the Administrative Agent in this Article IX (other than Section 9.07) with respect to any acts taken or omissions suffered by the Issuer in connection with Credits issued by it or proposed to be issued by it and the Issuer Documents pertaining to such Credits as fully as if the term "Administrative Agent" as used in this Article IX and in the definition of "Agent-Related Person" included the Issuer with respect to such acts or omissions, and (ii) as additionally provided herein with respect to the Issuer.

9.02 Delegation of Duties. The Administrative Agent may execute any of its duties under this Agreement or any other Credit Document by or through agents, employees or attorneys-in-fact and shall be entitled to advice of counsel and other consultants or experts concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects in the absence of gross negligence or willful misconduct.

- 9.03 Liability of Administrative Agent. No Agent-Related Person shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Credit Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct in connection with its duties expressly set forth herein), or (b) be responsible in any manner to any Lender or participant for any recital, statement, representation or warranty made by any Credit Party or any officer thereof, contained herein or in any other Credit Document, or in any certificate, report, statement or other document referred to or provided for in, or received by the Administrative Agent under or in connection with, this Agreement or any other Credit Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Credit Document, or for any failure of any Credit Party or any other party to any Credit Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Lender or participant to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Credit Document, or to inspect the properties, books or records of any Credit Party or any Affiliate thereof.
- **9.04** Reliance by Administrative Agent. (a) The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, electronic mail message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to any Credit Party), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under any Credit Document unless it shall first receive such advice or concurrence of the Required Credit-Linked Lenders as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Credit Document in accordance with a request or consent of the Required Lenders or Required Credit-Linked Lenders, as applicable (or such greater number of Lenders as may be expressly required hereby in any instance) and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Lenders.
- (b) For purposes of determining compliance with the conditions specified in Section 4.01, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.
- **9.05 Notice of Default.** The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to defaults in the payment of principal, interest and fees required to be paid to the Administrative Agent for the account of the Lenders, unless the Administrative Agent shall have received written notice from the Issuer, a Lender or the Company referring to this Agreement, describing

such Default or Event of Default and stating that such notice is a "notice of default." The Administrative Agent will notify the Lenders of its receipt of any such notice. The Administrative Agent shall take such action with respect to such Default or Event of Default as may be directed by the Required Credit-Linked Lenders in accordance with Article VIII; provided, however, that unless and until the Administrative Agent has received any such direction, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable or in the best interest of the Lenders.

9.06 Credit Decision; Disclosure of Information by Administrative Agent. Each Lender and the Issuer acknowledges that no Agent-Related Person has made any representation or warranty to it, and that no act by the Administrative Agent hereafter taken, including any consent to and acceptance of any assignment or review of the affairs of any Credit Party or any Affiliate thereof, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender as to any matter, including whether Agent-Related Persons have disclosed material information in their possession. Each Lender and the Issuer represents to the Administrative Agent that it has, independently and without reliance upon any Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of the Credit Parties and their respective Subsidiaries, and made its own decision to enter into this Agreement and to extend credit to the Company and any other Credit Parties hereunder. Each Lender and the Issuer hereby acknowledge that it has not relied on any Agent-Related Person (other than as a recipient of legal opinions of counsel to the Administrative Agent delivered pursuant to Section 4.01(a)(vi)) with respect to any bank or other regulatory Laws relating to the transactions contemplated hereby. Each Lender and the Issuer also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Credit Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of the Company and the other Credit Parties. Except for notices, reports and other documents expressly required to be furnished to the Issuer and the Lenders by the Administrative Agent herein, the Administrative Agent shall not have any duty or responsibility to provide the Issuer or any Lender with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of any of the Credit Parties or any of their respective Affiliates which may come into the possession of any Agent-Related Person.

9.07 Indemnification of Administrative Agent. The Lenders shall indemnify upon demand, in Dollars, the Administrative Agent, in its capacity as Administrative Agent, and the Affiliates (to the extent such Affiliates are acting for the Administrative Agent in such capacity), officers, directors, employees, agents and attorneys-in-fact thereof who are acting for or on behalf of the Administrative Agent (pursuant to the express authorization of, or by express delegation from, the Administrative Agent) in connection with this Agreement (to the extent not reimbursed by or on behalf of any Credit Party and without limiting the obligation of any Credit Party to do so), *pro rata* in accordance with its Pro Rata Share, and hold harmless each such

Person from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Person in any way relating to or arising out of or in connection with (before or after the Closing Date) the execution, delivery, enforcement, performance or administration of any Credit Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby (the "Agent Indemnified Liabilities"), incurred by any such Person in connection with the Administrative Agent's exercise of powers and performance of duties expressly provided for under this Agreement; provided, however, that no Lender shall be liable for the payment to any such Person of any portion of such Agent Indemnified Liabilities resulting from such Person's own gross negligence or willful misconduct; and provided further, however, that no action taken in accordance with the directions of the Required Lenders or Required Credit-Linked Lenders, as applicable, shall be deemed to constitute gross negligence or willful misconduct for purposes of this Section; and provided further, however, that the Agent Indemnified Liability was incurred by or asserted against the Administrative Agent in its capacity as such, or such Affiliate or other Person acting for the Administrative Agent in such capacity. Without limitation of the foregoing, each Lender shall reimburse the Administrative Agent upon demand for such Lender's Pro Rata Share of any costs or out-of-pocket expenses (including reasonable Attorney Costs) incurred by the Administrative Agent (i) in connection with any enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Credit Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Credit Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), or (ii) otherwise, while an Event of Default exists, to the extent that the Administrative Agent is not reimbursed for such expenses by or on behalf of the Company within 30 days after demand therefor; provided in each case that such cost or expense was incurred by the Administrative Agent in its capacity as such; and provided, however, in each case that before making any demand on the Lenders for reimbursement of any such cost or expense, the Administrative Agent shall have first requested the Company to pay such amounts and made a good faith determination that the Company will not honor such request. Notwithstanding the foregoing, no Lender shall be liable for any indirect, special, punitive or consequential damages claimed by any Person under this Section 9.07 (other than such damages claimed by such Person as a result of such damages being claimed against such Person by a third party). The undertaking in this Section shall survive termination of the Credit-Linked Deposits, the payment of all other Credit Obligations and other amounts due hereunder and the resignation of the Administrative Agent. Notwithstanding the foregoing, (a) no Lender shall be obligated to reimburse or indemnify the Administrative Agent, in its capacity as Administrative Agent, and the Affiliates (to the extent such Affiliates are acting for the Administrative Agent in such capacity), officers, directors, employees, agents and attorneys-in-fact thereof, in respect of the transactions contemplated by Section 10.19 resulting from a Deposit Account Interest Non-Payment Event or in respect of any other nonpayment of funds owing to the Lenders, if such other nonpayment results solely from the Administrative Agent's nonperformance of its paying agency duties hereunder (and not from any action or inaction of the Company or any other unrelated Person), and (b) no Lender shall be obligated to reimburse or indemnify BAS.

9.08 Administrative Agent in its Individual Capacity. Bank of America and its Affiliates may make loans to, issue letters of credit and guarantees or other undertakings for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with each of the Credit Parties and their respective Affiliates as though Bank of America were not the Administrative Agent or the Issuer hereunder and without notice to or consent of the Lenders. The Lenders acknowledge that, pursuant to such activities, Bank of America or its Affiliates may receive information regarding any Credit Party or its Affiliates (including information that may be subject to confidentiality obligations in favor of such Credit Party or such Affiliate) and acknowledge that the Administrative Agent shall be under no obligation to provide such information to them.

9.09 Successor Administrative Agent. The Administrative Agent may resign as Administrative Agent upon 30 days' notice to the Lenders; provided that any such resignation by Bank of America shall also constitute its resignation as Issuer under this Agreement; provided, however, the Administrative Agent may not resign if (i) it is the Issuer of any outstanding Credits under this Agreement or (ii) any Credit-Linked Deposits of any Lender exist. If the Administrative Agent resigns under this Agreement, the Required Credit-Linked Lenders shall appoint a successor administrative agent for the Lenders which successor administrative agent shall require the consent of the Company at all times other than during the existence of any Default or Event of Default (which consent of the Company shall not be unreasonably withheld or delayed). If no successor administrative agent is appointed prior to the effective date of the resignation of the Administrative Agent, the Administrative Agent may appoint, after consulting with the Credit-Linked Lenders and the Company, a successor administrative agent from among the Credit-Linked Lenders with the consent of such Credit-Linked Lender and the consent of the Company (which consent of the Company shall be required at all times other than during the existence of any Default or Event of Default and shall not be unreasonably withheld or delayed). The successor administrative agent under this Agreement, the Seven Year Agreement and the Five Year Agreement shall at all times be the same Person. Upon the acceptance of its appointment as successor administrative agent hereunder, the Person acting as such successor administrative agent, and the retiring Administrative Agent and the term "Administrative Agent" shall mean such successor administrative agent, and the retiring Administrative Agent and the term "Administrative Agent" shall be terminated. After any retiring Administrative Agent, and the retiring Administrative Agent hereunder as Administrative Agent, the provisions of this Articl

ARTICLE X MISCELLANEOUS

10.01 Amendments, Etc. No amendment or waiver of any provision of this Agreement or any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)), and no consent to any departure by the Company or any other Credit Party therefrom (other than a departure of Holdings pursuant to Section 11.09 of this Agreement), shall be effective unless (A) in writing signed by the Required Credit-Linked Lenders and the Company or the applicable Credit Party, as the case may be, and acknowledged by the Administrative

Agent and (B) a corresponding amendment, waiver or consent under (i) the Seven Year Agreement and (ii) the Five Year Agreement is also effective (it being understood that the same terms and conditions with respect to such amendment, waiver or consent shall apply to all Credit-Linked Lenders), and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no such amendment, waiver or consent shall, without the written consent of each Lender with Credit Obligations directly affected (whose consent shall be required for any such amendment, waiver or consent in addition to that of the Required Credit-Linked Lenders):

- (a) waive any condition set forth in Section 4.01;
- (b) extend the Maturity Date or increase the Credit-Linked Deposit of any Lender (or reinstate any obligation to make Credit Extensions terminated pursuant to Section 8.02 in which such Lender participates);
- (c) postpone any date fixed by this Agreement or any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)) for any payment of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b));
- (d) reduce the principal of, or the rate of interest specified herein on, any Borrowing, or (subject to clause (iii) of the proviso below) any fees or other amounts payable hereunder or under any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)); provided, however, that only the consent of the Required Credit-Linked Lenders shall be necessary to amend the definition of "Default Rate," the amount of any increase in the Applicable Rate pursuant to the definition thereof during the existence of any Event of Default, or the interest rate applicable from time to time to Borrowings under Section 2.04(a)(ii), or to waive any obligation of the Company to pay interest at the Default Rate or at the rate set forth in Section 2.04(a)(ii), or to increase the Applicable Rate during the existence of an Event of Default;
 - (e) change Section 2.08 or Section 8.03 in a manner that would alter the *pro rata* sharing of payments required thereby;
- (f) change any provision of this Section 10.01, or the definition of "Required Credit-Linked Lenders", "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder: or
- (g) release Holdings from any of the Guaranteed Obligations under the Holdings Guaranty (other than as expressly provided in Section 11.09 of this Agreement);

and, provided further, that (i) no amendment, waiver or consent shall, unless in writing and signed by the Issuer in addition to the Lenders required above, modify or waive any provision of Section 2.01(a) (including defined terms used therein) or otherwise affect the rights or duties of the Issuer under this Agreement or any Issuer Document relating to any Credit issued or to be issued by it; (ii) no amendment, waiver or consent shall, unless in writing and signed by the

Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Credit Document; and (iii) the Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed only by the respective parties thereto.

10.02 Notices and Other Communications; Facsimile Copies.

- (a) General. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission). All such written notices shall be mailed, faxed or delivered to the applicable address, facsimile number or (subject to subsection (c) below) electronic mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:
- (i) if to the Company, Holdings, the Administrative Agent or the Issuer, to the address, facsimile number, electronic mail address or telephone number specified for such Person on Schedule 10.02 or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties; and
- (ii) if to any other Lender, to the address, facsimile number, electronic mail address or telephone number specified in its Administrative Questionnaire or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the Company, the Administrative Agent and the Issuer.

All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (B) if delivered by mail, four Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of subsection (c) below), when delivered; provided, however, that notices and other communications to the Administrative Agent and the Issuer pursuant to Article II shall not be effective until actually received by any such Person. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.

- (b) Effectiveness of Facsimile Documents and Signatures. Credit Documents may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually-signed originals and shall be binding on all Credit Parties, the Administrative Agent, the Issuer and the Lenders. The Administrative Agent may also require that any such documents and signatures be confirmed by a manually-signed original thereof; provided, however, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.
- (c) Limited Use of Electronic Mail. Electronic mail and Internet and intranet websites may be used only to distribute routine communications, other than financial statements and other similar information, and to distribute Credit Documents for execution by the parties thereto, and may not be used for any other purpose.

- (d) Reliance by Administrative Agent, the Issuer and Lenders. The Administrative Agent, the Issuer and the Lenders shall be entitled to rely and act upon any notices purportedly given by or on behalf of the Company even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Company shall indemnify each Agent-Related Person, each Issuer-Related Person and each Lender from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Company. All telephonic notices to and other communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.
- **10.03 No Waiver; Cumulative Remedies.** No failure by any Lender, the Issuer or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein or therein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

10.04 Attorney Costs, Expenses and Taxes. The Company agrees (a) (i) to pay or reimburse each of the Administrative Agent, the Issuer and special counsel for the Lenders for all reasonable costs and expenses incurred in connection with the development, preparation, negotiation and execution of this Agreement and the other Credit Documents (whether or not the transactions contemplated hereby are consummated), and the consummation and administration of the transactions contemplated hereby, including all reasonable Attorney Costs, expenses and disbursements and (ii) to pay or reimburse each of the Administrative Agent, the Issuer and each Lender for all reasonable costs and expenses incurred in connection with any amendment, waiver, consent or other modification of the provisions of this Agreement and the other Credit Documents (whether or not the transactions contemplated thereby are consummated), including without limitation any out-of-pocket costs and expenses incurred in connection with the transactions contemplated by Section 10.19 of this Agreement, and the consummation and administration of the transactions contemplated thereby, including all reasonable Attorney Costs, expenses and disbursements, but excluding any such costs, expenses, attorneys' fees, and disbursements of the Issuer and the Administrative Agent in connection with the transactions contemplated in Section 10.19 resulting from a Deposit Account Interest Non-Payment Event; and (b) to pay or reimburse the Administrative Agent, the Issuer and each Lender for all costs and expenses incurred in connection with the enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Credit Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Credit Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all Attorney Costs. The foregoing costs and expenses shall include all search, filing, recording, title insurance and appraisal charges and fees and taxes related thereto, and other out-of-pocket expenses incurred by the Administrative Agent and the cost of independent public accountants and other outside experts retained by the Administrative Agent, the Issuer or any Lender. All amounts due under this Section 10.04 shall be payable within ten Business Days after demand therefor. The agreements in this Section 10.04 shall

survive the termination of the Credit-Linked Deposits and repayment of all Credit Obligations and all other amounts due hereunder.

10.05 Indemnification by the Company. Whether or not the transactions contemplated hereby are consummated, the Company shall indemnify and hold harmless each Agent-Related Person, each Issuer-Related Person, each Lender and their respective Affiliates, directors, officers, employees, counsel, agents and attorneys-in-fact (collectively, the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (before or after the Closing Date) (a) the execution, delivery, enforcement, performance or administration of any Credit Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, including without limitation the transactions contemplated by Section 10.19 of this Agreement but excluding any such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements of the Issuer and the Administrative Agent in connection with the transactions contemplated in Section 10.19 resulting from a Deposit Account Interest Non-Payment Event, (b) any Credit-Linked Deposit, Credit or the use or proposed use of the proceeds therefrom (including any refusal by the Issuer to honor a demand for payment under a Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Credit), or (c) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding) and regardless of whether any Indemnitee is a party thereto (all the foregoing, collectively, the "Indemnified Liabilities"), in all cases, whether or not caused by or arising, in whole or in part, out of the negligence of the Indemnitee; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. No Indemnitee shall be liable for any indirect, special, punitive or consequential damages relating to this Agreement or any other Credit Document or arising out of its activities in connection herewith or therewith (whether before or after the Closing Date). Notwithstanding the foregoing, the Company shall not be liable for any indirect, special, punitive or consequential damages claimed by an Indemnitee in connection with the transactions contemplated by Section 10.19 of this Agreement (other than such damages claimed by an Indemnitee as a result of such damages being claimed against such Indemnitee by a third party). All amounts due under this Section 10.05 shall be payable within ten Business Days after demand therefor. The agreements in this Section shall survive the resignation of the Administrative Agent, the replacement of any Lender, the termination of the Credit-Linked Deposits and the repayment, satisfaction or discharge of all the other Credit Obligations.

10.06 Payments Set Aside. To the extent that any payment by or on behalf of the Company is made to the Administrative Agent, the Issuer or any Lender, or the Administrative

Agent, the Issuer or any Lender exercises its right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, the Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred, and (b) the Issuer and each Lender severally agree to pay to the Administrative Agent upon demand its applicable share of any amount so recovered from or paid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect, in the applicable currency of such recovery or payment.

10.07 Successors and Assigns. (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Company may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Issuer and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of subsection (b) of this Section 10.07, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section 10.07, (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section 10.07 (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section 10.07 and, to the extent expressly contemplated hereby, the Indemnitees) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) Any Lender may at any time assign to another Lender or to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Credit-Linked Deposit and Credit Obligations at the time owing to it); provided that unless such assignment is to another Lender (i) upon such assignment, the aggregate principal amount of the Credit-Linked Deposit and Credit Obligations held by the assignee, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000; (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Credit Obligations and the Credit-Linked Deposit assigned; and (iii) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption. Subject to acceptance and recording thereof by the Administrative Agent pursuant to subsection (c) of this Section 10.07, from and after the effective date specified in each Assignment and Assumption, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the

case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 3.01, 3.02, 3.03, 10.04 and 10.05 with respect to facts and circumstances occurring prior to the effective date of such assignment). Upon request, the Company (at its expense) shall execute and deliver a Letter of Credit Linked Note and one or more Term Notes to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a Participation in such rights and obligations in accordance with subsection (d) of this Section 10.07.

- (c) The Administrative Agent, acting solely for this purpose as an agent of the Company, shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Credit-Linked Deposit of, and amounts of the Credit Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error and the Company, the Administrative Agent, the Issuer and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Company, the Issuer and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) Any Lender may at any time, without notice to or the consent of the Administrative Agent, sell participations to any Person (other than a natural person or the Company or any of the Company's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Credit-Linked Deposit and/or such Lender's Term Loans and/or Participations in Credit Obligations owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Company, the Administrative Agent, the Issuer and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to Section 10.01 that directly affects such Participant. Subject to subsection (e) of this Section 10.07, the Company agrees that each Participant shall be entitled to the benefits of Sections 3.01, 3.02 and 3.03 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section 10.07. To the extent permitted by Law, each Participant also shall be entitled to the benefits of Section 2.09 as though it were a Lender.
- (e) A Participant shall not be entitled to receive any greater payment under Section 3.01 or 3.05 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is

made with the Company's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of Section 3.01 unless the Company is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Company, to comply with Section 10.15 as though it were a Lender.

- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) Assignment of Credit-Linked Deposit. The Credit-Linked Deposit funded by any Lender under Section 2.01(c)(ii) shall not be released in connection with any assignment of its Credit-Linked Deposit, but shall instead be purchased by the relevant assignee and continue to be held for application (if not already applied in accordance with the terms of Section 2.01(c)(ii) or 2.03) pursuant to Section 2.01(c)(ii) in respect of such assignee's obligations under the Credit-Linked Deposit assigned to it.
- 10.08 Confidentiality. For the purposes of this Section 10.08, "Confidential Information" means information delivered to the Administrative Agent, the Issuer or any Lender by or on behalf of the Company or any Subsidiary in connection with the transactions contemplated by or otherwise pursuant to this Agreement that is proprietary in nature and that was clearly marked or labeled or otherwise adequately identified when received by it as being confidential information of the Company or such Subsidiary, provided that such term does not include information that (a) was publicly known or otherwise known to it prior to the time of such disclosure, (b) subsequently becomes publicly known through no act or omission by it or any Person acting on its behalf, (c) otherwise becomes known to it other than through disclosure by the Company or any Subsidiary or (d) constitutes financial statements delivered to it under Section 6.01 that are otherwise publicly available. Each of the Administrative Agent, the Issuer and the Lenders will maintain the confidentiality of such Confidential Information delivered to or received by it in accordance with procedures adopted by it in good faith to protect confidential information of third parties delivered to it, provided that the Administrative Agent, the Issuer or any Lender may deliver or disclose Confidential Information to (i) its directors, officers, employees, agents, attorneys and Affiliates (to the extent such disclosure reasonably relates to the administration of the Credit Documents by the Administrative Agent, the Issuer or any Lender), (ii) its financial advisors and other professional advisors who agree to hold confidential the Confidential Information substantially in accordance with the terms of this Section 10.08, (iii) any other party to this Agreement, (iv) any Eligible Assignee to which any Lender sells or offers to sell its Credit-Linked Deposit and Credit Obligations or any part thereof or any Participation therein (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 10.08), (v) any Person from which it offers to purchase any security of the Company (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 10.08), (vi) any federal or state regulatory authority having jurisdiction over it, (vii) the National Association of Insurance Commissioners or any similar organization, or any nationally recognized rating agency that requires access to information about its investment portfolio, or

(viii) any other Person to which such delivery or disclosure may be necessary or appropriate (w) to effect compliance with any law, rule, regulation or order applicable to it, (x) in response to any subpoena or other legal process, (y) in connection with any litigation to which it is a party or (z) if an Event of Default has occurred and is continuing, to the extent it may reasonably determine such delivery and disclosure to be necessary or appropriate in the enforcement or for the protection of the rights and remedies under the Letter of Credit Linked Notes, the Term Notes and this Agreement. Each Eligible Assignee, by its execution of an Assignment and Assumption, will be deemed to have agreed to be bound by and to be entitled to the benefits of this Section 10.08 as though it were a party to this Agreement. Notwithstanding anything herein to the contrary, "Confidential Information" shall not include, and the Issuer, the Administrative Agent and each Lender may disclose without limitation of any kind, any information with respect to the "tax treatment" and "tax structure" (in each case, within the meaning of Treasury Regulation Section 1.6011-4) of the transactions contemplated hereby and all materials of any kind (including opinions or other tax analyses) that are provided to the Issuer, the Administrative Agent or such Lender relating to such tax treatment and tax structure; provided that with respect to any document or similar item that in either case contains information concerning the tax treatment or tax structure of the Credits, Advances, Term Loans and transactions contemplated hereby.

10.09 [Intentionally Omitted].

- **10.10 Representations and Warranties of Each Lender.** Each Lender (for itself, and not for any other Lender) represents and warrants to the Issuer and the Company:
- (a) Purchase for Investment. Such Lender is an institutional "accredited investor" within the meaning of subparagraphs (1), (2), (3) or (7) of Rule 501(a) promulgated under the Securities Act. Such Lender represents that it is purchasing its Participation in this Agreement, its Letter of Credit Linked Note and its Term Notes, if any, for its own account or for one or more separate accounts maintained by it or for the account of one or more pension or trust funds and not with a view to the distribution thereof, provided that the disposition of its or their property shall at all times be within its or their control. Such Lender understands that the Letter of Credit Linked Notes and Term Notes have not been registered under the Securities Act and may be resold only if registered pursuant to the provisions of the Securities Act or if an exemption from registration is available, except under circumstances where neither such registration nor such an exemption is required by law, and that the Issuer, the Administrative Agent and the Company are not required to register the Letter of Credit Linked Notes and Term Notes.
- (b) Source of Funds. Such Lender represents that at least one of the following statements is an accurate representation as to each source of funds (a "Source") to be used by it to pay the purchase price of its Participation in this Agreement:
- (i) the Source is an "insurance company general account" (as the term is defined in the United States Department of Labor's Prohibited Transaction Exemption ("PTE") 95-60) in respect of which the reserves and liabilities (as defined by the annual statement for life insurance companies approved by the National Association of Insurance Commissioners (the "NAIC Annual Statement")) for the general account contract(s) held by or on behalf of any

employee benefit plan together with the amount of the reserves and liabilities for the general account contract(s) held by or on behalf of any other employee benefit plans maintained by the same employer (or affiliate thereof as defined in PTE 95-60) or by the same employee organization in the general account do not exceed 10% of the total reserves and liabilities of the general account (exclusive of separate account liabilities) plus surplus as set forth in the NAIC Annual Statement filed with such Lender's state of domicile; or

- (ii) the Source is a separate account that is maintained solely in connection with such Lender's fixed contractual obligations under which the amounts payable, or credited, to any employee benefit plan (or its related trust) that has any interest in such separate account (or to any participant or beneficiary of such plan (including any annuitant)) are not affected in any manner by the investment performance of the separate account; or
- (iii) the Source is either (A) an insurance company pooled separate account, within the meaning of PTE 90-1 or (B) a bank collective investment fund, within the meaning of the PTE 91-38 and, except as disclosed by such Lender to the Issuer and the Company in writing pursuant to this clause (iii), no employee benefit plan or group of plans maintained by the same employer or employee organization beneficially owns more than 10% of all assets allocated to such pooled separate account or collective investment fund; or
- (iv) the Source constitutes assets of an "investment fund" (within the meaning of Part V of PTE 84-14 (the "QPAM Exemption") managed by a "qualified professional asset manager" or "QPAM" (within the meaning of Part V of the QPAM Exemption), no employee benefit plan's assets that are included in such investment fund, when combined with the assets of all other employee benefit plans established or maintained by the same employer or by an affiliate (within the meaning of Section V(c)(1) of the QPAM Exemption) of such employer or by the same employee organization and managed by such QPAM, exceed 20% of the total client assets managed by such QPAM, the conditions of Section I(c) and (g) of the QPAM Exemption are satisfied, neither the QPAM nor a person controlling or controlled by the QPAM (applying the definition of "control" in Section V(e) of the QPAM Exemption) owns a 5% or more interest in the Issuer or the Company and (i) the identity of such QPAM and (ii) the names of all employee benefit plans whose assets are included in such investment fund have been disclosed to the Issuer and the Company in writing pursuant to this clause (iv); or
- (v) the Source constitutes assets of one or more "plans" (within the meaning of Part IV of PTE 96-23 (the "INHAM Exemption")) managed by an "inhouse asset manager" or "INHAM" (within the meaning of Part IV of the INHAM exemption), the conditions of Section I(a), (g) and (h) of the INHAM Exemption are satisfied, neither the INHAM nor a person controlling or controlled by the INHAM (applying the definition of "control" in Section IV(d)(3) of the INHAM Exemption) owns a 5% or more interest in the Issuer or the Company and (A) the identity of such INHAM and (B) the name(s) of the employee benefit plan(s) whose assets constitute the Source have been disclosed to the Issuer and the Company in writing pursuant to this clause (v); or

(vi) the Source is a governmental plan; or

- (vii) the Source is one or more employee benefit plans, or a separate account or trust fund comprised of one or more employee benefit plans, each of which has been identified to the Issuer and the Company in writing pursuant to this clause (vii); or
 - (viii) the Source does not include assets of any employee benefit plan, other than a plan exempt from the coverage of ERISA.

As used in this Section 10.10(b), the terms "employee benefit plan", "governmental plan", "party in interest" and "separate account" shall have the respective meanings assigned to such terms in Section 3 of ERISA.

- (c) Authorization; Enforceability. The execution, delivery and performance of this Agreement has been duly authorized by all necessary corporate action on the part of such Lender. This Agreement has been duly executed and delivered by such Lender and is the legally valid and binding obligation of such Lender, enforceable against such Lender in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles.
- (d) No Withholding. As of the date hereof and as of the date of any assignment under Section 10.07 (after giving effect thereto), no Lender is subject to United States federal withholding taxes on amounts payable pursuant to the terms of this Agreement. Each Lender has delivered to the Administrative Agent all forms required pursuant to Section 10.15(b) to ensure that the Administrative Agent shall not be required to effect any withholding with respect to any amounts payable pursuant to the terms of this Agreement.
- **10.11** Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- **10.12 Integration**. This Agreement, together with the other Credit Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. In the event of any conflict between the provisions of this Agreement and those of any other Credit Document, the provisions of this Agreement shall control; provided that the inclusion of supplemental rights or remedies in favor of the Administrative Agent, the Issuer or the Lenders in any other Credit Document shall not be deemed a conflict with this Agreement. Each Credit Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 10.13 Survival of Representations and Warranties. All representations and warranties made hereunder and in any other Credit Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent, the Issuer and each Lender, regardless of any investigation made by the Administrative Agent, the Issuer or any Lender or on their behalf and notwithstanding that the Administrative Agent, the Issuer or any Lender may have had notice or

knowledge of any Default or Event of Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Credit Obligation or any other Credit Obligation shall remain unpaid or unsatisfied or any Credit shall remain outstanding.

- **10.14 Severability**. If any provision of this Agreement or the other Credit Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Credit Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 10.15 Tax Forms. (a) (i) Each Lender and the Issuer that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code (a "Foreign Lender") shall deliver to the Administrative Agent, prior to receipt of any payment subject to withholding under the Code (or upon accepting an assignment of an interest herein), two duly signed completed copies of either IRS Form W-8BEN or any successor thereto (relating to such Foreign Lender and entitling it to an exemption from, or reduction of, withholding tax on all payments to be made to such Foreign Lender by the Company pursuant to this Agreement) or IRS Form W-8ECI or any successor thereto (relating to all payments to be made to such Foreign Lender by the Company pursuant to this Agreement), and, if necessary, such other evidence satisfactory to the Company and the Administrative Agent that such Foreign Lender is entitled to an exemption from, or reduction of, U.S. withholding tax, including any exemption pursuant to Section 881(c) of the Code. Thereafter and from time to time, each such Foreign Lender shall (A) promptly submit to the Administrative Agent additional duly completed and signed copies of one of such forms (or such successor forms as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then current United States laws and regulations to avoid, or such evidence as is satisfactory to the Company and the Administrative Agent of any available exemption from or reduction of, United States withholding taxes in respect of all payments to be made to such Foreign Lender by the Company pursuant to this Agreement, (B) promptly notify the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (C) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Foreign Lender, and as may be reasonably necessary (including the re-designation of its Lending Office) to avoid any requ
- (ii) Each Foreign Lender, to the extent it does not act or ceases to act for its own account with respect to any portion of any sums paid or payable to such Foreign Lender under any of the Credit Documents (for example, in the case of a typical participation by such Foreign Lender), shall deliver to the Administrative Agent on the date when such Foreign Lender ceases to act for its own account with respect to any portion of any such sums paid or payable, and at such other times as may be necessary in the determination of the Administrative Agent (in the reasonable exercise of its discretion), (A) two duly signed completed copies of the forms or statements required to be provided by such Foreign Lender as set forth above, to establish the

portion of any such sums paid or payable with respect to which such Foreign Lender acts for its own account that is exempt from, or subject to a reduced rate of, U.S. withholding tax, and (B) two duly signed completed copies of IRS Form W-8IMY (or any successor thereto), together with any information such Foreign Lender chooses to transmit with such form, and any other certificate or statement of exemption required under the Code, to establish that such Foreign Lender is not acting for its own account with respect to a portion of any such sums payable to such Foreign Lender.

- (iii) The Company shall not be required to pay any additional amount to any Foreign Lender under Section 3.01 (A) with respect to any Taxes required to be deducted or withheld on the basis of the information, certificates or statements of exemption such Foreign Lender transmits with an IRS Form W-8IMY pursuant to this Section 10.15(a) or (B) if such Foreign Lender shall have failed to satisfy the foregoing provisions of this Section 10.15(a); provided that if such Foreign Lender shall have satisfied the requirement of this Section 10.15(a) on the date such Foreign Lender became a Lender or ceased to act for its own account with respect to any payment under any of the Credit Documents, nothing in this Section 10.15(a) shall relieve the Company of its obligation to pay any amounts pursuant to Section 3.01 in the event that, as a result of any change in any applicable law, treaty or governmental rule, regulation or order, or any change in the interpretation, administration or application thereof, such Foreign Lender is no longer properly entitled to deliver forms, certificates or other evidence at a subsequent date establishing the fact that such Foreign Lender or other Person for the account of which such Foreign Lender receives any sums payable under any of the Credit Documents is not subject to withholding or is subject to withholding at a reduced rate.
- (iv) The Administrative Agent may, without reduction, withhold any Taxes required to be deducted and withheld from any payment under any of the Credit Documents with respect to which the Company is not required to pay additional amounts under this Section 10.15(a) or Section 3.01.
- (b) Upon the request of the Administrative Agent, the Issuer and each Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Administrative Agent two duly signed completed copies of IRS Form W-9. If the Issuer or such Lender fails to deliver such forms, then the Administrative Agent may withhold from any interest payment to the Issuer or such Lender an amount equivalent to the applicable back-up withholding tax imposed by the Code, without reduction.
- (c) If any Governmental Authority asserts that the Administrative Agent did not properly withhold or backup withhold, as the case may be, any tax or other amount from payments made to or for the account of the Issuer or any Lender, the Issuer or such Lender shall indemnify the Administrative Agent therefor, including all penalties and interest, any taxes imposed by any jurisdiction on the amounts payable to the Administrative Agent under this Section, and costs and expenses (including reasonable Attorney Costs) of the Administrative Agent. The obligation of the Issuer and the Lenders under this Section shall survive the termination of the Credit-Linked Deposits, repayment of all Credit Obligations and all other amounts due hereunder and the resignation or replacement of the Administrative Agent.

10.16 Governing Law. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE; PROVIDED THAT THE ADMINISTRATIVE AGENT. THE ISSUER AND EACH LENDER SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW.

ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK OR OF THE UNITED STATES FOR SOUTHERN DISTRICT OF SUCH STATE, AND BY EXECUTION AND DELIVERY OF THIS AGREEMENT, THE COMPANY CONSENTS, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THOSE COURTS. THE COMPANY IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF ANY CREDIT DOCUMENT OR OTHER DOCUMENT RELATED THERETO.

10.17 Waiver of Right to Trial by Jury. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY CREDIT DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY CREDIT DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

10.18 Time of the Essence. Time is of the essence of the Credit Documents.

10.19 Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event. (a) For purposes of this Section 10.19, (i) the term "Credit Rating Downgrade Event" means that the credit ratings established by S&P and Moody's with respect to the senior, unsecured long-term, non-credit enhanced debt securities ("Long-Term Debt") of Bank of America Corporation have been lowered below "A-" by S&P and below "A3" by Moody's or that the Long-Term Debt of Bank of America Corporation has ceased to be rated by S&P and Moody's for a period of more than ten (10) consecutive Business Days, in each case as evidenced by the public announcement thereof; provided that if the credit ratings established for the Long-Term Debt of Bank of America Corporation by either S&P or Moody's are unavailable for any reason, the rating established by the other rating agency shall be used for purposes of this Section 10.19; and (ii) the term "Deposit Account Interest Non-Payment Event" means that an

interest payment in respect of the Credit-Linked Deposits payable pursuant to Section 2.01(j)(i) shall not have been made by Bank of America and distributed by the Administrative Agent to the Lenders pursuant to Section 2.01(j)(i) within five (5) Business Days of when due. The Administrative Agent will promptly notify each Lender (A) after officers within the agency management group (or similar group performing the functions of the present agency management group) of Bank of America obtain knowledge, if a Credit Rating Downgrade Event has occurred, and (B) if a Deposit Interest Non-Payment Event has occurred.

- (b) In the event that a Credit Rating Downgrade Event or a Deposit Account Interest Non-Payment Event has occurred and is continuing, the Required Lenders may, by notice to the Administrative Agent and the Company (each a "Required Notice"), identify the Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event which then exists, and request that either of the following actions be taken, and upon such request the Company, the Administrative Agent, the Issuer and the Lenders agree to use reasonable commercial efforts to cause such action to be taken:
- (i) A new deposit account (the "New Account") shall be opened by the Administrative Agent at a bank which has a credit rating of at least "A+" from S&P and "A1" from Moody's (the "New Depository Bank"). The New Depository Bank, the Administrative Agent, the Issuer, the Company and the Lenders shall execute such documents as may be reasonably necessary or appropriate so that the Issuer is granted by the Administrative Agent and the Lenders a first priority perfected security interest in the New Account securing the Participations. Upon execution and delivery of such documents (including customary legal opinions), and evidence reasonably satisfactory to the Issuer of perfection with first priority of such security interest, all funds then held in the Credit-Linked Deposit Account shall be transferred in immediately available funds to the New Depository Bank for credit to the New Account. Thereafter, the New Account shall be deemed to be the "Credit-Linked Deposit Account" under this Agreement and the provisions of this Agreement relating to the Credit-Linked Deposit Account shall apply *mutatis mutandis* to the New Account. Except for the transfer of the Credit-Linked Deposit Account as provided above, the rights of the Administrative Agent, on behalf of the Issuer and the Lenders, in and to the Credit-Linked Deposit Account hereunder shall remain unchanged and in full force and effect.

In the event that the interest payable by the New Depository Bank on the funds on deposit in the New Account is less than the interest set forth in Section 2.01(j)(i) of this Agreement with respect to the Total Credit-Linked Deposit, then the Company shall pay such additional amounts at such times as may be required so that the Lenders receive no less than the interest payable under Section 2.01(j)(i), at the times set forth in Section 2.01(j)(i). If a Deposit Account Interest Non-Payment Event has triggered the procedures set forth in this Section 10.19(b), then Bank of America shall (A) indemnify the Company for all additional amounts the Company pays pursuant to the immediately preceding sentence and (B) reimburse the Company and each Lender for all costs and expenses, including Attorney Costs, incurred in connection with such Deposit Account Interest Non-Payment Event.

The parties to this Agreement shall execute and deliver an amendment to this Agreement and such other documents as may be reasonably required to accomplish the

foregoing. Notwithstanding anything to the contrary in this Section 10.19(b)(i), the documentation, terms and conditions utilized to establish and grant a security interest in the New Account and the other documentation (including legal opinions) and actions contemplated herein (including any amendment of this Agreement) shall be reasonably satisfactory in form and substance to the Company, the Issuer, the Administrative Agent and each Lender.

or

(ii) The Administrative Agent and the Lenders shall cause to be established, and beneficially owned by the Lenders, a trust (the "Trust") which shall in turn establish a securities account (the "Securities Account") at a securities intermediary (the "Securities Intermediary") having a credit rating of at least "A+" from S&P and at least "A1" from Moody's. The Securities Intermediary, the Administrative Agent, the Issuer, the Company and the Lenders shall execute such documents as may be reasonably necessary or appropriate so that the Issuer is granted by the Trust, the Administrative Agent and the Lenders, as appropriate, a first priority perfected security interest in the Securities Account securing the Participations. Upon execution and delivery of such documents (including customary legal opinions), and evidence reasonably satisfactory to the Issuer of perfection with first priority of such security interest, all funds then held in the Credit-Linked Deposit Account shall be transferred in immediately available funds to the Securities Intermediary for credit to the Securities Account. Such funds shall be invested in such LIBOR-based short-term debt investments, having a credit rating of at least "A+" from S&P and at least "A1" from Moody's as may be reasonably agreed upon by the Required Lenders, the Administrative Agent, the Issuer and the Company. Thereafter, the Securities Account shall be deemed to be the "Credit Linked Deposit Account" under this Agreement and the provisions of this Agreement relating to the Credit-Linked Deposit Account shall apply *mutatis mutandis* to the Securities Account. Except for the transfer of funds from the Credit-Linked Deposit Account to the Securities Account as provided above, the rights of the Administrative Agent, on behalf of the Issuer and the Lenders, in and to the Securities Account hereunder shall remain unchanged and in full force and effect.

In the event that the interest payable with respect to the investments in the Securities Account is less than the interest set forth in Section 2.01(j)(i) with respect to the Total Credit-Linked Deposit, then the Company shall pay such additional amounts at such times as may be required so that the Lenders receive no less than the interest payable under Section 2.01(j)(i), at the times set forth in Section 2.01(j)(i). If a Deposit Account Interest Non-Payment Event has triggered the procedures set forth in this Section 10.19(b), then Bank of America shall (A) indemnify the Company for all additional amounts the Company pays pursuant to the immediately preceding sentence and (B) reimburse the Company and each Lender for all costs and expenses, including Attorney Costs, incurred in connection with such Deposit Account Interest Non-Payment Event.

The parties to this Agreement shall execute and deliver an amendment to this Agreement and such other documents as may be reasonably required to accomplish the foregoing. Notwithstanding anything to the contrary in this Section 10.19(b)(ii), the documentation, terms and conditions utilized to establish the Trust and Securities Account and to grant a security interest in and to the Security Account and the other documentation (including

legal opinions) and actions contemplated herein (including any amendment of this Agreement) shall be reasonably satisfactory in form and substance to the Company, the Issuer, the Administrative Agent and each Lender.

(iii) The Company hereby agrees to use commercially reasonable efforts from time to time to inform the Administrative Agent, which shall promptly inform the Lenders upon receipt of such information, of the status of the transactions contemplated in Sections 10.19(b)(i) or (ii), as applicable. In the event that neither of the transactions contemplated in subsections (i) or (ii) above, or transactions substantially similar to the transactions in such sections, are consummated in form and substance reasonably satisfactory to the Issuer, the Administrative Agent, the Company and each Lender within 120 days (or such longer period as the Company and the Required Lenders may mutually agree) after the receipt by the Company of a Required Notice, then the Required Lenders may, by notice to the Administrative Agent, terminate the credit facility established under Section 2.01. The Required Lenders may, within 90 days of issuing a Required Notice, provide the Company written notice of their election to retract such Required Notice, but shall have no right to retract the Required Notice after such 90 day period. Upon delivery of a notice of retraction, the actions required by Section 10.19(b)(i) or (ii) shall no longer be required to be taken in connection with the event described in the Required Notice. After delivery of a retraction notice or in the event the credit facility is not terminated at the end of the 120 day period, the Required Lenders may not deliver another Required Notice under this Section 10.19 unless the credit ratings from S&P or Moody's on the Long-Term Debt of Bank of America are further lowered. Upon such termination, all funds then or at any time thereafter held in the Credit-Linked Deposit Account in excess of the sum of the then Outstanding Amount of all Credit Obligations (other than Advances and Term Loans) shall be withdrawn from the Credit-Linked Deposit Account by the Administrative Agent and distributed by the Administrative Agent to each Lender in accordance with its Pro Rata Share, subject to any notice required pursuant to the Money Market Account Agreement; provided that the Company may, pursuant to Section 2.03(b), request that the Lenders make Term Loans as provided in Section 2.03(b)(i); and with respect to all Term Loans (whether then outstanding or thereafter made), the restriction that Term Loans cannot be prepaid until after the second anniversary of the Closing Date set forth in Section 2.03(b)(vi) and the Prepayment Premium, if any, shall continue to apply.

ARTICLE XI HOLDINGS GUARANTY

11.01 Guaranty. For value received and hereby acknowledged and as an inducement to the Lenders and the Issuer to make the Credit Extensions available to the Company, Holdings hereby unconditionally and irrevocably guarantees (a) the full punctual payment when due, whether at stated maturity, by acceleration or otherwise, of all Credit Obligations of the Company now or hereafter existing whether for principal, interest, fees, expenses or otherwise, and (b) the strict performance and observance by the Company of all agreements, warranties and covenants applicable to the Company in the Credit Documents and (c) the obligations of the Company under the Credit Documents (such Credit Obligations collectively being hereafter referred to as the "Guaranteed Obligations"). The foregoing guaranty is a guaranty of payment

and not of collection. For purposes of this Article XI, the obligations of Holdings under this Article XI are referred to as the "Holdings Guaranty."

- 11.02 Guaranty Absolute. Holdings guarantees that the Guaranteed Obligations will be paid strictly in accordance with the terms hereof, regardless of any law, regulation or order now or hereafter in effect in any jurisdiction affecting any of such terms or the rights of any Lender, the Issuer or the Administrative Agent with respect thereto. The liability of Holdings under the Holdings Guaranty with regard to the Guaranteed Obligations shall be absolute and unconditional irrespective of
- (a) any change in the time, manner or place of payment of, or in any other term of, all or any of its Guaranteed Obligations or any other amendment or waiver of or any consent to departure from this Agreement or any other Credit Document (with regard to such Guaranteed Obligations);
 - (b) any release or amendment or waiver of or consent to departure from any other guaranty for all or any of its Guaranteed Obligations;
 - (c) any change in ownership of the Company;
 - (d) any acceptance of any partial payment(s) from the Company or Holdings; or
- (e) any other circumstance similar or dissimilar to the foregoing which might otherwise constitute a defense available to, or a discharge of, the Company in respect of its Credit Obligations under any Credit Document.

The Holdings Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any Guaranteed Obligation is rescinded or must otherwise be returned by the Lenders, the Issuer or the Administrative Agent upon the insolvency, bankruptcy or reorganization of the Company or otherwise, all as though such payment had not been made.

11.03 Effectiveness; Enforcement. The Holdings Guaranty shall be effective and shall be deemed to be made with respect to each Credit Extension as of the time it is made, issued or extended. No invalidity, irregularity or unenforceability by reason of any bankruptcy or similar law, or any law or order of any government or agency thereof purporting to reduce, amend or otherwise affect any liability of the Company, and no defect in or insufficiency or want of powers of the Company or irregular or improperly recorded exercise thereof, shall impair, affect, be a defense to or claim against such guaranty. The Holding Guaranty is a continuing guaranty and shall (a) survive any termination of this Agreement, and (b) remain in full force and effect until payment in full of, and performance of, all Guaranteed Obligations and all other amounts payable under this Agreement. The Holdings Guaranty is made for the benefit of the Administrative Agent, the Issuer and the Lenders and their successors and assigns, and may be enforced from time to time as often as occasion therefor may arise and without requirement on the part of the Administrative Agent, the Issuer or the Lenders first to exercise any rights against the Company, or to resort to any other source or means of obtaining payment of any of the said obligations or to elect any other remedy.

- 11.04 Waiver. Except as otherwise specifically provided in any of the Credit Documents, Holdings hereby waives promptness, diligence, protest, notice of protest, all suretyship defenses, notice of acceptance and any other notice with respect to any of its Guaranteed Obligations and the Holdings Guaranty and any requirement that the Lenders, the Issuer or the Administrative Agent protect, secure, perfect any security interest or Lien or any property subject thereto or exhaust any right or take any action against the Company or any other Person. Holdings also irrevocably waives, to the fullest extent permitted by law, all defenses which at any time may be available to it in respect of its Guaranteed Obligations by virtue of any statute of limitations, valuation, stay, moratorium law or other similar law now or hereafter in effect.
- 11.05 Expenses. Holdings hereby promises to reimburse (a) the Administrative Agent, the Issuer and the Lenders for all reasonable out-of-pocket fees and disbursements (including all reasonable Attorneys Costs), incurred or expended in connection with the preparation, filing or recording, or interpretation of the Holdings Guaranty, the other Credit Documents to which Holdings is a party, or any amendment, modification, approval, consent or waiver hereof or thereof, and (b) the Administrative Agent, the Issuer and the Lenders and their respective affiliates for all reasonable out-of-pocket fees and disbursements (including reasonable Attorney Costs), incurred or expended in connection with the enforcement of its Guaranteed Obligations (whether or not legal proceedings are instituted). Holdings will pay any taxes (including any interest and penalties in respect thereof) other than the Lenders' taxes based on overall income or profits, payable on or with respect to the transactions contemplated by the Holdings Guaranty, Holdings hereby agreeing jointly and severally to indemnify each Lender with respect thereto.

11.06 Concerning Joint and Several Liability of Holdings.

- (a) Holdings hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the Company, with respect to the payment and performance of all of its Guaranteed Obligations (including, without limitation, any Guaranteed Obligations arising under this Article XI), it being the intention of the parties hereto that all such Guaranteed Obligations shall be the joint and several Guaranteed Obligations of Holdings and the Company without preferences or distinction among them.
- (b) If and to the extent that the Company shall fail to make any payment with respect to any of its Credit Obligations as and when due or to perform any of its Guaranteed Obligations in accordance with the terns thereof, then in each such event Holdings will make such payment with respect to, or perform, such Guaranteed Obligation.
- (c) The Guaranteed Obligations of Holdings under the provisions of this Article XI constitute full recourse obligations of Holdings enforceable against Holdings to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of this Agreement or any other circumstance whatsoever.
- (d) Except as otherwise expressly provided in this Agreement, Holdings hereby waives notice of acceptance of its joint and several liability, notice of any Credit Extensions made or issued under this Agreement, notice of any action at any time taken or omitted by the Administrative Agent, the Issuer or the Lenders under or in respect of any of the Guaranteed

Obligations, and, generally, to the extent permitted by applicable law, all demands, notices and other formalities of every kind in connection with this Agreement. Holdings hereby assents to, and waives notice of, any extension or postponement of the time for the payment of any of the Guaranteed Obligations, the acceptance of any payment of any of the Guaranteed Obligations, the acceptance of any partial payment thereon, any waiver, consent or other action or acquiescence by the Administrative Agent, the Issuer or the Lenders at any time or times in respect of any Default or Event of Default by the Company or Holdings in the performance or satisfaction of any term, covenant, condition or provision of this Agreement or any other Credit Document, any and all other indulgences whatsoever by the Administrative Agent, the Issuer or the Lenders in respect of any of the Guaranteed Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Guaranteed Obligations or the addition, substitution or release, in whole or in part, of the Company or Holdings. Without limiting the generality of the foregoing, Holdings assents to any other action or delay in acting or failure to act on the part of the Lenders, the Issuer or the Administrative Agent with respect to the failure by the Company or Holdings to comply with its respective Credit Obligations or Guaranteed Obligations, including, without limitation, any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with applicable laws or regulations thereunder, which might, but for the provisions of this Article XI, afford grounds for terminating, discharging or relieving Holdings, in whole or in part, from any of the Guaranteed Obligations under this Article XI, it being the intention of Holdings that, so long as any of the Guaranteed Obligations hereunder remain unsatisfied, the Guaranteed Obligations of Holdings under this Article XI shall not be discharged except by performance and then only to the extent of such performance. The Guaranteed Obligations of Holdings under this Article XI shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, reconstruction or similar proceeding with respect to the Company or Holdings or the Lenders, the Issuer or the Administrative Agent. The joint and several liability of Holdings hereunder shall continue in full force and effect notwithstanding any absorption, merger, consolidation, amalgamation or any other change whatsoever in the name, membership, constitution or place of formation of the Company or Holdings, the Lenders, the Issuer or the Administrative Agent.

- (e) Holdings shall be liable under this Article XI only for the maximum amount of such liabilities that can be incurred under applicable law without rendering this Article XI voidable under applicable law relating to fraudulent conveyance and fraudulent transfer, and not for any greater amount. Accordingly, if any obligation under any provision under this Article XI shall be declared to be invalid or unenforceable in any respect or to any extent, it is the stated intention and agreement of Holdings, the Administrative Agent, the Issuer and the Lenders that any balance of the obligation created by such provision and all other obligations of Holdings under this Article XI to the Lenders, the Issuer or the Administrative Agent shall remain valid and enforceable, and that all sums not in excess of those permitted under applicable law shall remain fully collectible by the Lenders, the Issuer and the Administrative Agent from the Company or Holdings, as the case may be.
- (f) The provisions of this Article XI are made for the benefit of the Administrative Agent, the Issuer and the Lenders and their successors and assigns, and may be enforced in good faith by them from time to time against Holdings as often as occasion therefor may arise and

without requirement on the part of the Administrative Agent, the Issuer or the Lenders first to marshal any of their claims or to exercise any of their rights against the Company or Holdings or to exhaust any remedies available to them against the Company or Holdings or to resort to any other source or means of obtaining payment of any of the obligations hereunder or to elect any other remedy. The provisions of this Article XI shall remain in effect until all of the Guaranteed Obligations shall have been paid in full or otherwise fully satisfied, any remaining Credit-Linked Deposits in the Credit-Linked Deposit Account are withdrawn and distributed to the Lenders in accordance with their respective Pro Rata Shares, and all outstanding Credit shall have expired, matured or otherwise been terminated. If at any time, any payment, or any part thereof, made in respect of any of the Guaranteed Obligations, is rescinded or must otherwise be restored or returned by the Lenders, the Issuer or the Administrative Agent upon the insolvency, bankruptcy or reorganization of the Company or Holdings, or otherwise, the provisions of this Article XI will forthwith be reinstated in effect, as though such payment had not been made.

11.07 Waiver. Until the final payment and performance in full of all of the Credit Obligations, Holdings shall not exercise and Holdings hereby waives any rights Holdings may have against the Company arising as a result of payment by Holdings hereunder, by way of subrogation, reimbursement, restitution, contribution or otherwise, and will not prove any claim in competition with the Administrative Agent, the Issuer or any Lender in respect of any payment hereunder in any bankruptcy, insolvency or reorganization case or proceedings of any nature; Holdings will not claim any setoff, recoupment or counterclaim against the Company in respect of any liability of the Company to Holdings; and Holdings waives any benefit of and any right to participate in any collateral security which may be held by the Administrative Agent, the Issuer or any Lender.

11.08 Subrogation; Subordination. The payment of any amounts due with respect to any indebtedness of the Company for money borrowed or credit received now or hereafter owed to Holdings is hereby subordinated to the prior payment in full of all of the Credit Obligations. Holdings agrees that, after the occurrence of any default in the payment or performance of any of the Credit Obligations, Holdings will not demand, sue for or otherwise attempt to collect any such indebtedness of the Company to Holdings until all of the Credit Obligations shall have been paid in full. If, notwithstanding the foregoing sentence, Holdings shall collect, enforce or receive any amounts in respect of such indebtedness while any Credit Obligations are still outstanding, such amounts shall be collected, enforced and received by Holdings as trustee for the Lenders, the Issuer and the Administrative Agent and be paid over to the Administrative Agent at Default, for the benefit of the Lenders, the Issuer, and the Administrative Agent on account of the Credit Obligations without affecting in any manner the liability of Holdings under the other provisions hereof.

11.09 Release of Guaranty. The Administrative Agent, the Issuer and the Lenders hereby agree that the foregoing Holdings Guaranty shall remain in full force and effect until the occurrence of either (a) the consolidation or merger of Holdings into the Company or its successors, (b) the consolidation or merger of the Company or its successors into Holdings, or (c) the release of Holdings from all other guarantees of the Company's obligations under the Company's existing credit facilities or other senior indebtedness, at which time the foregoing

Holdings Guaranty shall be concurrently released without any further action on the part of the Lenders, the Administrative Agent or the Issuer.

ARTICLE XII ENTIRE AGREEMENT

12.01 Entire Agreement. THIS AGREEMENT AND THE OTHER CREDIT DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

(remainder of page intentionally left blank)

IN WITNESS WHEREOF, the parties hereto have caused this Ten Year Letter of Credit and Term Loan Agreement to be duly executed as of the date first above written.

WASTE MANAGEMENT, INC.

By: /s/ Ronald H. Jones

Name: Ronald H. Jones

Title: Vice President & Treasurer

WASTE MANAGEMENT HOLDINGS, INC.

(for the limited purposes of Article XI)

By: /s/ Ronald H. Jones

Name: Ronald H. Jones

Title: Vice President & Treasurer

By: /s/ Jay Clement

Name: Jay Clement Title: Assistant Treasurer

BANK OF AMERICA, N.A., as

Administrative Agent and Issuer

By: /s/ Brian D. Corum

Name: Brian D. Corum Title: Managing Director

LENDERS:

SUNAMERICA LIFE INSURANCE COMPANY

By: AIG Global Investment Corp.

Investment Advisor

By: /s/Victoria Chin

Victoria Chin, Vice President

TRANSAMERICA LIFE INSURANCE AND ANNUITY COMPANY

By: /s/ Mark E. Dunn

Name: Mark E. Dunn Title: Vice President

PRINCIPAL LIFE INSURANCE COMPANY

By: Principal Global Investors, LLC a Delaware limited liability company, its authorized signatory

By: /s/ Jon C. Heiny, Counsel

Its: Jon C. Heiny, Counsel

By: /s/ Debra Svoboda, EPP Counsel

Its: Debra Svoboda, EPP Counsel

\$15,000,000

FIVE YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT

Dated as of June 30, 2003

among

WASTE MANAGEMENT, INC.

WASTE MANAGEMENT HOLDINGS, INC.

BANK OF AMERICA, N.A.,

as Administrative Agent

BANK OF AMERICA, N.A.,

as Letter of Credit Issuer

and

The Lenders Party Hereto

BANC OF AMERICA SECURITIES LLC

Sole Lead Arranger and Sole Book Manager



TABLE OF CONTENTS

ARTICLE I	DEFINITIONS AND ACCOUNTING TERMS	Page 1
1.01	Defined Terms	1
1.02	Other Interpretive Provisions	17
1.03	Accounting Terms	17
1.03	Rounding	18
1.05	References to Agreements and Laws	18
1.06	Credit Amounts	18
ARTICLE II	THE CREDIT-LINKED DEPOSITS AND CREDIT EXTENSIONS	18
2.01	Credits	18
2.02	Cash Collateral	27
2.02	Reduction of Total Credit-Linked Deposits; Conversion to Funded Term Loans	28
2.04	Interest on Borrowings Other Than Term Loans	30
2.05	Fees and Other Charges	31
2.06	Computation of Interest and Fees	32
2.07	Evidence of Debt	32
2.08	Payments Generally	33
2.09	Sharing of Payments	33
ARTICLE III	TAXES AND YIELD PROTECTION	34
_		34
3.01	Taxes	
3.02	Increased Cost and Reduced Return; Capital Adequacy or Risk-Based Capital	35
3.03	Illegality	36
3.04	Funding Losses	36
3.05	Matters Applicable to All Requests for Compensation	37
3.06	Survival	37
3.07	Replacement of Lenders	37
ARTICLE IV	CONDITIONS PRECEDENT TO CREDIT EXTENSIONS	38
4.01	Conditions of Credit-Linked Deposits and Initial Credit Extension	38
4.02	Conditions to all Credit Extensions by the Issuer	40
4.03	Conditions to Term Loans	41
ARTICLE V	REPRESENTATIONS AND WARRANTIES	41
5.01	Organization; Power and Authority	41
5.02	Authorization, etc	42
5.03	Disclosure	42
5.04	Organization and Ownership of Shares of Subsidiaries; Affiliates	43
5.05	Financial Statements	43
5.06	Compliance with Laws, Other Instruments, etc	44
5.07	Governmental Authorizations, etc	44
5.08	Litigation; Observance of Agreements, Statutes and Orders	44
5.09	Taxes	45
5.10	Title to Property; Leases	45
5.11	Licenses, Permits, etc	45

TABLE OF CONTENTS

(continued)

5.12	Compliance with ERISA	Pag 46
5.13	Private Offering by Company	46
5.14	Use of Proceeds; Margin Regulations	47
5.15	Existing Indebtedness	47
5.16	Foreign Assets Control Regulations, etc	47
5.17	Status under Certain Statutes	48
5.18	Environmental Matters	48
5.19	Compliance With Issuance Limits	48
5.20	Tax Shelter Regulations	48
ARTICLE VI	AFFIRMATIVE COVENANTS	49
6.01	Financial and Business Information	49
6.02	Officer's Certificate	51
6.03	Inspection	52
6.04	Compliance with Laws	52
6.05	Insurance	52
6.06	Maintenance of Properties	52
6.07	Payment of Taxes and Claims	52
6.08	Corporate Existence, Etc	53
6.09	Books and Records	53
6.10	Purpose of Credits	53
ARTICLE VII	NEGATIVE COVENANTS	
7.01	Transactions with Affiliates	53 53
7.01		53 54
	Merger, Consolidation, Sale of Assets, etc	
7.03 7.04	Liens Limitation on Consolidated Debt	56
		58 58
7.05	Limitation on Priority Debt	
7.06 7.07	Nature of Business	58 58
	Designation of Unrestricted Subsidiaries	
7.08	Margin Stock	59
ARTICLE VIII	EVENTS OF DEFAULT AND REMEDIES	59
8.01	Events of Default	59
8.02	Remedies Upon Event of Default	61
8.03	Application of Funds	62
ARTICLE IX	ADMINISTRATIVE AGENT	63
9.01	Appointment and Authorization of Administrative Agent	63
9.02	Delegation of Duties	63
9.03	Liability of Administrative Agent	64
9.04	Reliance by Administrative Agent	64
9.05	Notice of Default	64
9.06	Credit Decision; Disclosure of Information by Administrative Agent	65
9.07	Indemnification of Administrative Agent	65
9.08	Administrative Agent in its Individual Capacity	67
9.09	Successor Administrative Agent	67

TABLE OF CONTENTS

(continued)

		Page
ARTICLE X	MISCELLANEOUS	67
10.01	Amendments, Etc	67
10.02	Notices and Other Communications; Facsimile Copies	69
10.03	No Waiver; Cumulative Remedies	70
10.04	Attorney Costs, Expenses and Taxes	70
10.05	Indemnification by the Company	71
10.06	Payments Set Aside	71
10.07	Successors and Assigns	72
10.08	Confidentiality	74
10.09	[Intentionally Omitted]	75
10.10	Representations and Warranties of Each Lender	75
10.11	Counterparts	77
10.12	Integration	77
10.13	Survival of Representations and Warranties	77
10.14	Severability	78
10.15	Tax Forms	78
10.16	Governing Law	80
10.17	Waiver of Right to Trial by Jury	80
10.18	Time of the Essence	80
10.19	Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event	80
ARTICLE XI	HOLDINGS GUARANTY	83
11.01	Guaranty	83
11.02	Guaranty Absolute	84
11.03	Effectiveness; Enforcement	84
11.04	Waiver	84
11.05	Expenses	85
11.06	Concerning Joint and Several Liability of Holdings	85
11.07	Waiver	87
11.08	Subrogation; Subordination	87
11.09	Release of Guaranty	87
ARTICLE XII	ENTIRE AGREEMENT	88
12.01	Entire Agreement	88
SIGNATURES		S-1

SCHEDULES	
A	Applicable Rates; Interest on Credit-Linked Deposits
1.01	Existing Credits
2.01	Credit-Linked Deposits and Pro Rata Shares
5.04	Organization and Ownership Shares of Subsidiaries; Affiliates
5.05	Financial Statements
5.08	Litigation; Observance of Agreements, Statutes and Orders
5.11	Licenses, Permits, etc.
5.15	Existing Indebtedness
5.17	Status under Certain Statutes
7.03	Liens
10.02	Administrative Agent's Office; Certain Addresses for Notices
EXHIBITS	
	Form of
A	Request for Credit Extension
В	Assignment and Assumption
C-1	Opinion of Internal Counsel to the Company and Holdings
C-2	Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer
B C-1	Request for Credit Extension Assignment and Assumption Opinion of Internal Counsel to the Company and Holdings

Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to the Administrative

C-3

C-4

E

F G

Η

Agent and the Issuer

Term Note Opinion

Term Note

Letter of Credit Linked Note

Money Market Account Agreement

Request for Term Loans

Opinion of Special Counsel to the Lenders

FIVE YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT

This FIVE YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT ("Agreement") is entered into as of June 30, 2003, among WASTE MANAGEMENT, INC., a Delaware corporation (the "Company"), WASTE MANAGEMENT HOLDINGS, INC., a Delaware corporation and wholly-owned Subsidiary of the Company ("Holdings") (executing this Agreement for the limited purposes set forth in Article XI hereof), BANK OF AMERICA, N.A., as issuer of letters of credit (the "Issuer"), the financial institutions having a Credit-Linked Deposit set forth opposite their names on Schedule 2.01 hereto under the heading "Credit-Linked Deposit" (collectively, the "Lenders" and individually, a "Lender"), and BANK OF AMERICA, N.A., as Administrative Agent.

The Company has requested that the Issuer and the Lenders provide a letter of credit facility, and the Issuer and the Lenders are willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I DEFINITIONS AND ACCOUNTING TERMS

1.01 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth below:

"Adjusted Consolidated EBITDA" means for any period, the total of the following calculated without duplication on a consolidated basis for such period: (i) Consolidated EBITDA, plus (ii) on a pro forma basis, the EBITDA of each Person acquired, or, as applicable, the EBITDA attributable to the assets acquired from any Person, by the Company or any Restricted Subsidiary in the subject period, for any portion of such subject period occurring prior to the date of the acquisition by the Company or any Restricted Subsidiary of such Person or related assets; provided, further, that any calculation pursuant to clause (ii) above of pro forma EBITDA for any subject period which gives effect to asset acquisitions and/or stock purchases during the subject period on a pro forma basis as provided above shall in any event be determined by the Company in accordance with sound financial practice in accordance with GAAP and on the basis, to the extent available, of appropriate financial statements and tax returns for the applicable period, and shall be set forth in a certificate of a Senior Financial Officer of the Company accompanied by calculations in reasonable detail showing the manner of determination thereof, which certificate shall be furnished to the Administrative Agent, the Issuer and each Lender not later than the certificate required to be furnished by the Company in respect of the fiscal period in which such date of determination occurs pursuant to Section 6.02 of this Agreement.

"Administrative Agent" means Bank of America in its capacity as administrative agent under any of the Credit Documents, or any successor administrative agent.

"Administrative Agent's Office" means the Administrative Agent's address and, as

appropriate, account as set forth on Schedule 10.02, or such other address or account as the Administrative Agent may from time to time specify by notice to the Company, the Issuer and the Lenders.

- "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.
- "Advance" means, with respect to each Lender, such Lender's funding of its participation in any Borrowing (other than Term Loans) in accordance with its Pro Rata Share.
- "Affiliate" means, at any time, (a) with respect to the Company and each of its Restricted Subsidiaries, any Person (other than the Company and each of its Restricted Subsidiaries) that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, the Company, (b) with respect to any Person (other than the Company and each of its Restricted Subsidiaries), any other Person that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, such first Person, and (c) any Person beneficially owning or holding, directly or indirectly, 10% or more of any class of voting or equity interests of the Company or any Subsidiary of the Company or any corporation of which the Company and its Subsidiaries beneficially own or hold, in the aggregate, directly or indirectly, 10% or more of any class of voting or equity interests. As used in this definition, "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. Unless the context otherwise clearly requires, any reference to an 'Affiliate' is a reference to an Affiliate of the Company.
 - "Agent Indemnified Liabilities" has the meaning specified in Section 9.07.
- "Agent-Related Persons" means the Administrative Agent or the Issuer, together with its Affiliates (including, in the case of Bank of America in its capacity as the Administrative Agent, BAS), and the officers, directors, employees, agents and attorneys-in-fact of such Persons and Affiliates.
 - "Agreement" means this Five Year Letter of Credit and Term Loan Agreement.
 - "Applicable Rate" has the meaning set forth on Schedule A annexed hereto.
 - "Arranger" means BAS, in its capacity as sole lead arranger and sole book manager.
 - "Assignment and Assumption" means an Assignment and Assumption substantially in the form of Exhibit B.
 - "Attorney Costs" means and includes all fees, expenses and disbursements of any law firm or other external counsel.
- "Audited Financial Statements" means the audited consolidated balance sheet of the Company and its Subsidiaries for the fiscal year ended December 31, 2002, and the related

consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year of the Company and its Subsidiaries, including the notes thereto.

"Authorized Representative" means any employee of a Credit Party involved principally in the financial administration, controllership or treasury function of such Credit Party who is authorized or designated as an Authorized Representative, as evidenced by the certificate delivered pursuant to Section 4.01(a)(iv) or from time to time by a certificate delivered by a Senior Financial Officer of such Credit Party to the Administrative Agent in form and substance satisfactory to it. Any document delivered hereunder that is signed by an Authorized Representative of a Credit Party shall be conclusively presumed to have been authorized by all necessary corporate action on the part of such Credit Party and such Authorized Representative shall be conclusively presumed to have acted on behalf of such Credit Party.

"Auto-Reinstating Credit" means any Credit containing any provisions providing for automatic reinstatement of the stated amount after any drawing thereunder.

"Auto-Renewal Credit" has the meaning specified in Section 2.01(b)(iii).

"Bank of America" means Bank of America, N.A. and its successors.

"BAS" means Banc of America Securities LLC.

"Base Rate" means for any day a fluctuating rate per annum equal to the higher of (a) the Federal Funds Rate plus 1/2 of 1% and (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate." Such rate is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such rate announced by Bank of America shall take effect at the opening of business on the day specified in the public announcement of such change.

"Beneficial Ownership" has the meaning set forth in Rule 13d-3 promulgated by the SEC under the Securities Exchange Act.

"Borrowing" means, without duplication, an extension of credit resulting from a drawing under any Credit which has not been reimbursed on the applicable Reimbursement Date, including the making of Advances, or the making of a Term Loan, as the context may require.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, New York or Dallas, or (if the Administrative Agent's Office is not located in New York or Dallas) the state where the Administrative Agent's Office is located, and, if such day relates to the Credit-Linked Deposit Account or any LIBO Rate Term Loan or LIBO Rate Advance, means any such day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Capital Leases" or "Capitalized Leases" means leases under which the Company or any

of its Restricted Subsidiaries is the lessee or obligor, the discounted future rental payments under which are required to be capitalized on the balance sheet of the lessee or obligor in accordance with GAAP.

"Cash Collateral" has the meaning specified in Section 2.02.

"Cash Collateralize" has the meaning specified in Section 2.02.

"Closing Date" means the first date all the conditions precedent in Section 4.01 are satisfied or waived in accordance with Section 4.01 (or, in the case of Section 4.01(b), waived by the Person entitled to receive the applicable payment).

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"Company" has the meaning specified in the introductory paragraph hereto.

"Consolidated Debt" means, as of any date of determination, the total of all Indebtedness of the Company and its Restricted Subsidiaries determined on a consolidated basis in accordance with GAAP.

"Consolidated EBITDA" means, for any period, Consolidated Net Income or Consolidated Net Loss, as the case may be, for such period, plus the sum of (a) interest expense; (b) income tax expense; (c) depletion and depreciation expense; and (d) amortization expense; in each case, which were deductible in determining Consolidated Net Income or Consolidated Net Loss for such period, determined on a consolidated basis for the Company and its Restricted Subsidiaries in accordance with GAAP.

"Consolidated Net Income or Consolidated Net Loss" means, respectively, for any period, the net income or loss for such period determined on a consolidated basis for the Company and its Restricted Subsidiaries in accordance with GAAP.

"Consolidated Net Worth" means, as of any date of determination, the consolidated stockholders' equity of the Company and its Restricted Subsidiaries (excluding any amounts attributable to mandatorily redeemable Preferred Stock) determined as of such date.

"Consolidated Tangible Assets" means, as of any date of determination, Consolidated Total Assets less the sum of (i) Intangible Assets and (ii) all amounts representing write-up in the book value of any assets of the Company and its Restricted Subsidiaries resulting from a revaluation thereof.

"Consolidated Total Assets" means all assets of the Company and its Restricted Subsidiaries determined on a consolidated basis in accordance with GAAP.

"Credit" means any standby letter of credit issued or outstanding hereunder (including a so-called "direct pay" standby letter of credit) that is a Permitted Credit and shall include the Existing Credits.

"Credit Documents" means this Agreement, the Fee Letter, each Request for Credit Extension, each Issuer Document, each Request for Term Loan, each Letter of Credit Linked Note and each Term Note and each other document, instrument or agreement from time to time executed by the Company or any of its Subsidiaries, Holdings, or any secretary, assistant secretary, Responsible Officer or Authorized Representative thereof and delivered in connection with this Agreement.

"Credit Expiration Date" means the day that is seven (7) Business Days prior to the Maturity Date (or, if such day is not a Business Day, the next preceding Business Day).

"Credit Extension" means, (a) with respect to any Credit, the issuance thereof or extension of the expiry date thereof, or the renewal, reinstatement or increase of the amount thereof and (b) the making of any Term Loans.

"Credit-Linked Deposit" shall have the meaning specified in Section 2.01(c)(ii).

"Credit-Linked Deposit Account" shall mean the deposit account established by the Administrative Agent in its name and under its sole and exclusive control maintained at the office of Bank of America at its principal office in Chicago, Illinois, designated as the "Bank of America N.A. as Administrative Agent Waste Management Inc Five Year LOC Facility Lender Credit Linked Notes Dep Acct Treasury" that shall be used solely for the purposes set forth in Sections 2.01(c) (ii) and (iii) and in Section 2.03(b)(iii), or such successor account established pursuant to Section 10.19 of this Agreement.

"Credit-Linked Lenders" means the Lenders under this Agreement and the lenders under the Seven Year Agreement and the Ten Year Agreement.

"Credit Obligations" means, without duplication, as at any date of determination, the aggregate undrawn face amount of all outstanding Credits plus the aggregate amount of all Unreimbursed Amounts in respect of any Credits, and, without duplication, all outstanding Borrowings arising from any Credits, plus the aggregate amount of the Term Loans. For all purposes of this Agreement, if on any date of determination a Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Credit Parties" means, collectively, the Company and Holdings.

"Credit Rating Downgrade Event" has the meaning specified in Section 10.19(a)(i).

"Debtor Relief Laws" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means an event or condition the occurrence or existence of which would, with

the lapse of time or the giving of notice or both, become an Event of Default.

"Default Rate" means (a) with respect to obligations other than Advances and Term Loans, the Base Rate plus 2.00% per annum, and (b) with respect to Advances and Term Loans, the LIBO Rate plus 3.00% per annum, in each case to the fullest extent permitted by applicable Laws.

"Deposit Account Interest Non-Payment Event" has the meaning specified in Section 10.19(a)(ii).

"Disclosure Documents" means the Company's financial statements referred to in Section 5.05 and filings made by the Company or Holdings with the SEC that were publicly available prior to the date of this Agreement and which were provided to the Issuer and the Lenders.

"Dollar" and "\$" means lawful money of the United States.

"EBITDA" means, (without duplication) with respect to any Person, division or business and with respect to any period, the sum of (i) (a) net income for such period from continuing operations after extraordinary items (excluding gains or losses from disposition of assets); (b) consolidated interest charges for such period; (c) the amount of taxes, based on or measured by income, used or included in the determination of such net income; (d) the amount of depreciation and amortization expense deducted in determining such net income; and (e) any extraordinary losses or non-recurring expenses not requiring the expenditure of cash decreasing net income for such period, minus (ii) any extraordinary gains or non-recurring items increasing net income for such period.

"Eligible Assignee" means any Institutional Investor.

"Environmental Laws" means any and all federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including but not limited to those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder from time to time in effect.

"ERISA Affiliate" means any trade or business (whether or not incorporated) that is treated as a single employer together with the Company under Section 414 of the Code.

"Escalating Credit" means a Credit which provides for a face amount that increases from time to time in accordance with its terms.

"Event of Default" has the meaning specified in Section 8.01.

"Existing Credits" means the standby letters of credit listed on Schedule 1.01.

"Facility Fee" has the meaning specified in Section 2.05(a).

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" has the meaning specified in Section 2.05(b).

"Foreign Lender" has the meaning specified in Section 10.15(a).

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Fee" has the meaning specified in Section 2.05(b).

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means

- (a) the government of
 - (i) the United States of America or any state or other political subdivision thereof, or
- (ii) any jurisdiction in which the Company or any Subsidiary conducts all or any part of its business, or which asserts jurisdiction over any properties of the Company or any Subsidiary, or
- (b) any entity exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, any such government.

"Guarantee" by any Person means any obligation, contingent or otherwise, of such Person directly or indirectly guaranteeing any Indebtedness of any other Person and, without limiting the generality of the foregoing, any obligation, direct or indirect, contingent or otherwise, of such Person (a) to purchase or pay (or advance or supply funds for the purchase or

payment of) such Indebtedness (whether arising by virtue of partnership arrangements, by agreement to keep-well, to purchase assets, goods, securities or services, to take-or-pay, or to maintain financial statement conditions or otherwise) or (b) entered into for the purpose of assuring in any other manner the obligee of such Indebtedness of the payment thereof or to protect such obligee against loss in respect thereof (in whole or in part); provided that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business. The term "Guarantee" used as a verb has a corresponding meaning.

"Guaranteed Obligations" has the meaning specified in Section 11.01.

"Hazardous Material" means any hazardous waste as defined by 42 U.S.C. §6903(5), any hazardous substance as defined by 42 U.S.C. §9601(14), any pollutant or contaminant as defined by 42 U.S.C. §9601(33) or any toxic substances, oil or other chemicals or substances regulated by the Environmental Laws, excluding household hazardous waste.

"Holdings" means Waste Management Holdings, Inc., a Delaware corporation and wholly-owned Subsidiary of the Company.

"Holdings Guaranty" means the meaning specified in Section 11.01.

"Indebtedness" means, collectively, without duplication, whether classified as Indebtedness, an investment or otherwise on the obligor's balance sheet, (a) all indebtedness for borrowed money, (b) all obligations for the deferred purchase price of property or services (other than trade payables incurred in the ordinary course of business which either (i) are not overdue by more than ninety (90) days, or (ii) are being disputed in good faith and for which adequate reserves have been established in accordance with GAAP), (c) all obligations evidenced by notes, bonds, debentures or other similar debt instruments, (d) all obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (e) all obligations, liabilities and indebtedness under Capitalized Leases, (f) all obligations, liabilities or indebtedness under drawn surety or performance bonds, or any other drawn bonding arrangement, (g) Guarantees with respect to all Indebtedness of others referred to in clauses (a) through (f) above, and (h) all Indebtedness of others referred to in clauses (a) through (f) above secured or supported by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured or supported by) any Lien on the property or assets of the Company or any Subsidiary, even though the owner of the property has not assumed or become liable, contractually or otherwise, for the payment of such Indebtedness; provided that if a Permitted Receivables Transaction is outstanding and is accounted for as a sale of accounts receivable under GAAP, Indebtedness shall also include the additional Indebtedness, determined on a consolidated basis, which would have been outstanding had such Permitted Receivables Transaction been accounted for as a borrowing.

"Indemnified Liabilities" has the meaning specified in Section 10.05.

"Indemnitees" has the meaning specified in Section 10.05.

"INHAM Exemption" has the meaning specified in Section 10.10(b)(v).

"Institutional Investor" means (a) any original Lender or any Affiliate of a Lender, (b) any Lender holding more than \$5,000,000 of the Credit-Linked Deposits and/or Credit Obligations then outstanding, and (c) any bank, trust company, savings and loans association or other financial institution, any pension plan, any investment company, any insurance company, any broker or dealer, any Person (other than a natural Person) engaged in making, purchasing, holding or otherwise investing in commercial loans in the ordinary course of its business, or any other similar financial institution or entity, regardless of legal form.

"Intangible Assets" means goodwill, the purchase price of acquired assets in excess of the fair value thereof, trademarks, trade names, service marks, customer lists, brand names, copyrights, patents and licenses, and rights with respect to the foregoing.

"Interest Period" means, (a) as to the Credit-Linked Deposits, the period commencing on the date such Credit-Linked Deposit is made and ending on the date three months thereafter; (b) as to Advances and Term Loans, the period commencing on the date any such Advance or Term Loan is disbursed (except that in respect of Term Loans made from the proceeds of the Credit-Linked Deposits pursuant to Sections 2.03(b)(i)(A) or (B), the Interest Period shall be deemed to commence on the same day as the Interest Period commencement date for the Credit-Linked Deposits, and in respect of Term Loans made through the conversion of Advances pursuant to Section 2.03(b)(i)(C), the Interest Period shall be deemed to commence on the same day as the Interest Period commencement date for the Credit-Linked Deposits, if any, on the date of such conversion) and ending on the earlier of (x) the date three months thereafter and (y) the last day of the Interest Period with respect to the Credit-Linked Deposits; and (c) after the initial Interest Periods described in (a) and (b) above, as to the Credit-Linked Deposits, Advances and Term Loans, the period commencing on the last day of the immediately preceding Interest Period and ending on the date three months thereafter; provided that:

- (i) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;
- (ii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and
 - (iii) no Interest Period shall extend beyond the Maturity Date;

provided further that after the occurrence and during the continuance of an Event of Default the Required Lenders, acting through the Administrative Agent, may designate one or more Interest Periods for the Advances and/or the Term Loans which are shorter than that set forth above, such designation to be effective after the last day of the applicable existing Interest Period.

"IRS" means the United States Internal Revenue Service.

"ISP" means, with respect to any Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance).

"Issuance Limits" has the meaning specified in Section 2.01(a)(i).

"Issuer" means Bank of America in its capacity as issuer of Credits hereunder, or any successors, and shall include any branch of the Issuer through which the Issuer may cause Credits to be issued hereunder.

"Issuer Documents" means with respect to any Credit issued by the Issuer, such Credit, any applications for issuance or amendment of such Credit, and any other document, agreement and instrument entered into by the Issuer and the Company or in favor of the Issuer and relating to any such Credit, including any of the Issuer's standard form documents for issuances and amendments of letters of credit and guarantees or other similar undertakings.

"Issuer-Related Persons" means as to the Issuer, together with its Affiliates, and the officers, directors, employees, agents and attorneys-in-fact of the Issuer and its Affiliates.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"Lender" has the meaning specified in the introductory paragraph hereto.

"Lending Office" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Company and the Administrative Agent.

"Letter of Credit Linked Note" has the meaning specified in Section 2.07(b).

"LIBO Rate" means for any Interest Period:

- (a) the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate that appears on the page of the Telerate screen (or any successor thereto) that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (b) if the rate referenced in the preceding clause (a) does not appear on such page or service or such page or service shall not be available, the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate on such other page or other service

that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or

(c) if the rates referenced in the preceding clauses (a) and (b) are not available, the rate per annum determined by the Administrative Agent as the rate of interest at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Total Credit-Linked Deposit or the average amount of the Advances of the Lenders or LIBO Rate Term Loans, as applicable, being made or continued and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 4:00 p.m. (London time) two Business Days prior to the first day of such Interest Period.

"LIBO Rate Term Loans" means Term Loans bearing interest based on the LIBO Rate.

"Lien" means, with respect to any Person, any mortgage, lien, pledge, charge, security interest or other encumbrance, or any interest or title of any vendor, lessor, lender or other secured party to or of such Person under any conditional sale or other title retention agreement or Capital Lease, upon or with respect to any property or asset of such Person (including in the case of stock, stockholder agreements, voting trust agreements and all similar arrangements).

"Long-Term Debt" has the meaning specified in Section 10.19(a).

"Material" or "Materially" means material or materially, as applicable, in relation to the business, operations, affairs, financial condition, assets, properties, or prospects of the Company and its Restricted Subsidiaries taken as a whole.

"Material Adverse Effect" means a material adverse effect on (a) the business, operations, affairs, financial condition, prospects, assets or properties of the Company and its Restricted Subsidiaries taken as a whole, (b) the ability of the Company to perform its obligations under this Agreement, the Letter of Credit Linked Notes or the Term Notes, or (c) the validity or enforceability of this Agreement, the Letter of Credit Linked Notes or the Term Notes.

"Material Debt" means any Indebtedness (other than any Indebtedness of the Company or Holdings in respect of the Credit Obligations) of any one or more of the Company, its Restricted Subsidiaries and Holdings in an aggregate principal amount for all such Persons exceeding \$50,000,000.

"Maturity Date" means (a) June 30, 2008, or (b) such earlier date upon which the obligations of the Issuer and the Lenders to make Credit Extensions may be terminated in accordance with the terms hereof.

"Memorandum" has the meaning specified in Section 5.03.

"Money Market Account Agreement" means the Money Market Account Agreement

between the Administrative Agent and Bank of America, as depository, substantially in the form of Exhibit G attached hereto, concerning the Credit-Linked Deposit Account.

"Moody's" means Moody's Investors Service, Inc., or any successor thereto that is a nationally-recognized rating agency.

"Multiemployer Plan" means any Plan that is a "multiemployer plan" (as such term is defined in Section 4001(a)(3) of ERISA).

"New Account" has the meaning specified in Section 10.19(b)(i).

"New Depository Bank" has the meaning specified in Section 10.19(b)(i).

"Nonrenewal Notice Date" has the meaning specified in Section 2.01(b)(iii).

"Officer's Certificate" means a certificate of a Senior Financial Officer or of any other officer of the Company whose responsibilities extend to the subject matter of such certificate.

"Other Taxes" has the meaning specified in Section 3.01(b).

"Outstanding Amount" means, with respect to any Credit Obligations on any date, the amount of such Credit Obligations on such date after giving effect to any Credit Extension occurring on such date and any other changes in the aggregate amount of the Credit Obligations as of such date, including as a result of any reimbursements of outstanding unpaid drawings under any Credits or any reductions in the maximum amount available for drawing under Credits taking effect on such date. Without limiting the provisions of Section 1.07, for avoidance of doubt, the portion of the Credit Obligations in respect of the undrawn face amount of any Escalating Credits shall be deemed to be the maximum aggregate amount available to be drawn under such Escalating Credits (after giving effect to all increases).

"Participant" has the meaning specified in Section 10.07(d).

"Participation" has the meaning specified in Section 2.01(c)(ii).

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA or any successor thereto.

"Permitted Credit" means a Credit used directly or indirectly to cover (a) a default in the performance of any non-financial or commercial obligations of the Company or any Subsidiary under specific contracts, including a Credit issued in favor of a bank or other surety who in connection therewith issues a guarantee or similar undertaking, performance bond, surety bond or other similar instrument that covers a default in the performance of any non-financial or commercial obligations under specific contract, or (b) the payment of any financial contractual obligation of the Company or any Subsidiary.

"Permitted Liens" has the meaning specified in Section 7.03.

"Permitted Receivables Transaction" means any sale or sales of, and/or securitization of,

any accounts receivable of the Company and/or any of its Restricted Subsidiaries (the "Receivables") pursuant to which (a) the Company and its Restricted Subsidiaries realize aggregate net proceeds of not more than \$750,000,000 at any one time outstanding, including, without limitation, any revolving purchase(s) of Receivables where the maximum aggregate uncollected purchase price (exclusive of any deferred purchase price) for such Receivables at any time outstanding does not exceed \$750,000,000, and (b) which Receivables shall not be discounted more than 25%.

"Person" means any individual, trustee, corporation, general partnership, limited partnership, limited liability company, joint stock company, trust, unincorporated organization, bank, business association, firm, joint venture, Governmental Authority or other legal entity.

"Plan" means an "employee benefit plan" (as defined in Section 3(3) of ERISA) that is or, within the preceding five years, has been established or maintained, or to which contributions are or, within the preceding five years, have been made or required to be made, by the Company or any ERISA Affiliate or with respect to which the Company or any ERISA Affiliate may have any liability.

"Preferred Stock" means any class of capital stock of a corporation that is preferred over any other class of capital stock of such corporation as to the payment of dividends or the payment of any amount upon liquidation or dissolution of such corporation.

"Priority Debt" means (without duplication) the sum of (a) unsecured Indebtedness of the Company's Restricted Subsidiaries other than (i) Indebtedness of the Company's Restricted Subsidiaries, including Holdings, listed on Schedule 5.15 and any extension, renewal or refinancing by such parties of such Indebtedness, provided that the terms of any such extension, renewal or refinancing are substantially the same as the terms and conditions in effect on the Closing Date, or more favorable to such parties, and the principal amount of such Indebtedness is not increased, (ii) Indebtedness owing to the Company or any other Restricted Subsidiary and (iii) any unsecured Indebtedness of Holdings consisting of Guarantees of the Company's Indebtedness existing as of the Closing Date and, so long as the Holdings Guaranty remains in full force and effect, Indebtedness consisting of Guarantees of the Company's Indebtedness incurred thereafter, plus (b) Indebtedness with respect to drawn landfill closure bonds, plus (c) Indebtedness with respect to Permitted Receivables Transactions, plus (d) Indebtedness of the Company or any of its Restricted Subsidiaries secured by a Lien not permitted by clauses (a) through (l) of Section 7.03.

"Pro Rata Share" means, with respect to each Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the Credit-Linked Deposit of such Lender at such time and the denominator of which is the amount of the Total Credit-Linked Deposit; provided that if the obligations of the Issuer to make Credit Extensions have been terminated pursuant to Section 8.02(a), the numerator of such fraction shall be such Lender's Outstanding Amount of all Credit Obligations at such time, and the denominator shall be the Outstanding Amount of all Credit Obligations of all Lenders at such time (with the aggregate amount of any Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Lender for purposes of this definition). The initial Pro Rata Share of each Lender is set forth opposite the name of such Lender on Schedule

2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"property" or "properties" means, unless otherwise specifically limited, real or personal property of any kind, tangible or intangible, existing or inchoate.

"PTE" has the meaning specified in Section 10.10(b)(i).

"QPAM Exemption" has the meaning specified in Section 10.10(b)(iv).

"Register" has the meaning specified in Section 10.07(c).

"Reimbursement Date" has the meaning specified in Section 2.01(c)(i).

"Replacement Lender" has the meaning specified in Section 3.07.

"Request for Credit Extension" means a request for the issuance or amendment of a Credit by the Issuer pursuant to a Request for Credit Extension substantially in the form of Exhibit A, with such changes thereto as the Administrative Agent or the Issuer may from time to time approve.

"Request for Term Loans" means a request for the making of Term Loans from the proceeds of Credit-Linked Deposits, substantially in the form of Exhibit E.

"Required Credit-Linked Lenders" means, Credit-Linked Lenders (other than the Company and its Affiliates) having Combined Pro Rata Shares aggregating more than 50%. For purposes hereof, the "Combined Pro Rata Share" of a Credit-Linked Lender means, with respect to such Credit-Linked Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the sum of (A) the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under this Agreement at such time plus (B) the sum of the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under the Seven Year Agreement at such time (as such terms are defined in the Seven Year Agreement), plus (C) the sum of the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under the Ten Year Agreement at such time (as such terms are defined in the Ten Year Agreement), and the denominator of which is the amount of the sum of (X) the Total Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations being deemed "held" by such time (with the aggregate amount of any Credit-Linked Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Credit-Linked Lenders under the Seven Year Agreement at such time (as such terms are defined in the Seven Year Agreement) (with the aggregate amount of any Credit-Linked Lender's risk participation and funded participation being deemed "held" by such Credit-Linked Lender for purposes of this definition) plus (Z) the Total Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of all Credit-Linked Lenders under the Ten Year Agreement at such time (as such terms are defined in the Ten Year Agreement) (with the

aggregate amount of any Credit-Linked Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Credit-Linked Lender for purposes of this definition).

"Required Lenders" means, as of any date of determination, Lenders (other than the Company and its Affiliates) having Pro Rata Shares aggregating more than 50%.

"Required Notice" has the meaning specified in Section 10.19(b).

"Responsible Officer" means any Senior Financial Officer and any other officer of a Credit Party with responsibility for the administration of the relevant portion of this Agreement. Any document delivered hereunder that is signed by a Responsible Officer of a Credit Party shall be conclusively presumed to have been authorized by all necessary corporate action on the part of such Credit Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Credit Party.

"Restricted Subsidiary" means any Subsidiary of the Company which is not designated as an Unrestricted Subsidiary in accordance with Section 7.07.

"S&P" means Standard & Poor's Rating Group, a division of The McGraw-Hill Companies, or any successor thereto that is a nationally-recognized rating agency.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Securities Account" has the meaning specified in Section 10.19(b)(ii).

"Securities Act" means the Securities Act of 1933.

"Securities Exchange Act" means Securities Exchange Act of 1934.

"Securities Intermediary" has the meaning specified in Section 10.19(b)(ii).

"Security" has the meaning specified in Section 2(1) of the Securities Act.

"Senior Financial Officer" means the chief financial officer, principal accounting officer, treasurer or controller of a Credit Party.

"Senior Responsible Officer" means any Senior Financial Officer, or any of the President, any Executive Vice President, any Senior Vice President or any Principal Vice President of a Credit Party.

"Seven Year Agreement" means the Seven Year Letter of Credit and Term Loan Agreement dated as of even date herewith (as amended, restated, supplemented or otherwise modified from time to time) among the Company, Holdings, each lender party thereto and Bank of America as administrative agent and issuer of letters of credit; provided, however, if all Credit Obligations (as such term is defined in the Seven Year Agreement) are paid in full and the Seven Year Agreement is terminated, then this definition of "Seven Year Agreement" shall no longer

be applicable to this Agreement.

"Source" has the meaning specified in Section 10.10(b).

"Subsidiary" means, as to any Person, any corporation, association or other business entity in which such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries owns sufficient equity or voting interests to enable it or them (as a group) ordinarily, in the absence of contingencies, to elect a majority of the directors (or Persons performing similar functions) of such entity, and any partnership or joint venture if more than a 50% interest in the profits or capital thereof is owned by such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries (unless such partnership or joint venture can and does ordinarily take major business actions without the prior approval of such Person or one or more of its Subsidiaries). Unless the context otherwise clearly requires, any reference to a "Subsidiary" is a reference to a Subsidiary of the Company.

"Taxes" has the meaning specified in Section 3.01(a).

"Ten Year Agreement" means the Ten Year Letter of Credit and Term Loan Agreement dated as of even date herewith (as amended, restated, supplemented or otherwise modified from time to time) among the Company, Holdings, each lender party thereto and Bank of America as administrative agent and issuer of letters of credit; provided, however, if all Credit Obligations (as such term is defined in the Ten Year Agreement) are paid in full and the Ten Year Agreement is terminated, then this definition of "Ten Year Agreement" shall no longer be applicable to this Agreement.

"Term Loan Issuance Date" has the meaning specified in Section 2.03(b)(iii).

"Term Loans" has the meaning specified in Section 2.03(b)(i).

"Term Note" has the meaning specified in Section 2.07(b).

"Total Credit-Linked Deposit" shall mean, at any time, the sum of all Lenders' Credit-Linked Deposits, as the same may be reduced from time to time.

"Trust" has the meaning specified in Section 10.19(b)(ii).

"United States" and "U.S." each means the United States of America.

"Unreimbursed Amount" has the meaning specified in Section 2.01(c)(i).

"Unrestricted Subsidiary" means any Subsidiary which is designated as such in writing by the Company to each Lender pursuant to Section 7.07.

"Unused Amount" has the meaning specified in Section 2.03(b)(i).

1.02 Other Interpretive Provisions. With reference to this Agreement and each other Credit Document, unless otherwise specified herein or in such other Credit Document:

- (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (b) (i) The words "herein," "hereof," and "hereunder" and words of similar import when used in any Credit Document shall refer to such Credit Document as a whole and not to any particular provision thereof.
 - (i) Article, Section, Exhibit and Schedule references are to the Credit Document in which such references appear.
 - (ii) The term "including" is by way of example and not limitation.
- (iii) The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.
- (c) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including."
- (d) Section headings herein and in the other Credit Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Credit Document.
- **1.03 Accounting Terms.** (a) All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Financial Statements, except as otherwise specifically prescribed herein.
- (b) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Credit Document, and either the Company or the Required Credit-Linked Lenders shall so request, the Administrative Agent, the Lenders and the Company shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Credit-Linked Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Company shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- **1.04 Rounding**. Any financial ratios required to be maintained by the Company pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such

ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

1.05 References to Agreements and Laws. Unless otherwise expressly provided herein, (a) references to agreements (including the Credit Documents) and other contractual instruments shall be deemed to include all subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Credit Document; and (b) references to any Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Law.

1.06 Credit Amounts. Unless otherwise specified, all references herein to the amount of a Credit at any time shall be deemed to mean the maximum face amount of such Credit after giving effect to all increases thereof contemplated by such Credit or the Issuer Documents related thereto, whether or not such maximum face amount is in effect at such time.

ARTICLE II THE CREDIT-LINKED DEPOSITS AND CREDIT EXTENSIONS

2.01 Credits.

- (a) The Credits.
- (i) Subject to the terms and conditions set forth herein, (A) the Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.01, from time to time on any Business Day during the period from the Closing Date until the Credit Expiration Date, to issue Credits denominated in Dollars for the account of the Company or any Subsidiary, and (B) the Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.01, (1) to amend, extend, reinstate or renew Credits previously issued by it, in accordance with subsection (b) below, and (2) to honor drawings under the Credits; and (C) the Lenders severally agree to participate in Credits issued or outstanding hereunder for the account of the Company or any Subsidiary as provided in Section 2.01(c), and all reimbursement obligations and rights hereunder in respect thereof and the Issuer Documents with respect thereto and the Issuer hereby grants to the Lenders such participation interests in such Credits and all reimbursement obligations and rights hereunder in respect thereof and such related Issuer Documents; provided that the Company shall not request the issuance of any Credit, the Issuer shall not make any Credit Extension with respect to any Credit, and no Lender shall participate in any Credit, to the extent that, as of the date of such Credit Extension:
 - (A) the sum, without duplication, of the applicable Outstanding Amount of all (i) Credits and (ii) Unreimbursed Amounts for which Advances have not yet been made would exceed the Total Credit-Linked Deposit;
 - (B) any Lender's Pro Rata Share of the sum, without duplication, of the applicable Outstanding Amount of all (i) Credits and (ii) Unreimbursed Amounts for which Advances have not yet been made would exceed such Lender's Credit-Linked Deposit; or

(C) the expiry date of any such requested Credit would occur after the Credit Expiration Date (or, in the case of an Auto-Renewal Credit, no Nonrenewal Notice Date would occur on or prior to the Credit Expiration Date);

and provided further that in determining the availability hereunder with respect to any Escalating Credits issued or outstanding hereunder, the Total Credit-Linked Deposit will be deemed to be utilized in respect of such Escalating Credits in the aggregate amount equal to the maximum aggregate amount available to be drawn under all such Escalating Credits (after giving effect to all increases).

Within the foregoing limits (the "Issuance Limits"), and subject to the terms and conditions hereof, the Company's ability to obtain Credits shall be fully revolving, and accordingly the Company may, during the foregoing period, obtain Credits to replace Credits that have expired, been terminated or cancelled, or that have been drawn upon and reimbursed. All Existing Credits up to the respective amounts specified in Schedule 1.01 shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof. Notwithstanding anything to the contrary set forth herein, if any portion of the Total Credit-Linked Deposit is utilized to reimburse the Issuer for any amounts drawn under any Credit (as contemplated by Section 2.01(c) hereof) or to make Term Loans (as contemplated by Section 2.03(b) hereof), the Total Credit-Linked Deposit shall be permanently reduced by the corresponding amount utilized to reimburse the Issuer or to make Term Loans and shall not be reinstated.

- (ii) The Issuer shall not be under any obligation to issue any Credit if:
- (A) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain the Issuer from issuing such Credit, or any Law applicable to the Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over the Issuer shall prohibit, or request that the Issuer refrain from, the issuance of letters of credit generally or such Credit in particular or shall impose upon the Issuer with respect to such Credit any restriction, reserve or capital requirement (in each case, for which the Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon the Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which the Issuer in good faith deems material to it;
 - (B) the Administrative Agent or the Issuer reasonably determines that such Credit is not a Permitted Credit;
 - (C) the issuance of such Credit would violate any Laws or any policies of the Issuer; or
- (D) except as otherwise agreed by the Administrative Agent and the Issuer, such Credit is in a face amount less than \$1,000,000, or is to be denominated in a currency other than Dollars.

- (iii) The Issuer shall not be under any obligation to amend any Credit if (A) the Issuer would have no obligation at such time to issue such Credit in its amended form under the terms hereof, or (B) the beneficiary of such Credit does not accept the proposed amendment to such Credit.
- (iv) The Issuer shall not issue or amend any Credit if the Issuer has received written notice from any Lender, the Administrative Agent or any Credit Party, on or prior to the Business Day prior to the requested date of issuance or amendment of such Credit, that one or more applicable conditions contained in Article IV shall not then be satisfied.
 - (b) Procedures for Issuance and Amendment of Credits; Auto-Renewal Credits.
- (i) Each Credit shall be issued or amended, as the case may be, upon the request of the Company delivered to the Issuer (with a copy concurrently delivered to the Administrative Agent) in the form of a Request for Credit Extension, appropriately completed and signed by a Responsible Officer or Authorized Representative of the Company. Any such Request for Credit Extension must be received by the Issuer and the Administrative Agent not later than 1:00 p.m., Dallas time, at least five Business Days prior to the proposed issuance date or date of amendment, as the case may be; or such other date or time as the Issuer and the Administrative Agent may agree in a particular instance in their sole discretion. In the case of a request for an initial issuance of a Credit, such request shall be in the form of a Request for Credit Extension and shall specify in form and detail satisfactory to the Issuer and the Administrative Agent: (A) the proposed issuance date of the requested Credit (which shall be a Business Day); (B) the face amount thereof; (C) the expiry date thereof; (D) the purpose and nature of the Credit; (E) the Person whose obligations are supported thereby (in the case of a Credit supporting the obligations of a Subsidiary); (F) the name and address of the beneficiary thereof; (G) any required text to be contained in the Credit; (H) the delivery instructions with respect to the Credit; and (J) whether such Credit will be an Escalating Credit, and if so, the maximum face amount of such Credit after giving effect to all increases. In the case of a request for an amendment of any outstanding Credit, such request shall be in the form of a Request for Credit Extension and shall specify in form and detail satisfactory to the Issuer and the Administrative Agent: (1) the Credit to be amended; (2) the proposed date of amendment thereof (which shall be a Business Day); (3) the nature of the proposed amendment; and (4) delivery instructions with respect to the amendment. Additionally, the Company shall furnish to the Issuer and the Administrative Agent such other documents and information pertaining to such requested Credit issuance or amendment, including any Issuer Documents as the Issuer or the Administrative Agent may reasonably require. Without limiting the generality of the foregoing, the delivery of a Request for Credit Extension is additional to, and not in replacement or substitution of, any other Issuer Document that the Issuer may require. Upon the effectiveness of any issuance or amendment of a Credit that will constitute a Credit hereunder, the Administrative Agent and the Lenders shall be entitled to assume that the Issuer has obtained such Issuer Documents as it shall have requested, executed by the relevant parties thereto to the extent required hereby.
- (ii) Promptly after receipt of any Request for Credit Extension by the Issuer, the Issuer will confirm with the Administrative Agent (by telephone or in writing) that the

Administrative Agent has received a copy of such Request for Credit Extension from the Company and, if not, the Issuer will provide the Administrative Agent with a copy thereof. Upon receipt by the Issuer and the Company of confirmation from the Administrative Agent that the requested issuance or amendment is permitted in accordance with the terms hereof, then, subject to the terms and conditions hereof, the Issuer shall, on the requested date, issue a Credit for the account of the Company or the applicable Subsidiary, or enter into the applicable amendment, as the case may be, in each case in accordance with the Issuer's usual and customary business practices.

(iii) If the Company so requests in any applicable Request for Credit Extension, the Issuer agrees, subject to the terms and conditions hereof, to issue a Credit that has automatic renewal provisions (each, an "Auto-Renewal Credit"); provided that any such Auto-Renewal Credit must permit the Issuer to prevent any such renewal at least once during the term thereof (commencing with the date of issuance of such Credit) by giving prior notice to the beneficiary thereof not later than a specified date to be agreed upon at the time such Credit is issued, which shall occur and be effective prior to the Credit Expiration Date (the "Nonrenewal Notice Date"). Unless otherwise directed by the Issuer, the Company shall not be required to make a specific request to the Issuer for any renewal of an Auto-Renewal Credit, but the Issuer shall be required to provide prior notice to the Administrative Agent of any pending renewal of an Auto-Renewal Credit at least ten Business Days before the applicable Nonrenewal Notice Date. Once an Auto-Renewal Credit has been issued (or is permitted to be outstanding hereunder in the case of an Existing Credit that is an Auto-Renewal Credit), the Lenders shall be deemed to have authorized (but may not require) the Issuer to permit the renewal of such Credit at any time to a date not later than the Credit Expiration Date; provided, however, that the Issuer shall not permit any renewal of an Auto-Renewal Credit if (A) the Issuer has determined that it would have no obligation at such time to issue such Credit in its renewed form under the terms hereof (by reason of the provisions of Section 2.01(a)(ii), Section 4.02 or otherwise), (B) after giving effect to any such renewal, the earlier of the (x) expiry date of such Auto-Renewal Credit and (y) next occurring Nonrenewal Notice Date of such Auto-Renewal Credit would occur after the Credit Expiration Date, or (C) it has received notice (which may be by telephone or in writing) on or before the day that is two Business Days before the Nonrenewal N

(iv) If the Issuer shall issue any Credit or amend any Credit (if the effect thereof is to increase the face amount), without obtaining prior consent from the Administrative Agent (as provided in clause (ii) above), or if the Issuer shall permit the extension or renewal of an Auto-Renewal Credit without giving timely prior notice to the Administrative Agent or when such extension or renewal is not permitted hereunder (as provided in clause (iii) above), or if the Issuer shall permit the reinstatement of an Auto-Reinstating Credit when such reinstatement is not permitted hereunder (as provided in clause (vi) below), such Credit (in the case of any such amendment, to the extent of the increased face amount thereof, and in the case of any reinstatement, to the extent of the reinstated amount thereof) (A) shall for all purposes be deemed to have been issued by the Issuer solely for its own account and risk, and (B) shall not be considered a Credit outstanding under this Agreement, and no Lender shall be deemed to have

any participation therein, effective as of the date of such issuance, amendment, reinstatement, extension or renewal, as the case may be, unless the Required Lenders expressly consent thereto.

- (v) Promptly after its delivery of any Credit or any amendment to a Credit to an advising bank with respect thereto or to the beneficiary thereof, the Issuer will also deliver to the Company and the Administrative Agent a true and complete copy of such Credit or amendment.
- (vi) If any Credit contains provisions providing for automatic reinstatement of the stated amount after any drawing thereunder, the Administrative Agent and the Lenders hereby authorize and direct the Issuer to permit such automatic reinstatement, whether or not a Default or Event of Default then exists, unless the Issuer has received a notice (which may be by telephone or in writing) on or before the day that is two Business Days before the reinstatement date from the Administrative Agent or the Required Lenders that one or more applicable conditions specified in Sections 2.01(a)(ii) or 4.02 is not then satisfied and directing it to cease permitting such automatic reinstatements.
 - (c) Drawings and Reimbursements; Credit-Linked Deposits.
- (i) Upon receipt from the beneficiary of any Credit of any notice of a drawing under such Credit, the Issuer of such Credit shall promptly notify the Company thereof. If the Company obtains written notice (confirmed by telephone) from the Issuer of a drawing under a Credit prior to 1:00 p.m., Dallas time, on any Business Day, the Company shall reimburse the Issuer on such Business Day. If the Company obtains written notice (confirmed by telephone) from the Issuer of a drawing under a Credit at or after 1:00 p.m., Dallas time, on any Business Day, the Company shall reimburse the Issuer on the Business Day immediately following the Business Day upon which such notice was received by the Company. Such reimbursement shall be made directly to the Issuer in an amount equal to the unreimbursed drawing (the "Unreimbursed Amount"). Without affecting the Company's obligations to so reimburse the Issuer on any such Business Day, which shall be absolute and unconditional, if the Company fails to so reimburse the Issuer by such time referenced above on the applicable due date for reimbursement specified above (the "Reimbursement Date"), the Issuer shall so notify the Administrative Agent (with a copy to the Company), which notice shall be provided on a Business Day, and specify in such notice the amount of the Unreimbursed Amount. Immediately upon receipt of such notice from the Issuer, the Administrative Agent shall promptly notify each Lender of the Reimbursement Date, the Unreimbursed Amount, the amount of such Lender's Pro Rata Share thereof, and shall state that Advances will be funded by application of such Lender's Pro Rata Share of the Credit-Linked Deposits in the Credit-Linked Deposit Account on the sixth Business Day after such Reimbursement Date to the extent the Unreimbursed Amount (or any portion thereof) remains outstanding on such day.
- (ii) The Issuer hereby irrevocably grants to each Lender, and each Lender hereby irrevocably accepts and purchases from the Issuer on the terms and conditions hereinafter stated, for such Lender's own account and risk an undivided interest equal to its Pro Rata Share in the Issuer's obligations and rights with respect to the Credits issued pursuant to this Agreement (as to each Lender, its "Participation"), which purchase price when paid to the Administrative Agent shall not be a deposit obligation of the Administrative Agent or Bank of

America. The consideration for the Participation of each Lender shall consist of the payment by such Lender to the Administrative Agent of an amount equal to the Dollar amount set forth opposite such Lender's name in Schedule 2.01 under the heading "Credit-Linked Deposit" (as the same may be reduced from time to time in accordance with this Agreement, such Lender's "Credit-Linked Deposit"), subject to the conditions precedent set forth in Section 4.01 hereof. Each Lender shall pay to the Administrative Agent its Credit-Linked Deposit in full on the Closing Date. The Credit-Linked Deposits of all Lenders shall be held by, and in the name of, the Administrative Agent in the Credit-Linked Deposit Account under the sole dominion and control of the Administrative Agent, Each Lender unconditionally and irrevocably agrees with the Administrative Agent and the Issuer that, if a drawing is paid under any Credit for which the Issuer is not reimbursed in full by the Company in cash within five Business Days after the applicable Reimbursement Date or converted into a Term Loan under Section 2.03(b), such Lender hereby authorizes the Administrative Agent to reimburse to the Issuer the Unreimbursed Amount (or the outstanding portion thereof) related to such drawing on or after the sixth Business Day after such Reimbursement Date, to the extent of such Lender's Pro Rata Share of the Unreimbursed Amount, solely from such Lender's Pro Rata Share of the Credit-Linked Deposits in the Credit-Linked Deposit Account, and each Lender hereby irrevocably authorizes the Administrative Agent to charge the Credit-Linked Deposit Account for such purpose, in satisfaction of such Lender's reimbursement obligation arising with respect to such drawing hereunder. Without limiting the generality of Section 9.01, in charging the Credit-Linked Deposit Account or otherwise exercising any rights of set-off with respect thereto, the Administrative Agent acts as the agent of the Issuer. The amount of each Lender's Pro Rata Share of such Unreimbursed Amount (or portion thereof) which is paid the Issuer as set forth above shall be deemed to be an Advance by such Lender to the Company hereunder. The outstanding principal amount of each Advance, together with interest thereon as provided in Section 2.04, shall be due and payable, in Dollars, on demand. Each Lender shall be subrogated to the rights and remedies of the Issuer against the Company and any Subsidiary liable under such Credit to the extent such Lender has reimbursed the Issuer as set forth in this Section 2.01(c)(ii). The Issuer shall reasonably cooperate in exercising and enforcing such rights and remedies as may be requested by the Required Lenders (and such cooperation shall be subject to any applicable indemnification set forth in Section 9.07 or Section 10.05 of this Agreement). The Issuer shall have no right to withdraw or set-off against monies on deposit in the Credit-Linked Deposit Account other than as set forth in this Section 2.01(c)(ii).

(iii) The Credit-Linked Deposits shall be held by the Administrative Agent in its name in the Credit-Linked Deposit Account and no Person other than the Administrative Agent shall have a right of withdrawal from the Credit-Linked Deposit Account nor any other right or power with respect to the Credit-Linked Deposits or the Credit-Linked Deposit Account. Notwithstanding anything in this Agreement to the contrary, the sole funding obligation of each Lender in respect of its Participation shall be satisfied upon funding of its Credit-Linked Deposit.

(iv) Until an Advance is funded pursuant to this Section 2.01(c) to reimburse the Issuer for the Unreimbursed Amount in respect of any Credit, or such Unreimbursed Amount is converted to a Term Loan pursuant to Section 2.03, interest in respect of each Lender's Pro Rata Share of the Borrowing in the amount of such Unreimbursed Amount shall be solely for the account of the Issuer.

- (v) Except as expressly provided herein, each Lender's agreement to fund Advances, by application of such Lender's Pro Rata Share of the Credit-Linked Deposits, to reimburse the Issuer for amounts drawn under Credits issued by the Issuer, as contemplated by this Section 2.01(c), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against the Issuer, the Company, any Subsidiary or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or Event of Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing.
- (vi) Notwithstanding that a Credit issued or outstanding hereunder is in support of any obligations of a Subsidiary, the Company shall be obligated to reimburse the Issuer hereunder for the full amount of any and all drawings under such Credit.
- (vii) In the event of any voluntary payment or prepayment (other than through the making of a Term Loan as provided for in Section 2.03(b)) of an Advance at any time prior to the second anniversary of the Closing Date, the Company shall pay on the date of such prepayment to the Administrative Agent, for the account of the Lenders in accordance with their Pro Rata Shares, a premium in an amount equal to three percent (3.00%) of the amount so prepaid.
 - (d) Repayment of Participations.
- (i) At any time after the Issuer has made a payment under any Credit and has received from the Credit-Linked Deposit the proceeds of Advances by the Lenders in respect of such payment in accordance with Section 2.01(c), if the Administrative Agent receives for the account of the Issuer any payment in respect of the related Unreimbursed Amount or interest thereon (whether directly from the Company or otherwise, including proceeds of Cash Collateral applied thereto by the Administrative Agent), the Administrative Agent will distribute to each Lender its Pro Rata Share thereof (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's Advance was outstanding) in Dollars and in the same funds as those received by the Administrative Agent. If a Borrowing has occurred hereunder, the Company shall repay the Unreimbursed Amount and interest thereon as provided in Sections 2.01(c) and 2.04 and shall pay any prepayment premium required to be paid pursuant to Section 2.01(c)(vii). Notwithstanding the foregoing, if the Issuer shall have received from the Credit-Linked Deposit the proceeds of Advances by the Lenders and thereafter shall receive any direct payment from the Company in respect of a Borrowing, the Issuer shall immediately pay the amount received to the Administrative Agent for distribution to the Lenders in accordance with this Section 2.01(d).
- (ii) If any payment received by the Administrative Agent for the account of the Issuer pursuant to Section 2.01(d)(i) and distributed to the Lenders by the Administrative Agent is required to be returned under any of the circumstances described in Section 10.06, each Lender shall pay to the Administrative Agent for the account of the Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect.

- (e) Obligations Absolute. The obligation of the Company to reimburse the Issuer for each drawing under each Credit and to repay each Borrowing, shall be absolute, unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following:
 - (i) any lack of validity or enforceability of such Credit, this Agreement, or any other Credit Document;
- (ii) the existence of any claim, counterclaim, set-off, defense or other right that the Company or any Subsidiary may have at any time against any beneficiary or any transferee of such Credit (or any Person for whom any such beneficiary or any such transferee may be acting), the Issuer or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by such Credit or any agreement or instrument relating thereto, or any unrelated transaction;
- (iii) any draft, demand, certificate or other document presented under such Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under such Credit;
- (iv) any payment by the Issuer under such Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Credit; or any payment made by the Issuer under such Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Credit, including any arising in connection with any proceeding under any Debtor Relief Law; or
- (v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Company or any Subsidiary.

The Company shall promptly examine a copy of each Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Company's instructions or other irregularity, the Company will promptly notify the Issuer. The Company shall be conclusively deemed to have waived any such claim against the Issuer and its correspondents unless such notice is given as aforesaid.

(f) Role of Issuer. Each Lender and the Company agree that, in paying any drawing under a Credit, the Issuer shall not have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by the Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. None of the Issuer, any Issuer-Related Person nor any of the respective correspondents, participants or assignees of the Issuer shall be liable to any Lender for (i) any action taken or omitted in connection herewith at the request or with the approval of the Required Lenders or the Required Credit-Linked Lenders, as applicable; (ii) any action taken or omitted in the absence of gross negligence or willful misconduct; or (iii) the due execution, effectiveness, validity or enforceability of any document or instrument related to any

Credit or Request for Credit Extension. The Company hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Company's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuer, any Issuer-Related Person, nor any of the respective correspondents, participants or assignees of the Issuer, shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.01(e); provided, however, that anything in such clauses to the contrary notwithstanding, the Company may have a claim against the Issuer, and the Issuer may be liable to the Company, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Company which the Company proves were caused by the Issuer's willful misconduct or gross negligence or the Issuer's willful failure to pay under any Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Credit. In furtherance and not in limitation of the foregoing, the Issuer may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the Issuer shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

- (g) Applicability of ISP98. Unless otherwise expressly agreed by the Issuer and the Company when a Credit is issued (including any such agreement applicable to an Existing Credit), the rules of the ISP shall apply to each Credit.
- (h) Issuer Documents. In the event of any conflict between the terms hereof, the terms of any Request for Credit Extension and the terms of any Issuer Document (other than any Credit), the terms hereof shall control.
 - (i) Existing Credits. The Company and the Issuer certify that Schedule 1.01 accurately and completely sets forth the Existing Credits.
 - (j) Interest on Credit-Linked Deposit; Repayment of Credit-Linked Deposit.
- (i) The Administrative Agent hereby agrees to cause Bank of America to pay interest to the Administrative Agent, as holder of the Credit-Linked Deposit Account, and the Administrative Agent, as holder of the Credit-Linked Deposit Account, agrees to distribute to each Lender, from such interest payments received from Bank of America, such Lender's Pro Rata Share of interest on the outstanding amount of the Credit-Linked Deposits in the Credit-Linked Deposit Account at the rate per annum set forth on Schedule A annexed hereto. Such interest will be distributed to the Lenders by the Administrative Agent quarterly in arrears on the first Business Day following the end of each Interest Period relating to the Credit-Linked Deposits. The Administrative Agent shall compute all amounts due under this Section 2.01(j)(i) and shall notify the Company and such Lender of each such amount due.
- (ii) Subject to, and to the extent of, the Company's compliance with the cash-collateralization requirements set forth in Section 2.02, the Administrative Agent shall distribute the aggregate amount of any remaining Credit-Linked Deposits to the Lenders on the Maturity

Date to the extent of the amount credited to the Credit-Linked Deposit Account. The Credit-Linked Deposit Account is a deposit account in the name of the Administrative Agent under its sole dominion and control, and held by it with Bank of America subject to the terms and conditions of this Agreement. No Lender, whether upon the receivership or liquidation of such Lender or otherwise, shall be entitled to its Credit-Linked Deposit other than as expressly provided in this Agreement.

(iii) The Company shall have no right, title or interest in or to the Credit-Linked Deposits or the Credit-Linked Deposit Account and no obligations with respect thereto, it being acknowledged and agreed by the parties hereto that the making of the Credit-Linked Deposits by the Lenders, the deposit by the Administrative Agent of the Credit-Linked Deposits in the Credit-Linked Deposits of this Section 2.01(j) and the application of the Credit-Linked Deposits in the manner contemplated by Section 2.01(c) constitute agreements among the Administrative Agent, the Issuer and each Lender in respect of the funding obligations of each Lender in respect of its Participation in Credits, and do not constitute any loan or extension of credit to the Company.

2.02 Cash Collateral

- (a) Upon the request of the Administrative Agent, (i) if, as of the Credit Expiration Date, (A) any Credit for any reason remains outstanding and partially or wholly undrawn or (B) any amount remains available to be drawn under any Credit by reason of the operation of Rule 3.14 of the ISP, or (ii) upon the occurrence of the circumstances described in Section 8.02(c) requiring the Company to Cash Collateralize Credits, the Company shall within two Business Days after receipt of such request Cash Collateralize the then Outstanding Amount of all Credits (in an amount equal to the Outstanding Amount thereof). For purposes hereof, "Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Issuer and the Lenders, as collateral for the Credit Obligations, cash or deposit account balances ("Cash Collateral") pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent (which documents are hereby consented to by the Lenders). Derivatives of such term have corresponding meanings.
- (b) The Company hereby grants the Administrative Agent, for the benefit of the Issuer and the Lenders, a security interest in all such Cash Collateral. Cash Collateral shall be maintained in blocked, non-interest bearing deposit accounts at Bank of America and shall be subject to such Lien documentation as the Administrative Agent shall reasonably request. The Company hereby authorizes the filing by the Administrative Agent of any UCC-1 financing statements in connection therewith.

2.03 Reduction of Total Credit-Linked Deposits; Conversion to Funded Term Loans.

(a) Reduction. The Company may, on or after the second anniversary of the Closing Date, upon notice from the Company to the Administrative Agent, permanently reduce the available amount of the Total Credit-Linked Deposit to an amount not less than the sum of the then Outstanding Amount of all Credit Obligations (other than Advances and Term Loans); provided that (i) any such notice shall be received by the Administrative Agent not later than

1:00 p.m., Dallas time, ten Business Days prior to the date of termination or reduction, and (ii) any such partial reduction shall be in an aggregate amount of \$10,000,000 or any whole multiple of \$1,000,000 in excess thereof, or such lesser amount as would reduce the Total Credit-Linked Deposit to \$0. Such notice shall specify the proposed effective date of such reduction. The Administrative Agent will promptly notify the Lenders of any such notice of reduction of the Total Credit-Linked Deposit. Once reduced in accordance with this Section, the Total Credit-Linked Deposit may not be increased. Any reduction of the Total Credit-Linked Deposit shall be applied to each Lender according to its Pro Rata Share. All Facility Fees and Fronting Fees shall accrue until the effective date of any reduction of the Total Credit-Linked Deposit and shall be paid on the effective date of such reduction if the Total Credit-Linked Deposit is \$0 after giving effect to such reduction. Upon any such reduction, the Administrative Agent shall promptly remit to each Lender its Pro Rata Share of the amount of the reduction in the Total Credit-Linked Deposit.

(b) Conversion to Funded Term Loans.

(i) Subject to the terms and conditions set forth herein (including satisfaction of the applicable conditions specified in Sections 4.02 and 4.03), the Company may elect to (A) cause the Lenders to make, and each Lender agrees to make, funded loans ("Term Loans") to the Company in an aggregate principal amount up to the excess of (x) the Total Credit-Linked Deposit over (y) the Outstanding Amount of all Credit Obligations (other than Advances and Term Loans) (the amount of such excess being the "Unused Amount"), or (B) cause the Lenders to make, and each Lender agrees to make, one or more funded Term Loans in an aggregate principal amount up to the amount of a forthcoming drawing on any Credit, or (C) cause the Lenders to make, and each Lender agrees to make, one or more funded Term Loans in the aggregate amount of any Advances then outstanding by converting such Advances to Term Loans; in each case in accordance with the procedures set forth in this Section 2.03(b). Any Term Loans made under this Section 2.03(b) and subsequently repaid or prepaid may not be reborrowed.

(ii) The Company may irrevocably request the making of one or more Term Loans in an aggregate amount up to the Unused Amount or in an aggregate amount up to the anticipated drawing on a Credit from the proceeds of Credit-Linked Deposits, or in an aggregate amount up to the aggregate amount of Advances then outstanding by converting such Advances to Term Loans, as applicable, by delivering a Request for Term Loans therefor to the Administrative Agent, appropriately completed and signed by a Responsible Officer or Authorized Representative of the Company, not later than 10 Business Days prior to the proposed date of making of the Term Loans; provided that the Term Loans requested pursuant to Section 2.03(b)(i)(A) shall only be made on a day which is the last day of an Interest Period with respect to the Credit-Linked Deposits and the Company shall in any event pay the amounts, if any, payable under Section 3.04; and provided further that the minimum aggregate amount of Term Loans made on any date shall be (A) \$25,000,000, in the case of Term Loans requested pursuant to Section 2.03(b)(i)(A), and (B) \$10,000,000, in the case of Term Loans requested pursuant to Sections 2.03(b)(i)(B) and 2.03(b)(i)(C), or in each case of clauses (A) and (B) of this clause (ii), such lesser amount which, when added to the applicable Outstanding Amount of

all (x) Credits and (y) the Unreimbursed Amounts for which Advances have not been made would equal the Total Credit-Linked Deposit.

- (iii) Following receipt of a Request for Term Loans, the Administrative Agent shall promptly notify each Lender of its Pro Rata Share thereof. Upon satisfaction of the applicable conditions set forth in Sections 4.02 and 4.03, the Administrative Agent shall (A) in the case of Term Loans made pursuant to Section 2.03(b)(i)(A), debit the Credit-Linked Deposit Account in an amount equal to the aggregate amount of the proposed Term Loans (but not exceeding the Unused Amount) and shall make such funds available to the Company either by (i) crediting the account of the Company on the books of Bank of America with the amount of such funds or (ii) wire transferring such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Company, (B) in the case of Term Loans made pursuant to Section 2.03(b)(i)(B), debit the Credit-Linked Deposit Account in an amount equal to the aggregate amount of the proposed Term Loans and transfer such funds to the Issuer in reimbursement of the applicable drawing on a Credit, and (C) in the case of Term Loans made pursuant to Section 2.03(b)(i)(C), record in its books and records and on the Register the conversion of the applicable Advances into Term Loans, which credit or transfer or recordation, as applicable, shall be deemed to be the making of the Term Loans by the Lenders hereunder, and such date shall be the "Term Loan Issuance Date" (and the Administrative Agent shall forthwith deliver a notice to each Lender informing such Lender of the applicable Term Loan Issuance Date). Each Lender hereby irrevocably authorizes the Administrative Agent to charge the Credit-Linked Deposit Account for such purpose. No Lender shall have any obligation to make Term Loans to the Company other than from the proceeds of its Credit-Linked Deposits.
 - (iv) Interest Rate on Term Loans.
 - (A) Subject to the provisions of subsection (B) below, each Term Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the LIBO Rate for such Interest Period plus the Applicable Rate.
 - (B) If any amount of the Term Loans is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Law. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.
 - (C) Interest on each Term Loan shall be due and payable in arrears on the first Business Day following the end of each Interest Period applicable thereto, at such other times as may be specified herein and on the Maturity Date. To the fullest extent permitted by applicable Law, interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.

- (v) Repayment of Term Loans. The Company shall repay to the Administrative Agent for the account of the Lenders on the Maturity Date the aggregate principal amount of the Term Loans outstanding on such date.
- (vi) Optional Prepayment of Term Loans. The Company may, on or after the second anniversary of the Closing Date, at its option, upon notice to the Administrative Agent as provided below, prepay on any Business Day all, or from time to time any part of, the Term Loans (subject to the limitations sets forth below); provided that (i) any such notice shall be received by the Administrative Agent not later than 1:00 p.m., Dallas time, ten Business Days prior to the date of prepayment, and (ii) any such partial prepayment shall be in an aggregate amount of \$1,000,000 or any whole multiple of \$1,000,000 in excess thereof, or such lesser amount as would reduce the Term Loans to \$0. Such notice shall specify the proposed effective date of such prepayment, the aggregate principal amount of the Term Notes to be prepaid on such date, the interest to be paid on the prepayment date with respect to such principal amount being prepaid, which shall be the last day of the Interest Period therefor. The Administrative Agent will promptly notify the Lenders of any such notice of prepayment of the Term Loans. Once prepaid in accordance with this Section, such Term Loans may not be reborrowed. Upon any such prepayment, the Administrative Agent shall promptly remit to each Lender its Pro Rata Share of the amount of the Prepayment of the Term Loans.

2.04 Interest on Borrowings Other Than Term Loans.

- (a) Each Borrowing (other than the Term Loans) shall bear interest on the outstanding principal amount thereof (i) from and including the applicable Borrowing date to and excluding the date six Business Days thereafter at an interest rate per annum equal at all times to the Base Rate plus 1.00% per annum, and (ii) from and including the sixth Business Day after the applicable Borrowing date at an interest rate per annum equal at all times to the Default Rate.
- (b) If any other amount payable by the Company under any Credit Document is not paid when due (without regard to any applicable grace periods) whether at stated maturity, by acceleration or otherwise, each such amount shall thereafter bear interest at an interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Law.
 - (c) Accrued and unpaid interest on all such unpaid amounts (including interest on past due interest) shall be due and payable upon demand.
- (d) Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.

2.05 Fees and Other Charges.

(a) Facility Fee. The Company shall pay to the Administrative Agent for the account of each Lender in accordance with its Pro Rata Share, a fee (the "Facility Fee") equal to the Applicable Rate, expressed on a daily basis, times the actual daily amount of such Lender's Credit-Linked Deposit. The Facility Fee shall accrue effective as of the Closing Date and shall be calculated quarterly in arrears by the Administrative Agent. If there is any change in the

Applicable Rate during any quarter, the actual daily Credit-Linked Deposit amount shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect. The Facility Fee shall be due and payable quarterly in arrears on the first Business Day following the end of each Interest Period relating to the Credit-Linked Deposits, commencing with the first such date to occur after the Closing Date, and on the Maturity Date; provided that, in connection with any reduction of the Total Credit-Linked Deposit under Section 2.03, the accrued Facility Fee calculated for the period ending on such date shall also be paid on the date of such reduction, and the following quarterly payment shall be calculated on the basis of the period from such reduction date to such quarterly payment date. The Facility Fee shall accrue at all times, including at any time during which one or more of the conditions in Article IV is not met.

- (b) Fronting Fee. The Company shall pay directly to the Issuer for its own account, a fee (the "Fronting Fee") equal to the rate per annum set forth in the letter agreement dated May 8, 2003 (the "Fee Letter"), among the Company, BAS and the Administrative Agent, times the daily amount of the Total Credit-Linked Deposit, due and payable quarterly in arrears on the first Business Day immediately following the last Business Day of each March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the Maturity Date; provided that, in connection with any reduction of the Total Credit-Linked Deposit under Section 2.03, the accrued Fronting Fee calculated for the period ending on such date shall also be paid on the date of such reduction, and the following quarterly payment shall be calculated on the basis of the period from such reduction date to such quarterly payment date. At any time that an Event of Default exists, the Fronting Fee shall be increased by adding 2% to the amount thereof determined as provided in the Fee Letter. The Fronting Fee shall accrue at all times, including at any time during which one or more of the conditions in Article IV is not met.
- (c) Documentary and Processing Charges Payable to Issuer. The Company shall pay directly to the Issuer for its own account the customary and standard issuance, commission, presentation, amendment and other processing fees, and other standard costs and charges (if any), of the Issuer or any Affiliate of the Issuer, relating to its Credits as from time to time in effect and agreed upon between the Company and the Issuer. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.
- (d) Placement and Administrative Fees. The Company shall pay a placement fee to BAS, and shall pay an administrative fee to the Administrative Agent for the Administrative Agent's own account, in the amounts and at the times specified in the Fee Letter, and if a successor Administrative Agent is appointed pursuant to Section 9.09 hereof, the Company shall pay the administrative fees of such successor Administrative Agent.

2.06 Computation of Interest and Fees.

(a) All computations of interest when the Base Rate is determined by reference to Bank of America's "prime rate" shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. Computation of all other types of interest and all fees shall be calculated on the basis of a year of 360 days, and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day or 366-day year). Interest shall accrue on each Advance or Term Loan for the day on which the Advance or

Term Loan is made, and shall not accrue on an Advance or Term Loan, or any portion thereof, for the day on which the Advance, such Term Loan or such portion is paid, provided that any Advance or Term Loan that is repaid on the same day on which it is made shall bear interest for one day.

(b) The Administrative Agent shall promptly notify the Company and the Lenders of the LIBO Rate applicable to any Interest Period upon determination of such interest rate. The determination of LIBO Rate by the Administrative Agent shall be conclusive in the absence of manifest error. At any time that Advances or Term Loans accruing interest based on the Base Rate are outstanding, the Administrative Agent shall notify the Company and the Lenders of any change in Bank of America's prime rate used in determining the Base Rate promptly following the public announcement of such change.

2.07 Evidence of Debt.

- (a) The Credit-Linked Deposits, the Credit Extensions made by the Issuer, the participations of each Lender therein and any Advances and Term Loans made by each Lender shall be evidenced by one or more accounts or records maintained by the Issuer, such Lender and by the Administrative Agent in the ordinary course of business. The accounts or records maintained by the Issuer, the Administrative Agent and each Lender shall be conclusive absent manifest error of the amount of the Credit-Linked Deposits, the Credit Extensions made by the Issuer and any Advances and Term Loans made by each Lender and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of the Company hereunder to pay any amount owing with respect to the Credit Obligations. In the event of any conflict between the accounts and records maintained by the Issuer or any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error.
- (b) In addition to the accounts and records referred to in subsection (a), (i) the Credit-Linked Deposit of, and the obligation of the Company to pay Advances to, each Lender shall be evidenced by a Letter of Credit Linked Note in the form of Exhibit D attached hereto (a "Letter of Credit Linked Note"), executed by the Administrative Agent and by the Company, and (ii) the Term Loans of a Lender shall be evidenced by a Term Note in the form of Exhibit F attached hereto (a "Term Note"), executed by the Company.

2.08 Payments Generally.

(a) All payments to be made by the Company to any Lender, the Administrative Agent, the Issuer or any other Person hereunder shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by the Company hereunder to be made directly to the Issuer shall be made to the Issuer in accordance with its payment instructions in Dollars and in immediately available funds not later than the times and on the dates specified herein. Except as otherwise expressly provided herein and except with respect to such payments to be made directly to the Issuer, all payments by the Company hereunder shall be made to the Administrative Agent, for the account of the respective Lenders or the Issuer to which such payment is owed, at the Administrative

Agent's Office in Dollars and in immediately available funds not later than 1:00 p.m., Dallas time, on the date specified herein. The Administrative Agent will promptly distribute to the Issuer its applicable share as provided herein, or to each Lender its Pro Rata Share (or other applicable share as provided herein), of such payment in like funds as received by wire transfer to the Issuer in accordance with its payment instructions or to such Lender at its Lending Office. All payments received by the Administrative Agent after 1:00 p.m., Dallas time, or by the Issuer after the time specified herein or in any Issuer Document, shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.

- (b) If any payment to be made by the Company shall come due on a day other than a Business Day, or any other date specified hereunder would otherwise occur on a day other than a Business Day, then except as otherwise provided herein, such payment shall be made, or other date shall occur, on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.
- (c) The obligations of the Lenders hereunder to fund Participations in Credits are several and not joint. The failure of any Lender to fund its Participation on the Closing Date with its Credit-Linked Deposit shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so purchase its Participation.
- (d) Nothing herein shall be deemed to obligate any Lender to obtain the funds for the Credit-Linked Deposit in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for the Credit-Linked Deposit in any particular place or manner.
- 2.09 Sharing of Payments. If, other than as expressly provided elsewhere herein, any Lender shall obtain on account of the Term Loans or Advances made by it or participations in Credit Obligations held by it, any payment (whether voluntary, involuntary, through the exercise of any right of set-off, or otherwise) in excess of its Pro Rata Share (or other share contemplated hereunder) thereof, such Lender shall immediately (a) notify the Administrative Agent of such fact, and (b) purchase from the other Lenders such subparticipations in the Term Loans or Advances made by them or in the participations in Credit Obligations held by them as shall be necessary to cause such purchasing Lender to share the excess payment in respect thereof *pro rata* with each of them; provided, however, that if all or any portion of such excess payment is thereafter recovered from the purchasing Lender under any of the circumstances described in Section 10.06 (including pursuant to any settlement entered into by the purchasing Lender in its discretion), such purchase shall to that extent be rescinded and each other Lender shall repay to the purchasing Lender the purchase price paid therefor, together with an amount equal to such paying Lender's ratable share (according to the proportion of (i) the amount of such paying Lender's required repayment to (ii) the total amount so recovered from the purchasing Lender) of any interest or other amount paid or payable by the purchasing Lender in respect of the total amount so recovered, without further interest thereon. The Company agrees that any Lender so purchasing a subparticipation from another Lender may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of set-off) with respect to such subparticipation as fully as if such Lender were the direct creditor of the Company in the amount

of such subparticipation. The Administrative Agent will keep records (which shall be conclusive and binding in the absence of manifest error) of subparticipations purchased under this Section and will in each case notify the Lenders following any such purchases or repayments. Each Lender that purchases a subparticipation pursuant to this Section shall from and after such purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Credit Obligations purchased to the same extent as though the purchasing Lender were the original owner of the Credit Obligations purchased.

ARTICLE III TAXES AND YIELD PROTECTION

- 3.01 Taxes. (a) Subject to Section 10.15(a)(iii), any and all payments by the Company to or for the account of the Administrative Agent, the Issuer or any Lender under any Credit Document shall be made free and clear of and without deduction for any and all present or future taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and all liabilities with respect thereto, excluding, in the case of the Administrative Agent, the Issuer and each Lender, taxes imposed on or measured by its overall net income, and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the Laws of which the Administrative Agent, the Issuer or such Lender, as the case may be, is organized or maintains a lending office (all such non-excluded taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and liabilities being hereinafter referred to as "Taxes"). If the Company shall be required by any Laws to deduct any Taxes from or in respect of any sum payable under any Credit Document to the Administrative Agent, the Issuer or any Lender, (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section), the Administrative Agent, the Issuer or such Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) the Company shall make such deductions, (iii) the Company shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable Laws, and (iv) within 30 days after the date of such payment, the Company shall furnish to the Administrative Agent (which shall promptly forward the same to the Issuer or such Lender) the original or a certified copy of a receipt evidencing payment thereof.
- (b) In addition, the Company agrees to pay any and all present or future stamp, court or documentary taxes and any other excise or property taxes or charges or similar levies which arise from any payment made under any Credit Document or from the execution, delivery, performance, enforcement or registration of, or otherwise with respect to, any Credit Document (hereinafter referred to as "Other Taxes").
- (c) If the Company shall be required to deduct or pay any Taxes or Other Taxes from or in respect of any sum payable under any Credit Document to the Administrative Agent, the Issuer or any Lender, the Company shall also pay to the Administrative Agent or to such Lender, as the case may be, at the time interest is paid, such additional amount that the Administrative Agent, the Issuer or any Lender specifies as necessary to preserve the after-tax yield (after factoring in all taxes, including taxes imposed on or measured by net income) that the

Administrative Agent, the Issuer or any Lender would have received if such Taxes or Other Taxes had not been imposed.

- (d) The Company agrees to indemnify the Administrative Agent, the Issuer and each Lender for (i) the full amount of Taxes and Other Taxes (including any Taxes or Other Taxes imposed or asserted by any jurisdiction on amounts payable under this Section) paid by the Administrative Agent, the Issuer and such Lender, (ii) amounts payable under Section 3.01(c) and (iii) any liability (including additions to tax, penalties, interest and expenses) arising therefrom or with respect thereto, in each case whether or not such Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. Payment under this subsection (d) shall be made within 30 days after the date the Issuer, such Lender or the Administrative Agent makes a demand therefor.
- 3.02 Increased Cost and Reduced Return; Capital Adequacy or Risk-Based Capital. (a) If the Issuer or any Lender determines that as a result of the introduction of or any change in or in the interpretation of any Law, or the Issuer's or such Lender's compliance therewith, there shall be any increase in the cost to the Issuer or such Lender of issuing or participating in Credits or the funding or maintaining Term Loans or Advances, or a reduction in the amount received or receivable by the Issuer or such Lender in connection with the foregoing (excluding for purposes of this subsection (a) any such increased costs or reduction in amount resulting from (i) Taxes or Other Taxes (as to which Section 3.01 shall govern), or (ii) changes in the basis of taxation of overall net income or overall gross income by the United States or any foreign jurisdiction or any political subdivision of either thereof under the Laws of which the Issuer or such Lender is organized or has its Lending Office), then from time to time upon demand of the Issuer or such Lender (with a copy of such demand to the Administrative Agent), the Company shall pay to the Issuer or such Lender such additional amounts as will compensate the Issuer or such Lender for such increased cost or reduction; provided that such compensation will be limited to (A) the period commencing not more than 120 days prior to the date of such demand or (B) any longer period of retroactive effect of any such change or requirement for compliance if such demand is made 120 days or less after such change or requirement for compliance.
- (b) If the Issuer or any Lender determines that the introduction of any Law regarding capital adequacy or risk-based capital or any change therein or in the interpretation thereof, or compliance by the Issuer or such Lender (or its Lending Office) therewith, has the effect of reducing the rate of return on the capital of the Issuer or such Lender or any corporation controlling the Issuer or such Lender as a consequence of the Issuer's or such Lender's obligations (for avoidance of doubt not including outstanding Term Loans and Advances) hereunder (taking into consideration its policies with respect to capital adequacy or risk-based capital and the Issuer's or such Lender's desired return on capital), then from time to time upon demand of the Issuer or such Lender (with a copy of such demand to the Administrative Agent), the Company shall pay to the Issuer or such Lender such additional amounts as will compensate the Issuer or such Lender for such reduction; provided that such compensation will be limited to (A) the period commencing not more than 120 days prior to the date of such demand or (B) any longer period of retroactive effect of any such change or requirement for compliance if such demand is made 120 days or less after such change or requirement for compliance.

3.03 Illegality. If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender to make, maintain or fund LIBO Rate Term Loans, or to determine or charge interest rates based upon the LIBO Rate, then, on notice thereof by such Lender to the Company through the Administrative Agent, any obligation of such Lender to make or continue LIBO Rate Term Loans shall be suspended until such Lender notifies the Administrative Agent and the Company that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Company shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay the Term Loans of such Lender, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such LIBO Rate Term Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such LIBO Rate Term Loans. Upon any such prepayment, the Company shall also pay accrued interest on the amount so prepaid.

3.04 Funding Losses. Upon demand of any Lender (with a copy to the Administrative Agent), or demand of the Administrative Agent, from time to time, the Company shall promptly compensate such Lender or the Administrative Agent, as applicable, for and hold such Lender or the Administrative Agent harmless from any loss, cost or expense incurred by it as a result of:

- (a) any payment or prepayment of any LIBO Rate Term Loan or Advance on a day other than the last day of the Interest Period for such Term Loan or Advance (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);
 - (b) any reduction of the Total Credit-Linked Deposits on a day other than the last day of an Interest Period therefor;
 - (c) any failure by the Company to prepay or borrow any Term Loan on the date or in the amount notified by the Company; or
- (d) any assignment of a LIBO Rate Term Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Company pursuant to Section 3.07;

including any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Term Loan or from fees payable to terminate the deposits from which such funds were obtained; provided, however, that such Lender shall use commercially reasonable efforts to mitigate any such loss, cost or expense.

For purposes of calculating amounts payable by the Company to the Lenders under this Section 3.04, each Lender shall be deemed to have funded each LIBO Rate Term Loan made by it at the LIBO Rate used in determining the LIBO Rate for such Term Loan by a matching deposit or other borrowing in the London interbank LIBOR market for a comparable amount and for a comparable period, whether or not such LIBO Rate Term Loan was in fact so funded.

3.05 Matters Applicable to All Requests for Compensation. A certificate of the Administrative Agent, the Issuer or any Lender claiming compensation under this Article III and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. In determining such amount, the Administrative Agent, the Issuer or any Lender may use any reasonable averaging and attribution methods.

3.06 Survival. All of the Company's obligations under this Article III shall survive termination of the Credit-Linked Deposits and repayment of all Credit Obligations and all other amounts due hereunder for a period of one (1) year; provided, however, any and all of the Company's obligations under this Article III related to Taxes shall survive for a period of three (3) years.

3.07 Replacement of Lenders. On each occasion that a Lender either makes a demand for compensation pursuant to Section 3.02 in an amount in excess of the amount that the Company would have had to pay pursuant to such Section if such Lender's Credit-Linked Deposit or Term Loan was held by a Lender with respect to which no such amounts would then be payable pursuant to such Section, the Company may, upon at least ten Business Days' prior written notice to the Issuer, each such Lender and the Administrative Agent, in whole permanently replace the Credit-Linked Deposit or Term Loan of such Lender; provided that, the Company shall replace such Credit-Linked Deposit or Term Loan with the Credit-Linked Deposit or Term Loan of (a) any other Lender, or, if no other Lender shall agree to replace such Lender (within 30 days after being offered such option), (b) a substitute lending institution which is an Eligible Assignee (in each case, a "Replacement Lender"). Such Replacement Lender shall, upon the effective date of replacement, purchase the Credit-Linked Deposit or Term Loan of, and the Credit Obligations owed to, such replaced Lender for the aggregate amount thereof and shall thereupon for all purposes become a "Lender" hereunder. Such notice from the Company shall specify an effective date for the replacement of such Lender, which date shall not be earlier than the tenth day after the day such notice is given. On the effective date of any replacement of such Lender pursuant to this Section 3.07 and in accordance with the terms of Section 10.07, the Company shall pay to the Administrative Agent for the account of such Lender (i) any fees or other amounts due to such Lender to the date of such replacement, (ii) accrued interest on the principal amount of outstanding Borrowings and the Credit-Linked Deposit or Term Loan held by such Lender to the date of such replacement and (iii) any other amount or amounts payable to such Lender pursuant to this Section 3.07. Upon the effective date of replacement of any Lender pursuant to this Section 3.07, such Lender shall cease to be a "Lender" hereunder, without any other or further act or deed on the part of any other Lender. No such replacement of any such Lender, the purchase of such Lender's Borrowings, Credit-Linked Deposit or Term Loans pursuant to this Section 3.07 shall affect (x) any liability or obligation of the Company or any other Lender to such replaced Lender which accrued on or prior to the date of such termination or (y) such replaced Lender's rights hereunder in respect of any such liability or obligation.

ARTICLE IV CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

4.01 Conditions of Credit-Linked Deposits and Initial Credit Extension. The obligation of each Lender to fund its Credit-Linked Deposit and the obligation of the Issuer to make its initial Credit Extension hereunder are subject to satisfaction or waiver of the following conditions precedent:

(a) The Administrative Agent's and Lenders' special counsel's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless

otherwise specified, each properly executed by a Responsible Officer or Secretary or assistant Secretary, as applicable of the signing Credit Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance reasonably satisfactory to the Administrative Agent, the Issuer, the Lenders and their respective legal counsel:

- (i) executed counterparts of this Agreement, sufficient in number for distribution to the Administrative Agent, the Issuer, each Lender and the Credit Parties;
- (ii) certified copies of the certificate or articles of incorporation (including all amendments thereto) of each Credit Party, certified as of a recent date prior to the Closing Date by the Secretary of State of the respective states of incorporation of such Credit Party;
- (iii) copies of the bylaws (including all amendments thereto) of each Credit Party, certified as of the Closing Date by the Secretary or assistant Secretary of such Credit Party;
- (iv) incumbency certificates of the Secretary or assistant Secretary of each Credit Party evidencing the identity, authority and capacity of each Responsible Officer or Authorized Representative thereof authorized to act as a Responsible Officer or Authorized Representative in connection with this Agreement and the other Credit Documents to which such Credit Party is a party;
- (v) certificates of the Secretary of State or other appropriate official of the respective states of incorporation of each the Credit Party, certifying as of a recent date prior to the Closing Date that such Credit Party is duly organized or formed and validly existing and in good standing in the jurisdiction of its organization or formation as of such date, and certificates of the Secretary of State or other appropriate official of the State of Texas, certifying as of a recent date prior to the Closing Date that such Credit Party is in good standing and qualified to engage in business in the State of Texas;
- (vi) true and correct copies of the resolutions adopted by the respective boards of directors of each Credit Party, duly certified as of the Closing Date by the Secretary or assistant Secretary of such Credit Party, (i) approving the transactions contemplated by (and the form of) this Agreement and the other Credit Documents and all other documents to be entered into in connection with this Agreement to be delivered by such Credit Party hereunder and (ii) authorizing the officers of such Credit Party, and each of them, acting individually or together, to execute and deliver each such document with such changes as such officer or officers shall approve;
- (vii) a certificate signed by a Responsible Officer of the Company certifying that (a) the representations and warranties contained in Article V, or any other Credit Document are true and correct, (b) no Default or Event of Default has occurred and is continuing, and (c) since the date of the Audited Financial Statements, no event or circumstance that either individually or in the aggregate has resulted in or could reasonably be expected to result in a Material Adverse Effect;

- (viii) opinions of counsel to each Credit Party, addressed to the Administrative Agent, the Issuer and each Lender, as to the matters set forth in Exhibit C-1 and such other matters concerning the Credit Parties and the Credit Documents as the Issuer or the Required Credit-Linked Lenders may reasonably request;
- (ix) opinions of counsel to the Administrative Agent and the Issuer, addressed to the Administrative Agent, the Issuer and each Lender, as to the matters set forth in Exhibit C-2 and Exhibit C-3 and such other matters as the Required Credit-Linked Lenders may reasonably request; and
- (x) opinion of special counsel to the Lenders, addressed to each Lender, as to the matters set forth in Exhibit C-4 and such other matters as the Required Credit-Linked Lenders may reasonably request.
 - (b) Any fees required to be paid on or before the Closing Date shall have been paid.
- (c) The Company shall have paid all reasonable Attorney Costs of (i) the Administrative Agent, and (ii) the special counsel to the Lenders, in each case to the extent invoiced prior to or on the Closing Date.
 - (d) The Closing Date shall have occurred on or prior to June 30, 2003.
- (e) On the Closing Date, each Lender's purchase of its Participation shall (i) be permitted by the laws and regulations of each jurisdiction to which such Lender is subject, without recourse to provisions (such as Section 1405(a)(8) of the New York Insurance Law) permitting limited investments by insurance companies without restriction as to the character of the particular investment, (ii) not violate any applicable law or regulation (including, without limitation, Regulation T, U or X of the Board of Governors of the Federal Reserve System) and (iii) not subject such Lender to any tax, penalty or liability under or pursuant to any applicable law or regulation, which law or regulation was not in effect on the date hereof. If requested by each Lender, such Lender shall have received an Officer's Certificate certifying as to such matters of fact as such Lender may reasonably specify to enable such Lender to determine whether such purchase is so permitted. The wire transfer of funds on the Closing Date by each Lender in the amount of its Credit-Linked Deposit to the Administrative Agent shall constitute its confirmation as to satisfaction of this condition with respect to such Lender.
- (f) A Private Placement number issued by Standard & Poor's CUSIP Service Bureau (in cooperation with the Securities Valuation Office of the National Association of Insurance Commissioners) shall have been obtained for (i) the Letter of Credit Linked Notes and (ii) the Term Notes, which Private Placement number shall not be in the series of numbers issued to Bank of America Corporation and its affiliates, including but not limited to Bank of America.
 - (g) The Issuer shall have received evidence satisfactory to it that each Lender, in respect of its Participation hereunder, has funded its Credit-Linked Deposit.
 - (h) The Company and the Administrative Agent shall have executed and delivered to each Lender such Lender's Letter of Credit Linked Note.

- (i) The Administrative Agent and Bank of America in its capacity as depository shall have executed and delivered the Money Market Account Agreement and special counsel for the Lenders shall have received a copy thereof.
- **4.02 Conditions to all Credit Extensions by the Issuer.** The obligation of the Issuer to make any Credit Extension or honor any Request for Credit Extension is subject to the following conditions precedent:
- (a) The representations and warranties of the Company and each other Credit Party contained in Article V, or any other Credit Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct on and as of the date of such Credit Extension, except to the extent that such representations and warranties relate expressly and solely to an earlier date, and except to the extent of changes resulting from transactions contemplated or permitted by the Credit Documents and changes occurring in the ordinary course of business which either individually or in the aggregate do not result in a Material Adverse Effect.
 - (b) No Default or Event of Default shall exist, or would result from such proposed Credit Extension.
- (c) There has occurred since the date of the Audited Financial Statements, no event or circumstance that either individually or in the aggregate has resulted in or could reasonably be expected to result in a Material Adverse Effect.
- (d) Except in the case of any automatic renewal of any Auto-Renewal Credit or any automatic reinstatement of any Auto-Reinstating Credit, the Administrative Agent and the Issuer shall have received a Request for Credit Extension in accordance with the requirements hereof.
- (e) The Administrative Agent shall have received, in form and substance satisfactory to it, such other assurances, certificates, documents or consents related to the foregoing as the Administrative Agent, the Issuer or the Required Lenders reasonably may require.

Each Request for Credit Extension submitted by the Company shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a), (b) and (c) have been satisfied on and as of the date of the applicable Credit Extension.

- **4.03 Conditions to Term Loans.** The obligation of the Lenders to make any Term Loans is subject to the following conditions precedent:
- (a) The Administrative Agent shall have received in accordance with the provisions of Section 2.03(b) hereof an originally executed Request for Term Loans.
- (b) The Company, by a duly authorized Senior Responsible Officer, shall have executed and delivered to the Administrative Agent on the applicable Term Loan Issuance Date, for delivery to the Lenders, Term Notes to evidence such Term Loans, dated as of the Term Loan Issuance Date, in the aggregate principal amount of the Term Loans and with other appropriate insertions.

- (c) The Administrative Agent shall have received an opinion of in-house counsel of the Company, addressed to the Administrative Agent and each Lender, substantially in the form of Exhibit H attached hereto.
- (d) Conditions Satisfied. The Company shall have delivered to the Administrative Agent an Officer's Certificate, dated as of the applicable Term Loan Issuance Date, in form and substance satisfactory to the Administrative Agent, certifying that all conditions precedent set forth in Sections 4.02(a) and (b) and 4.03 have been satisfied on and as of the applicable Term Loan Issuance Date.

Each Request for Term Loans submitted by the Company shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a) and (b) have been satisfied on and as of the date of the applicable Term Loan Issuance Date.

ARTICLE V REPRESENTATIONS AND WARRANTIES

The Company represents and warrants to the Administrative Agent and the Lenders that:

5.01 Organization; Power and Authority.

- (a) The Company is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by Law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Company has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Credit Documents to which it is a party and to perform the provisions hereof and thereof.
- (b) Holdings is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by Law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a material adverse effect on Holdings and its Subsidiaries, taken as a whole. Holdings has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Credit Documents to which it is a party and to perform the provisions hereof and thereof.

5.02 Authorization, etc.

(a) This Agreement and each other Credit Document have been duly authorized by all necessary corporate action on the part of the Company, and this Agreement constitutes, and each other Credit Document to which the Company is a party when executed and delivered will constitute, a legal, valid and binding obligation of the Company enforceable against the

Company in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

- (b) This Agreement and each other Credit Document have been duly authorized by all necessary corporate action on the part of Holdings, and this Agreement constitutes, and each other Credit Document to which Holdings is a party when executed and delivered will constitute, a legal, valid and binding obligation of Holdings enforceable against Holdings in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (c) To the extent Credits are being issued for the account of Subsidiaries, at the time of each such Credit Extension, such Credits do and will inure to the benefit of the Company, and the Company's business derives and will derive substantial benefits from the businesses of such Subsidiaries.
- **5.03 Disclosure**. The Company, through its agent, BAS, has delivered to each Lender a copy of a Private Placement Memorandum, dated May 2003 (the "Memorandum"), relating to the transactions contemplated hereby. As of its date, the Memorandum fairly describes, in all material respects, the general nature of the business and principal properties of the Company and its Subsidiaries. As of the Closing Date, this Agreement, the Memorandum, the documents, certificates or other writings delivered to the Lenders by or on behalf of the Company in connection with the transactions contemplated hereby and the financial statements listed in Schedule 5.05, taken as a whole, do not contain any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein not misleading in light of the circumstances under which they were made. Since December 31, 2002, there has been no change in the financial condition, operations, business or properties of the Company or any Subsidiary except changes that individually or in the aggregate could not reasonably be expected to have a Material Adverse Effect. There is no fact known to the Company that could reasonably be expected to have a Material Adverse Effect that has not been disclosed herein or in the Memorandum, in the Disclosure Documents, or in the other documents, certificates and other writings delivered to the Lenders by or on behalf of the Company specifically for use in connection with the transactions contemplated hereby.

5.04 Organization and Ownership of Shares of Subsidiaries; Affiliates.

- (a) Schedule 5.04 contains (except as noted therein) complete and correct lists as of the Closing Date of (i) each of the Company's Restricted Subsidiaries, showing, as to each Restricted Subsidiary, the correct name thereof and the jurisdiction of its organization, (ii) each of the Company's Affiliates, and (iii) the Company's directors and Senior Responsible Officers. As of the Closing Date, there are no Unrestricted Subsidiaries.
- (b) As of the Closing Date, all of the outstanding shares of capital stock or similar equity interests of each Restricted Subsidiary owned by the Company and its Subsidiaries have

been validly issued, are fully paid and nonassessable and are owned by the Company or another Subsidiary free and clear of any Lien (except for any Permitted Lien or as otherwise disclosed in Schedule 5.04).

- (c) Each Subsidiary identified on Schedule 5.04 is a corporation or other legal entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, and is duly qualified as a foreign corporation or other legal entity and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each such Subsidiary has the corporate or other power and authority to own or hold under lease the properties it purports to own or hold under lease and to transact the business it transacts and proposes to transact.
- (d) As of the Closing Date, no Subsidiary is a party to, or otherwise subject to any legal restriction or any agreement (other than the agreements listed on Schedule 5.04 and customary limitations imposed by corporate law statutes) restricting the ability of such Subsidiary to pay dividends out of profits or make any other similar distributions of profits to the Company or any of its Subsidiaries that owns outstanding shares of capital stock or similar equity interests of such Subsidiary.

5.05 Financial Statements.

The Company has delivered to the Issuer and each Lender the financial statements of the Company and its Subsidiaries listed on Schedule 5.05. All of said financial statements (including in each case the related schedules and notes) fairly present in all material respects the consolidated financial position of the Company and its Subsidiaries as of the respective dates specified in such Schedule and the consolidated results of their operations and cash flows for the respective periods so specified and have been prepared in accordance with GAAP consistently applied throughout the periods involved except as set forth in the notes thereto (subject, in the case of any interim financial statements, to normal year-end adjustments).

5.06 Compliance with Laws, Other Instruments, etc.

(a) The execution, delivery and performance by the Company of this Agreement, the Letter of Credit Linked Notes and the Term Notes will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of the Company or any Subsidiary under, any indenture, mortgage, deed of trust, loan, note purchase or credit agreement, or lease, its corporate charter or by-laws, or any other agreement or instrument to which the Company or any Subsidiary is bound or by which the Company or any Subsidiary or any of their respective properties may be bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to the Company or any Subsidiary or (iii) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to the Company or any Subsidiary.

(b) The execution, delivery and performance by Holdings of this Agreement will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of Holdings under, any indenture, mortgage, deed of trust, loan, note purchase or credit agreement, lease, corporate charter or by-laws, or any agreement or instrument to which Holdings is bound or by which Holdings or any of its properties may be bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to Holdings, or (iii) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to Holdings.

5.07 Governmental Authorizations, etc.

- (a) No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by the Company of this Agreement, the Letter of Credit Linked Notes or the Term Notes.
- (b) No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by Holdings of this Agreement.

5.08 Litigation; Observance of Agreements, Statutes and Orders.

- (a) Except as disclosed in Schedule 5.08 or in the Disclosure Documents, there are no actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company or any Subsidiary or any property of the Company or any Subsidiary in any court or before any arbitrator of any kind or before or by any Governmental Authority that, individually or in the aggregate, could reasonably be expected, after offset for claims against insurers, to have a Material Adverse Effect.
- (b) Except as disclosed in Schedule 5.08 or in the Disclosure Documents, neither the Company nor any Subsidiary is in default under any term of any agreement or instrument to which it is a party or by which it is bound, or any order, judgment, decree or ruling of any court, arbitrator or Governmental Authority or is in violation of any applicable law, ordinance, rule or regulation (including without limitation Environmental Laws) of any Governmental Authority, which default or violation, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

5.09 Taxes. The Company and its Subsidiaries have filed all tax returns that are required to have been filed in any jurisdiction, and have paid all taxes shown to be due and payable on such returns and all other taxes and assessments levied upon them or their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent, except for any taxes and assessments (i) the amount of which is not individually or in the aggregate Material or (ii) the amount, applicability or validity of which is currently being contested in good faith by appropriate proceedings and with respect to which the Company or a Subsidiary, as the case may be, has established adequate reserves in accordance with GAAP. The Company knows of no basis for any other tax or

assessment that could reasonably be expected to have a Material Adverse Effect. The charges, accruals and reserves on the books of the Company and its Subsidiaries in respect of federal, state or other taxes for all fiscal periods are adequate in accordance with GAAP. The federal income tax liabilities of the Company and its Subsidiaries have been determined by the IRS and paid for all fiscal years up to and including the fiscal year ended December 31, 1987.

5.10 Title to Property; Leases. The Company and its Restricted Subsidiaries have good and sufficient title to their respective properties that the Company and the Subsidiaries own or purport to own that individually or in the aggregate are Material, including all such properties reflected in the most recent audited balance sheet referred to in Section 5.05 or purported to have been acquired by the Company or any Restricted Subsidiary after said date (except as sold or otherwise disposed of in the ordinary course of business), in each case free and clear of Liens prohibited by this Agreement. All leases that individually or in the aggregate are Material are valid and subsisting and are in full force and effect in all material respects.

5.11 Licenses, Permits, etc. Except as disclosed in Schedule 5.11,

- (a) the Company and its Restricted Subsidiaries own or possess all licenses, permits, franchises, authorizations, patents, copyrights, service marks, trademarks and trade names, or rights thereto, that individually or in the aggregate are Material, without known material conflict with the rights of others;
- (b) to the best knowledge of the Company, no product of the Company or any of its Restricted Subsidiaries infringes in any Material respect any license, permit, franchise, authorization, patent, copyright, service mark, trademark, trade name or other right owned by any other Person; and
- (c) to the best knowledge of the Company, there is no Material violation by any Person of any right of the Company or any of its Restricted Subsidiaries with respect to any patent, copyright, service mark, trademark, trade name or other right owned or used by the Company or any of its Restricted Subsidiaries.

5.12 Compliance with ERISA.

(a) The Company and each ERISA Affiliate have operated and administered each Plan in compliance with all applicable laws except for such instances of noncompliance as have not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any ERISA Affiliate has incurred any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans (as defined in Section 3 of ERISA), and no event, transaction or condition has occurred or exists that could reasonably be expected to result in the incurrence of any such liability by the Company or any ERISA Affiliate, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate, in either case pursuant to Title I or IV of ERISA or to such penalty or excise tax provisions or to Section 401(a)(29) or 412 of the Code, other than such liabilities or Liens as would not be individually or in the aggregate Material.

- (b) The present value of the aggregate accrued benefits under each of the Plans that are subject to Title IV of ERISA (other than Multiemployer Plans), determined as of the valuation date for the most recently ended plan year on the basis of the actuarial assumptions specified for funding purposes in such Plan's most recent actuarial valuation report, did not exceed the aggregate current value of the assets of such Plan allocable to such benefits (i) in the case of any single Plan, by an amount which, if said amount were immediately contributed to the Plans, could have a Material Adverse Effect, and (ii) in the case of all said Plans, by an amount which, if said amount were immediately contributed to the Plans, could have a Material Adverse Effect.
- (c) The Company and its ERISA Affiliates have not incurred withdrawal liabilities (and are not subject to contingent withdrawal liabilities) under Section 4201 or 4204 of ERISA in respect of Multiemployer Plans that individually or in the aggregate are Material.
- (d) The expected obligation for postretirement benefits other than pensions (determined as of the last day of the Company's most recently ended fiscal year in accordance with Financial Accounting Standards Board Statement No. 106, without regard to liabilities attributable to continuation coverage mandated by Section 4980B of the Code and after taking into account any governmental reimbursements) of the Company and its Subsidiaries is not Material.
- (e) The execution and delivery of this Agreement and the issuance and sale of the Letter of Credit Linked Notes and the Term Notes will not involve any transaction that is subject to the prohibitions of Section 406 of ERISA or in connection with which a tax could be imposed pursuant to Section 4975(c)(1)(A)-(D) of the Code. The representations by the Company in the first sentence of this Section 5.12(e) are made in reliance upon and subject to the accuracy of each Lender's representations in Section 10.10(b) as to the sources of the funds used to pay the purchase price of its Participation under this Agreement.
- **5.13 Private Offering by Company**. Neither the Company nor anyone acting on its behalf has offered the Letter of Credit Linked Notes or Term Notes or any similar Securities for sale to, or solicited any offer to buy any of the same from, or otherwise approached or negotiated in respect thereof with, any Person other than the Lenders and not more than seventeen (17) other Institutional Investors, each of which has been offered the Letter of Credit Linked Notes and Term Notes at a private sale for investment. Neither the Company nor anyone acting on its behalf has taken, or will take, any action that would subject the issuance or sale of the Letter of Credit Linked Notes or Term Notes to the registration requirements of Section 5 of the Securities Act.
- **5.14** Use of Proceeds; Margin Regulations. The Company will apply the proceeds from the Advances and the Term Loans hereunder for general corporate purposes of the Company and its Subsidiaries. No part of the proceeds from the Credit Extensions hereunder will be used, directly or indirectly, for the purpose of buying or carrying any margin stock within the meaning of Regulation U of the FRB (12 CFR 221), or for the purpose of buying or carrying or trading in any Securities under such circumstances as to involve the Company in a violation of Regulation X of said Board (12 CFR 224) or to involve any broker or dealer in a violation of Regulation T of the FRB (12 CFR 220). Margin stock does not constitute more than 5.0% of the

value of the consolidated assets of the Company and its Subsidiaries and the Company does not have any present intention that margin stock will constitute more than 5.0% of the value of such assets. As used in this Section, the terms "margin stock" and "purpose of buying or carrying" shall have the meanings assigned to them in said Regulation U.

- **5.15** Existing Indebtedness. Except as described therein, Schedule 5.15 sets forth a complete and correct list of all outstanding Indebtedness of the Company and its Restricted Subsidiaries as of March 31, 2003, specifying whether such Indebtedness is secured or unsecured. From March 31, 2003 to the Closing Date, there has been no material change in the amounts, interest rates, sinking funds, installment payments or maturities of such Indebtedness of the Company or its Restricted Subsidiaries. As of the Closing Date, neither the Company nor any Restricted Subsidiary is in default and no waiver of default is currently in effect, in the payment of any principal or interest on any such Indebtedness of the Company or such Restricted Subsidiary, and no event or condition exists with respect to any such Indebtedness of the Company or any Restricted Subsidiary, that would permit (or that with notice or the lapse of time, or both, would permit) one or more Persons to cause such Indebtedness to become due and payable before its stated maturity or before its regularly scheduled dates of payment.
- **5.16 Foreign Assets Control Regulations, etc.** Neither any Credit Extension to the Company hereunder nor its use of the proceeds thereof will violate the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto. Without limiting the foregoing, neither the Company nor any of its Subsidiaries or its Affiliates (a) is or will become a Person whose property or interests in property are blocked pursuant to Section 1 of Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)) or (b) engages or will engage in any dealings or transactions, or be otherwise associated, with any such Person. The Company and its Subsidiaries and its Affiliates are in compliance, in all Material respects, with the Uniting And Strengthening America By Providing Appropriate Tools Required To Intercept And Obstruct Terrorism (USA Patriot Act of 2001). No part of the proceeds from any Credit Extension hereunder will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.
- **5.17 Status under Certain Statutes**. Neither the Company nor any Restricted Subsidiary is subject to regulation under the Investment Company Act of 1940, as amended, the Public Utility Holding Company Act of 1935, as amended, or, except as disclosed in Schedule 5.17, the Federal Power Act, as amended.
- **5.18** Environmental Matters. Except as disclosed in the Disclosure Documents, neither the Company nor any Restricted Subsidiary has knowledge of any Material claim or has received any notice of any Material claim, and no proceeding has been instituted raising any Material claim against the Company or any of its Restricted Subsidiaries or any of their

respective real properties now or formerly owned, leased or operated by any of them or other assets, alleging any damage to the environment or violation of any Environmental Laws, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect. Except as otherwise disclosed to each Lender in writing,

- (a) neither the Company nor any Restricted Subsidiary has knowledge of any facts which would give rise to any claim, public or private, of violation of Environmental Laws or damage to the environment emanating from, occurring on or in any way related to real properties now or formerly owned, leased or operated by any of them or to other assets or their use, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect;
- (b) neither the Company nor any of its Restricted Subsidiaries has stored any Hazardous Materials on real properties now or formerly owned, leased or operated by any of them in a manner contrary to any Environmental Laws and has not disposed of any Hazardous Materials in a manner contrary to any Environmental Laws, in each case in any manner that could reasonably be expected to result in a Material Adverse Effect; and
- (c) all buildings on all real properties now owned, leased or operated by the Company or any of its Restricted Subsidiaries are in compliance with applicable Environmental Laws, except where failure to comply could not reasonably be expected to result in a Material Adverse Effect.
- **5.19 Compliance With Issuance Limits**. Both before and after giving effect to any Credit Extension, all Credits issued and outstanding hereunder are in full compliance with the Issuance Limits.
- **5.20 Tax Shelter Regulations**. The Company does not intend to treat the Advances, Term Loans and Credits and related transactions as being a "reportable transaction" (within the meaning of Treasury Regulation Section 1.6011-4).

ARTICLE VI AFFIRMATIVE COVENANTS

So long as any Lender shall have any Credit-Linked Deposit, Letter of Credit Linked Note or Term Note hereunder or any Credit hereunder shall remain outstanding:

- **6.01 Financial and Business Information**. The Company shall deliver to the Administrative Agent, the Issuer and to each of the Lenders, in form and detail satisfactory to the Administrative Agent, the Issuer and the Required Credit-Linked Lenders:
- (a) Quarterly Statements within 60 days after the end of each quarterly fiscal period in each fiscal year of the Company (other than the last quarterly fiscal period of each such fiscal year), duplicate copies of,
 - (i) a consolidated balance sheet of the Company and its Subsidiaries as at the end of such quarter, and

(ii) consolidated statements of income, changes in shareholders' equity and cash flows of the Company and its Subsidiaries, for such quarter and (in the case of the second and third quarters) for the portion of the fiscal year ending with such quarter,

setting forth in each case in comparative form the figures for the corresponding periods in the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP applicable to quarterly financial statements generally, and certified by a Senior Financial Officer as fairly presenting, in all material respects, the financial position of the companies being reported on and their results of operations and cash flows, subject to changes resulting from year-end adjustments, provided that delivery within the time period specified above of copies of the Company's Quarterly Report on Form 10-Q prepared in compliance with the requirements therefor and filed with the SEC, if applicable, shall be deemed to satisfy the requirements of this Section 6.01(a) so long as such Quarterly Report on Form 10-Q includes the financial statements identified in clauses (i) and (ii) above;

- (b) Annual Statements within 100 days after the end of each fiscal year of the Company, duplicate copies of,
 - (i) a consolidated balance sheet of the Company and its Subsidiaries, as at the end of such year, and
 - (ii) consolidated statements of income, shareholder's equity and cash flows of the Company and its Subsidiaries, for such year,

setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP, and accompanied by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that such financial statements present fairly, in all material respects, the financial position of the companies being reported upon and their results of operations and cash flows and have been prepared in conformity with GAAP, and that the examination of such accountants in connection with such financial statements has been made in accordance with generally accepted auditing standards, and that such audit provides a reasonable basis for such opinion in the circumstances; provided that the delivery within the time period specified above of the Company's Annual Report on Form 10-K for such fiscal year (together with the Company's annual report to shareholders, if any, prepared pursuant to Rule 14a-3 under the Exchange Act) prepared in accordance with the requirements therefor and filed with the SEC, if applicable, shall be deemed to satisfy the requirements of this Section 6.01(b) so long as such Annual Report on Form 10-K includes the financial statements identified in clauses (i) and (ii) above;

- (c) SEC and Other Reports promptly, following the filing or mailing thereof, copies of all material of a financial nature filed with the SEC or sent to the Company's and its Subsidiaries' stockholders generally;
- (d) Outstanding Credits not later than five (5) Business Days after the end of each quarter, the information with respect to the outstanding Credits as is required under Section 2.01(j);

- (e) [intentionally omitted]
- (f) Notice of Default or Event of Default promptly, and in any event within five (5) Business Days after a Responsible Officer becoming aware of the existence of any Default or Event of Default, a written notice specifying the nature and period of existence thereof and what action the Company is taking or proposes to take with respect thereto;
- (g) ERISA Matters promptly, and in any event within five (5) Business Days after a Responsible Officer becoming aware of any of the following, a written notice setting forth the nature thereof and the action, if any, that the Company or an ERISA Affiliate proposes to take with respect thereto:
- (i) with respect to any Plan, any reportable event, as defined in Section 4043(b) of ERISA and the regulations thereunder, for which notice thereof has not been waived pursuant to such regulations as in effect on the date hereof; or
- (ii) the taking by the PBGC of steps to institute, or the threatening by the PBGC of the institution of, proceedings under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, or the receipt by the Company or any ERISA Affiliate of a notice from a Multiemployer Plan that such action has been taken by the PBGC with respect to such Multiemployer Plan; or
- (iii) any event, transaction or condition that would result in the incurrence of any liability by the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or such penalty or excise tax provisions, if such liability or Lien, taken together with any other such liabilities or Liens then existing, could reasonably be expected to have a Material Adverse Effect;
- (h) Notices from Governmental Authority promptly, and in any event within 30 days of receipt thereof, copies of any notice to the Company or any Subsidiary from any federal or state Governmental Authority relating to any order, ruling, statute or other law or regulation that the Company reasonably believes could have a Material Adverse Effect;
- (i) Additional Reporting Requirement in the event that Unrestricted Subsidiaries account for more than 10% of the Consolidated Total Assets of the Company and its Subsidiaries, or more than 10% of the consolidated revenue of the Company and its Subsidiaries, in each case as of the last day of the fiscal quarter or fiscal year for which the Company is delivering financial information pursuant to Sections 6.01(a) and (b), then each such set of financial information delivered pursuant to Sections 6.01(a) and (b) shall be accompanied by unaudited financial statements for all Unrestricted Subsidiaries of the Company taken as a group, together with consolidating statements reflecting eliminations or adjustments required to reconcile such group statements to the consolidated financial statements of the Company and its Subsidiaries;

(j) Requested Information — with reasonable promptness, such other data and information relating to the business, operations, affairs, financial condition, assets or properties of the Company or any its Subsidiaries or relating to the ability of the Company to perform its obligations hereunder, under the Letter of Credit Linked Notes and under the Term Notes or the ability of Holdings to perform its obligations hereunder as from time to time may be reasonably requested by any Lender, including without limitation and subject to Section 10.07, such information as is required by SEC Rule 144A under the Securities Act to be delivered to a prospective transferee of the Letter of Credit Linked Notes or Term Notes.

6.02 Officer's Certificate. Each set of financial statements delivered to a Lender pursuant to Section 6.01(a) or Section 6.01(b) hereof shall be accompanied by a certificate of a Senior Financial Officer setting forth:

- (a) Covenant Compliance the information (including reasonably detailed calculations) required in order to establish whether the Company was in compliance with the requirements of Sections 7.02(c)(iv), 7.03(m), 7.04 and 7.05, during the quarterly or annual period covered by the statements then being furnished (including with respect to each such Section, where applicable, the calculations of the maximum or minimum amount, ratio or percentage, as the case may be, permissible under the terms of such Sections, and the calculation of the amount, ratio or percentage then in existence); and
- (b) Event of Default a statement that such officer has reviewed the relevant terms hereof and has made, or caused to be made, under his or her supervision, a review of the transactions and conditions of the Company and its Subsidiaries from the beginning of the quarterly or annual period covered by the statements then being furnished to the date of the certificate and that such review shall not have disclosed the existence during such period of any condition or event that constitutes a Default or an Event of Default or, if any such condition or event existed or exists (including, without limitation, any such event or condition resulting from the failure of the Company or any Subsidiary to comply with any Environmental Law), specifying the nature and period of existence thereof and what action the Company shall have taken or proposes to take with respect thereto.
 - **6.03 Inspection**. The Company shall permit the representatives of each Lender:
- (a) No Default if no Default or Event of Default then exists, at the expense of such Lender, to visit the principal executive office of the Company and to discuss the affairs, finances and accounts of the Company and its Restricted Subsidiaries with the Company's officers, all at such reasonable times during normal business hours and as often as may be reasonably requested in writing; and
- (b) Default if a Default or Event of Default then exists, at the expense of the Company and upon reasonable prior notice, to visit the principal executive office of the Company, to examine all relevant books of account, records, reports and other papers relating to the Company's and Holdings' performance of their respective obligations under this Agreement, the Letter of Credit Linked Notes and the Term Notes and to make copies and abstracts therefrom, and to discuss the Company's and its Restricted Subsidiaries' affairs, finances and accounts with the Company's officers and independent public accountants (and by this provision

the Company authorizes said accountants to discuss the affairs, finances and accounts of the Company and its Restricted Subsidiaries), all at such reasonable times during normal business hours and as often as may be reasonably requested.

6.04 Compliance with Laws. The Company will, and will cause each of its Subsidiaries to, comply with all laws, ordinances or governmental rules or regulations to which each of them is subject, including, without limitation, Environmental Laws, and will obtain and maintain in effect all licenses, certificates, permits, franchises and other governmental authorizations necessary to the ownership of their respective properties or to the conduct of their respective businesses, in each case to the extent necessary to ensure that non-compliance with such laws, ordinances or governmental rules or regulations or failures to obtain or maintain in effect such licenses, certificates, permits, franchises and other governmental authorizations could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

6.05 Insurance. The Company will, and will cause each of its Restricted Subsidiaries to, maintain, with financially sound and reputable insurers, insurance with respect to their respective properties and businesses against such risks, of such types, on such terms and in such amounts as is customary in the case of entities of established reputations engaged in the same or a similar business and similarly situated.

6.06 Maintenance of Properties. The Company will, and will cause each of its Restricted Subsidiaries to, maintain and keep, or cause to be maintained and kept, their respective properties in good repair, working order and condition (other than ordinary wear and tear), so that the business carried on in connection therewith may be properly conducted at all times, provided that this Section shall not prevent the Company or any Restricted Subsidiary from discontinuing the operation and the maintenance of any of its properties or failing to carry on any such business if the Company has concluded that such discontinuance or failure could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

6.07 Payment of Taxes and Claims. The Company will, and will cause each of its Subsidiaries to, file all income tax and similar tax returns required to be filed in any jurisdiction and to pay and discharge all taxes shown to be due and payable on such returns and pay all other taxes, assessments, governmental charges, or levies imposed on them or any of their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent and all claims for which sums have become due and payable that have or might become a Lien on properties or assets of the Company or any Subsidiary, provided that neither the Company nor any Subsidiary need file any such returns or pay any such tax or assessment or claims (i) if the amount, applicability or validity thereof is contested by the Company or such Subsidiary on a timely basis in good faith and in appropriate proceedings, diligently conducted, and the Company or a Subsidiary has established adequate reserves therefor in accordance with GAAP on the books of the Company or such Subsidiary, or (ii) to the extent that the failure to file such returns and the nonpayment of all such taxes, assessments and claims, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

6.08 Corporate Existence, Etc. Subject to Section 7.02, the Company will at all times preserve and keep in full force and effect its corporate existence, and the Company will at

all times preserve and keep in full force and effect the corporate existence of each of its Restricted Subsidiaries (unless merged into the Company or a Restricted Subsidiary) and all rights and franchises of the Company and its Restricted Subsidiaries, unless, in the good faith judgment of the Company, the termination of or failure to preserve and keep in full force and effect such corporate existence, right or franchise could not, individually or in the aggregate, have a Material Adverse Effect.

6.09 Books and Records. The Company will, and will cause each Restricted Subsidiary to, (a) maintain proper books of record and account, in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of the Company or such Restricted Subsidiary, as the case may be; and (b) maintain such books of record and account in material conformity with all applicable requirements of any Governmental Authority having regulatory jurisdiction over the Company or such Restricted Subsidiary, as the case may be.

6.10 Purpose of Credits. Each Credit shall be used directly or indirectly to cover (a) a default in the performance of any non-financial or commercial obligations of the Company or any Subsidiary under specific contracts, including a Credit issued in favor of a bank or other surety who in connection therewith issues a guarantee or similar undertaking, performance bond, surety bond or other similar instrument that covers a default in the performance of any non-financial or commercial obligations under specific contracts or the payment of any financial contractual obligation, or (b) the payment of any financial contractual obligation of the Company or any Subsidiary.

ARTICLE VII NEGATIVE COVENANTS

So long as any Lender shall have any Credit-Linked Deposit, any Letter of Credit Linked Note or any Term Note hereunder or any Credit shall remain outstanding:

7.01 Transactions with Affiliates. The Company will not, and will not permit any Restricted Subsidiary to, enter into directly or indirectly any Material transaction or Material group of related transactions (including without limitation the purchase, lease, sale or exchange of properties of any kind or the rendering of any service) with any Affiliate (other than the Company or another Restricted Subsidiary), except in the ordinary course and pursuant to the reasonable requirements of the Company's or such Restricted Subsidiary's business and upon fair and reasonable terms no less favorable to the Company or such Restricted Subsidiary than would be obtainable in a comparable arm's-length transaction with a Person not an Affiliate.

7.02 Merger, Consolidation, Sale of Assets, etc.

- (a) The Company will not, and will not permit any Restricted Subsidiary to, consolidate with or merge with any other Person unless immediately after giving effect to any consolidation or merger no Default or Event of Default would exist and:
- (i) in the case of a consolidation or merger of a Restricted Subsidiary, either (x) the surviving or continuing corporation is the Company or another Restricted Subsidiary or

- (y) such consolidation or merger, if considered as the sale of the assets of such Restricted Subsidiary to such other Person, would be permitted by Section 7.02(c); and
- (ii) in the case of a consolidation or merger of the Company, the successor corporation which results from such consolidation or merger (the "surviving corporation"), if not the Company, shall have executed and delivered to the Issuer, the Administrative Agent and each Lender in writing its assumption of the due and punctual payment of all Credit Obligations, and the due and punctual performance and observation of all covenants in this Agreement, the Letter of Credit Linked Notes and the Term Notes, to be performed or observed by the Company hereunder and under the Credit Documents and shall furnish to the Issuer, the Administrative Agent and such Lenders a customary opinion of counsel solely to the effect that the instrument of assumption has been duly authorized, executed and delivered and constitutes the legal, valid and binding contract and agreement of the surviving corporation enforceable in accordance with its terms, except as enforcement of such terms may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the enforcement of creditors' rights generally and by general equitable principles.
- (b) The Company will not sell, lease (as lessor) or otherwise transfer all or substantially all of its assets in a single transaction or series of transactions to any Person unless:
 - (i) immediately after giving effect to such transaction, no Default or Event of Default shall have occurred and be continuing; and
- (ii) the successor corporation to which all or substantially all of the Company's assets have been sold, leased or transferred (the "successor corporation") shall have executed and delivered to the Issuer, the Administrative Agent and each Lender in writing its assumption of the due and punctual payment of all Credit Obligations, and the due and punctual performance and observation of all covenants in this Agreement, the Letter of Credit Linked Notes and the Term Notes, to be performed or observed by the Company hereunder and under the Credit Documents and shall furnish to the Issuer, the Administrative Agent and such Lenders a customary opinion of counsel solely to the effect that the instrument of assumption has been duly authorized, executed and delivered and constitutes the legal, valid and binding contract and agreement of the successor corporation enforceable in accordance with its terms, except as enforcement of such terms may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the enforcement of creditors' rights generally and by general equitable principles.

No such conveyance, transfer or lease of all or substantially all of the assets of the Company shall have the effect of releasing the Company or any successor corporation that shall theretofore have become such in the manner prescribed in this Section 7.02 from its liability under this Agreement or the Credit Documents.

(c) The Company will not, and will not permit any Restricted Subsidiary to, sell, lease (as lessor), transfer, abandon or otherwise dispose of (collectively, a "Disposition") any assets in one or a series of transactions (including, without limitation, in any sale-leaseback transaction), other than those assets consisting of accounts receivable sold through any Permitted Receivables Transaction, to any Person other than:

- (i) Dispositions in the ordinary course of business of inventory or of equipment, fixtures, supplies or materials no longer required in the business of the Company or any of its Restricted Subsidiaries or that is obsolete;
 - (ii) Dispositions by the Company to a Restricted Subsidiary or by a Restricted Subsidiary to the Company or to another Restricted Subsidiary;
 - (iii) the sale of assets for cash or other property if all of the following conditions are met:
 - (A) such assets (valued at book value) do not constitute a "substantial part" of the assets of the Company and its Restricted Subsidiaries;
 - (B) in the opinion of a Responsible Officer of the Company, the sale is for fair value and is in the best interests of the Company; and
 - (C) immediately after the consummation of the transaction and after giving effect thereto, no Default or Event of Default would exist; or
- (iv) the sale of assets meeting the conditions set forth in clauses (B) and (C) of clause (iii) above, as long as the cash proceeds from such sale in excess of a substantial part of the assets of the Company and its Restricted Subsidiaries (net of expenses of sale and an amount reasonably estimated by the Company to represent incremental income taxes due as a result of such sale) are (i) applied within 365 days of the date of receipt to the acquisition of productive assets useful and intended to be used in the operation or expansion of the business of the Company or its Restricted Subsidiaries, and/or (ii) used to repay any Indebtedness of the Company or its Restricted Subsidiaries (other than (x) Indebtedness that is in any manner subordinated in right of payment in any respect to Indebtedness arising or incurred under this Agreement, (y) Indebtedness owing to the Company, any of its Subsidiaries or any Affiliate, and (z) Indebtedness in respect of any revolving credit or similar credit facility providing the Company or any of its Restricted Subsidiaries with the right to obtain loans or other extensions of credit from time to time, except to the extent that in connection with such payment of Indebtedness the availability of credit under such credit facility is permanently reduced by an amount not less than the amount of such proceeds applied to the payment of such Indebtedness).
- (d) For purposes of Section 7.02(c), a sale of assets will be deemed to involve a "substantial part" of the assets of the Company and its Restricted Subsidiaries if the book value of such assets, together with all other assets sold, including mergers effected pursuant to Section 7.02(a)(i)(y) and the issuance, sale, transfer or disposition of stock or other Securities of Restricted Subsidiaries (except to the Company or another Restricted Subsidiary) effected pursuant to Section 7.02(e), during the same fiscal year (except (1) those assets sold pursuant to clauses (i) and (ii) of Section 7.02(c), and (2) those assets consisting of receivables sold through any Permitted Receivables Transaction permitted under Section 7.03(i)) equals at least 15.00% of the Consolidated Total Assets of the Company and its Restricted Subsidiaries determined as of the end of the immediately preceding fiscal year.

- (e) The Company will not permit any Restricted Subsidiary to issue shares of stock or other Securities to any Person other than the Company or another Restricted Subsidiary. The Company will not, and will not permit any Restricted Subsidiary to, issue, sell, transfer or otherwise dispose of its interest in any stock or any other Securities (or any options or warrants to purchase stock or other Securities exchangeable for or convertible into stock or other Securities) of any Restricted Subsidiary (except to the Company or another Restricted Subsidiary) unless such issuance, sale, transfer or disposition (valued at book value) would be permitted under Section 7.02(c).
- **7.03** Liens. The Company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly create, incur, assume or permit to exist (upon the happening of a contingency or otherwise) any Lien on or with respect to any of its properties or assets, whether now owned or hereafter acquired, except for the following (which are collectively referred to as "Permitted Liens"):
- (a) Mechanics', materialmen's, carriers' or other like liens arising in the ordinary course of business (including construction of facilities) in respect of obligations that are not due or which are being contested in good faith;
- (b) Liens arising by reason of deposits with, or the giving of any form of security to, any governmental agency or any body created or approved by law or governmental regulation, which is required by law or governmental regulation as a condition to the transaction of any business or the exercise of any privilege, franchise or license (including, without limitation, any Lien arising by reason of one or more letters of credit in connection with any international waste management contract to be performed by the Company or any of its Subsidiaries or their respective Affiliates);
- (c) Liens for taxes, assessments or governmental charges or levies not yet delinquent or Liens for taxes, assessments or governmental charges or levies already delinquent but the validity of which is being contested in good faith;
- (d) Liens (including judgment liens) arising in connection with legal proceedings so long as such proceedings are being contested in good faith and, in the case of judgment liens, execution thereon is stayed;
 - (e) Landlords' liens on fixtures located on premises leased by the Company or any Restricted Subsidiary in the ordinary course of business;
 - (f) [intentionally omitted]
- (g) Liens arising by reason of deposits to qualify the Company or any Restricted Subsidiary to conduct business, to maintain self-insurance, or to obtain the benefit of, or comply with, Laws;
- (h) Liens that secure any Indebtedness of a Restricted Subsidiary owing to the Company or another Restricted Subsidiary or by the Company to a Restricted Subsidiary;

- (i) Liens created or deemed to be created under Permitted Receivables Transactions at any time provided such Liens do not extend to any property or assets other than the trade receivables sold pursuant to such Permitted Receivables Transactions, interests in the goods or products (including returned goods and products), if any, relating to the sales giving rise to such trade receivables; any security interests or other Liens and property subject thereto (other than on any leases or related lease payment rights or receivables between the Company and any of its Restricted Subsidiaries, as lessors or sublessors) from time to time purporting to secure the payment by the obligors describing the collateral securing such trade receivables (together with any financing statements signed by such obligors describing the collateral securing such trade receivables) pursuant to such Permitted Receivables Transactions;
 - (j) Liens existing at Closing and reflected on Schedule 7.03 hereto;
- (k) (i) Liens upon property or assets which are created prior to or contemporaneously with, or within 360 days after, (1) in the case of the acquisition of such property or assets, the completion of such acquisition; and (2) in the case of the construction, development or improvement of such property or assets, the later to occur of the completion of such construction, development or improvement or the commencement of operation or use of the property or assets, which Lien secures or provides for the payment, financing or refinancing, directly or indirectly, of all or any part of the acquisition cost of such property or assets or the cost of construction, development or improvement thereof; or (ii) any Lien upon property or assets existing at the time of the acquisition thereof, which Lien secures obligations assumed by the Company or any Restricted Subsidiary; or (iii) any conditional sales agreement or other title retention agreement with respect to any property or assets acquired by the Company or any Restricted Subsidiary; or (iv) any Lien existing on the property or assets or shares of stock of a corporation or firm at the time such corporation or firm is merged into or consolidated with the Company or any Restricted Subsidiary or at the time such corporation becomes a Restricted Subsidiary; or (v) any Lien existing on the property, assets or shares of stock of any successor which shall have become the Company in accordance with the provisions set forth in Section 7.02; provided, in each case, that any such Lien described in the foregoing clauses (ii), (iii), (iv) or (v) does not attach to or affect property or assets owned by the Company or any Restricted Subsidiary prior to the event referred to in such clauses; provided further (for avoidance of doubt) that any such Lien described in the foregoing clauses (ii), (iii) or (iii) may be created in favor of any Governmental Authority.
- (l) Any extension, renewal or refunding (or successive extensions, renewals or refundings) in whole or in part of any Indebtedness secured by any Lien referred to in the foregoing clauses, inclusive, provided that the principal amount of the Indebtedness secured is not increased or the maturity thereof reduced and the Lien securing such Indebtedness shall be limited to the property or assets that, immediately prior to such extension, renewal or refunding, secured such Indebtedness and additions to such property or assets; and
- (m) Liens securing Priority Debt, provided that Priority Debt shall not at any time exceed 15.00% of Consolidated Tangible Assets as of the then most recently ended fiscal quarter

or fiscal year of the Company for which financial information is reportable pursuant to Sections 6.01(a) and (b).

7.04 Limitation on Consolidated Debt. The Company will not permit at any time the ratio of (a) Consolidated Debt to (b) Adjusted Consolidated EBITDA for the most recently ended period of four consecutive fiscal quarters of the Company to be greater than 3.75 to 1.00.

7.05 Limitation on Priority Debt. The Company will not permit Priority Debt at any time to exceed 15.00% of Consolidated Tangible Assets as of the end of the most recently ended fiscal quarter or fiscal year of the Company for which financial information is reportable pursuant to Sections 6.01(a) and (b).

7.06 Nature of Business. The Company will not, and will not permit any Restricted Subsidiary to, engage in any business if, as a result, the general nature of the business of the Company and its Restricted Subsidiaries, taken as a whole, which would then be engaged in by the Company and its Restricted Subsidiaries would be substantially changed from the nature of the business engaged in by the Company and its Restricted Subsidiaries, taken as a whole, on the Closing Date.

7.07 Designation of Unrestricted Subsidiaries. The Company may designate any newly-formed or newly-acquired Subsidiary as a Restricted Subsidiary or an Unrestricted Subsidiary; provided that if the Company shall fail to so designate by the end of the fiscal quarter in which such Subsidiary is formed or acquired, such Subsidiary shall be deemed to be a Restricted Subsidiary. In addition, the Company may designate any Restricted Subsidiary as an Unrestricted Subsidiary and any Unrestricted Subsidiary as a Restricted Subsidiary; provided that, (a) if such Subsidiary initially is a Restricted Subsidiary, then such Restricted Subsidiary may be subsequently redesignated as a Restricted Subsidiary may be subsequently designated as a Restricted Subsidiary may be subsequently designated as a Restricted Subsidiary may be subsequently redesignated as an Unrestricted Subsidiary, but no further changes in designation may be made; (c) immediately before and after designation of a Restricted Subsidiary as an Unrestricted Subsidiary or of an Unrestricted Subsidiary as a Restricted Subsidiary there exists no Default or Event of Default; and (d) Holdings may not be designated as an Unrestricted Subsidiary.

7.08 Margin Stock. The Company shall not use the Credits, the Advances or the Term Loans, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock with in the meaning of Regulation U of the FRB.

ARTICLE VIII EVENTS OF DEFAULT AND REMEDIES

8.01 Events of Default. Any of the following shall constitute an Event of Default:

(a) Non-Payment of Drawing or Principal Amount. The Company fails to pay when and as required to be paid herein or in any other Credit Document (i) within six Business Days

after the applicable Reimbursement Date of any drawing under any Credit, the amount of such drawing (unless within such six Business Day period a Request for Term Loans has been received pursuant to Section 2.03(b), and such Term Loans are made in accordance with the terms and conditions of Sections 2.03, 4.02 and 4.03 within 15 Business Days of the receipt by the Administrative Agent of such Request for Term Loans), or (ii) on the Maturity Date, the aggregate principal amount of any Term Loan or Advances outstanding; or

- (b) Non-Payment of Interest and Other Amounts. The Company defaults in the payment of any interest on any Letter of Credit Linked Note or Term Note, any Facility Fee or Fronting Fee due hereunder, or other amount payable hereunder or under any other Credit Document for more than five (5) Business Days after the same becomes due and payable; or
- (c) Specific Covenants. The Company defaults in the performance of or compliance with (i) Section 7.05, which default is not remedied within ten (10) Business Days after a Responsible Officer obtaining actual knowledge of such default, or (ii) Sections 7.02, 7.03, or 7.04; or
- (d) Other Defaults. The Company defaults in the performance of or compliance with any term contained herein (other than those referred to in paragraphs (a), (b) and (c) of this Section 8.01) and such default is not remedied within thirty (30) days after a Responsible Officer obtaining actual knowledge of such default; or
- (e) Representations and Warranties. Any representation or warranty made in writing by or on behalf of the Company or Holdings or by any officer of the Company or Holdings in this Agreement or in any writing furnished in connection with the transactions contemplated hereby proves to have been false or incorrect in any material respect on the date as of which made; or
- (f) Cross-Defaults. (i) A default in the payment of any principal of or premium or make-whole amount or interest on any Material Debt has occurred and continues beyond any period of grace provided with respect thereto, or (ii) any other default in the performance of or compliance with any term of any evidence of any Material Debt or of any mortgage, indenture or other agreement relating thereto exists, and as a consequence of such default such Material Debt has become, or has been declared (or one or more Persons are entitled to declare such Material Debt to be), due and payable before its stated maturity or before its regularly scheduled dates of payment, or (iii) as a consequence of the occurrence or continuation of any event, (x) the Company or any Restricted Subsidiary has become obligated to purchase or repay any Material Debt before its regular maturity or before its regularly scheduled dates of payment, or (y) one or more Persons have the right to require the Company or any Restricted Subsidiary so to purchase or repay any Material Debt, or (iv) an event or condition shall occur which results in an "Event of Default" under the Seven Year Agreement or the Ten Year Agreement; or
- (g) Failure to Pay Debts; Voluntary Proceedings. The Company, any Restricted Subsidiary or Holdings (i) is generally not paying, or admits in writing its inability to pay, its debts as they become due, (ii) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy, for liquidation or to take advantage of any bankruptcy, insolvency, reorganization, moratorium or

other similar law of any jurisdiction, (iii) makes an assignment for the benefit of its creditors, (iv) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, (v) is adjudicated as insolvent or to be liquidated, or (vi) takes corporate action for the purpose of any of the foregoing; or

- (h) Involuntary Proceedings. A court or governmental authority of competent jurisdiction enters an order appointing, without consent by the Company, any Restricted Subsidiary or Holdings, as applicable, a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, or constituting an order for relief or approving a petition for relief or reorganization or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy or insolvency law of any jurisdiction, or ordering the dissolution, winding-up or liquidation of the Company, any Restricted Subsidiary or Holdings, or any such petition shall be filed against the Company, any Restricted Subsidiary or Holdings and such petition shall not be dismissed within 60 days; or
- (i) Judgments. A final judgment or judgments for the payment of money aggregating in excess of 10.00% of Consolidated Net Worth are rendered against one or more of the Company, any Restricted Subsidiary or Holdings and which judgments are not, within 60 days after entry thereof, bonded, discharged or stayed pending appeal, or are not discharged within 60 days after the expiration of such stay; or
- (j) Invalidity of Holdings Guaranty. If for any reason (other than payment in full of all obligations under this Agreement, the Letter of Credit Linked Notes and the Term Notes) the Holdings Guaranty shall cease to be in full force and effect (other than as expressly provided in Section 11.09 of this Agreement), or the Company or Holdings (or any other Person on behalf of the Company or Holdings) shall contest the validity or enforceability of such Holdings Guaranty, deny that Holdings has any further liability hereunder, or seek to revoke such Holdings Guaranty; or
- (k) ERISA. If (i) any Plan shall fail to satisfy the minimum funding standards of ERISA or the Code for any plan year or part thereof or a waiver of such standards or extension of any amortization period is sought or granted under Section 412 of the Code, (ii) a notice of intent to terminate any Plan shall have been or is reasonably expected to be filed with the PBGC or the PBGC shall have instituted proceedings under ERISA Section 4042 to terminate or appoint a trustee to administer any Plan or the PBGC shall have notified the Company or any ERISA Affiliate that a Plan may become a subject of any such proceedings, (iii) the aggregate "amount of unfunded benefit liabilities" (within the meaning of Section 4001(a)(18) of ERISA) under all Plans, determined in accordance with Title IV of ERISA, shall exceed 10.00% of Consolidated Net Worth, (iv) the Company or any ERISA Affiliate shall have incurred or is reasonably expected to incur any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, (v) the Company or any ERISA Affiliate withdraws from any Multiemployer Plan, or (vi) the Company or any Subsidiary establishes or amends any employee welfare benefit plan that provides post-employment welfare benefits in a manner that would increase the liability of the Company or any Subsidiary thereunder; and any such event or events described in clauses (i) through (vi) above,

either individually or together with any other such event or events, could reasonably be expected to have a Material Adverse Effect; or

(l) Change in Control. At any time any Person or group of Persons (within the meaning of Section 13 or 14 of the Securities Exchange Act of 1934, as amended) shall have acquired beneficial ownership (within the meaning of Rule 13d-3 promulgated by the Securities and Exchange Commission under said Act) of 25% or more of the outstanding shares of common voting stock of the Company; or during any period of twelve consecutive calendar months, individuals who were directors of the Company on the first day of such period (together with any new directors whose election by such board or whose nomination for election by the shareholders of the Company was approved by a vote of a majority of the directors still in office who were either directors at the beginning of such period or whose election or nomination for election was previously so approved) shall cease to constitute a majority of the board of directors of the Company.

As used in Section 8.01(k), the terms "employee benefit plan" and "employee welfare benefit plan" shall have the respective meanings assigned to such terms in Section 3 of ERISA.

- **8.02 Remedies Upon Event of Default**. If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Credit-Linked Lenders, take any or all of the following actions, which actions shall be taken simultaneously under this Agreement, the Seven Year Agreement and the Ten Year Agreement:
- (a) declare any obligation of the Issuer to make Credit Extensions and the obligations of the Lenders to make Term Loans to be terminated, whereupon such commitments and obligation shall be terminated;
- (b) declare all amounts owing or payable hereunder or under any other Credit Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Company;
- (c) require that the Company Cash Collateralize the Credit Obligations (other than Advances and Term Loans) (in an amount equal to the Outstanding Amount thereof, as such amount may vary from time to time); and
- (d) exercise on behalf of itself, the Issuer and the Lenders all rights and remedies available to it, the Issuer and the Lenders under the Credit Documents or applicable law;

provided, however, that upon the occurrence of an Event of Default pursuant to Sections 8.01(g) or (h), any obligation of the Issuer or the Lenders to make Credit Extensions shall automatically terminate, all unpaid amounts as aforesaid shall automatically become due and payable, and the obligation of the Company to Cash Collateralize the Credit Obligations (other than Advances and Term Loans) as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent or any Lender.

Thereafter, at the request of the Required Lenders, the Administrative Agent shall (x)

withdraw from the Credit-Linked Deposit Account and distribute to the Lenders an amount equal to the excess of the Total Credit-Linked Deposit over the Outstanding Amount of the Credit Obligations (other than Advances and Term Loans), and (y) subject to, and to the extent of, the Company's compliance with any Cash Collateralization requirements as aforesaid, withdraw from the Credit-Linked Deposit Account and distribute to the Lenders the aggregate amount of any remaining Credit-Linked Deposits to the extent of the amount credited to the Credit-Linked Deposit Account; in each case subject to any notice required pursuant to the Money Market Account Agreement.

8.03 Application of Funds. After the exercise of remedies provided for in Section 8.02 (or after the obligations of the Issuer to make Credit Extensions have been automatically terminated and the Credit Obligations have automatically been required to be Cash Collateralized as set forth in the proviso to Section 8.02), any amounts received by the Administrative Agent, the Issuer or the Lenders on account of the Credit Obligations shall be applied by the Administrative Agent in the following order:

First, to payment of fees, indemnities, expenses and other amounts (including interest thereon, Attorney Costs and amounts payable under Article III) payable to the Administrative Agent in its capacity as such;

Second, to payment of fees, indemnities and other amounts (excluding principal and interest on any Borrowings but including interest, if any, on any such fees, indemnities and other amounts) payable to the Issuer and the Lenders (including Attorney Costs and amounts payable under Article III), ratably among them in proportion to the amounts described in this clause Second payable to them;

Third, to payment of that portion of the Credit Obligations constituting accrued and unpaid interest on the Borrowings, to the Issuer and ratably among the Lenders in proportion to the respective amounts described in this clause Third payable to them;

Fourth, to payment of that portion of the Credit Obligations constituting unpaid principal of the Borrowings, to the Issuer and ratably among the Lenders in proportion to the respective amounts described in this clause Fourth held by them;

Fifth, to the Administrative Agent for the account of the Issuer, to Cash Collateralize that portion of Credit Obligations comprised of the aggregate undrawn amount of Credits; and

Last, the balance, if any, after all of the Credit Obligations have been paid in full, to the Company or as otherwise required by Law.

Subject to Section 2.01(c), amounts used to Cash Collateralize the aggregate undrawn amount of Credits pursuant to clause Fifth above shall be applied to satisfy drawings under such Credits as they occur. If any amount remains on deposit as Cash Collateral after all Credits have either been fully drawn or expired, such remaining amount shall be applied to the other Credit Obligations, if any, in the order set forth above.

ARTICLE IX ADMINISTRATIVE AGENT

9.01 Appointment and Authorization of Administrative Agent. (a) Each of the Lenders and the Issuer hereby irrevocably appoints, designates and authorizes the Administrative Agent to take such action on its behalf under the provisions of this Agreement and each other Credit Document and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement or any other Credit Document, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Credit Document, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall the Administrative Agent have or be deemed to have any fiduciary relationship with the Issuer, any Lender or participant, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Credit Document or otherwise exist against the Administrative Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Credit Documents with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable Law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.

(b) The Issuer shall act on behalf of the Lenders with respect to any Credits issued by it and the Issuer Documents associated therewith, and the Issuer shall have all of the benefits and immunities (i) provided to the Administrative Agent in this Article IX (other than Section 9.07) with respect to any acts taken or omissions suffered by the Issuer in connection with Credits issued by it or proposed to be issued by it and the Issuer Documents pertaining to such Credits as fully as if the term "Administrative Agent" as used in this Article IX and in the definition of "Agent-Related Person" included the Issuer with respect to such acts or omissions, and (ii) as additionally provided herein with respect to the Issuer.

9.02 Delegation of Duties. The Administrative Agent may execute any of its duties under this Agreement or any other Credit Document by or through agents, employees or attorneys-in-fact and shall be entitled to advice of counsel and other consultants or experts concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects in the absence of gross negligence or willful misconduct.

9.03 Liability of Administrative Agent. No Agent-Related Person shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Credit Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct in connection with its duties expressly set forth herein), or (b) be responsible in any manner to any Lender or participant for any recital, statement, representation or warranty made by any Credit Party or any officer thereof, contained herein or in any other Credit Document, or in any certificate, report, statement or other document referred to or provided for in, or received by the Administrative Agent under or in connection with, this Agreement or any other Credit Document, or the validity, effectiveness, genuineness,

enforceability or sufficiency of this Agreement or any other Credit Document, or for any failure of any Credit Party or any other party to any Credit Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Lender or participant to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Credit Document, or to inspect the properties, books or records of any Credit Party or any Affiliate thereof.

9.04 Reliance by Administrative Agent. (a) The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, electronic mail message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to any Credit Party), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under any Credit Document unless it shall first receive such advice or concurrence of the Required Credit-Linked Lenders as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Credit Document in accordance with a request or consent of the Required Lenders or Required Credit-Linked Lenders, as applicable (or such greater number of Lenders as may be expressly required hereby in any instance) and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Lenders.

(b) For purposes of determining compliance with the conditions specified in Section 4.01, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

9.05 Notice of Default. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to defaults in the payment of principal, interest and fees required to be paid to the Administrative Agent for the account of the Lenders, unless the Administrative Agent shall have received written notice from the Issuer, a Lender or the Company referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default." The Administrative Agent will notify the Lenders of its receipt of any such notice. The Administrative Agent shall take such action with respect to such Default or Event of Default as may be directed by the Required Credit-Linked Lenders in accordance with Article VIII; provided, however, that unless and until the Administrative Agent has received any such direction, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable or in the best interest of the Lenders.

9.06 Credit Decision; Disclosure of Information by Administrative Agent. Each Lender and the Issuer acknowledges that no Agent-Related Person has made any representation or warranty to it, and that no act by the Administrative Agent hereafter taken, including any consent to and acceptance of any assignment or review of the affairs of any Credit Party or any Affiliate thereof, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender as to any matter, including whether Agent-Related Persons have disclosed material information in their possession. Each Lender and the Issuer represents to the Administrative Agent that it has, independently and without reliance upon any Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of the Credit Parties and their respective Subsidiaries, and made its own decision to enter into this Agreement and to extend credit to the Company and any other Credit Parties hereunder. Each Lender and the Issuer hereby acknowledge that it has not relied on any Agent-Related Person (other than as a recipient of legal opinions of counsel to the Administrative Agent delivered pursuant to Section 4.01(a)(vi)) with respect to any bank or other regulatory Laws relating to the transactions contemplated hereby. Each Lender and the Issuer also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Credit Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of the Company and the other Credit Parties. Except for notices, reports and other documents expressly required to be furnished to the Issuer and the Lenders by the Administrative Agent herein, the Administrative Agent shall not have any duty or responsibility to provide the Issuer or any Lender with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of any of the Credit Parties or any of their respective Affiliates which may come into the possession of any Agent-Related Person.

9.07 Indemnification of Administrative Agent. The Lenders shall indemnify upon demand, in Dollars, the Administrative Agent, in its capacity as Administrative Agent, and the Affiliates (to the extent such Affiliates are acting for the Administrative Agent in such capacity), officers, directors, employees, agents and attorneys-in-fact thereof who are acting for or on behalf of the Administrative Agent (pursuant to the express authorization of, or by express delegation from, the Administrative Agent) in connection with this Agreement (to the extent not reimbursed by or on behalf of any Credit Party and without limiting the obligation of any Credit Party to do so), *pro rata* in accordance with its Pro Rata Share, and hold harmless each such Person from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Person in any way relating to or arising out of or in connection with (before or after the Closing Date) the execution, delivery, enforcement, performance or administration of any Credit Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby (the "Agent Indemnified Liabilities"), incurred by any such Person in connection with the Administrative Agent's exercise of powers

and performance of duties expressly provided for under this Agreement; provided, however, that no Lender shall be liable for the payment to any such Person of any portion of such Agent Indemnified Liabilities resulting from such Person's own gross negligence or willful misconduct; and provided further, however, that no action taken in accordance with the directions of the Required Lenders or Required Credit-Linked Lenders, as applicable, shall be deemed to constitute gross negligence or willful misconduct for purposes of this Section; and provided further, however, that the Agent Indemnified Liability was incurred by or asserted against the Administrative Agent in its capacity as such, or such Affiliate or other Person acting for the Administrative Agent in such capacity. Without limitation of the foregoing, each Lender shall reimburse the Administrative Agent upon demand for such Lender's Pro Rata Share of any costs or out-of-pocket expenses (including reasonable Attorney Costs) incurred by the Administrative Agent (i) in connection with any enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Credit Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Credit Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), or (ii) otherwise, while an Event of Default exists, to the extent that the Administrative Agent is not reimbursed for such expenses by or on behalf of the Company within 30 days after demand therefor; provided in each case that such cost or expense was incurred by the Administrative Agent in its capacity as such; and provided, however, in each case that before making any demand on the Lenders for reimbursement of any such cost or expense, the Administrative Agent shall have first requested the Company to pay such amounts and made a good faith determination that the Company will not honor such request. Notwithstanding the foregoing, no Lender shall be liable for any indirect, special, punitive or consequential damages claimed by any Person under this Section 9.07 (other than such damages claimed by such Person as a result of such damages being claimed against such Person by a third party). The undertaking in this Section shall survive termination of the Credit-Linked Deposits, the payment of all other Credit Obligations and other amounts due hereunder and the resignation of the Administrative Agent. Notwithstanding the foregoing, (a) no Lender shall be obligated to reimburse or indemnify the Administrative Agent, in its capacity as Administrative Agent, and the Affiliates (to the extent such Affiliates are acting for the Administrative Agent in such capacity), officers, directors, employees, agents and attorneys-in-fact thereof, in respect of the transactions contemplated by Section 10.19 resulting from a Deposit Account Interest Non-Payment Event or in respect of any other nonpayment of funds owing to the Lenders, if such other nonpayment results solely from the Administrative Agent's nonperformance of its paying agency duties hereunder (and not from any action or inaction of the Company or any other unrelated Person), and (b) no Lender shall be obligated to reimburse or indemnify BAS.

9.08 Administrative Agent in its Individual Capacity. Bank of America and its Affiliates may make loans to, issue letters of credit and guarantees or other undertakings for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with each of the Credit Parties and their respective Affiliates as though Bank of America were not the Administrative Agent or the Issuer hereunder and without notice to or consent of the Lenders. The Lenders acknowledge that, pursuant to such activities, Bank of America or its Affiliates may receive information regarding any Credit Party or its Affiliates (including information that may be subject to

confidentiality obligations in favor of such Credit Party or such Affiliate) and acknowledge that the Administrative Agent shall be under no obligation to provide such information to them.

9.09 Successor Administrative Agent. The Administrative Agent may resign as Administrative Agent upon 30 days' notice to the Lenders; provided that any such resignation by Bank of America shall also constitute its resignation as Issuer under this Agreement; provided, however, the Administrative Agent may not resign if (i) it is the Issuer of any outstanding Credits under this Agreement or (ii) any Credit-Linked Deposits of any Lender exist. If the Administrative Agent resigns under this Agreement, the Required Credit-Linked Lenders shall appoint a successor administrative agent for the Lenders which successor administrative agent shall require the consent of the Company at all times other than during the existence of any Default or Event of Default (which consent of the Company shall not be unreasonably withheld or delayed). If no successor administrative agent is appointed prior to the effective date of the resignation of the Administrative Agent, the Administrative Agent may appoint, after consulting with the Credit-Linked Lenders and the Company, a successor administrative agent from among the Credit-Linked Lenders with the consent of such Credit-Linked Lender and the consent of the Company (which consent of the Company shall be required at all times other than during the existence of any Default or Event of Default and shall not be unreasonably withheld or delayed). The successor administrative agent under this Agreement, the Seven Year Agreement and the Ten Year Agreement shall at all times be the same Person. Upon the acceptance of its appointment as successor administrative agent shall succeed to all the rights, powers and duties of the retiring Administrative Agent and the term "Administrative Agent" shall mean such successor administrative agent, and the retiring Administrative Agent and the term "Administrative Agent" shall be terminated. After any retiring Administrative Agent in the retirin

ARTICLE X MISCELLANEOUS

10.01 Amendments, Etc. No amendment or waiver of any provision of this Agreement or any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)), and no consent to any departure by the Company or any other Credit Party therefrom (other than a departure of Holdings pursuant to Section 11.09 of this Agreement), shall be effective unless (A) in writing signed by the Required Credit-Linked Lenders and the Company or the applicable Credit Party, as the case may be, and acknowledged by the Administrative Agent and (B) a corresponding amendment, waiver or consent under (i) the Seven Year Agreement and (ii) the Ten Year Agreement is also effective (it being understood that the same terms and conditions with respect to such amendment, waiver or consent shall apply to all Credit-Linked Lenders), and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no such amendment, waiver or consent shall, without the written consent of each Lender with Credit Obligations directly affected (whose consent shall be required for any such amendment, waiver or consent in addition to that of the Required Credit-Linked Lenders):

- (a) waive any condition set forth in Section 4.01;
- (b) extend the Maturity Date or increase the Credit-Linked Deposit of any Lender (or reinstate any obligation to make Credit Extensions terminated pursuant to Section 8.02 in which such Lender participates);
- (c) postpone any date fixed by this Agreement or any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)) for any payment of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b));
- (d) reduce the principal of, or the rate of interest specified herein on, any Borrowing, or (subject to clause (iii) of the proviso below) any fees or other amounts payable hereunder or under any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)); provided, however, that only the consent of the Required Credit-Linked Lenders shall be necessary to amend the definition of "Default Rate," the amount of any increase in the Applicable Rate pursuant to the definition thereof during the existence of any Event of Default, or the interest rate applicable from time to time to Borrowings under Section 2.04(a)(ii), or to waive any obligation of the Company to pay interest at the Default Rate or at the rate set forth in Section 2.04(a)(ii), or to increase the Applicable Rate during the existence of an Event of Default;
 - (e) change Section 2.08 or Section 8.03 in a manner that would alter the *pro rata* sharing of payments required thereby;
- (f) change any provision of this Section 10.01, or the definition of "Required Credit-Linked Lenders", "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder; or
- (g) release Holdings from any of the Guaranteed Obligations under the Holdings Guaranty (other than as expressly provided in Section 11.09 of this Agreement);

and, provided further, that (i) no amendment, waiver or consent shall, unless in writing and signed by the Issuer in addition to the Lenders required above, modify or waive any provision of Section 2.01(a) (including defined terms used therein) or otherwise affect the rights or duties of the Issuer under this Agreement or any Issuer Document relating to any Credit issued or to be issued by it; (ii) no amendment, waiver or consent shall, unless in writing and signed by the Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Credit Document; and (iii) the Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed only by the respective parties thereto.

10.02 Notices and Other Communications; Facsimile Copies.

(a) General. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission).

All such written notices shall be mailed, faxed or delivered to the applicable address, facsimile number or (subject to subsection (c) below) electronic mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

- (i) if to the Company, Holdings, the Administrative Agent or the Issuer, to the address, facsimile number, electronic mail address or telephone number specified for such Person on Schedule 10.02 or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties; and
- (ii) if to any other Lender, to the address, facsimile number, electronic mail address or telephone number specified in its Administrative Questionnaire or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the Company, the Administrative Agent and the Issuer.

All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (B) if delivered by mail, four Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of subsection (c) below), when delivered; provided, however, that notices and other communications to the Administrative Agent and the Issuer pursuant to Article II shall not be effective until actually received by any such Person. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.

- (b) Effectiveness of Facsimile Documents and Signatures. Credit Documents may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually-signed originals and shall be binding on all Credit Parties, the Administrative Agent, the Issuer and the Lenders. The Administrative Agent may also require that any such documents and signatures be confirmed by a manually-signed original thereof; provided, however, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.
- (c) Limited Use of Electronic Mail. Electronic mail and Internet and intranet websites may be used only to distribute routine communications, other than financial statements and other similar information, and to distribute Credit Documents for execution by the parties thereto, and may not be used for any other purpose.
- (d) Reliance by Administrative Agent, the Issuer and Lenders. The Administrative Agent, the Issuer and the Lenders shall be entitled to rely and act upon any notices purportedly given by or on behalf of the Company even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Company shall indemnify each Agent-Related Person, each Issuer-Related Person and each Lender from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Company. All

telephonic notices to and other communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

10.03 No Waiver; Cumulative Remedies. No failure by any Lender, the Issuer or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein or therein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

10.04 Attorney Costs, Expenses and Taxes. The Company agrees (a) (i) to pay or reimburse each of the Administrative Agent, the Issuer and special counsel for the Lenders for all reasonable costs and expenses incurred in connection with the development, preparation, negotiation and execution of this Agreement and the other Credit Documents (whether or not the transactions contemplated hereby are consummated), and the consummation and administration of the transactions contemplated hereby, including all reasonable Attorney Costs, expenses and disbursements and (ii) to pay or reimburse each of the Administrative Agent, the Issuer and each Lender for all reasonable costs and expenses incurred in connection with any amendment, waiver, consent or other modification of the provisions of this Agreement and the other Credit Documents (whether or not the transactions contemplated thereby are consummated), including without limitation any out-of-pocket costs and expenses incurred in connection with the transactions contemplated by Section 10.19 of this Agreement, and the consummation and administration of the transactions contemplated thereby, including all reasonable Attorney Costs, expenses and disbursements, but excluding any such costs, expenses, attorneys' fees, and disbursements of the Issuer and the Administrative Agent in connection with the transactions contemplated in Section 10.19 resulting from a Deposit Account Interest Non-Payment Event; and (b) to pay or reimburse the Administrative Agent, the Issuer and each Lender for all costs and expenses incurred in connection with the enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Credit Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Credit Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all Attorney Costs. The foregoing costs and expenses shall include all search, filing, recording, title insurance and appraisal charges and fees and taxes related thereto, and other out-of-pocket expenses incurred by the Administrative Agent and the cost of independent public accountants and other outside experts retained by the Administrative Agent, the Issuer or any Lender. All amounts due under this Section 10.04 shall be payable within ten Business Days after demand therefor. The agreements in this Section 10.04 shall survive the termination of the Credit-Linked Deposits and repayment of all Credit Obligations and all other amounts due hereunder.

10.05 Indemnification by the Company. Whether or not the transactions contemplated hereby are consummated, the Company shall indemnify and hold harmless each Agent-Related Person, each Issuer-Related Person, each Lender and their respective Affiliates, directors, officers, employees, counsel, agents and attorneys-in-fact (collectively, the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties,

claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (before or after the Closing Date) (a) the execution, delivery, enforcement, performance or administration of any Credit Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, including without limitation the transactions contemplated by Section 10.19 of this Agreement but excluding any such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements of the Issuer and the Administrative Agent in connection with the transactions contemplated in Section 10.19 resulting from a Deposit Account Interest Non-Payment Event, (b) any Credit-Linked Deposit, Credit or the use or proposed use of the proceeds therefrom (including any refusal by the Issuer to honor a demand for payment under a Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Credit), or (c) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding) and regardless of whether any Indemnitee is a party thereto (all the foregoing, collectively, the "Indemnified Liabilities"), in all cases, whether or not caused by or arising, in whole or in part, out of the negligence of the Indemnitee; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. No Indemnitee shall be liable for any indirect, special, punitive or consequential damages relating to this Agreement or any other Credit Document or arising out of its activities in connection herewith or therewith (whether before or after the Closing Date). Notwithstanding the foregoing, the Company shall not be liable for any indirect, special, punitive or consequential damages claimed by an Indemnitee in connection with the transactions contemplated by Section 10.19 of this Agreement (other than such damages claimed by an Indemnitee as a result of such damages being claimed against such Indemnitee by a third party). All amounts due under this Section 10.05 shall be payable within ten Business Days after demand therefor. The agreements in this Section shall survive the resignation of the Administrative Agent, the replacement of any Lender, the termination of the Credit-Linked Deposits and the repayment, satisfaction or discharge of all the other Credit Obligations.

10.06 Payments Set Aside. To the extent that any payment by or on behalf of the Company is made to the Administrative Agent, the Issuer or any Lender, or the Administrative Agent, the Issuer or any Lender exercises its right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, the Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred, and (b) the Issuer and each Lender severally

agree to pay to the Administrative Agent upon demand its applicable share of any amount so recovered from or paid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect, in the applicable currency of such recovery or payment.

10.07 Successors and Assigns. (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Company may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Issuer and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of subsection (b) of this Section 10.07, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section 10.07, (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section 10.07 (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section 10.07 and, to the extent expressly contemplated hereby, the Indemnitees) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) Any Lender may at any time assign to another Lender or to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Credit-Linked Deposit and Credit Obligations at the time owing to it); provided that unless such assignment is to another Lender (i) upon such assignment, the aggregate principal amount of the Credit-Linked Deposit and Credit Obligations held by the assignee, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000; (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Credit Obligations and the Credit-Linked Deposit assigned; and (iii) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption. Subject to acceptance and recording thereof by the Administrative Agent pursuant to subsection (c) of this Section 10.07, from and after the effective date specified in each Assignment and Assumption, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 3.01, 3.02, 3.03, 10.04 and 10.05 with respect to facts and circumstances occurring prior to the effective date of such assignment). Upon request, the Company (at its expense) shall execute and deliver a Letter of Credit Linked Note and one or more Term Notes to the assignee Lender. Any assignment or transfer by a Lender of rights

purposes of this Agreement as a sale by such Lender of a Participation in such rights and obligations in accordance with subsection (d) of this Section 10.07.

- (c) The Administrative Agent, acting solely for this purpose as an agent of the Company, shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Credit-Linked Deposit of, and amounts of the Credit Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error and the Company, the Administrative Agent, the Issuer and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Company, the Issuer and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) Any Lender may at any time, without notice to or the consent of the Administrative Agent, sell participations to any Person (other than a natural person or the Company or any of the Company's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Credit-Linked Deposit and/or such Lender's Term Loans and/or Participations in Credit Obligations owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Company, the Administrative Agent, the Issuer and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to Section 10.01 that directly affects such Participant. Subject to subsection (e) of this Section 10.07, the Company agrees that each Participant shall be entitled to the benefits of Sections 3.01, 3.02 and 3.03 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section 10.07. To the extent permitted by Law, each Participant also shall be entitled to the benefits of Section 2.09 as though it were a Lender.
- (e) A Participant shall not be entitled to receive any greater payment under Section 3.01 or 3.05 than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Company's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of Section 3.01 unless the Company is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Company, to comply with Section 10.15 as though it were a Lender.

- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) **Assignment of Credit-Linked Deposit**. The Credit-Linked Deposit funded by any Lender under Section 2.01(c)(ii) shall not be released in connection with any assignment of its Credit-Linked Deposit, but shall instead be purchased by the relevant assignee and continue to be held for application (if not already applied in accordance with the terms of Section 2.01(c)(ii) or 2.03) pursuant to Section 2.01(c)(ii) in respect of such assignee's obligations under the Credit-Linked Deposit assigned to it.

10.08 Confidentiality. For the purposes of this Section 10.08, "Confidential Information" means information delivered to the Administrative Agent, the Issuer or any Lender by or on behalf of the Company or any Subsidiary in connection with the transactions contemplated by or otherwise pursuant to this Agreement that is proprietary in nature and that was clearly marked or labeled or otherwise adequately identified when received by it as being confidential information of the Company or such Subsidiary, provided that such term does not include information that (a) was publicly known or otherwise known to it prior to the time of such disclosure, (b) subsequently becomes publicly known through no act or omission by it or any Person acting on its behalf, (c) otherwise becomes known to it other than through disclosure by the Company or any Subsidiary or (d) constitutes financial statements delivered to it under Section 6.01 that are otherwise publicly available. Each of the Administrative Agent, the Issuer and the Lenders will maintain the confidentiality of such Confidential Information delivered to or received by it in accordance with procedures adopted by it in good faith to protect confidential information of third parties delivered to it, provided that the Administrative Agent, the Issuer or any Lender may deliver or disclose Confidential Information to (i) its directors, officers, employees, agents, attorneys and Affiliates (to the extent such disclosure reasonably relates to the administration of the Credit Documents by the Administrative Agent, the Issuer or any Lender), (ii) its financial advisors and other professional advisors who agree to hold confidential the Confidential Information substantially in accordance with the terms of this Section 10.08, (iii) any other party to this Agreement, (iv) any Eligible Assignee to which any Lender sells or offers to sell its Credit-Linked Deposit and Credit Obligations or any part thereof or any Participation therein (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 10.08), (v) any Person from which it offers to purchase any security of the Company (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 10.08), (vi) any federal or state regulatory authority having jurisdiction over it, (vii) the National Association of Insurance Commissioners or any similar organization, or any nationally recognized rating agency that requires access to information about its investment portfolio, or (viii) any other Person to which such delivery or disclosure may be necessary or appropriate (w) to effect compliance with any law, rule, regulation or order applicable to it, (x) in response to any subpoena or other legal process, (y) in connection with any litigation to which it is a party or (z) if an Event of Default has occurred and is continuing, to the extent it may reasonably determine such delivery and disclosure to be necessary or appropriate in the enforcement or for

the protection of the rights and remedies under the Letter of Credit Linked Notes, the Term Notes and this Agreement. Each Eligible Assignee, by its execution of an Assignment and Assumption, will be deemed to have agreed to be bound by and to be entitled to the benefits of this Section 10.08 as though it were a party to this Agreement. Notwithstanding anything herein to the contrary, "Confidential Information" shall not include, and the Issuer, the Administrative Agent and each Lender may disclose without limitation of any kind, any information with respect to the "tax treatment" and "tax structure" (in each case, within the meaning of Treasury Regulation Section 1.6011-4) of the transactions contemplated hereby and all materials of any kind (including opinions or other tax analyses) that are provided to the Issuer, the Administrative Agent or such Lender relating to such tax treatment and tax structure; provided that with respect to any document or similar item that in either case contains information concerning the tax treatment or tax structure of the transaction as well as other information, this sentence shall only apply to such portions of the document or similar item that relate to the tax treatment or tax structure of the Credits, Advances, Term Loans and transactions contemplated hereby.

10.09 [Intentionally Omitted].

10.10 Representations and Warranties of Each Lender. Each Lender (for itself, and not for any other Lender) represents and warrants to the Issuer and the Company:

- (a) Purchase for Investment. Such Lender is an institutional "accredited investor" within the meaning of subparagraphs (1), (2), (3) or (7) of Rule 501(a) promulgated under the Securities Act. Such Lender represents that it is purchasing its Participation in this Agreement, its Letter of Credit Linked Note and its Term Notes, if any, for its own account or for one or more separate accounts maintained by it or for the account of one or more pension or trust funds and not with a view to the distribution thereof, provided that the disposition of its or their property shall at all times be within its or their control. Such Lender understands that the Letter of Credit Linked Notes and Term Notes have not been registered under the Securities Act and may be resold only if registered pursuant to the provisions of the Securities Act or if an exemption from registration is available, except under circumstances where neither such registration nor such an exemption is required by law, and that the Issuer, the Administrative Agent and the Company are not required to register the Letter of Credit Linked Notes and Term Notes.
- (b) Source of Funds. Such Lender represents that at least one of the following statements is an accurate representation as to each source of funds (a "Source") to be used by it to pay the purchase price of its Participation in this Agreement:
- (i) the Source is an "insurance company general account" (as the term is defined in the United States Department of Labor's Prohibited Transaction Exemption ("PTE") 95-60) in respect of which the reserves and liabilities (as defined by the annual statement for life insurance companies approved by the National Association of Insurance Commissioners (the "NAIC Annual Statement")) for the general account contract(s) held by or on behalf of any employee benefit plan together with the amount of the reserves and liabilities for the general account contract(s) held by or on behalf of any other employee benefit plans maintained by the same employer (or affiliate thereof as defined in PTE 95-60) or by the same employee organization in the general account do not exceed 10% of the total reserves and liabilities of the

general account (exclusive of separate account liabilities) plus surplus as set forth in the NAIC Annual Statement filed with such Lender's state of domicile; or

- (ii) the Source is a separate account that is maintained solely in connection with such Lender's fixed contractual obligations under which the amounts payable, or credited, to any employee benefit plan (or its related trust) that has any interest in such separate account (or to any participant or beneficiary of such plan (including any annuitant)) are not affected in any manner by the investment performance of the separate account; or
- (iii) the Source is either (A) an insurance company pooled separate account, within the meaning of PTE 90-1 or (B) a bank collective investment fund, within the meaning of the PTE 91-38 and, except as disclosed by such Lender to the Issuer and the Company in writing pursuant to this clause (iii), no employee benefit plan or group of plans maintained by the same employer or employee organization beneficially owns more than 10% of all assets allocated to such pooled separate account or collective investment fund; or
- (iv) the Source constitutes assets of an "investment fund" (within the meaning of Part V of PTE 84-14 (the "QPAM Exemption") managed by a "qualified professional asset manager" or "QPAM" (within the meaning of Part V of the QPAM Exemption), no employee benefit plan's assets that are included in such investment fund, when combined with the assets of all other employee benefit plans established or maintained by the same employer or by an affiliate (within the meaning of Section V(c)(1) of the QPAM Exemption) of such employer or by the same employee organization and managed by such QPAM, exceed 20% of the total client assets managed by such QPAM, the conditions of Section I(c) and (g) of the QPAM Exemption are satisfied, neither the QPAM nor a person controlling or controlled by the QPAM (applying the definition of "control" in Section V(e) of the QPAM Exemption) owns a 5% or more interest in the Issuer or the Company and (i) the identity of such QPAM and (ii) the names of all employee benefit plans whose assets are included in such investment fund have been disclosed to the Issuer and the Company in writing pursuant to this clause (iv); or
- (v) the Source constitutes assets of one or more "plans" (within the meaning of Part IV of PTE 96-23 (the "INHAM Exemption")) managed by an "inhouse asset manager" or "INHAM" (within the meaning of Part IV of the INHAM exemption), the conditions of Section I(a), (g) and (h) of the INHAM Exemption are satisfied, neither the INHAM nor a person controlling or controlled by the INHAM (applying the definition of "control" in Section IV(d)(3) of the INHAM Exemption) owns a 5% or more interest in the Issuer or the Company and (A) the identity of such INHAM and (B) the name(s) of the employee benefit plan(s) whose assets constitute the Source have been disclosed to the Issuer and the Company in writing pursuant to this clause (v); or
 - (vi) the Source is a governmental plan; or
- (vii) the Source is one or more employee benefit plans, or a separate account or trust fund comprised of one or more employee benefit plans, each of which has been identified to the Issuer and the Company in writing pursuant to this clause (vii); or

(viii) the Source does not include assets of any employee benefit plan, other than a plan exempt from the coverage of ERISA.

As used in this Section 10.10(b), the terms "employee benefit plan", "governmental plan", "party in interest" and "separate account" shall have the respective meanings assigned to such terms in Section 3 of ERISA.

- (c) Authorization; Enforceability. The execution, delivery and performance of this Agreement has been duly authorized by all necessary corporate action on the part of such Lender. This Agreement has been duly executed and delivered by such Lender and is the legally valid and binding obligation of such Lender, enforceable against such Lender in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles.
- (d) No Withholding. As of the date hereof and as of the date of any assignment under Section 10.07 (after giving effect thereto), no Lender is subject to United States federal withholding taxes on amounts payable pursuant to the terms of this Agreement. Each Lender has delivered to the Administrative Agent all forms required pursuant to Section 10.15(b) to ensure that the Administrative Agent shall not be required to effect any withholding with respect to any amounts payable pursuant to the terms of this Agreement.
- **10.11 Counterparts**. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- **10.12 Integration**. This Agreement, together with the other Credit Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. In the event of any conflict between the provisions of this Agreement and those of any other Credit Document, the provisions of this Agreement shall control; provided that the inclusion of supplemental rights or remedies in favor of the Administrative Agent, the Issuer or the Lenders in any other Credit Document shall not be deemed a conflict with this Agreement. Each Credit Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 10.13 Survival of Representations and Warranties. All representations and warranties made hereunder and in any other Credit Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent, the Issuer and each Lender, regardless of any investigation made by the Administrative Agent, the Issuer or any Lender or on their behalf and notwithstanding that the Administrative Agent, the Issuer or any Lender may have had notice or knowledge of any Default or Event of Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Credit Obligation or any other Credit Obligation shall remain unpaid or unsatisfied or any Credit shall remain outstanding.

10.14 Severability. If any provision of this Agreement or the other Credit Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Credit Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

10.15 Tax Forms. (a) (i) Each Lender and the Issuer that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code (a "Foreign Lender") shall deliver to the Administrative Agent, prior to receipt of any payment subject to withholding under the Code (or upon accepting an assignment of an interest herein), two duly signed completed copies of either IRS Form W-8BEN or any successor thereto (relating to such Foreign Lender and entitling it to an exemption from, or reduction of, withholding tax on all payments to be made to such Foreign Lender by the Company pursuant to this Agreement) or IRS Form W-8ECI or any successor thereto (relating to all payments to be made to such Foreign Lender by the Company pursuant to this Agreement), and, if necessary, such other evidence satisfactory to the Company and the Administrative Agent that such Foreign Lender is entitled to an exemption from, or reduction of, U.S. withholding tax, including any exemption pursuant to Section 881(c) of the Code. Thereafter and from time to time, each such Foreign Lender shall (A) promptly submit to the Administrative Agent such additional duly completed and signed copies of one of such forms (or such successor forms as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then current United States laws and regulations to avoid, or such evidence as is satisfactory to the Company and the Administrative Agent of any available exemption from or reduction of, United States withholding taxes in respect of all payments to be made to such Foreign Lender by the Company pursuant to this Agreement, (B) promptly notify the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (C) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Foreign Lender, and as may be reasonably necessary (including the re-designation of its Lending Office) to avoid any

(ii) Each Foreign Lender, to the extent it does not act or ceases to act for its own account with respect to any portion of any sums paid or payable to such Foreign Lender under any of the Credit Documents (for example, in the case of a typical participation by such Foreign Lender), shall deliver to the Administrative Agent on the date when such Foreign Lender ceases to act for its own account with respect to any portion of any such sums paid or payable, and at such other times as may be necessary in the determination of the Administrative Agent (in the reasonable exercise of its discretion), (A) two duly signed completed copies of the forms or statements required to be provided by such Foreign Lender as set forth above, to establish the portion of any such sums paid or payable with respect to which such Foreign Lender acts for its own account that is exempt from, or subject to a reduced rate of, U.S. withholding tax, and (B) two duly signed completed copies of IRS Form W-8IMY (or any successor thereto), together with any information such Foreign Lender chooses to transmit with such form, and any other

certificate or statement of exemption required under the Code, to establish that such Foreign Lender is not acting for its own account with respect to a portion of any such sums payable to such Foreign Lender.

- (iii) The Company shall not be required to pay any additional amount to any Foreign Lender under Section 3.01 (A) with respect to any Taxes required to be deducted or withheld on the basis of the information, certificates or statements of exemption such Foreign Lender transmits with an IRS Form W-8IMY pursuant to this Section 10.15(a) or (B) if such Foreign Lender shall have failed to satisfy the foregoing provisions of this Section 10.15(a); provided that if such Foreign Lender shall have satisfied the requirement of this Section 10.15(a) on the date such Foreign Lender became a Lender or ceased to act for its own account with respect to any payment under any of the Credit Documents, nothing in this Section 10.15(a) shall relieve the Company of its obligation to pay any amounts pursuant to Section 3.01 in the event that, as a result of any change in any applicable law, treaty or governmental rule, regulation or order, or any change in the interpretation, administration or application thereof, such Foreign Lender is no longer properly entitled to deliver forms, certificates or other evidence at a subsequent date establishing the fact that such Foreign Lender or other Person for the account of which such Foreign Lender receives any sums payable under any of the Credit Documents is not subject to withholding or is subject to withholding at a reduced rate.
- (iv) The Administrative Agent may, without reduction, withhold any Taxes required to be deducted and withheld from any payment under any of the Credit Documents with respect to which the Company is not required to pay additional amounts under this Section 10.15(a) or Section 3.01.
- (b) Upon the request of the Administrative Agent, the Issuer and each Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Administrative Agent two duly signed completed copies of IRS Form W-9. If the Issuer or such Lender fails to deliver such forms, then the Administrative Agent may withhold from any interest payment to the Issuer or such Lender an amount equivalent to the applicable back-up withholding tax imposed by the Code, without reduction.
- (c) If any Governmental Authority asserts that the Administrative Agent did not properly withhold or backup withhold, as the case may be, any tax or other amount from payments made to or for the account of the Issuer or any Lender, the Issuer or such Lender shall indemnify the Administrative Agent therefor, including all penalties and interest, any taxes imposed by any jurisdiction on the amounts payable to the Administrative Agent under this Section, and costs and expenses (including reasonable Attorney Costs) of the Administrative Agent. The obligation of the Issuer and the Lenders under this Section shall survive the termination of the Credit-Linked Deposits, repayment of all Credit Obligations and all other amounts due hereunder and the resignation or replacement of the Administrative Agent.

10.16 Governing Law. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE; PROVIDED THAT THE ADMINISTRATIVE AGENT, THE ISSUER AND EACH LENDER SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW.

ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK OR OF THE UNITED STATES FOR SOUTHERN DISTRICT OF SUCH STATE, AND BY EXECUTION AND DELIVERY OF THIS AGREEMENT, THE COMPANY CONSENTS, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THOSE COURTS. THE COMPANY IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF ANY CREDIT DOCUMENT OR OTHER DOCUMENT RELATED THERETO.

10.17 Waiver of Right to Trial by Jury. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY CREDIT DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY CREDIT DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

10.18 Time of the Essence. Time is of the essence of the Credit Documents.

10.19 Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event. (a) For purposes of this Section 10.19, (i) the term "Credit Rating Downgrade Event" means that the credit ratings established by S&P and Moody's with respect to the senior, unsecured long-term, non-credit enhanced debt securities ("Long-Term Debt") of Bank of America Corporation have been lowered below "A-" by S&P and below "A3" by Moody's or that the Long-Term Debt of Bank of America Corporation has ceased to be rated by S&P and Moody's for a period of more than ten (10) consecutive Business Days, in each case as evidenced by the public announcement thereof; provided that if the credit ratings established for the Long-Term Debt of Bank of America Corporation by either S&P or Moody's are unavailable for any reason, the rating established by the other rating agency shall be used for purposes of this Section 10.19; and (ii) the term "Deposit Account Interest Non-Payment Event" means that an interest payment in respect of the Credit-Linked Deposits payable pursuant to Section 2.01(j)(i) shall not have been made by Bank of America and distributed by the Administrative Agent to the Lenders pursuant to Section 2.01(j)(i) within five (5) Business Days of when due. The Administrative Agent will promptly notify each Lender (A) after officers within the agency management group (or similar group performing the functions of the present agency

management group) of Bank of America obtain knowledge, if a Credit Rating Downgrade Event has occurred, and (B) if a Deposit Interest Non-Payment Event has occurred.

- (b) In the event that a Credit Rating Downgrade Event or a Deposit Account Interest Non-Payment Event has occurred and is continuing, the Required Lenders may, by notice to the Administrative Agent and the Company (each a "Required Notice"), identify the Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event which then exists, and request that either of the following actions be taken, and upon such request the Company, the Administrative Agent, the Issuer and the Lenders agree to use reasonable commercial efforts to cause such action to be taken:
- (i) A new deposit account (the "New Account") shall be opened by the Administrative Agent at a bank which has a credit rating of at least "A+" from S&P and "A1" from Moody's (the "New Depository Bank"). The New Depository Bank, the Administrative Agent, the Issuer, the Company and the Lenders shall execute such documents as may be reasonably necessary or appropriate so that the Issuer is granted by the Administrative Agent and the Lenders a first priority perfected security interest in the New Account securing the Participations. Upon execution and delivery of such documents (including customary legal opinions), and evidence reasonably satisfactory to the Issuer of perfection with first priority of such security interest, all funds then held in the Credit-Linked Deposit Account shall be transferred in immediately available funds to the New Depository Bank for credit to the New Account. Thereafter, the New Account shall be deemed to be the "Credit-Linked Deposit Account" under this Agreement and the provisions of this Agreement relating to the Credit-Linked Deposit Account shall apply *mutatis mutandis* to the New Account. Except for the transfer of the Credit-Linked Deposit Account as provided above, the rights of the Administrative Agent, on behalf of the Issuer and the Lenders, in and to the Credit-Linked Deposit Account hereunder shall remain unchanged and in full force and effect.

In the event that the interest payable by the New Depository Bank on the funds on deposit in the New Account is less than the interest set forth in Section 2.01(j)(i) of this Agreement with respect to the Total Credit-Linked Deposit, then the Company shall pay such additional amounts at such times as may be required so that the Lenders receive no less than the interest payable under Section 2.01(j)(i), at the times set forth in Section 2.01(j)(i). If a Deposit Account Interest Non-Payment Event has triggered the procedures set forth in this Section 10.19(b), then Bank of America shall (A) indemnify the Company for all additional amounts the Company pays pursuant to the immediately preceding sentence and (B) reimburse the Company and each Lender for all costs and expenses, including Attorney Costs, incurred in connection with such Deposit Account Interest Non-Payment Event.

The parties to this Agreement shall execute and deliver an amendment to this Agreement and such other documents as may be reasonably required to accomplish the foregoing. Notwithstanding anything to the contrary in this Section 10.19(b)(i), the documentation, terms and conditions utilized to establish and grant a security interest in the New Account and the other documentation (including legal opinions) and actions contemplated herein (including any amendment of this Agreement) shall be reasonably satisfactory in form and substance to the Company, the Issuer, the Administrative Agent and each Lender.

(ii) The Administrative Agent and the Lenders shall cause to be established, and beneficially owned by the Lenders, a trust (the "Trust") which shall in turn establish a securities account (the "Securities Account") at a securities intermediary (the "Securities Intermediary") having a credit rating of at least "A+" from S&P and at least "A1" from Moody's. The Securities Intermediary, the Administrative Agent, the Issuer, the Company and the Lenders shall execute such documents as may be reasonably necessary or appropriate so that the Issuer is granted by the Trust, the Administrative Agent and the Lenders, as appropriate, a first priority perfected security interest in the Securities Account securing the Participations. Upon execution and delivery of such documents (including customary legal opinions), and evidence reasonably satisfactory to the Issuer of perfection with first priority of such security interest, all funds then held in the Credit-Linked Deposit Account shall be transferred in immediately available funds to the Securities Intermediary for credit to the Securities Account. Such funds shall be invested in such LIBOR-based short-term debt investments, having a credit rating of at least "A+" from S&P and at least "A1" from Moody's as may be reasonably agreed upon by the Required Lenders, the Administrative Agent, the Issuer and the Company. Thereafter, the Securities Account shall be deemed to be the "Credit Linked Deposit Account" under this Agreement and the provisions of this Agreement relating to the Credit-Linked Deposit Account shall apply *mutatis mutandis* to the Securities Account. Except for the transfer of funds from the Credit-Linked Deposit Account to the Securities Account as provided above, the rights of the Administrative Agent, on behalf of the Issuer and the Lenders, in and to the Securities Account hereunder shall remain unchanged and in full force and effect.

In the event that the interest payable with respect to the investments in the Securities Account is less than the interest set forth in Section 2.01(j)(i) with respect to the Total Credit-Linked Deposit, then the Company shall pay such additional amounts at such times as may be required so that the Lenders receive no less than the interest payable under Section 2.01(j)(i), at the times set forth in Section 2.01(j)(i). If a Deposit Account Interest Non-Payment Event has triggered the procedures set forth in this Section 10.19(b), then Bank of America shall (A) indemnify the Company for all additional amounts the Company pays pursuant to the immediately preceding sentence and (B) reimburse the Company and each Lender for all costs and expenses, including Attorney Costs, incurred in connection with such Deposit Account Interest Non-Payment Event.

The parties to this Agreement shall execute and deliver an amendment to this Agreement and such other documents as may be reasonably required to accomplish the foregoing. Notwithstanding anything to the contrary in this Section 10.19(b)(ii), the documentation, terms and conditions utilized to establish the Trust and Securities Account and to grant a security interest in and to the Security Account and the other documentation (including legal opinions) and actions contemplated herein (including any amendment of this Agreement) shall be reasonably satisfactory in form and substance to the Company, the Issuer, the Administrative Agent and each Lender.

(iii) The Company hereby agrees to use commercially reasonable efforts from time to time to inform the Administrative Agent, which shall promptly inform the Lenders upon receipt of such information, of the status of the transactions contemplated in Sections 10.19(b)(i) or (ii), as applicable. In the event that neither of the transactions contemplated in subsections (i) or (ii) above, or transactions substantially similar to the transactions in such sections, are consummated in form and substance reasonably satisfactory to the Issuer, the Administrative Agent, the Company and each Lender within 120 days (or such longer period as the Company and the Required Lenders may mutually agree) after the receipt by the Company of a Required Notice, then the Required Lenders may, by notice to the Administrative Agent, terminate the credit facility established under Section 2.01. The Required Lenders may, within 90 days of issuing a Required Notice, provide the Company written notice of their election to retract such Required Notice, but shall have no right to retract the Required Notice after such 90 day period. Upon delivery of a notice of retraction, the actions required by Section 10.19(b)(i) or (ii) shall no longer be required to be taken in connection with the event described in the Required Notice. After delivery of a retraction notice or in the event the credit facility is not terminated at the end of the 120 day period, the Required Lenders may not deliver another Required Notice under this Section 10.19 unless the credit ratings from S&P or Moody's on the Long-Term Debt of Bank of America are further lowered. Upon such termination, all funds then or at any time thereafter held in the Credit-Linked Deposit Account in excess of the sum of the then Outstanding Amount of all Credit Obligations (other than Advances and Term Loans) shall be withdrawn from the Credit-Linked Deposit Account by the Administrative Agent and distributed by the Administrative Agent to each Lender in accordance with its Pro Rata Share, subject to any notice required pursuant to the Money Market Account Agreement; provided that the Company may, pursuant to Section 2.03(b), request that the Lenders make Term Loans as provided in Section 2.03(b)(i); and with respect to all Term Loans (whether then outstanding or thereafter made), the restriction that Term Loans cannot be prepaid until after the second anniversary of the Closing Date set forth in Section 2.03(b)(vi) shall continue to apply.

ARTICLE XI HOLDINGS GUARANTY

11.01 Guaranty. For value received and hereby acknowledged and as an inducement to the Lenders and the Issuer to make the Credit Extensions available to the Company, Holdings hereby unconditionally and irrevocably guarantees (a) the full punctual payment when due, whether at stated maturity, by acceleration or otherwise, of all Credit Obligations of the Company now or hereafter existing whether for principal, interest, fees, expenses or otherwise, and (b) the strict performance and observance by the Company of all agreements, warranties and covenants applicable to the Company in the Credit Documents and (c) the obligations of the Company under the Credit Documents (such Credit Obligations collectively being hereafter referred to as the "Guaranteed Obligations"). The foregoing guaranty is a guaranty of payment and not of collection. For purposes of this Article XI, the obligations of Holdings under this Article XI are referred to as the "Holdings Guaranty."

11.02 Guaranty Absolute. Holdings guarantees that the Guaranteed Obligations will be paid strictly in accordance with the terms hereof, regardless of any law, regulation or order now or hereafter in effect in any jurisdiction affecting any of such terms or the rights of any

Lender, the Issuer or the Administrative Agent with respect thereto. The liability of Holdings under the Holdings Guaranty with regard to the Guaranteed Obligations shall be absolute and unconditional irrespective of

- (a) any change in the time, manner or place of payment of, or in any other term of, all or any of its Guaranteed Obligations or any other amendment or waiver of or any consent to departure from this Agreement or any other Credit Document (with regard to such Guaranteed Obligations);
 - (b) any release or amendment or waiver of or consent to departure from any other guaranty for all or any of its Guaranteed Obligations;
 - (c) any change in ownership of the Company;
 - (d) any acceptance of any partial payment(s) from the Company or Holdings; or
- (e) any other circumstance similar or dissimilar to the foregoing which might otherwise constitute a defense available to, or a discharge of, the Company in respect of its Credit Obligations under any Credit Document.

The Holdings Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any Guaranteed Obligation is rescinded or must otherwise be returned by the Lenders, the Issuer or the Administrative Agent upon the insolvency, bankruptcy or reorganization of the Company or otherwise, all as though such payment had not been made.

11.03 Effectiveness; Enforcement. The Holdings Guaranty shall be effective and shall be deemed to be made with respect to each Credit Extension as of the time it is made, issued or extended. No invalidity, irregularity or unenforceability by reason of any bankruptcy or similar law, or any law or order of any government or agency thereof purporting to reduce, amend or otherwise affect any liability of the Company, and no defect in or insufficiency or want of powers of the Company or irregular or improperly recorded exercise thereof, shall impair, affect, be a defense to or claim against such guaranty. The Holding Guaranty is a continuing guaranty and shall (a) survive any termination of this Agreement, and (b) remain in full force and effect until payment in full of, and performance of, all Guaranteed Obligations and all other amounts payable under this Agreement. The Holdings Guaranty is made for the benefit of the Administrative Agent, the Issuer and the Lenders and their successors and assigns, and may be enforced from time to time as often as occasion therefor may arise and without requirement on the part of the Administrative Agent, the Issuer or the Lenders first to exercise any rights against the Company, or to resort to any other source or means of obtaining payment of any of the said obligations or to elect any other remedy.

11.04 Waiver. Except as otherwise specifically provided in any of the Credit Documents, Holdings hereby waives promptness, diligence, protest, notice of protest, all suretyship defenses, notice of acceptance and any other notice with respect to any of its Guaranteed Obligations and the Holdings Guaranty and any requirement that the Lenders, the Issuer or the Administrative Agent protect, secure, perfect any security interest or Lien or any property subject thereto or exhaust any right or take any action against the Company or any other

Person. Holdings also irrevocably waives, to the fullest extent permitted by law, all defenses which at any time may be available to it in respect of its Guaranteed Obligations by virtue of any statute of limitations, valuation, stay, moratorium law or other similar law now or hereafter in effect.

11.05 Expenses. Holdings hereby promises to reimburse (a) the Administrative Agent, the Issuer and the Lenders for all reasonable out-of-pocket fees and disbursements (including all reasonable Attorneys Costs), incurred or expended in connection with the preparation, filing or recording, or interpretation of the Holdings Guaranty, the other Credit Documents to which Holdings is a party, or any amendment, modification, approval, consent or waiver hereof or thereof, and (b) the Administrative Agent, the Issuer and the Lenders and their respective affiliates for all reasonable out-of-pocket fees and disbursements (including reasonable Attorney Costs), incurred or expended in connection with the enforcement of its Guaranteed Obligations (whether or not legal proceedings are instituted). Holdings will pay any taxes (including any interest and penalties in respect thereof) other than the Lenders' taxes based on overall income or profits, payable on or with respect to the transactions contemplated by the Holdings Guaranty, Holdings hereby agreeing jointly and severally to indemnify each Lender with respect thereto.

11.06 Concerning Joint and Several Liability of Holdings.

- (a) Holdings hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the Company, with respect to the payment and performance of all of its Guaranteed Obligations (including, without limitation, any Guaranteed Obligations arising under this Article XI), it being the intention of the parties hereto that all such Guaranteed Obligations shall be the joint and several Guaranteed Obligations of Holdings and the Company without preferences or distinction among them.
- (b) If and to the extent that the Company shall fail to make any payment with respect to any of its Credit Obligations as and when due or to perform any of its Guaranteed Obligations in accordance with the terns thereof, then in each such event Holdings will make such payment with respect to, or perform, such Guaranteed Obligation.
- (c) The Guaranteed Obligations of Holdings under the provisions of this Article XI constitute full recourse obligations of Holdings enforceable against Holdings to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of this Agreement or any other circumstance whatsoever.
- (d) Except as otherwise expressly provided in this Agreement, Holdings hereby waives notice of acceptance of its joint and several liability, notice of any Credit Extensions made or issued under this Agreement, notice of any action at any time taken or omitted by the Administrative Agent, the Issuer or the Lenders under or in respect of any of the Guaranteed Obligations, and, generally, to the extent permitted by applicable law, all demands, notices and other formalities of every kind in connection with this Agreement. Holdings hereby assents to, and waives notice of, any extension or postponement of the time for the payment of any of the Guaranteed Obligations, the acceptance of any payment of any of the Guaranteed Obligations, the acceptance of any payment thereon, any waiver, consent or other action or acquiescence by the Administrative Agent, the Issuer or the Lenders at any time or times in

respect of any Default or Event of Default by the Company or Holdings in the performance or satisfaction of any term, covenant, condition or provision of this Agreement or any other Credit Document, any and all other indulgences whatsoever by the Administrative Agent, the Issuer or the Lenders in respect of any of the Guaranteed Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Guaranteed Obligations or the addition, substitution or release, in whole or in part, of the Company or Holdings. Without limiting the generality of the foregoing, Holdings assents to any other action or delay in acting or failure to act on the part of the Lenders, the Issuer or the Administrative Agent with respect to the failure by the Company or Holdings to comply with its respective Credit Obligations or Guaranteed Obligations, including, without limitation, any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with applicable laws or regulations thereunder, which might, but for the provisions of this Article XI, afford grounds for terminating, discharging or relieving Holdings, in whole or in part, from any of the Guaranteed Obligations under this Article XI, at he intention of Holdings that, so long as any of the Guaranteed Obligations hereunder remain unsatisfied, the Guaranteed Obligations of Holdings under this Article XI shall not be discharged except by performance and then only to the extent of such performance. The Guaranteed Obligations of Holdings under this Article XI shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, reconstruction or similar proceeding with respect to the Company or Holdings or the Lenders, the Issuer or the Administrative Agent. The joint and several liability of Holdings hereunder shall continue in full force and effect notwithstanding any absorption, merger, consolidation, amalgamation or any other change whatso

- (e) Holdings shall be liable under this Article XI only for the maximum amount of such liabilities that can be incurred under applicable law without rendering this Article XI voidable under applicable law relating to fraudulent conveyance and fraudulent transfer, and not for any greater amount. Accordingly, if any obligation under any provision under this Article XI shall be declared to be invalid or unenforceable in any respect or to any extent, it is the stated intention and agreement of Holdings, the Administrative Agent, the Issuer and the Lenders that any balance of the obligation created by such provision and all other obligations of Holdings under this Article XI to the Lenders, the Issuer or the Administrative Agent shall remain valid and enforceable, and that all sums not in excess of those permitted under applicable law shall remain fully collectible by the Lenders, the Issuer and the Administrative Agent from the Company or Holdings, as the case may be.
- (f) The provisions of this Article XI are made for the benefit of the Administrative Agent, the Issuer and the Lenders and their successors and assigns, and may be enforced in good faith by them from time to time against Holdings as often as occasion therefor may arise and without requirement on the part of the Administrative Agent, the Issuer or the Lenders first to marshal any of their claims or to exercise any of their rights against the Company or Holdings or to exhaust any remedies available to them against the Company or Holdings or to resort to any other source or means of obtaining payment of any of the obligations hereunder or to elect any other remedy. The provisions of this Article XI shall remain in effect until all of the Guaranteed Obligations shall have been paid in full or otherwise fully satisfied, any remaining Credit-Linked

Deposits in the Credit-Linked Deposit Account are withdrawn and distributed to the Lenders in accordance with their respective Pro Rata Shares, and all outstanding Credit shall have expired, matured or otherwise been terminated. If at any time, any payment, or any part thereof, made in respect of any of the Guaranteed Obligations, is rescinded or must otherwise be restored or returned by the Lenders, the Issuer or the Administrative Agent upon the insolvency, bankruptcy or reorganization of the Company or Holdings, or otherwise, the provisions of this Article XI will forthwith be reinstated in effect, as though such payment had not been made.

11.07 Waiver. Until the final payment and performance in full of all of the Credit Obligations, Holdings shall not exercise and Holdings hereby waives any rights Holdings may have against the Company arising as a result of payment by Holdings hereunder, by way of subrogation, reimbursement, restitution, contribution or otherwise, and will not prove any claim in competition with the Administrative Agent, the Issuer or any Lender in respect of any payment hereunder in any bankruptcy, insolvency or reorganization case or proceedings of any nature; Holdings will not claim any setoff, recoupment or counterclaim against the Company in respect of any liability of the Company to Holdings; and Holdings waives any benefit of and any right to participate in any collateral security which may be held by the Administrative Agent, the Issuer or any Lender.

11.08 Subrogation; Subordination. The payment of any amounts due with respect to any indebtedness of the Company for money borrowed or credit received now or hereafter owed to Holdings is hereby subordinated to the prior payment in full of all of the Credit Obligations. Holdings agrees that, after the occurrence of any default in the payment or performance of any of the Credit Obligations, Holdings will not demand, sue for or otherwise attempt to collect any such indebtedness of the Company to Holdings until all of the Credit Obligations shall have been paid in full. If, notwithstanding the foregoing sentence, Holdings shall collect, enforce or receive any amounts in respect of such indebtedness while any Credit Obligations are still outstanding, such amounts shall be collected, enforced and received by Holdings as trustee for the Lenders, the Issuer and the Administrative Agent and be paid over to the Administrative Agent at Default, for the benefit of the Lenders, the Issuer, and the Administrative Agent on account of the Credit Obligations without affecting in any manner the liability of Holdings under the other provisions hereof.

11.09 Release of Guaranty. The Administrative Agent, the Issuer and the Lenders hereby agree that the foregoing Holdings Guaranty shall remain in full force and effect until the occurrence of either (a) the consolidation or merger of Holdings into the Company or its successors, (b) the consolidation or merger of the Company or its successors into Holdings, or (c) the release of Holdings from all other guarantees of the Company's obligations under the Company's existing credit facilities or other senior indebtedness, at which time the foregoing Holdings Guaranty shall be concurrently released without any further action on the part of the Lenders, the Administrative Agent or the Issuer.

ARTICLE XII ENTIRE AGREEMENT

12.01 Entire Agreement. THIS AGREEMENT AND THE OTHER CREDIT DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND

MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

(remainder of page intentionally left blank)

IN WITNESS WHEREOF, the parties hereto have caused this Five Year Letter of Credit and Term Loan Agreement to be duly executed as of the date first above written.

WASTE MANAGEMENT, INC.

By: /s/ Ronald H. Jones

Name: Ronald H. Jones

Title: Vice President & treasurer

WASTE MANAGEMENT HOLDINGS, INC.

(for the limited purposes of Article XI)

By: /s/ Ronald H. Jones

Name: Ronald H. Jones

Title: Vice President & Treasurer

By: /s/ Jay Clement

Name: Jay Clement Title: Assistant Treasurer

BANK OF AMERICA, N.A., as

Administrative Agent and Issuer

By: /s/ Brian D. Corum

Name: Brian D. Corum Title: Managing Director

LENDERS:	•
----------	---

TIAA GLOBAL MARKETS, INC.

By: /s/ Helen Armbrust

Name: Helen Armbrust Title: Managing Director

\$175,000,000

SEVEN YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT

Dated as of June 30, 2003

among

WASTE MANAGEMENT, INC.

WASTE MANAGEMENT HOLDINGS, INC.

BANK OF AMERICA, N.A.,

as Administrative Agent

BANK OF AMERICA, N.A.,

as Letter of Credit Issuer

and

The Lenders Party Hereto

BANC OF AMERICA SECURITIES LLC

Sole Lead Arranger and Sole Book Manager



TABLE OF CONTENTS

ARTICLE I	DEFINITIONS AND ACCOUNTING TERMS	Page 1
1.01	Defined Terms	1
1.02	Other Interpretive Provisions	17
1.03	Accounting Terms	17
1.04	Rounding	18
1.05	References to Agreements and Laws	18
1.06	Credit Amounts	18
ARTICLE II	THE CREDIT-LINKED DEPOSITS AND CREDIT EXTENSIONS	18
2.01	Credits	18
2.02	Cash Collateral	27
2.03	Reduction of Total Credit-Linked Deposits; Conversion to Funded Term Loans	28
2.04	Interest on Borrowings Other Than Term Loans	30
2.05	Fees and Other Charges	31
2.06	Computation of Interest and Fees	32
2.07	Evidence of Debt	32
2.08	Payments Generally	33
2.09	Sharing of Payments	33
ARTICLE III	TAXES AND YIELD PROTECTION	34
3.01	Taxes	34
3.02		35
3.03	Increased Cost and Reduced Return; Capital Adequacy or Risk-Based Capital	36
3.04	Illegality	36
3.05	Funding Losses Matters Applicable to All Requests for Compensation	37
3.06	Matters Applicable to All Requests for Compensation Survival	
3.07		37
=	Replacement of Lenders	37
ARTICLE IV	CONDITIONS PRECEDENT TO CREDIT EXTENSIONS	38
4.01	Conditions of Credit-Linked Deposits and Initial Credit Extension	38
4.02	Conditions to all Credit Extensions by the Issuer	40
4.03	Conditions to Term Loans	41
ARTICLE V	REPRESENTATIONS AND WARRANTIES	41
5.01	Organization; Power and Authority	41
5.02	Authorization, etc.	42
5.03	Disclosure	42
5.04	Organization and Ownership of Shares of Subsidiaries; Affiliates	43
5.05	Financial Statements	43
5.06	Compliance with Laws, Other Instruments, etc.	44
5.07	Governmental Authorizations, etc.	44
5.08	Litigation; Observance of Agreements, Statutes and Orders	44
5.09	Taxes	45
5.10	Title to Property; Leases	45
5.11	Licenses, Permits, etc.	45

TABLE OF CONTENTS (Continued)

5.12	Compliance with ERISA	Page 46
5.13	Private Offering by Company	46
5.14	Use of Proceeds; Margin Regulations	47
5.15	Existing Indebtedness	47
5.16	Foreign Assets Control Regulations, etc.	47
5.17	Status under Certain Statutes	48
5.18	Environmental Matters	48
5.19	Compliance With Issuance Limits	48
5.20	Tax Shelter Regulations	48
ARTICLE VI	AFFIRMATIVE COVENANTS	49
6.01	Financial and Business Information	49
6.02	Officer's Certificate	51
6.03	Inspection	52
6.04	Compliance with Laws	52
6.05	Insurance	52
6.06	Maintenance of Properties	52
6.07	Payment of Taxes and Claims	52
6.08	Corporate Existence, etc.	53
6.09	Books and Records	53
6.10	Purpose of Credits	53
ARTICLE VII	NEGATIVE COVENANTS	53
7.01	Transactions with Affiliates	53
7.02	Merger, Consolidation, Sale of Assets, etc.	54
7.03	Liens	56
7.04	Limitation on Consolidated Debt	58
7.05	Limitation on Priority Debt	58
7.06	Nature of Business	58
7.07	Designation of Unrestricted Subsidiaries	58
7.08	Margin Stock	59
ARTICLE VIII	EVENTS OF DEFAULT AND REMEDIES	59
8.01	Events of Default	59
8.02	Remedies Upon Event of Default	61
8.03	Application of Funds	62
ARTICLE IX	ADMINISTRATIVE AGENT	63
9.01	Appointment and Authorization of Administrative Agent	63
9.02	Delegation of Duties	63
9.03	Liability of Administrative Agent	64
9.04	Reliance by Administrative Agent	64
9.05	Notice of Default	64
9.06	Credit Decision; Disclosure of Information by Administrative Agent	65
9.07	Indemnification of Administrative Agent	65
9.08	Administrative Agent in its Individual Capacity	67
9.09	Successor Administrative Agent	67

TABLE OF CONTENTS (Continued)

		Page
ARTICLE X	MISCELLANEOUS	67
10.01	Amendments, Etc.	67
10.02	Notices and Other Communications; Facsimile Copies	69
10.03	No Waiver; Cumulative Remedies	70
10.04	Attorney Costs, Expenses and Taxes	70
10.05	Indemnification by the Company	71
10.06	Payments Set Aside	71
10.07	Successors and Assigns	72
10.08	Confidentiality	74
10.09	[Intentionally Omitted]	75
10.10	Representations and Warranties of Each Lender	75
10.11	Counterparts	77
10.12	Integration	77
10.13	Survival of Representations and Warranties	77
10.14	Severability	78
10.15	Tax Forms	78
10.16	Governing Law	80
10.17	Waiver of Right to Trial by Jury	80
10.18	Time of the Essence	80
10.19	Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event	80
ARTICLE XI	HOLDINGS GUARANTY	83
11.01	Guaranty	83
11.02	Guaranty Absolute	84
11.03	Effectiveness; Enforcement	84
11.04	Waiver	84
11.05	Expenses	85
11.06	Concerning Joint and Several Liability of Holdings	85
11.07	Waiver	87
11.08	Subrogation; Subordination	87
11.09	Release of Guaranty	87
ARTICLE XII	ENTIRE AGREEMENT	88
12.01	Entire Agreement	88
SIGNATURES		S-1

SCHEDULES

A	Applicable Rates; Interest on Credit-Linked Deposits
1.01	Existing Credits
2.01	Credit-Linked Deposits and Pro Rata Shares
5.04	Organization and Ownership Shares of Subsidiaries; Affiliates
5.05	Financial Statements
5.08	Litigation; Observance of Agreements, Statutes and Orders
5.11	Licenses, Permits, etc.
5.15	Existing Indebtedness
5.17	Status under Certain Statutes
7.03	Liens
10.02	Administrative Agent's Office; Certain Addresses for Notices

EXHIBITS

Form of

A	Request for Credit Extension
В	Assignment and Assumption
C-1	Opinion of Internal Counsel to the Company and Holdings
C-2	Opinion of Morrison & Foerster LLP, counsel to the Administrative Agent and the Issuer
C-3	Opinion of Assistant General Counsel of Bank of America Legal Department, internal counsel to
	the Administrative Agent and the Issuer
C-4	Opinion of Special Counsel to the Lenders
D	Letter of Credit Linked Note
E	Request for Term Loans
F	Term Note
G	Money Market Account Agreement
Н	Term Note Opinion

SEVEN YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT

This SEVEN YEAR LETTER OF CREDIT AND TERM LOAN AGREEMENT ("Agreement") is entered into as of June 30, 2003, among WASTE MANAGEMENT, INC., a Delaware corporation (the "Company"), WASTE MANAGEMENT HOLDINGS, INC., a Delaware corporation and wholly-owned Subsidiary of the Company ("Holdings") (executing this Agreement for the limited purposes set forth in Article XI hereof), BANK OF AMERICA, N.A., as issuer of letters of credit (the "Issuer"), the financial institutions having a Credit-Linked Deposit set forth opposite their names on Schedule 2.01 hereto under the heading "Credit-Linked Deposit" (collectively, the "Lenders" and individually, a "Lender"), and BANK OF AMERICA, N.A., as Administrative Agent.

The Company has requested that the Issuer and the Lenders provide a letter of credit facility, and the Issuer and the Lenders are willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I DEFINITIONS AND ACCOUNTING TERMS

1.01 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth below:

"Adjusted Consolidated EBITDA" means for any period, the total of the following calculated without duplication on a consolidated basis for such period: (i) Consolidated EBITDA, plus (ii) on a pro forma basis, the EBITDA of each Person acquired, or, as applicable, the EBITDA attributable to the assets acquired from any Person, by the Company or any Restricted Subsidiary in the subject period, for any portion of such subject period occurring prior to the date of the acquisition by the Company or any Restricted Subsidiary of such Person or related assets; provided, further, that any calculation pursuant to clause (ii) above of pro forma EBITDA for any subject period which gives effect to asset acquisitions and/or stock purchases during the subject period on a pro forma basis as provided above shall in any event be determined by the Company in accordance with sound financial practice in accordance with GAAP and on the basis, to the extent available, of appropriate financial statements and tax returns for the applicable period, and shall be set forth in a certificate of a Senior Financial Officer of the Company accompanied by calculations in reasonable detail showing the manner of determination thereof, which certificate shall be furnished to the Administrative Agent, the Issuer and each Lender not later than the certificate required to be furnished by the Company in respect of the fiscal period in which such date of determination occurs pursuant to Section 6.02 of this Agreement.

"Administrative Agent" means Bank of America in its capacity as administrative agent under any of the Credit Documents, or any successor administrative agent.

"Administrative Agent's Office" means the Administrative Agent's address and, as

appropriate, account as set forth on Schedule 10.02, or such other address or account as the Administrative Agent may from time to time specify by notice to the Company, the Issuer and the Lenders.

- "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.
- "Advance" means, with respect to each Lender, such Lender's funding of its participation in any Borrowing (other than Term Loans) in accordance with its Pro Rata Share.
- "Affiliate" means, at any time, (a) with respect to the Company and each of its Restricted Subsidiaries, any Person (other than the Company and each of its Restricted Subsidiaries) that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, the Company, (b) with respect to any Person (other than the Company and each of its Restricted Subsidiaries), any other Person that at such time directly or indirectly through one or more intermediaries Controls, or is Controlled by, or is under common Control with, such first Person, and (c) any Person beneficially owning or holding, directly or indirectly, 10% or more of any class of voting or equity interests of the Company or any Subsidiary of the Company or any corporation of which the Company and its Subsidiaries beneficially own or hold, in the aggregate, directly or indirectly, 10% or more of any class of voting or equity interests. As used in this definition, "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise. Unless the context otherwise clearly requires, any reference to an 'Affiliate' is a reference to an Affiliate of the Company.
 - "Agent Indemnified Liabilities" has the meaning specified in Section 9.07.
- "Agent-Related Persons" means the Administrative Agent or the Issuer, together with its Affiliates (including, in the case of Bank of America in its capacity as the Administrative Agent, BAS), and the officers, directors, employees, agents and attorneys-in-fact of such Persons and Affiliates.
 - "Agreement" means this Seven Year Letter of Credit and Term Loan Agreement.
 - "Applicable Rate" has the meaning set forth on Schedule A annexed hereto.
 - "Arranger" means BAS, in its capacity as sole lead arranger and sole book manager.
 - "Assignment and Assumption" means an Assignment and Assumption substantially in the form of Exhibit B.
 - "Attorney Costs" means and includes all fees, expenses and disbursements of any law firm or other external counsel.
- "Audited Financial Statements" means the audited consolidated balance sheet of the Company and its Subsidiaries for the fiscal year ended December 31, 2002, and the related

consolidated statements of income or operations, shareholders' equity and cash flows for such fiscal year of the Company and its Subsidiaries, including the notes thereto.

"Authorized Representative" means any employee of a Credit Party involved principally in the financial administration, controllership or treasury function of such Credit Party who is authorized or designated as an Authorized Representative, as evidenced by the certificate delivered pursuant to Section 4.01(a)(iv) or from time to time by a certificate delivered by a Senior Financial Officer of such Credit Party to the Administrative Agent in form and substance satisfactory to it. Any document delivered hereunder that is signed by an Authorized Representative of a Credit Party shall be conclusively presumed to have been authorized by all necessary corporate action on the part of such Credit Party and such Authorized Representative shall be conclusively presumed to have acted on behalf of such Credit Party.

"Auto-Reinstating Credit" means any Credit containing any provisions providing for automatic reinstatement of the stated amount after any drawing thereunder.

"Auto-Renewal Credit" has the meaning specified in Section 2.01(b)(iii).

"Bank of America" means Bank of America, N.A. and its successors.

"BAS" means Banc of America Securities LLC.

"Base Rate" means for any day a fluctuating rate per annum equal to the higher of (a) the Federal Funds Rate plus 1/2 of 1% and (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate." Such rate is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such rate announced by Bank of America shall take effect at the opening of business on the day specified in the public announcement of such change.

"Beneficial Ownership" has the meaning set forth in Rule 13d-3 promulgated by the SEC under the Securities Exchange Act.

"Borrowing" means, without duplication, an extension of credit resulting from a drawing under any Credit which has not been reimbursed on the applicable Reimbursement Date, including the making of Advances, or the making of a Term Loan, as the context may require.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, New York or Dallas, or (if the Administrative Agent's Office is not located in New York or Dallas) the state where the Administrative Agent's Office is located, and, if such day relates to the Credit-Linked Deposit Account or any LIBO Rate Term Loan or LIBO Rate Advance, means any such day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Capital Leases" or "Capitalized Leases" means leases under which the Company or any

of its Restricted Subsidiaries is the lessee or obligor, the discounted future rental payments under which are required to be capitalized on the balance sheet of the lessee or obligor in accordance with GAAP.

"Cash Collateral" has the meaning specified in Section 2.02.

"Cash Collateralize" has the meaning specified in Section 2.02.

"Closing Date" means the first date all the conditions precedent in Section 4.01 are satisfied or waived in accordance with Section 4.01 (or, in the case of Section 4.01(b), waived by the Person entitled to receive the applicable payment).

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and the rules and regulations promulgated thereunder from time to time.

"Company" has the meaning specified in the introductory paragraph hereto.

"Consolidated Debt" means, as of any date of determination, the total of all Indebtedness of the Company and its Restricted Subsidiaries determined on a consolidated basis in accordance with GAAP.

"Consolidated EBITDA" means, for any period, Consolidated Net Income or Consolidated Net Loss, as the case may be, for such period, plus the sum of (a) interest expense; (b) income tax expense; (c) depletion and depreciation expense; and (d) amortization expense; in each case, which were deductible in determining Consolidated Net Income or Consolidated Net Loss for such period, determined on a consolidated basis for the Company and its Restricted Subsidiaries in accordance with GAAP.

"Consolidated Net Income or Consolidated Net Loss" means, respectively, for any period, the net income or loss for such period determined on a consolidated basis for the Company and its Restricted Subsidiaries in accordance with GAAP.

"Consolidated Net Worth" means, as of any date of determination, the consolidated stockholders' equity of the Company and its Restricted Subsidiaries (excluding any amounts attributable to mandatorily redeemable Preferred Stock) determined as of such date.

"Consolidated Tangible Assets" means, as of any date of determination, Consolidated Total Assets less the sum of (i) Intangible Assets and (ii) all amounts representing write-up in the book value of any assets of the Company and its Restricted Subsidiaries resulting from a revaluation thereof.

"Consolidated Total Assets" means all assets of the Company and its Restricted Subsidiaries determined on a consolidated basis in accordance with GAAP.

"Credit" means any standby letter of credit issued or outstanding hereunder (including a so-called "direct pay" standby letter of credit) that is a Permitted Credit and shall include the Existing Credits.

"Credit Documents" means this Agreement, the Fee Letter, each Request for Credit Extension, each Issuer Document, each Request for Term Loan, each Letter of Credit Linked Note and each Term Note and each other document, instrument or agreement from time to time executed by the Company or any of its Subsidiaries, Holdings, or any secretary, assistant secretary, Responsible Officer or Authorized Representative thereof and delivered in connection with this Agreement.

"Credit Expiration Date" means the day that is seven (7) Business Days prior to the Maturity Date (or, if such day is not a Business Day, the next preceding Business Day).

"Credit Extension" means, (a) with respect to any Credit, the issuance thereof or extension of the expiry date thereof, or the renewal, reinstatement or increase of the amount thereof and (b) the making of any Term Loans.

"Credit-Linked Deposit" shall have the meaning specified in Section 2.01(c)(ii).

"Credit-Linked Deposit Account" shall mean the deposit account established by the Administrative Agent in its name and under its sole and exclusive control maintained at the office of Bank of America at its principal office in Chicago, Illinois, designated as the "Bank of America N.A. as Administrative Agent Waste Management Inc Seven Year LOC Facility Lender Credit Linked Notes Dep Acct Treasury" that shall be used solely for the purposes set forth in Sections 2.01(c) (ii) and (iii) and in Section 2.03(b)(iii), or such successor account established pursuant to Section 10.19 of this Agreement.

"Credit-Linked Lenders" means the Lenders under this Agreement and the lenders under the Ten Year Agreement and the Five Year Agreement.

"Credit Obligations" means, without duplication, as at any date of determination, the aggregate undrawn face amount of all outstanding Credits plus the aggregate amount of all Unreimbursed Amounts in respect of any Credits, and, without duplication, all outstanding Borrowings arising from any Credits, plus the aggregate amount of the Term Loans. For all purposes of this Agreement, if on any date of determination a Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Credit Parties" means, collectively, the Company and Holdings.

"Credit Rating Downgrade Event" has the meaning specified in Section 10.19(a)(i).

"Debtor Relief Laws" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means an event or condition the occurrence or existence of which would, with

the lapse of time or the giving of notice or both, become an Event of Default.

"Default Rate" means (a) with respect to obligations other than Advances and Term Loans, the Base Rate plus 2.00% per annum, and (b) with respect to Advances and Term Loans, the LIBO Rate plus 3.00% per annum, in each case to the fullest extent permitted by applicable Laws.

"Deposit Account Interest Non-Payment Event" has the meaning specified in Section 10.19(a)(ii).

"Disclosure Documents" means the Company's financial statements referred to in Section 5.05 and filings made by the Company or Holdings with the SEC that were publicly available prior to the date of this Agreement and which were provided to the Issuer and the Lenders.

"Dollar" and "\$" means lawful money of the United States.

"EBITDA" means, (without duplication) with respect to any Person, division or business and with respect to any period, the sum of (i) (a) net income for such period from continuing operations after extraordinary items (excluding gains or losses from disposition of assets); (b) consolidated interest charges for such period; (c) the amount of taxes, based on or measured by income, used or included in the determination of such net income; (d) the amount of depreciation and amortization expense deducted in determining such net income; and (e) any extraordinary losses or non-recurring expenses not requiring the expenditure of cash decreasing net income for such period, minus (ii) any extraordinary gains or non-recurring items increasing net income for such period.

"Eligible Assignee" means any Institutional Investor.

"Environmental Laws" means any and all federal, state, local, and foreign statutes, laws, regulations, ordinances, rules, judgments, orders, decrees, permits, concessions, grants, franchises, licenses, agreements or governmental restrictions relating to pollution and the protection of the environment or the release of any materials into the environment, including but not limited to those related to hazardous substances or wastes, air emissions and discharges to waste or public systems.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the rules and regulations promulgated thereunder from time to time in effect.

"ERISA Affiliate" means any trade or business (whether or not incorporated) that is treated as a single employer together with the Company under Section 414 of the Code.

"Escalating Credit" means a Credit which provides for a face amount that increases from time to time in accordance with its terms.

"Event of Default" has the meaning specified in Section 8.01.

"Existing Credits" means the standby letters of credit listed on Schedule 1.01.

"Facility Fee" has the meaning specified in Section 2.05(a).

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" has the meaning specified in Section 2.05(b).

"Five Year Agreement" means the Five Year Letter of Credit and Term Loan Agreement dated as of even date herewith (as amended, restated, supplemented or otherwise modified from time to time) among the Company, Holdings, each lender party thereto and Bank of America as administrative agent and issuer of letters of credit; provided, however, if all Credit Obligations (as such term is defined in the Five Year Agreement) are paid in full and the Five Year Agreement is terminated, then this definition of "Five Year Agreement" shall no longer be applicable to this Agreement.

"Foreign Lender" has the meaning specified in Section 10.15(a).

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Fee" has the meaning specified in Section 2.05(b).

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means

- (a) the government of
 - (i) the United States of America or any state or other political subdivision thereof, or

- (ii) any jurisdiction in which the Company or any Subsidiary conducts all or any part of its business, or which asserts jurisdiction over any properties of the Company or any Subsidiary, or
- (b) any entity exercising executive, legislative, judicial, regulatory or administrative functions of, or pertaining to, any such government.

"Guarantee" by any Person means any obligation, contingent or otherwise, of such Person directly or indirectly guaranteeing any Indebtedness of any other Person and, without limiting the generality of the foregoing, any obligation, direct or indirect, contingent or otherwise, of such Person (a) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness (whether arising by virtue of partnership arrangements, by agreement to keep-well, to purchase assets, goods, securities or services, to take-or-pay, or to maintain financial statement conditions or otherwise) or (b) entered into for the purpose of assuring in any other manner the obligee of such Indebtedness of the payment thereof or to protect such obligee against loss in respect thereof (in whole or in part); provided that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business. The term "Guarantee" used as a verb has a corresponding meaning.

"Guaranteed Obligations" has the meaning specified in Section 11.01.

"Hazardous Material" means any hazardous waste as defined by 42 U.S.C. §6903(5), any hazardous substance as defined by 42. U.S.C. §9601(14), any pollutant or contaminant as defined by 42 U.S.C. §9601(33) or any toxic substances, oil or other chemicals or substances regulated by the Environmental Laws, excluding household hazardous waste.

"Holdings" means Waste Management Holdings, Inc., a Delaware corporation and wholly-owned Subsidiary of the Company.

"Holdings Guaranty" means the meaning specified in Section 11.01.

"Indebtedness" means, collectively, without duplication, whether classified as Indebtedness, an investment or otherwise on the obligor's balance sheet, (a) all indebtedness for borrowed money, (b) all obligations for the deferred purchase price of property or services (other than trade payables incurred in the ordinary course of business which either (i) are not overdue by more than ninety (90) days, or (ii) are being disputed in good faith and for which adequate reserves have been established in accordance with GAAP), (c) all obligations evidenced by notes, bonds, debentures or other similar debt instruments, (d) all obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (e) all obligations, liabilities and indebtedness under Capitalized Leases, (f) all obligations, liabilities or indebtedness under drawn surety or performance bonds, or any other drawn bonding arrangement, (g) Guarantees with respect to all Indebtedness of others referred to in clauses (a) through (f) above, and (h) all Indebtedness of others referred to in clauses (a) through (f) above secured or supported by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured or supported by) any Lien on the property or assets of the Company or any Subsidiary,

even though the owner of the property has not assumed or become liable, contractually or otherwise, for the payment of such Indebtedness; provided that if a Permitted Receivables Transaction is outstanding and is accounted for as a sale of accounts receivable under GAAP, Indebtedness shall also include the additional Indebtedness, determined on a consolidated basis, which would have been outstanding had such Permitted Receivables Transaction been accounted for as a borrowing.

"Indemnified Liabilities" has the meaning specified in Section 10.05.

"Indemnitees" has the meaning specified in Section 10.05.

"INHAM Exemption" has the meaning specified in Section 10.10(b)(v).

"Institutional Investor" means (a) any original Lender or any Affiliate of a Lender, (b) any Lender holding more than \$5,000,000 of the Credit-Linked Deposits and/or Credit Obligations then outstanding, and (c) any bank, trust company, savings and loans association or other financial institution, any pension plan, any investment company, any insurance company, any broker or dealer, any Person (other than a natural Person) engaged in making, purchasing, holding or otherwise investing in commercial loans in the ordinary course of its business, or any other similar financial institution or entity, regardless of legal form.

"Intangible Assets" means goodwill, the purchase price of acquired assets in excess of the fair value thereof, trademarks, trade names, service marks, customer lists, brand names, copyrights, patents and licenses, and rights with respect to the foregoing.

"Interest Period" means, (a) as to the Credit-Linked Deposits, the period commencing on the date such Credit-Linked Deposit is made and ending on the date three months thereafter; (b) as to Advances and Term Loans, the period commencing on the date any such Advance or Term Loan is disbursed (except that in respect of Term Loans made from the proceeds of the Credit-Linked Deposits pursuant to Sections 2.03(b)(i)(A) or (B), the Interest Period shall be deemed to commence on the same day as the Interest Period commencement date for the Credit-Linked Deposits, and in respect of Term Loans made through the conversion of Advances pursuant to Section 2.03(b)(i)(C), the Interest Period shall be deemed to commence on the same day as the Interest Period commencement date for the Credit-Linked Deposits, if any, on the date of such conversion) and ending on the earlier of (x) the date three months thereafter and (y) the last day of the Interest Period with respect to the Credit-Linked Deposits; and (c) after the initial Interest Periods described in (a) and (b) above, as to the Credit-Linked Deposits, Advances and Term Loans, the period commencing on the last day of the immediately preceding Interest Period and ending on the date three months thereafter; provided that:

- (i) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;
- (ii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at

the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and

(iii) no Interest Period shall extend beyond the Maturity Date;

provided further that after the occurrence and during the continuance of an Event of Default the Required Lenders, acting through the Administrative Agent, may designate one or more Interest Periods for the Advances and/or the Term Loans which are shorter than that set forth above, such designation to be effective after the last day of the applicable existing Interest Period.

"IRS" means the United States Internal Revenue Service.

"ISP" means, with respect to any Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice (or such later version thereof as may be in effect at the time of issuance).

"Issuance Limits" has the meaning specified in Section 2.01(a)(i).

"Issuer" means Bank of America in its capacity as issuer of Credits hereunder, or any successors, and shall include any branch of the Issuer through which the Issuer may cause Credits to be issued hereunder.

"Issuer Documents" means with respect to any Credit issued by the Issuer, such Credit, any applications for issuance or amendment of such Credit, and any other document, agreement and instrument entered into by the Issuer and the Company or in favor of the Issuer and relating to any such Credit, including any of the Issuer's standard form documents for issuances and amendments of letters of credit and guarantees or other similar undertakings.

"Issuer-Related Persons" means as to the Issuer, together with its Affiliates, and the officers, directors, employees, agents and attorneys-in-fact of the Issuer and its Affiliates.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

"Lender" has the meaning specified in the introductory paragraph hereto.

"Lending Office" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Company and the Administrative Agent.

"Letter of Credit Linked Note" has the meaning specified in Section 2.07(b).

"LIBO Rate" means for any Interest Period:

- (a) the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate that appears on the page of the Telerate screen (or any successor thereto) that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (b) if the rate referenced in the preceding clause (a) does not appear on such page or service or such page or service shall not be available, the rate per annum equal to the rate determined by the Administrative Agent to be the offered rate on such other page or other service that displays an average British Bankers Association Interest Settlement Rate for deposits in Dollars (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period, determined as of approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, or
- (c) if the rates referenced in the preceding clauses (a) and (b) are not available, the rate per annum determined by the Administrative Agent as the rate of interest at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Total Credit-Linked Deposit or the average amount of the Advances of the Lenders or LIBO Rate Term Loans, as applicable, being made or continued and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 4:00 p.m. (London time) two Business Days prior to the first day of such Interest Period.

"LIBO Rate Term Loans" means Term Loans bearing interest based on the LIBO Rate.

"Lien" means, with respect to any Person, any mortgage, lien, pledge, charge, security interest or other encumbrance, or any interest or title of any vendor, lessor, lender or other secured party to or of such Person under any conditional sale or other title retention agreement or Capital Lease, upon or with respect to any property or asset of such Person (including in the case of stock, stockholder agreements, voting trust agreements and all similar arrangements).

"Long-Term Debt" has the meaning specified in Section 10.19(a).

"Material" or "Materially" means material or materially, as applicable, in relation to the business, operations, affairs, financial condition, assets, properties, or prospects of the Company and its Restricted Subsidiaries taken as a whole.

"Material Adverse Effect" means a material adverse effect on (a) the business, operations, affairs, financial condition, prospects, assets or properties of the Company and its Restricted Subsidiaries taken as a whole, (b) the ability of the Company to perform its obligations under this Agreement, the Letter of Credit Linked Notes or the Term Notes, or (c) the validity or enforceability of this Agreement, the Letter of Credit Linked Notes or the Term Notes.

"Material Debt" means any Indebtedness (other than any Indebtedness of the Company or Holdings in respect of the Credit Obligations) of any one or more of the Company, its Restricted

Subsidiaries and Holdings in an aggregate principal amount for all such Persons exceeding \$50,000,000.

"Maturity Date" means (a) June 30, 2010, or (b) such earlier date upon which the obligations of the Issuer and the Lenders to make Credit Extensions may be terminated in accordance with the terms hereof.

"Memorandum" has the meaning specified in Section 5.03.

"Money Market Account Agreement" means the Money Market Account Agreement between the Administrative Agent and Bank of America, as depository, substantially in the form of Exhibit G attached hereto, concerning the Credit-Linked Deposit Account.

"Moody's" means Moody's Investors Service, Inc., or any successor thereto that is a nationally-recognized rating agency.

"Multiemployer Plan" means any Plan that is a "multiemployer plan" (as such term is defined in Section 4001(a)(3) of ERISA).

"New Account" has the meaning specified in Section 10.19(b)(i).

"New Depository Bank" has the meaning specified in Section 10.19(b)(i).

"Nonrenewal Notice Date" has the meaning specified in Section 2.01(b)(iii).

"Officer's Certificate" means a certificate of a Senior Financial Officer or of any other officer of the Company whose responsibilities extend to the subject matter of such certificate.

"Other Taxes" has the meaning specified in Section 3.01(b).

"Outstanding Amount" means, with respect to any Credit Obligations on any date, the amount of such Credit Obligations on such date after giving effect to any Credit Extension occurring on such date and any other changes in the aggregate amount of the Credit Obligations as of such date, including as a result of any reimbursements of outstanding unpaid drawings under any Credits or any reductions in the maximum amount available for drawing under Credits taking effect on such date. Without limiting the provisions of Section 1.07, for avoidance of doubt, the portion of the Credit Obligations in respect of the undrawn face amount of any Escalating Credits shall be deemed to be the maximum aggregate amount available to be drawn under such Escalating Credits (after giving effect to all increases).

"Participant" has the meaning specified in Section 10.07(d).

"Participation" has the meaning specified in Section 2.01(c)(ii).

"PBGC" means the Pension Benefit Guaranty Corporation referred to and defined in ERISA or any successor thereto.

"Permitted Credit" means a Credit used directly or indirectly to cover (a) a default in the

performance of any non-financial or commercial obligations of the Company or any Subsidiary under specific contracts, including a Credit issued in favor of a bank or other surety who in connection therewith issues a guarantee or similar undertaking, performance bond, surety bond or other similar instrument that covers a default in the performance of any non-financial or commercial obligations under specific contract, or (b) the payment of any financial contractual obligation of the Company or any Subsidiary.

"Permitted Liens" has the meaning specified in Section 7.03.

"Permitted Receivables Transaction" means any sale or sales of, and/or securitization of, any accounts receivable of the Company and/or any of its Restricted Subsidiaries (the "Receivables") pursuant to which (a) the Company and its Restricted Subsidiaries realize aggregate net proceeds of not more than \$750,000,000 at any one time outstanding, including, without limitation, any revolving purchase(s) of Receivables where the maximum aggregate uncollected purchase price (exclusive of any deferred purchase price) for such Receivables at any time outstanding does not exceed \$750,000,000, and (b) which Receivables shall not be discounted more than 25%.

"Person" means any individual, trustee, corporation, general partnership, limited partnership, limited liability company, joint stock company, trust, unincorporated organization, bank, business association, firm, joint venture, Governmental Authority or other legal entity.

"Plan" means an "employee benefit plan" (as defined in Section 3(3) of ERISA) that is or, within the preceding five years, has been established or maintained, or to which contributions are or, within the preceding five years, have been made or required to be made, by the Company or any ERISA Affiliate or with respect to which the Company or any ERISA Affiliate may have any liability.

"Preferred Stock" means any class of capital stock of a corporation that is preferred over any other class of capital stock of such corporation as to the payment of dividends or the payment of any amount upon liquidation or dissolution of such corporation.

"Prepayment Premium" shall mean in connection with any prepayment of Advances, Term Loans or reduction of the available amount of the Total Credit-Linked Deposit an amount equal to the applicable percentage of the principal amount of the Advances, Term Loans or Total Credit-Linked Deposit so prepaid or reduced as follows: (i) on or after the second anniversary of the Closing Date and prior to the third anniversary of the Closing Date, a premium of 2%, and (ii) on or after the third anniversary of the Closing Date and prior to the fourth anniversary of the Closing Date, a premium of 1%.

"Priority Debt" means (without duplication) the sum of (a) unsecured Indebtedness of the Company's Restricted Subsidiaries other than (i) Indebtedness of the Company's Restricted Subsidiaries, including Holdings, listed on Schedule 5.15 and any extension, renewal or refinancing by such parties of such Indebtedness, provided that the terms of any such extension, renewal or refinancing are substantially the same as the terms and conditions in effect on the Closing Date, or more favorable to such parties, and the principal amount of such Indebtedness is not increased, (ii) Indebtedness owing to the Company or any other Restricted Subsidiary and

(iii) any unsecured Indebtedness of Holdings consisting of Guarantees of the Company's Indebtedness existing as of the Closing Date and, so long as the Holdings Guaranty remains in full force and effect, Indebtedness consisting of Guarantees of the Company's Indebtedness incurred thereafter, plus (b) Indebtedness with respect to drawn landfill closure bonds, plus (c) Indebtedness with respect to Permitted Receivables Transactions, plus (d) Indebtedness of the Company or any of its Restricted Subsidiaries secured by a Lien not permitted by clauses (a) through (l) of Section 7.03.

"Pro Rata Share" means, with respect to each Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the Credit-Linked Deposit of such Lender at such time and the denominator of which is the amount of the Total Credit-Linked Deposit; provided that if the obligations of the Issuer to make Credit Extensions have been terminated pursuant to Section 8.02(a), the numerator of such fraction shall be such Lender's Outstanding Amount of all Credit Obligations at such time, and the denominator shall be the Outstanding Amount of all Credit Obligations of all Lenders at such time (with the aggregate amount of any Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Lender for purposes of this definition). The initial Pro Rata Share of each Lender is set forth opposite the name of such Lender on Schedule 2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"property" or "properties" means, unless otherwise specifically limited, real or personal property of any kind, tangible or intangible, existing or inchoate.

"PTE" has the meaning specified in Section 10.10(b)(i).

"QPAM Exemption" has the meaning specified in Section 10.10(b)(iv).

"Register" has the meaning specified in Section 10.07(c).

"Reimbursement Date" has the meaning specified in Section 2.01(c)(i).

"Replacement Lender" has the meaning specified in Section 3.07.

"Request for Credit Extension" means a request for the issuance or amendment of a Credit by the Issuer pursuant to a Request for Credit Extension substantially in the form of Exhibit A, with such changes thereto as the Administrative Agent or the Issuer may from time to time approve.

"Request for Term Loans" means a request for the making of Term Loans from the proceeds of Credit-Linked Deposits, substantially in the form of Exhibit E.

"Required Credit-Linked Lenders" means, Credit-Linked Lenders (other than the Company and its Affiliates) having Combined Pro Rata Shares aggregating more than 50%. For purposes hereof, the "Combined Pro Rata Share" of a Credit-Linked Lender means, with respect to such Credit-Linked Lender at any time, a fraction (expressed as a percentage, carried out to the ninth decimal place), the numerator of which is the amount of the sum of (A) the Credit-

Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under this Agreement at such time plus (B) the sum of the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under the Ten Year Agreement at such time (as such terms are defined in the Ten Year Agreement), plus (C) the sum of the Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of such Credit-Linked Lender under the Five Year Agreement at such time (as such terms are defined in the Five Year Agreement), and the denominator of which is the amount of the sum of (X) the Total Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of all Credit-Linked Lenders under this Agreement at such time (with the aggregate amount of any Credit-Linked Lender's risk participation and funded participation in Credit Obligations being deemed "held" by such Credit-Linked Lenders under the Ten Year Agreement at such time (as such terms are defined in the Ten Year Agreement) (with the aggregate amount of any Credit-Linked Lender's risk participation and funded participations being deemed "held" by such Credit-Linked Lender for purposes of this definition) plus (Z) the Total Credit-Linked Deposit plus the Outstanding Amount of all Credit Obligations of all Credit-Linked Lenders under the Five Year Agreement at such time (as such terms are defined in the Five Year Agreement) (with the aggregate amount of any Credit-Linked Lender's risk participation in Credit Obligations being deemed "held" by such Credit-Linked Lender for purposes of this definition).

"Required Lenders" means, as of any date of determination, Lenders (other than the Company and its Affiliates) having Pro Rata Shares aggregating more than 50%.

"Required Notice" has the meaning specified in Section 10.19(b).

"Responsible Officer" means any Senior Financial Officer and any other officer of a Credit Party with responsibility for the administration of the relevant portion of this Agreement. Any document delivered hereunder that is signed by a Responsible Officer of a Credit Party shall be conclusively presumed to have been authorized by all necessary corporate action on the part of such Credit Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Credit Party.

"Restricted Subsidiary" means any Subsidiary of the Company which is not designated as an Unrestricted Subsidiary in accordance with Section 7.07.

"S&P" means Standard & Poor's Rating Group, a division of The McGraw-Hill Companies, or any successor thereto that is a nationally-recognized rating agency.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Securities Account" has the meaning specified in Section 10.19(b)(ii).

"Securities Act" means the Securities Act of 1933.

"Securities Exchange Act" means Securities Exchange Act of 1934.

"Securities Intermediary" has the meaning specified in Section 10.19(b)(ii).

"Security" has the meaning specified in Section 2(1) of the Securities Act.

"Senior Financial Officer" means the chief financial officer, principal accounting officer, treasurer or controller of a Credit Party.

"Senior Responsible Officer" means any Senior Financial Officer, or any of the President, any Executive Vice President, any Senior Vice President or any Principal Vice President of a Credit Party.

"Source" has the meaning specified in Section 10.10(b).

"Subsidiary" means, as to any Person, any corporation, association or other business entity in which such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries owns sufficient equity or voting interests to enable it or them (as a group) ordinarily, in the absence of contingencies, to elect a majority of the directors (or Persons performing similar functions) of such entity, and any partnership or joint venture if more than a 50% interest in the profits or capital thereof is owned by such Person or one or more of its Subsidiaries or such Person and one or more of its Subsidiaries (unless such partnership or joint venture can and does ordinarily take major business actions without the prior approval of such Person or one or more of its Subsidiaries). Unless the context otherwise clearly requires, any reference to a "Subsidiary" is a reference to a Subsidiary of the Company.

"Taxes" has the meaning specified in Section 3.01(a).

"Ten Year Agreement" means the Ten Year Letter of Credit and Term Loan Agreement dated as of even date herewith (as amended, restated, supplemented or otherwise modified from time to time) among the Company, Holdings, each lender party thereto and Bank of America as administrative agent and issuer of letters of credit; provided, however, if all Credit Obligations (as such term is defined in the Ten Year Agreement) are paid in full and the Ten Year Agreement is terminated, then this definition of "Ten Year Agreement" shall no longer be applicable to this Agreement.

"Term Loan Issuance Date" has the meaning specified in Section 2.03(b)(iii).

"Term Loans" has the meaning specified in Section 2.03(b)(i).

"Term Note" has the meaning specified in Section 2.07(b).

"Total Credit-Linked Deposit" shall mean, at any time, the sum of all Lenders' Credit-Linked Deposits, as the same may be reduced from time to time.

"Trust" has the meaning specified in Section 10.19(b)(ii).

"United States" and "U.S." each means the United States of America.

"Unreimbursed Amount" has the meaning specified in Section 2.01(c)(i).

"Unrestricted Subsidiary" means any Subsidiary which is designated as such in writing by the Company to each Lender pursuant to Section 7.07.

"Unused Amount" has the meaning specified in Section 2.03(b)(i).

- **1.02 Other Interpretive Provisions**. With reference to this Agreement and each other Credit Document, unless otherwise specified herein or in such other Credit Document:
 - (a) The meanings of defined terms are equally applicable to the singular and plural forms of the defined terms.
- (b) (i) The words "herein," "hereof," and "hereunder" and words of similar import when used in any Credit Document shall refer to such Credit Document as a whole and not to any particular provision thereof.
 - (ii) Article, Section, Exhibit and Schedule references are to the Credit Document in which such references appear.
 - (iii) The term "including" is by way of example and not limitation.
- (iv) The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form.
- (c) In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including."
- (d) Section headings herein and in the other Credit Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Credit Document.
- 1.03 Accounting Terms. All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Audited Financial Statements, except as otherwise specifically prescribed herein.
- (b) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Credit Document, and either the Company or the Required Credit-Linked Lenders shall so request, the Administrative Agent, the Lenders and the Company shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Credit-Linked Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Company shall provide to the Administrative Agent and the Lenders financial statements and other documents

required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

- **1.04 Rounding.** Any financial ratios required to be maintained by the Company pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).
- **1.05 References to Agreements and Laws**. Unless otherwise expressly provided herein, (a) references to agreements (including the Credit Documents) and other contractual instruments shall be deemed to include all subsequent amendments, restatements, extensions, supplements and other modifications thereto, but only to the extent that such amendments, restatements, extensions, supplements and other modifications are not prohibited by any Credit Document; and (b) references to any Law shall include all statutory and regulatory provisions consolidating, amending, replacing, supplementing or interpreting such Law.
- **1.06 Credit Amounts.** Unless otherwise specified, all references herein to the amount of a Credit at any time shall be deemed to mean the maximum face amount of such Credit after giving effect to all increases thereof contemplated by such Credit or the Issuer Documents related thereto, whether or not such maximum face amount is in effect at such time.

ARTICLE II THE CREDIT-LINKED DEPOSITS AND CREDIT EXTENSIONS

2.01 Credits.

- (a) The Credits.
- (i) Subject to the terms and conditions set forth herein, (A) the Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.01, from time to time on any Business Day during the period from the Closing Date until the Credit Expiration Date, to issue Credits denominated in Dollars for the account of the Company or any Subsidiary, and (B) the Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.01, (1) to amend, extend, reinstate or renew Credits previously issued by it, in accordance with subsection (b) below, and (2) to honor drawings under the Credits; and (C) the Lenders severally agree to participate in Credits issued or outstanding hereunder for the account of the Company or any Subsidiary as provided in Section 2.01(c), and all reimbursement obligations and rights hereunder in respect thereof and the Issuer Documents with respect thereto and the Issuer hereby grants to the Lenders such participation interests in such Credits and all reimbursement obligations and rights hereunder in respect thereof and such related Issuer Documents; provided that the Company shall not request the issuance of any Credit, the Issuer shall not make any Credit Extension with respect to any Credit, and no Lender shall participate in any Credit, to the extent that, as of the date of such Credit Extension:

- (A) the sum, without duplication, of the applicable Outstanding Amount of all (i) Credits and (ii) Unreimbursed Amounts for which Advances have not yet been made would exceed the Total Credit-Linked Deposit;
- (B) any Lender's Pro Rata Share of the sum, without duplication, of the applicable Outstanding Amount of all (i) Credits and (ii) Unreimbursed Amounts for which Advances have not yet been made would exceed such Lender's Credit-Linked Deposit; or
- (C) the expiry date of any such requested Credit would occur after the Credit Expiration Date (or, in the case of an Auto-Renewal Credit, no Nonrenewal Notice Date would occur on or prior to the Credit Expiration Date);

and provided further that in determining the availability hereunder with respect to any Escalating Credits issued or outstanding hereunder, the Total Credit-Linked Deposit will be deemed to be utilized in respect of such Escalating Credits in the aggregate amount equal to the maximum aggregate amount available to be drawn under all such Escalating Credits (after giving effect to all increases).

Within the foregoing limits (the "Issuance Limits"), and subject to the terms and conditions hereof, the Company's ability to obtain Credits shall be fully revolving, and accordingly the Company may, during the foregoing period, obtain Credits to replace Credits that have expired, been terminated or cancelled, or that have been drawn upon and reimbursed. All Existing Credits up to the respective amounts specified in Schedule 1.01 shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof. Notwithstanding anything to the contrary set forth herein, if any portion of the Total Credit-Linked Deposit is utilized to reimburse the Issuer for any amounts drawn under any Credit (as contemplated by Section 2.01(c) hereof) or to make Term Loans (as contemplated by Section 2.03(b) hereof), the Total Credit-Linked Deposit shall be permanently reduced by the corresponding amount utilized to reimburse the Issuer or to make Term Loans and shall not be reinstated.

- (ii) The Issuer shall not be under any obligation to issue any Credit if:
 - (A) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain the Issuer from issuing such Credit, or any Law applicable to the Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over the Issuer shall prohibit, or request that the Issuer refrain from, the issuance of letters of credit generally or such Credit in particular or shall impose upon the Issuer with respect to such Credit any restriction, reserve or capital requirement (in each case, for which the Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon the Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which the Issuer in good faith deems material to it;
 - (B) the Administrative Agent or the Issuer reasonably determines that such Credit is not a Permitted Credit;

- (C) the issuance of such Credit would violate any Laws or any policies of the Issuer; or
- (D) except as otherwise agreed by the Administrative Agent and the Issuer, such Credit is in a face amount less than \$1,000,000, or is to be denominated in a currency other than Dollars.
- (iii) The Issuer shall not be under any obligation to amend any Credit if (A) the Issuer would have no obligation at such time to issue such Credit in its amended form under the terms hereof, or (B) the beneficiary of such Credit does not accept the proposed amendment to such Credit.
- (iv) The Issuer shall not issue or amend any Credit if the Issuer has received written notice from any Lender, the Administrative Agent or any Credit Party, on or prior to the Business Day prior to the requested date of issuance or amendment of such Credit, that one or more applicable conditions contained in Article IV shall not then be satisfied.
 - (b) Procedures for Issuance and Amendment of Credits; Auto-Renewal Credits.
- (i) Each Credit shall be issued or amended, as the case may be, upon the request of the Company delivered to the Issuer (with a copy concurrently delivered to the Administrative Agent) in the form of a Request for Credit Extension, appropriately completed and signed by a Responsible Officer or Authorized Representative of the Company. Any such Request for Credit Extension must be received by the Issuer and the Administrative Agent not later than 1:00 p.m., Dallas time, at least five Business Days prior to the proposed issuance date or date of amendment, as the case may be; or such other date or time as the Issuer and the Administrative Agent may agree in a particular instance in their sole discretion. In the case of a request for an initial issuance of a Credit, such request shall be in the form of a Request for Credit Extension and shall specify in form and detail satisfactory to the Issuer and the Administrative Agent: (A) the proposed issuance date of the requested Credit (which shall be a Business Day); (B) the face amount thereof; (C) the expiry date thereof; (D) the purpose and nature of the Credit; (E) the Person whose obligations are supported thereby (in the case of a Credit supporting the obligations of a Subsidiary); (F) the name and address of the beneficiary thereof; (G) any required text to be contained in the Credit; (H) the delivery instructions with respect to the Credit; and (J) whether such Credit will be an Escalating Credit, and if so, the maximum face amount of such Credit after giving effect to all increases. In the case of a request for an amendment of any outstanding Credit, such request shall be in the form of a Request for Credit Extension and shall specify in form and detail satisfactory to the Issuer and the Administrative Agent: (1) the Credit to be amended; (2) the proposed date of amendment thereof (which shall be a Business Day); (3) the nature of the proposed amendment; and (4) delivery instructions with respect to the amendment. Additionally, the Company shall furnish to the Issuer and the Administrative Agent such other documents and information pertaining to such requested Credit issuance or amendment, including any Issuer Documents as the Issuer or the Administrative Agent may reasonably require. Without limiting the generality of the foregoing, the delivery of a Request for Credit Extension is additional to, and not in replacement or substitution of, any other Issuer Document that the Issuer may require. Upon the effectiveness of any issuance or amendment of a Credit that will constitute a Credit hereunder, the

Administrative Agent and the Lenders shall be entitled to assume that the Issuer has obtained such Issuer Documents as it shall have requested, executed by the relevant parties thereto to the extent required hereby.

- (ii) Promptly after receipt of any Request for Credit Extension by the Issuer, the Issuer will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has received a copy of such Request for Credit Extension from the Company and, if not, the Issuer will provide the Administrative Agent with a copy thereof. Upon receipt by the Issuer and the Company of confirmation from the Administrative Agent that the requested issuance or amendment is permitted in accordance with the terms hereof, then, subject to the terms and conditions hereof, the Issuer shall, on the requested date, issue a Credit for the account of the Company or the applicable Subsidiary, or enter into the applicable amendment, as the case may be, in each case in accordance with the Issuer's usual and customary business practices.
- (iii) If the Company so requests in any applicable Request for Credit Extension, the Issuer agrees, subject to the terms and conditions hereof, to issue a Credit that has automatic renewal provisions (each, an "Auto-Renewal Credit"); provided that any such Auto-Renewal Credit must permit the Issuer to prevent any such renewal at least once during the term thereof (commencing with the date of issuance of such Credit) by giving prior notice to the beneficiary thereof not later than a specified date to be agreed upon at the time such Credit is issued, which shall occur and be effective prior to the Credit Expiration Date (the "Nonrenewal Notice Date"). Unless otherwise directed by the Issuer, the Company shall not be required to make a specific request to the Issuer for any renewal of an Auto-Renewal Credit, but the Issuer shall be required to provide prior notice to the Administrative Agent of any pending renewal of an Auto-Renewal Credit at least ten Business Days before the applicable Nonrenewal Notice Date. Once an Auto-Renewal Credit has been issued (or is permitted to be outstanding hereunder in the case of an Existing Credit that is an Auto-Renewal Credit), the Lenders shall be deemed to have authorized (but may not require) the Issuer to permit the renewal of such Credit at any time to a date not later than the Credit Expiration Date; provided, however, that the Issuer shall not permit any renewal of an Auto-Renewal Credit if (A) the Issuer has determined that it would have no obligation at such time to issue such Credit in its renewed form under the terms hereof (by reason of the provisions of Section 2.01(a)(ii), Section 4.02 or otherwise), (B) after giving effect to any such renewal, the earlier of the (x) expiry date of such Auto-Renewal Credit would occur after the Credit Expiration Date, or (C) it has received notice (which may be by telephone or in writing) on or before the day that is two Business Days before the Nonrenewal Notice Date from the Administrative Agent, any Lender or the Company that o
- (iv) If the Issuer shall issue any Credit or amend any Credit (if the effect thereof is to increase the face amount), without obtaining prior consent from the Administrative Agent (as provided in clause (ii) above), or if the Issuer shall permit the extension or renewal of an Auto-Renewal Credit without giving timely prior notice to the Administrative Agent or when such extension or renewal is not permitted hereunder (as provided in clause (iii) above), or if the Issuer shall permit the reinstatement of an Auto-Reinstating Credit when such reinstatement is

not permitted hereunder (as provided in clause (vi) below), such Credit (in the case of any such amendment, to the extent of the increased face amount thereof, and in the case of any reinstatement, to the extent of the reinstated amount thereof) (A) shall for all purposes be deemed to have been issued by the Issuer solely for its own account and risk, and (B) shall not be considered a Credit outstanding under this Agreement, and no Lender shall be deemed to have any participation therein, effective as of the date of such issuance, amendment, reinstatement, extension or renewal, as the case may be, unless the Required Lenders expressly consent thereto.

- (v) Promptly after its delivery of any Credit or any amendment to a Credit to an advising bank with respect thereto or to the beneficiary thereof, the Issuer will also deliver to the Company and the Administrative Agent a true and complete copy of such Credit or amendment.
- (vi) If any Credit contains provisions providing for automatic reinstatement of the stated amount after any drawing thereunder, the Administrative Agent and the Lenders hereby authorize and direct the Issuer to permit such automatic reinstatement, whether or not a Default or Event of Default then exists, unless the Issuer has received a notice (which may be by telephone or in writing) on or before the day that is two Business Days before the reinstatement date from the Administrative Agent or the Required Lenders that one or more applicable conditions specified in Sections 2.01(a)(ii) or 4.02 is not then satisfied and directing it to cease permitting such automatic reinstatements.
 - (c) Drawings and Reimbursements; Credit-Linked Deposits.
- (i) Upon receipt from the beneficiary of any Credit of any notice of a drawing under such Credit, the Issuer of such Credit shall promptly notify the Company thereof. If the Company obtains written notice (confirmed by telephone) from the Issuer of a drawing under a Credit prior to 1:00 p.m., Dallas time, on any Business Day, the Company shall reimburse the Issuer on such Business Day. If the Company obtains written notice (confirmed by telephone) from the Issuer of a drawing under a Credit at or after 1:00 p.m., Dallas time, on any Business Day, the Company shall reimburse the Issuer on the Business Day immediately following the Business Day upon which such notice was received by the Company. Such reimbursement shall be made directly to the Issuer in an amount equal to the unreimbursed drawing (the "Unreimbursed Amount"). Without affecting the Company's obligations to so reimburse the Issuer on any such Business Day, which shall be absolute and unconditional, if the Company fails to so reimburse the Issuer by such time referenced above on the applicable due date for reimbursement specified above (the "Reimbursement Date"), the Issuer shall so notify the Administrative Agent (with a copy to the Company), which notice shall be provided on a Business Day, and specify in such notice the amount of the Unreimbursed Amount. Immediately upon receipt of such notice from the Issuer, the Administrative Agent shall promptly notify each Lender of the Reimbursement Date, the Unreimbursed Amount, the amount of such Lender's Pro Rata Share thereof, and shall state that Advances will be funded by application of such Lender's Pro Rata Share of the Credit-Linked Deposits in the Credit-Linked Deposit Account on the sixth Business Day after such Reimbursement Date to the extent the Unreimbursed Amount (or any portion thereof) remains outstanding on such day.

(ii) The Issuer hereby irrevocably grants to each Lender, and each Lender hereby irrevocably accepts and purchases from the Issuer on the terms and conditions hereinafter stated, for such Lender's own account and risk an undivided interest equal to its Pro Rata Share in the Issuer's obligations and rights with respect to the Credits issued pursuant to this Agreement (as to each Lender, its "Participation"), which purchase price when paid to the Administrative Agent shall not be a deposit obligation of the Administrative Agent or Bank of America. The consideration for the Participation of each Lender shall consist of the payment by such Lender to the Administrative Agent of an amount equal to the Dollar amount set forth opposite such Lender's name in Schedule 2.01 under the heading "Credit-Linked Deposit" (as the same may be reduced from time to time in accordance with this Agreement, such Lender's "Credit-Linked Deposit"), subject to the conditions precedent set forth in Section 4.01 hereof. Each Lender shall pay to the Administrative Agent its Credit-Linked Deposit in full on the Closing Date. The Credit-Linked Deposits of all Lenders shall be held by, and in the name of, the Administrative Agent in the Credit-Linked Deposit Account under the sole dominion and control of the Administrative Agent. Each Lender unconditionally and irrevocably agrees with the Administrative Agent and the Issuer that, if a drawing is paid under any Credit for which the Issuer is not reimbursed in full by the Company in cash within five Business Days after the applicable Reimbursement Date or converted into a Term Loan under Section 2.03(b), such Lender hereby authorizes the Administrative Agent to reimburse to the Issuer the Unreimbursed Amount (or the outstanding portion thereof) related to such drawing on or after the sixth Business Day after such Reimbursement Date, to the extent of such Lender's Pro Rata Share of the Unreimbursed Amount, solely from such Lender's Pro Rata Share of the Credit-Linked Deposits in the Credit-Linked Deposit Account, and each Lender hereby irrevocably authorizes the Administrative Agent to charge the Credit-Linked Deposit Account for such purpose, in satisfaction of such Lender's reimbursement obligation arising with respect to such drawing hereunder. Without limiting the generality of Section 9.01, in charging the Credit-Linked Deposit Account or otherwise exercising any rights of set-off with respect thereto, the Administrative Agent acts as the agent of the Issuer. The amount of each Lender's Pro Rata Share of such Unreimbursed Amount (or portion thereof) which is paid the Issuer as set forth above shall be deemed to be an Advance by such Lender to the Company hereunder. The outstanding principal amount of each Advance, together with interest thereon as provided in Section 2.04, shall be due and payable, in Dollars, on demand. Each Lender shall be subrogated to the rights and remedies of the Issuer against the Company and any Subsidiary liable under such Credit to the extent such Lender has reimbursed the Issuer as set forth in this Section 2.01(c)(ii). The Issuer shall reasonably cooperate in exercising and enforcing such rights and remedies as may be requested by the Required Lenders (and such cooperation shall be subject to any applicable indemnification set forth in Section 9.07 or Section 10.05 of this Agreement). The Issuer shall have no right to withdraw or set-off against monies on deposit in the Credit-Linked Deposit Account other than as set forth in this Section 2.01(c)(ii).

(iii) The Credit-Linked Deposits shall be held by the Administrative Agent in its name in the Credit-Linked Deposit Account and no Person other than the Administrative Agent shall have a right of withdrawal from the Credit-Linked Deposit Account nor any other right or power with respect to the Credit-Linked Deposits or the Credit-Linked Deposit Account. Notwithstanding anything in this Agreement to the contrary, the sole funding obligation of each Lender in respect of its Participation shall be satisfied upon funding of its Credit-Linked Deposit.

- (iv) Until an Advance is funded pursuant to this Section 2.01(c) to reimburse the Issuer for the Unreimbursed Amount in respect of any Credit, or such Unreimbursed Amount is converted to a Term Loan pursuant to Section 2.03, interest in respect of each Lender's Pro Rata Share of the Borrowing in the amount of such Unreimbursed Amount shall be solely for the account of the Issuer.
- (v) Except as expressly provided herein, each Lender's agreement to fund Advances, by application of such Lender's Pro Rata Share of the Credit-Linked Deposits, to reimburse the Issuer for amounts drawn under Credits issued by the Issuer, as contemplated by this Section 2.01(c), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against the Issuer, the Company, any Subsidiary or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or Event of Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing.
- (vi) Notwithstanding that a Credit issued or outstanding hereunder is in support of any obligations of a Subsidiary, the Company shall be obligated to reimburse the Issuer hereunder for the full amount of any and all drawings under such Credit.
- (vii) In the event of any voluntary payment or prepayment (other than through the making of a Term Loan as provided for in Section 2.03(b)) of an Advance at any time prior to the second anniversary of the Closing Date, the Company shall pay on the date of such prepayment to the Administrative Agent, for the account of the Lenders in accordance with their Pro Rata Shares, a premium in an amount equal to three percent (3.00%) of the amount so prepaid and any prepayment of the Advances thereafter shall be accompanied by the Prepayment Premium, if any.

(d) Repayment of Participations.

(i) At any time after the Issuer has made a payment under any Credit and has received from the Credit-Linked Deposit the proceeds of Advances by the Lenders in respect of such payment in accordance with Section 2.01(c), if the Administrative Agent receives for the account of the Issuer any payment in respect of the related Unreimbursed Amount or interest thereon (whether directly from the Company or otherwise, including proceeds of Cash Collateral applied thereto by the Administrative Agent), the Administrative Agent will distribute to each Lender its Pro Rata Share thereof (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's Advance was outstanding) in Dollars and in the same funds as those received by the Administrative Agent. If a Borrowing has occurred hereunder, the Company shall repay the Unreimbursed Amount and interest thereon as provided in Sections 2.01(c) and 2.04 and shall pay any prepayment premium required to be paid pursuant to Section 2.01(c)(vii). Notwithstanding the foregoing, if the Issuer shall have received from the Credit-Linked Deposit the proceeds of Advances by the Lenders and thereafter shall receive any direct payment from the Company in respect of a Borrowing, the Issuer shall immediately pay the amount received to the Administrative Agent for distribution to the Lenders in accordance with this Section 2.01(d).

- (ii) If any payment received by the Administrative Agent for the account of the Issuer pursuant to Section 2.01(d)(i) and distributed to the Lenders by the Administrative Agent is required to be returned under any of the circumstances described in Section 10.06, each Lender shall pay to the Administrative Agent for the account of the Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect.
- (e) Obligations Absolute. The obligation of the Company to reimburse the Issuer for each drawing under each Credit and to repay each Borrowing, shall be absolute, unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following:
 - (i) any lack of validity or enforceability of such Credit, this Agreement, or any other Credit Document;
- (ii) the existence of any claim, counterclaim, set-off, defense or other right that the Company or any Subsidiary may have at any time against any beneficiary or any transferee of such Credit (or any Person for whom any such beneficiary or any such transferee may be acting), the Issuer or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by such Credit or any agreement or instrument relating thereto, or any unrelated transaction:
- (iii) any draft, demand, certificate or other document presented under such Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under such Credit;
- (iv) any payment by the Issuer under such Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Credit; or any payment made by the Issuer under such Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Credit, including any arising in connection with any proceeding under any Debtor Relief Law; or
- (v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Company or any Subsidiary.

The Company shall promptly examine a copy of each Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Company's instructions or other irregularity, the Company will promptly notify the Issuer. The Company shall be conclusively deemed to have waived any such claim against the Issuer and its correspondents unless such notice is given as aforesaid.

(f) Role of Issuer. Each Lender and the Company agree that, in paying any drawing under a Credit, the Issuer shall not have any responsibility to obtain any document (other than

any sight draft, certificates and documents expressly required by the Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. None of the Issuer, any Issuer-Related Person nor any of the respective correspondents, participants or assignees of the Issuer shall be liable to any Lender for (i) any action taken or omitted in connection herewith at the request or with the approval of the Required Lenders or the Required Credit-Linked Lenders, as applicable; (ii) any action taken or omitted in the absence of gross negligence or willful misconduct; or (iii) the due execution, effectiveness, validity or enforceability of any document or instrument related to any Credit or Request for Credit Extension. The Company hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Company's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuer, any Issuer-Related Person, nor any of the respective correspondents, participants or assignees of the Issuer, shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.01(e); provided, however, that anything in such clauses to the contrary notwithstanding, the Company may have a claim against the Issuer, and the Issuer may be liable to the Company, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Company which the Company proves were caused by the Issuer's willful misconduct or gross negligence or the Issuer's willful failure to pay under any Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Credit. In furtherance and not in limitation of the foregoing, the Issuer may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the Issuer shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

- (g) Applicability of ISP98. Unless otherwise expressly agreed by the Issuer and the Company when a Credit is issued (including any such agreement applicable to an Existing Credit), the rules of the ISP shall apply to each Credit.
- (h) Issuer Documents. In the event of any conflict between the terms hereof, the terms of any Request for Credit Extension and the terms of any Issuer Document (other than any Credit), the terms hereof shall control.
 - (i) Existing Credits. The Company and the Issuer certify that Schedule 1.01 accurately and completely sets forth the Existing Credits.
 - (j) Interest on Credit-Linked Deposit; Repayment of Credit-Linked Deposit.
- (i) The Administrative Agent hereby agrees to cause Bank of America to pay interest to the Administrative Agent, as holder of the Credit-Linked Deposit Account, and the Administrative Agent, as holder of the Credit-Linked Deposit Account, agrees to distribute to each Lender, from such interest payments received from Bank of America, such Lender's Pro Rata Share of interest on the outstanding amount of the Credit-Linked Deposits in the Credit-Linked Deposit Account at the rate per annum set forth on Schedule A annexed hereto. Such

interest will be distributed to the Lenders by the Administrative Agent quarterly in arrears on the first Business Day following the end of each Interest Period relating to the Credit-Linked Deposits. The Administrative Agent shall compute all amounts due under this Section 2.01(j)(i) and shall notify the Company and such Lender of each such amount due.

- (ii) Subject to, and to the extent of, the Company's compliance with the cash-collateralization requirements set forth in Section 2.02, the Administrative Agent shall distribute the aggregate amount of any remaining Credit-Linked Deposits to the Lenders on the Maturity Date to the extent of the amount credited to the Credit-Linked Deposit Account. The Credit-Linked Deposit Account in the name of the Administrative Agent under its sole dominion and control, and held by it with Bank of America subject to the terms and conditions of this Agreement. No Lender, whether upon the receivership or liquidation of such Lender or otherwise, shall be entitled to its Credit-Linked Deposit other than as expressly provided in this Agreement.
- (iii) The Company shall have no right, title or interest in or to the Credit-Linked Deposits or the Credit-Linked Deposit Account and no obligations with respect thereto, it being acknowledged and agreed by the parties hereto that the making of the Credit-Linked Deposits by the Lenders, the deposit by the Administrative Agent of the Credit-Linked Deposits in the Credit-Linked Deposits Account, the provisions of this Section 2.01(j) and the application of the Credit-Linked Deposits in the manner contemplated by Section 2.01(c) constitute agreements among the Administrative Agent, the Issuer and each Lender in respect of the funding obligations of each Lender in respect of its Participation in Credits, and do not constitute any loan or extension of credit to the Company.

2.02 Cash Collateral

- (a) Upon the request of the Administrative Agent, (i) if, as of the Credit Expiration Date, (A) any Credit for any reason remains outstanding and partially or wholly undrawn or (B) any amount remains available to be drawn under any Credit by reason of the operation of Rule 3.14 of the ISP, or (ii) upon the occurrence of the circumstances described in Section 8.02(c) requiring the Company to Cash Collateralize Credits, the Company shall within two Business Days after receipt of such request Cash Collateralize the then Outstanding Amount of all Credits (in an amount equal to the Outstanding Amount thereof). For purposes hereof, "Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Issuer and the Lenders, as collateral for the Credit Obligations, cash or deposit account balances ("Cash Collateral") pursuant to documentation in form and substance reasonably satisfactory to the Administrative Agent (which documents are hereby consented to by the Lenders). Derivatives of such term have corresponding meanings.
- (b) The Company hereby grants the Administrative Agent, for the benefit of the Issuer and the Lenders, a security interest in all such Cash Collateral. Cash Collateral shall be maintained in blocked, non-interest bearing deposit accounts at Bank of America and shall be subject to such Lien documentation as the Administrative Agent shall reasonably request. The Company hereby authorizes the filing by the Administrative Agent of any UCC-1 financing statements in connection therewith.

2.03 Reduction of Total Credit-Linked Deposits; Conversion to Funded Term Loans.

(a) Reduction. The Company may, on or after the second anniversary of the Closing Date, upon notice from the Company to the Administrative Agent, permanently reduce the available amount of the Total Credit-Linked Deposit to an amount not less than the sum of the then Outstanding Amount of all Credit Obligations (other than Advances and Term Loans); provided that (i) any such notice shall be received by the Administrative Agent not later than 1:00 p.m., Dallas time, ten Business Days prior to the date of termination or reduction, (ii) any such partial reduction shall be in an aggregate amount of \$10,000,000 or any whole multiple of \$1,000,000 in excess thereof, or such lesser amount as would reduce the Total Credit-Linked Deposit to \$0; and (iii) a Prepayment Premium, if any, shall be payable with respect to any amounts reduced. Such notice shall specify the proposed effective date of such reduction and the amount of the Prepayment Premium, if any. The Administrative Agent will promptly notify the Lenders of any such notice of reduction of the Total Credit-Linked Deposit. Once reduced in accordance with this Section, the Total Credit-Linked Deposit may not be increased. Any reduction of the Total Credit-Linked Deposit shall be applied to each Lender according to its Pro Rata Share. All Facility Fees and Fronting Fees shall accrue until the effective date of any reduction of the Total Credit-Linked Deposit and shall be paid on the effective date of such reduction if the Total Credit-Linked Deposit is \$0 after giving effect to such reduction. Upon any such reduction, the Administrative Agent shall promptly remit to each Lender its Pro Rata Share of the amount of the reduction in the Total Credit-Linked Deposit and Prepayment Premium, if any.

(b) Conversion to Funded Term Loans.

- (i) Subject to the terms and conditions set forth herein (including satisfaction of the applicable conditions specified in Sections 4.02 and 4.03), the Company may elect to (A) cause the Lenders to make, and each Lender agrees to make, funded loans ("Term Loans") to the Company in an aggregate principal amount up to the excess of (x) the Total Credit-Linked Deposit over (y) the Outstanding Amount of all Credit Obligations (other than Advances and Term Loans) (the amount of such excess being the "Unused Amount"), or (B) cause the Lenders to make, and each Lender agrees to make, one or more funded Term Loans in an aggregate principal amount up to the amount of a forthcoming drawing on any Credit, or (C) cause the Lenders to make, and each Lender agrees to make, one or more funded Term Loans in the aggregate amount of any Advances then outstanding by converting such Advances to Term Loans; in each case in accordance with the procedures set forth in this Section 2.03(b). Any Term Loans made under this Section 2.03(b) and subsequently repaid or prepaid may not be reborrowed.
- (ii) The Company may irrevocably request the making of one or more Term Loans in an aggregate amount up to the Unused Amount or in an aggregate amount up to the anticipated drawing on a Credit from the proceeds of Credit-Linked Deposits, or in an aggregate amount up to the aggregate amount of Advances then outstanding by converting such Advances to Term Loans, as applicable, by delivering a Request for Term Loans therefor to the Administrative Agent, appropriately completed and signed by a Responsible Officer or

Authorized Representative of the Company, not later than 10 Business Days prior to the proposed date of making of the Term Loans; provided that the Term Loans requested pursuant to Section 2.03(b)(i)(A) shall only be made on a day which is the last day of an Interest Period with respect to the Credit-Linked Deposits and the Company shall in any event pay the amounts, if any, payable under Section 3.04; and provided further that the minimum aggregate amount of Term Loans made on any date shall be (A) \$25,000,000, in the case of Term Loans requested pursuant to Section 2.03(b)(i)(A), and (B) \$10,000,000, in the case of Term Loans requested pursuant to Sections 2.03(b)(i)(B) and 2.03(b)(i)(C), or in each case of clauses (A) and (B) of this clause (ii), such lesser amount which, when added to the applicable Outstanding Amount of all (x) Credits and (y) the Unreimbursed Amounts for which Advances have not been made would equal the Total Credit-Linked Deposit.

(iii) Following receipt of a Request for Term Loans, the Administrative Agent shall promptly notify each Lender of its Pro Rata Share thereof. Upon satisfaction of the applicable conditions set forth in Sections 4.02 and 4.03, the Administrative Agent shall (A) in the case of Term Loans made pursuant to Section 2.03(b)(i)(A), debit the Credit-Linked Deposit Account in an amount equal to the aggregate amount of the proposed Term Loans (but not exceeding the Unused Amount) and shall make such funds available to the Company either by (i) crediting the account of the Company on the books of Bank of America with the amount of such funds or (ii) wire transferring such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Company, (B) in the case of Term Loans made pursuant to Section 2.03(b)(i)(B), debit the Credit-Linked Deposit Account in an amount equal to the aggregate amount of the proposed Term Loans and transfer such funds to the Issuer in reimbursement of the applicable drawing on a Credit, and (C) in the case of Term Loans made pursuant to Section 2.03(b)(i)(C), record in its books and records and on the Register the conversion of the applicable Advances into Term Loans, which credit or transfer or recordation, as applicable, shall be deemed to be the making of the Term Loans by the Lenders hereunder, and such date shall be the "Term Loan Issuance Date" (and the Administrative Agent shall forthwith deliver a notice to each Lender informing such Lender of the applicable Term Loan Issuance Date). Each Lender hereby irrevocably authorizes the Administrative Agent to charge the Credit-Linked Deposit Account for such purpose. No Lender shall have any obligation to make Term Loans to the Company other than from the proceeds of its Credit-Linked Deposits.

(iv) Interest Rate on Term Loans.

- (A) Subject to the provisions of subsection (B) below, each Term Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the LIBO Rate for such Interest Period plus the Applicable Rate.
- (B) If any amount of the Term Loans is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable

Law. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.

- (C) Interest on each Term Loan shall be due and payable in arrears on the first Business Day following the end of each Interest Period applicable thereto, at such other times as may be specified herein and on the Maturity Date. To the fullest extent permitted by applicable Law, interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.
- (v) Repayment of Term Loans. The Company shall repay to the Administrative Agent for the account of the Lenders on the Maturity Date the aggregate principal amount of the Term Loans outstanding on such date.
- (vi) Optional Prepayment of Term Loans. The Company may, on or after the second anniversary of the Closing Date, at its option, upon notice to the Administrative Agent as provided below, prepay on any Business Day all, or from time to time any part of, the Term Loans (subject to the limitations sets forth below); provided that (i) any such notice shall be received by the Administrative Agent not later than 1:00 p.m., Dallas time, ten Business Days prior to the date of prepayment, (ii) any such partial prepayment shall be in an aggregate amount of \$1,000,000 or any whole multiple of \$1,000,000 in excess thereof, or such lesser amount as would reduce the Term Loans to \$0; and (iii) a Prepayment Premium, if any, shall be payable with respect to any amounts prepaid. Such notice shall specify the proposed effective date of such prepayment, the aggregate principal amount of the Term Notes to be prepaid on such date, the interest to be paid on the prepayment date with respect to such principal amount being prepaid, which shall be the last day of the Interest Period therefor, and the amount of the Prepayment Premium, if any. The Administrative Agent will promptly notify the Lenders of any such notice of prepayment of the Term Loans. Once prepaid in accordance with this Section, such Term Loans may not be reborrowed. Upon any such prepayment, the Administrative Agent shall promptly remit to each Lender its Pro Rata Share of the amount of the Prepayment of the Term Loans and Prepayment Premium, if any.

2.04 Interest on Borrowings Other Than Term Loans.

- (a) Each Borrowing (other than the Term Loans) shall bear interest on the outstanding principal amount thereof (i) from and including the applicable Borrowing date to and excluding the date six Business Days thereafter at an interest rate per annum equal at all times to the Base Rate plus 1.00% per annum, and (ii) from and including the sixth Business Day after the applicable Borrowing date at an interest rate per annum equal at all times to the Default Rate.
- (b) If any other amount payable by the Company under any Credit Document is not paid when due (without regard to any applicable grace periods) whether at stated maturity, by acceleration or otherwise, each such amount shall thereafter bear interest at an interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Law.
 - (c) Accrued and unpaid interest on all such unpaid amounts (including interest on past due interest) shall be due and payable upon demand.

(d) Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.

2.05 Fees and Other Charges.

- (a) Facility Fee. The Company shall pay to the Administrative Agent for the account of each Lender in accordance with its Pro Rata Share, a fee (the "Facility Fee") equal to the Applicable Rate, expressed on a daily basis, times the actual daily amount of such Lender's Credit-Linked Deposit. The Facility Fee shall accrue effective as of the Closing Date and shall be calculated quarterly in arrears by the Administrative Agent. If there is any change in the Applicable Rate during any quarter, the actual daily Credit-Linked Deposit amount shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect. The Facility Fee shall be due and payable quarterly in arrears on the first Business Day following the end of each Interest Period relating to the Credit-Linked Deposits, commencing with the first such date to occur after the Closing Date, and on the Maturity Date; provided that, in connection with any reduction of the Total Credit-Linked Deposit under Section 2.03, the accrued Facility Fee calculated for the period ending on such date shall also be paid on the date of such reduction, and the following quarterly payment shall be calculated on the basis of the period from such reduction date to such quarterly payment date. The Facility Fee shall accrue at all times, including at any time during which one or more of the conditions in Article IV is not met.
- (b) Fronting Fee. The Company shall pay directly to the Issuer for its own account, a fee (the "Fronting Fee") equal to the rate per annum set forth in the letter agreement dated May 8, 2003 (the "Fee Letter"), among the Company, BAS and the Administrative Agent, times the daily amount of the Total Credit-Linked Deposit, due and payable quarterly in arrears on the first Business Day immediately following the last Business Day of each March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the Maturity Date; provided that, in connection with any reduction of the Total Credit-Linked Deposit under Section 2.03, the accrued Fronting Fee calculated for the period ending on such date shall also be paid on the date of such reduction, and the following quarterly payment shall be calculated on the basis of the period from such reduction date to such quarterly payment date. At any time that an Event of Default exists, the Fronting Fee shall be increased by adding 2% to the amount thereof determined as provided in the Fee Letter. The Fronting Fee shall accrue at all times, including at any time during which one or more of the conditions in Article IV is not met.
- (c) Documentary and Processing Charges Payable to Issuer. The Company shall pay directly to the Issuer for its own account the customary and standard issuance, commission, presentation, amendment and other processing fees, and other standard costs and charges (if any), of the Issuer or any Affiliate of the Issuer, relating to its Credits as from time to time in effect and agreed upon between the Company and the Issuer. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.
- (d) Placement and Administrative Fees. The Company shall pay a placement fee to BAS, and shall pay an administrative fee to the Administrative Agent for the Administrative Agent's own account, in the amounts and at the times specified in the Fee Letter, and if a

successor Administrative Agent is appointed pursuant to Section 9.09 hereof, the Company shall pay the administrative fees of such successor Administrative Agent.

2.06 Computation of Interest and Fees.

- (a) All computations of interest when the Base Rate is determined by reference to Bank of America's "prime rate" shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. Computation of all other types of interest and all fees shall be calculated on the basis of a year of 360 days, and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day or 366-day year). Interest shall accrue on each Advance or Term Loan for the day on which the Advance or Term Loan is made, and shall not accrue on an Advance or Term Loan, or any portion thereof, for the day on which the Advance, such Term Loan or such portion is paid, provided that any Advance or Term Loan that is repaid on the same day on which it is made shall bear interest for one day.
- (b) The Administrative Agent shall promptly notify the Company and the Lenders of the LIBO Rate applicable to any Interest Period upon determination of such interest rate. The determination of LIBO Rate by the Administrative Agent shall be conclusive in the absence of manifest error. At any time that Advances or Term Loans accruing interest based on the Base Rate are outstanding, the Administrative Agent shall notify the Company and the Lenders of any change in Bank of America's prime rate used in determining the Base Rate promptly following the public announcement of such change.

2.07 Evidence of Debt.

- (a) The Credit-Linked Deposits, the Credit Extensions made by the Issuer, the participations of each Lender therein and any Advances and Term Loans made by each Lender shall be evidenced by one or more accounts or records maintained by the Issuer, such Lender and by the Administrative Agent in the ordinary course of business. The accounts or records maintained by the Issuer, the Administrative Agent and each Lender shall be conclusive absent manifest error of the amount of the Credit-Linked Deposits, the Credit Extensions made by the Issuer and any Advances and Term Loans made by each Lender and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of the Company hereunder to pay any amount owing with respect to the Credit Obligations. In the event of any conflict between the accounts and records maintained by the Issuer or any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error.
- (b) In addition to the accounts and records referred to in subsection (a), (i) the Credit-Linked Deposit of, and the obligation of the Company to pay Advances to, each Lender shall be evidenced by a Letter of Credit Linked Note in the form of Exhibit D attached hereto (a "Letter of Credit Linked Note"), executed by the Administrative Agent and by the Company, and (ii) the Term Loans of a Lender shall be evidenced by a Term Note in the form of Exhibit F attached hereto (a "Term Note"), executed by the Company.

2.08 Payments Generally.

- (a) All payments to be made by the Company to any Lender, the Administrative Agent, the Issuer or any other Person hereunder shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by the Company hereunder to be made directly to the Issuer shall be made to the Issuer in accordance with its payment instructions in Dollars and in immediately available funds not later than the times and on the dates specified herein. Except as otherwise expressly provided herein and except with respect to such payments to be made directly to the Issuer, all payments by the Company hereunder shall be made to the Administrative Agent, for the account of the respective Lenders or the Issuer to which such payment is owed, at the Administrative Agent's Office in Dollars and in immediately available funds not later than 1:00 p.m., Dallas time, on the date specified herein. The Administrative Agent will promptly distribute to the Issuer its applicable share as provided herein, or to each Lender its Pro Rata Share (or other applicable share as provided herein), of such payment in like funds as received by wire transfer to the Issuer in accordance with its payment instructions or to such Lender at its Lending Office. All payments received by the Administrative Agent after 1:00 p.m., Dallas time, or by the Issuer after the time specified herein or in any Issuer Document, shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue.
- (b) If any payment to be made by the Company shall come due on a day other than a Business Day, or any other date specified hereunder would otherwise occur on a day other than a Business Day, then except as otherwise provided herein, such payment shall be made, or other date shall occur, on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.
- (c) The obligations of the Lenders hereunder to fund Participations in Credits are several and not joint. The failure of any Lender to fund its Participation on the Closing Date with its Credit-Linked Deposit shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so purchase its Participation.
- (d) Nothing herein shall be deemed to obligate any Lender to obtain the funds for the Credit-Linked Deposit in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for the Credit-Linked Deposit in any particular place or manner.
- **2.09 Sharing of Payments**. If, other than as expressly provided elsewhere herein, any Lender shall obtain on account of the Term Loans or Advances made by it or participations in Credit Obligations held by it, any payment (whether voluntary, involuntary, through the exercise of any right of set-off, or otherwise) in excess of its Pro Rata Share (or other share contemplated hereunder) thereof, such Lender shall immediately (a) notify the Administrative Agent of such fact, and (b) purchase from the other Lenders such subparticipations in the Term Loans or Advances made by them or in the participations in Credit Obligations held by them as shall be necessary to cause such purchasing Lender to share the excess payment in respect thereof *pro rata* with each of them; provided, however, that if all or any portion of such excess payment is thereafter recovered from the purchasing Lender under any of the circumstances described in

Section 10.06 (including pursuant to any settlement entered into by the purchasing Lender in its discretion), such purchase shall to that extent be rescinded and each other Lender shall repay to the purchasing Lender the purchase price paid therefor, together with an amount equal to such paying Lender's ratable share (according to the proportion of (i) the amount of such paying Lender's required repayment to (ii) the total amount so recovered from the purchasing Lender) of any interest or other amount paid or payable by the purchasing Lender in respect of the total amount so recovered, without further interest thereon. The Company agrees that any Lender so purchasing a subparticipation from another Lender may, to the fullest extent permitted by law, exercise all its rights of payment (including the right of set-off) with respect to such subparticipation as fully as if such Lender were the direct creditor of the Company in the amount of such subparticipation. The Administrative Agent will keep records (which shall be conclusive and binding in the absence of manifest error) of subparticipations purchased under this Section and will in each case notify the Lenders following any such purchases or repayments. Each Lender that purchases a subparticipation pursuant to this Section shall from and after such purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Credit Obligations purchased to the same extent as though the purchasing Lender were the original owner of the Credit Obligations purchased.

ARTICLE III TAXES AND YIELD PROTECTION

3.01 Taxes. (a) Subject to Section 10.15(a)(iii), any and all payments by the Company to or for the account of the Administrative Agent, the Issuer or any Lender under any Credit Document shall be made free and clear of and without deduction for any and all present or future taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and all liabilities with respect thereto, excluding, in the case of the Administrative Agent, the Issuer and each Lender, taxes imposed on or measured by its overall net income, and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the Laws of which the Administrative Agent, the Issuer or such Lender, as the case may be, is organized or maintains a lending office (all such non-excluded taxes, duties, levies, imposts, deductions, assessments, fees, withholdings or similar charges, and liabilities being hereinafter referred to as "Taxes"). If the Company shall be required by any Laws to deduct any Taxes from or in respect of any sum payable under any Credit Document to the Administrative Agent, the Issuer or any Lender, (i) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section), the Administrative Agent, the Issuer or such Lender receives an amount equal to the sum it would have received had no such deductions been made, (ii) the Company shall make such deductions, (iii) the Company shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable Laws, and (iv) within 30 days after the date of such payment, the Company shall furnish to the Administrative Agent (which shall promptly forward the same to the Issuer or such Lender) the original or a certified copy of a receipt evidencing payment thereof.

(b) In addition, the Company agrees to pay any and all present or future stamp, court or documentary taxes and any other excise or property taxes or charges or similar levies which

arise from any payment made under any Credit Document or from the execution, delivery, performance, enforcement or registration of, or otherwise with respect to, any Credit Document (hereinafter referred to as "Other Taxes").

- (c) If the Company shall be required to deduct or pay any Taxes or Other Taxes from or in respect of any sum payable under any Credit Document to the Administrative Agent, the Issuer or any Lender, the Company shall also pay to the Administrative Agent or to such Lender, as the case may be, at the time interest is paid, such additional amount that the Administrative Agent, the Issuer or any Lender specifies as necessary to preserve the after-tax yield (after factoring in all taxes, including taxes imposed on or measured by net income) that the Administrative Agent, the Issuer or any Lender would have received if such Taxes or Other Taxes had not been imposed.
- (d) The Company agrees to indemnify the Administrative Agent, the Issuer and each Lender for (i) the full amount of Taxes and Other Taxes (including any Taxes or Other Taxes imposed or asserted by any jurisdiction on amounts payable under this Section) paid by the Administrative Agent, the Issuer and such Lender, (ii) amounts payable under Section 3.01(c) and (iii) any liability (including additions to tax, penalties, interest and expenses) arising therefrom or with respect thereto, in each case whether or not such Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. Payment under this subsection (d) shall be made within 30 days after the date the Issuer, such Lender or the Administrative Agent makes a demand therefor.
- 3.02 Increased Cost and Reduced Return; Capital Adequacy or Risk-Based Capital. (a) If the Issuer or any Lender determines that as a result of the introduction of or any change in or in the interpretation of any Law, or the Issuer's or such Lender's compliance therewith, there shall be any increase in the cost to the Issuer or such Lender of issuing or participating in Credits or the funding or maintaining Term Loans or Advances, or a reduction in the amount received or receivable by the Issuer or such Lender in connection with the foregoing (excluding for purposes of this subsection (a) any such increased costs or reduction in amount resulting from (i) Taxes or Other Taxes (as to which Section 3.01 shall govern), or (ii) changes in the basis of taxation of overall net income or overall gross income by the United States or any foreign jurisdiction or any political subdivision of either thereof under the Laws of which the Issuer or such Lender is organized or has its Lending Office), then from time to time upon demand of the Issuer or such Lender (with a copy of such demand to the Administrative Agent), the Company shall pay to the Issuer or such Lender such additional amounts as will compensate the Issuer or such Lender for such increased cost or reduction; provided that such compensation will be limited to (A) the period commencing not more than 120 days prior to the date of such demand or (B) any longer period of retroactive effect of any such change or requirement for compliance if such demand is made 120 days or less after such change or requirement for compliance.
- (b) If the Issuer or any Lender determines that the introduction of any Law regarding capital adequacy or risk-based capital or any change therein or in the interpretation thereof, or compliance by the Issuer or such Lender (or its Lending Office) therewith, has the effect of reducing the rate of return on the capital of the Issuer or such Lender or any corporation

controlling the Issuer or such Lender as a consequence of the Issuer's or such Lender's obligations (for avoidance of doubt not including outstanding Term Loans and Advances) hereunder (taking into consideration its policies with respect to capital adequacy or risk-based capital and the Issuer's or such Lender's desired return on capital), then from time to time upon demand of the Issuer or such Lender (with a copy of such demand to the Administrative Agent), the Company shall pay to the Issuer or such Lender such additional amounts as will compensate the Issuer or such Lender for such reduction; provided that such compensation will be limited to (A) the period commencing not more than 120 days prior to the date of such demand or (B) any longer period of retroactive effect of any such change or requirement for compliance if such demand is made 120 days or less after such change or requirement for compliance.

- 3.03 Illegality. If any Lender determines that any Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender to make, maintain or fund LIBO Rate Term Loans, or to determine or charge interest rates based upon the LIBO Rate, then, on notice thereof by such Lender to the Company through the Administrative Agent, any obligation of such Lender to make or continue LIBO Rate Term Loans shall be suspended until such Lender notifies the Administrative Agent and the Company that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, the Company shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay the Term Loans of such Lender, either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such LIBO Rate Term Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such LIBO Rate Term Loans. Upon any such prepayment, the Company shall also pay accrued interest on the amount so prepaid.
- **3.04 Funding Losses**. Upon demand of any Lender (with a copy to the Administrative Agent), or demand of the Administrative Agent, from time to time, the Company shall promptly compensate such Lender or the Administrative Agent, as applicable, for and hold such Lender or the Administrative Agent harmless from any loss, cost or expense incurred by it as a result of:
- (a) any payment or prepayment of any LIBO Rate Term Loan or Advance on a day other than the last day of the Interest Period for such Term Loan or Advance (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);
 - (b) any reduction of the Total Credit-Linked Deposits on a day other than the last day of an Interest Period therefor;
 - (c) any failure by the Company to prepay or borrow any Term Loan on the date or in the amount notified by the Company; or
- (d) any assignment of a LIBO Rate Term Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Company pursuant to Section 3.07;

including any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Term Loan or from fees payable to terminate the deposits from which such funds were obtained; provided, however, that such Lender shall use commercially reasonable efforts to mitigate any such loss, cost or expense.

For purposes of calculating amounts payable by the Company to the Lenders under this Section 3.04, each Lender shall be deemed to have funded each LIBO Rate Term Loan made by it at the LIBO Rate used in determining the LIBO Rate for such Term Loan by a matching deposit or other borrowing in the London interbank LIBOR market for a comparable amount and for a comparable period, whether or not such LIBO Rate Term Loan was in fact so funded.

- **3.05 Matters Applicable to All Requests for Compensation**. A certificate of the Administrative Agent, the Issuer or any Lender claiming compensation under this Article III and setting forth the additional amount or amounts to be paid to it hereunder shall be conclusive in the absence of manifest error. In determining such amount, the Administrative Agent, the Issuer or any Lender may use any reasonable averaging and attribution methods.
- **3.06** Survival. All of the Company's obligations under this Article III shall survive termination of the Credit-Linked Deposits and repayment of all Credit Obligations and all other amounts due hereunder for a period of one (1) year; provided, however, any and all of the Company's obligations under this Article III related to Taxes shall survive for a period of three (3) years.
- **3.07 Replacement of Lenders.** On each occasion that a Lender either makes a demand for compensation pursuant to Section 3.02 in an amount in excess of the amount that the Company would have had to pay pursuant to such Section if such Lender's Credit-Linked Deposit or Term Loan was held by a Lender with respect to which no such amounts would then be payable pursuant to such Section, the Company may, upon at least ten Business Days' prior written notice to the Issuer, each such Lender and the Administrative Agent, in whole permanently replace the Credit-Linked Deposit or Term Loan of such Lender; provided that, the Company shall replace such Credit-Linked Deposit or Term Loan with the Credit-Linked Deposit or Term Loan of (a) any other Lender, or, if no other Lender shall agree to replace such Lender (within 30 days after being offered such option), (b) a substitute lending institution which is an Eligible Assignee (in each case, a "Replacement Lender"). Such Replacement Lender shall, upon the effective date of replacement, purchase the Credit-Linked Deposit or Term Loan of, and the Credit Obligations owed to, such replaced Lender for the aggregate amount thereof and shall thereupon for all purposes become a "Lender" hereunder. Such notice from the Company shall specify an effective date for the replacement of such Lender, which date shall not be earlier than the tenth day after the day such notice is given. On the effective date of any replacement of such Lender pursuant to this Section 3.07 and in accordance with the terms of Section 10.07, the Company shall pay to the Administrative Agent for the account of such Lender (i) any fees or other amounts due to such Lender to the date of such replacement, (ii) accrued interest on the principal amount of outstanding Borrowings and the Credit-Linked Deposit or Term Loan held by such Lender to the date of such replacement and (iii) any other amount or amounts payable to such Lender pursuant to this Section 3.07. Upon the effective date of replacement of any Lender pursuant to this Section 3.07, such Lender shall cease to be a "Lender" hereunder, without any other or further act or deed on the part of any other Lender. No such replacement of any such Lender, the purchase of such Lender's Borrowings, Credit-Linked Deposit or Term Loans pursuant to this Section 3.07 shall affect (x) any liability or obligation of the Company or any other Lender to such replaced Lender which accrued on or prior to the date

of such termination or (y) such replaced Lender's rights hereunder in respect of any such liability or obligation.

ARTICLE IV CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

- **4.01 Conditions of Credit-Linked Deposits and Initial Credit Extension**. The obligation of each Lender to fund its Credit-Linked Deposit and the obligation of the Issuer to make its initial Credit Extension hereunder are subject to satisfaction or waiver of the following conditions precedent:
- (a) The Administrative Agent's and Lenders' special counsel's receipt of the following, each of which shall be originals or facsimiles (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer or Secretary or assistant Secretary, as applicable of the signing Credit Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance reasonably satisfactory to the Administrative Agent, the Issuer, the Lenders and their respective legal counsel:
- (i) executed counterparts of this Agreement, sufficient in number for distribution to the Administrative Agent, the Issuer, each Lender and the Credit Parties;
- (ii) certified copies of the certificate or articles of incorporation (including all amendments thereto) of each Credit Party, certified as of a recent date prior to the Closing Date by the Secretary of State of the respective states of incorporation of such Credit Party;
- (iii) copies of the bylaws (including all amendments thereto) of each Credit Party, certified as of the Closing Date by the Secretary or assistant Secretary of such Credit Party;
- (iv) incumbency certificates of the Secretary or assistant Secretary of each Credit Party evidencing the identity, authority and capacity of each Responsible Officer or Authorized Representative thereof authorized to act as a Responsible Officer or Authorized Representative in connection with this Agreement and the other Credit Documents to which such Credit Party is a party;
- (v) certificates of the Secretary of State or other appropriate official of the respective states of incorporation of each the Credit Party, certifying as of a recent date prior to the Closing Date that such Credit Party is duly organized or formed and validly existing and in good standing in the jurisdiction of its organization or formation as of such date, and certificates of the Secretary of State or other appropriate official of the State of Texas, certifying as of a recent date prior to the Closing Date that such Credit Party is in good standing and qualified to engage in business in the State of Texas;
- (vi) true and correct copies of the resolutions adopted by the respective boards of directors of each Credit Party, duly certified as of the Closing Date by the Secretary or assistant Secretary of such Credit Party, (i) approving the transactions contemplated by (and the

form of) this Agreement and the other Credit Documents and all other documents to be entered into in connection with this Agreement to be delivered by such Credit Party hereunder and (ii) authorizing the officers of such Credit Party, and each of them, acting individually or together, to execute and deliver each such document with such changes as such officer or officers shall approve;

- (vii) a certificate signed by a Responsible Officer of the Company certifying that (a) the representations and warranties contained in Article V, or any other Credit Document are true and correct, (b) no Default or Event of Default has occurred and is continuing, and (c) since the date of the Audited Financial Statements, no event or circumstance that either individually or in the aggregate has resulted in or could reasonably be expected to result in a Material Adverse Effect:
- (viii) opinions of counsel to each Credit Party, addressed to the Administrative Agent, the Issuer and each Lender, as to the matters set forth in Exhibit C-1 and such other matters concerning the Credit Parties and the Credit Documents as the Issuer or the Required Credit-Linked Lenders may reasonably request;
- (ix) opinions of counsel to the Administrative Agent and the Issuer, addressed to the Administrative Agent, the Issuer and each Lender, as to the matters set forth in Exhibit C-2 and Exhibit C-3 and such other matters as the Required Credit-Linked Lenders may reasonably request; and
- (x) opinion of special counsel to the Lenders, addressed to each Lender, as to the matters set forth in Exhibit C-4 and such other matters as the Required Credit-Linked Lenders may reasonably request.
 - (b) Any fees required to be paid on or before the Closing Date shall have been paid.
- (c) The Company shall have paid all reasonable Attorney Costs of (i) the Administrative Agent, and (ii) the special counsel to the Lenders, in each case to the extent invoiced prior to or on the Closing Date.
 - (d) The Closing Date shall have occurred on or prior to June 30, 2003.
- (e) On the Closing Date, each Lender's purchase of its Participation shall (i) be permitted by the laws and regulations of each jurisdiction to which such Lender is subject, without recourse to provisions (such as Section 1405(a)(8) of the New York Insurance Law) permitting limited investments by insurance companies without restriction as to the character of the particular investment, (ii) not violate any applicable law or regulation (including, without limitation, Regulation T, U or X of the Board of Governors of the Federal Reserve System) and (iii) not subject such Lender to any tax, penalty or liability under or pursuant to any applicable law or regulation, which law or regulation was not in effect on the date hereof. If requested by each Lender, such Lender shall have received an Officer's Certificate certifying as to such matters of fact as such Lender may reasonably specify to enable such Lender to determine whether such purchase is so permitted. The wire transfer of funds on the Closing Date by each

Lender in the amount of its Credit-Linked Deposit to the Administrative Agent shall constitute its confirmation as to satisfaction of this condition with respect to such Lender.

- (f) A Private Placement number issued by Standard & Poor's CUSIP Service Bureau (in cooperation with the Securities Valuation Office of the National Association of Insurance Commissioners) shall have been obtained for (i) the Letter of Credit Linked Notes and (ii) the Term Notes, which Private Placement number shall not be in the series of numbers issued to Bank of America Corporation and its affiliates, including but not limited to Bank of America.
 - (g) The Issuer shall have received evidence satisfactory to it that each Lender, in respect of its Participation hereunder, has funded its Credit-Linked Deposit.
 - (h) The Company and the Administrative Agent shall have executed and delivered to each Lender such Lender's Letter of Credit Linked Note.
- (i) The Administrative Agent and Bank of America in its capacity as depository shall have executed and delivered the Money Market Account Agreement and special counsel for the Lenders shall have received a copy thereof.
- **4.02 Conditions to all Credit Extensions by the Issuer**. The obligation of the Issuer to make any Credit Extension or honor any Request for Credit Extension is subject to the following conditions precedent:
- (a) The representations and warranties of the Company and each other Credit Party contained in Article V, or any other Credit Document, or which are contained in any document furnished at any time under or in connection herewith or therewith, shall be true and correct on and as of the date of such Credit Extension, except to the extent that such representations and warranties relate expressly and solely to an earlier date, and except to the extent of changes resulting from transactions contemplated or permitted by the Credit Documents and changes occurring in the ordinary course of business which either individually or in the aggregate do not result in a Material Adverse Effect.
 - (b) No Default or Event of Default shall exist, or would result from such proposed Credit Extension.
- (c) There has occurred since the date of the Audited Financial Statements, no event or circumstance that either individually or in the aggregate has resulted in or could reasonably be expected to result in a Material Adverse Effect.
- (d) Except in the case of any automatic renewal of any Auto-Renewal Credit or any automatic reinstatement of any Auto-Reinstating Credit, the Administrative Agent and the Issuer shall have received a Request for Credit Extension in accordance with the requirements hereof.
- (e) The Administrative Agent shall have received, in form and substance satisfactory to it, such other assurances, certificates, documents or consents related to the foregoing as the Administrative Agent, the Issuer or the Required Lenders reasonably may require.

Each Request for Credit Extension submitted by the Company shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a), (b) and (c) have been satisfied on and as of the date of the applicable Credit Extension.

- **4.03 Conditions to Term Loans**. The obligation of the Lenders to make any Term Loans is subject to the following conditions precedent:
- (a) The Administrative Agent shall have received in accordance with the provisions of Section 2.03(b) hereof an originally executed Request for Term Loans.
- (b) The Company, by a duly authorized Senior Responsible Officer, shall have executed and delivered to the Administrative Agent on the applicable Term Loan Issuance Date, for delivery to the Lenders, Term Notes to evidence such Term Loans, dated as of the Term Loan Issuance Date, in the aggregate principal amount of the Term Loans and with other appropriate insertions.
- (c) The Administrative Agent shall have received an opinion of in-house counsel of the Company, addressed to the Administrative Agent and each Lender, substantially in the form of Exhibit H attached hereto.
- (d) Conditions Satisfied. The Company shall have delivered to the Administrative Agent an Officer's Certificate, dated as of the applicable Term Loan Issuance Date, in form and substance satisfactory to the Administrative Agent, certifying that all conditions precedent set forth in Sections 4.02(a) and (b) and 4.03 have been satisfied on and as of the applicable Term Loan Issuance Date.

Each Request for Term Loans submitted by the Company shall be deemed to be a representation and warranty that the conditions specified in Sections 4.02(a) and (b) have been satisfied on and as of the date of the applicable Term Loan Issuance Date.

ARTICLE V REPRESENTATIONS AND WARRANTIES

The Company represents and warrants to the Administrative Agent and the Lenders that:

5.01 Organization; Power and Authority.

(a) The Company is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by Law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Company has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Credit Documents to which it is a party and to perform the provisions hereof and thereof.

(b) Holdings is a corporation duly organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation, and is duly qualified as a foreign corporation and is in good standing in each jurisdiction in which such qualification is required by Law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a material adverse effect on Holdings and its Subsidiaries, taken as a whole. Holdings has the corporate power and authority to own or hold under lease the properties it purports to own or hold under lease, to transact the business it transacts and proposes to transact, to execute and deliver this Agreement and the Credit Documents to which it is a party and to perform the provisions hereof and thereof.

5.02 Authorization, etc.

- (a) This Agreement and each other Credit Document have been duly authorized by all necessary corporate action on the part of the Company, and this Agreement constitutes, and each other Credit Document to which the Company is a party when executed and delivered will constitute, a legal, valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (b) This Agreement and each other Credit Document have been duly authorized by all necessary corporate action on the part of Holdings, and this Agreement constitutes, and each other Credit Document to which Holdings is a party when executed and delivered will constitute, a legal, valid and binding obligation of Holdings enforceable against Holdings in accordance with its terms, except as such enforceability may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting the enforcement of creditors' rights generally and (ii) general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- (c) To the extent Credits are being issued for the account of Subsidiaries, at the time of each such Credit Extension, such Credits do and will inure to the benefit of the Company, and the Company's business derives and will derive substantial benefits from the businesses of such Subsidiaries.
- **5.03 Disclosure**. The Company, through its agent, BAS, has delivered to each Lender a copy of a Private Placement Memorandum, dated May 2003 (the "Memorandum"), relating to the transactions contemplated hereby. As of its date, the Memorandum fairly describes, in all material respects, the general nature of the business and principal properties of the Company and its Subsidiaries. As of the Closing Date, this Agreement, the Memorandum, the documents, certificates or other writings delivered to the Lenders by or on behalf of the Company in connection with the transactions contemplated hereby and the financial statements listed in Schedule 5.05, taken as a whole, do not contain any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein not misleading in light of the circumstances under which they were made. Since December 31, 2002, there has been no change in the financial condition, operations, business or properties of the Company or any Subsidiary except changes that individually or in the aggregate could not reasonably be expected

to have a Material Adverse Effect. There is no fact known to the Company that could reasonably be expected to have a Material Adverse Effect that has not been disclosed herein or in the Memorandum, in the Disclosure Documents, or in the other documents, certificates and other writings delivered to the Lenders by or on behalf of the Company specifically for use in connection with the transactions contemplated hereby.

5.04 Organization and Ownership of Shares of Subsidiaries; Affiliates.

- (a) Schedule 5.04 contains (except as noted therein) complete and correct lists as of the Closing Date of (i) each of the Company's Restricted Subsidiaries, showing, as to each Restricted Subsidiary, the correct name thereof and the jurisdiction of its organization, (ii) each of the Company's Affiliates, and (iii) the Company's directors and Senior Responsible Officers. As of the Closing Date, there are no Unrestricted Subsidiaries.
- (b) As of the Closing Date, all of the outstanding shares of capital stock or similar equity interests of each Restricted Subsidiary owned by the Company and its Subsidiaries have been validly issued, are fully paid and nonassessable and are owned by the Company or another Subsidiary free and clear of any Lien (except for any Permitted Lien or as otherwise disclosed in Schedule 5.04).
- (c) Each Subsidiary identified on Schedule 5.04 is a corporation or other legal entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, and is duly qualified as a foreign corporation or other legal entity and is in good standing in each jurisdiction in which such qualification is required by law, other than those jurisdictions as to which the failure to be so qualified or in good standing could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each such Subsidiary has the corporate or other power and authority to own or hold under lease the properties it purports to own or hold under lease and to transact the business it transacts and proposes to transact.
- (d) As of the Closing Date, no Subsidiary is a party to, or otherwise subject to any legal restriction or any agreement (other than the agreements listed on Schedule 5.04 and customary limitations imposed by corporate law statutes) restricting the ability of such Subsidiary to pay dividends out of profits or make any other similar distributions of profits to the Company or any of its Subsidiaries that owns outstanding shares of capital stock or similar equity interests of such Subsidiary.

5.05 Financial Statements.

The Company has delivered to the Issuer and each Lender the financial statements of the Company and its Subsidiaries listed on Schedule 5.05. All of said financial statements (including in each case the related schedules and notes) fairly present in all material respects the consolidated financial position of the Company and its Subsidiaries as of the respective dates specified in such Schedule and the consolidated results of their operations and cash flows for the respective periods so specified and have been prepared in accordance with GAAP consistently applied throughout the periods involved except as set forth in the notes thereto (subject, in the case of any interim financial statements, to normal year-end adjustments).

5.06 Compliance with Laws, Other Instruments, etc.

- (a) The execution, delivery and performance by the Company of this Agreement, the Letter of Credit Linked Notes and the Term Notes will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of the Company or any Subsidiary under, any indenture, mortgage, deed of trust, loan, note purchase or credit agreement, or lease, its corporate charter or by-laws, or any other agreement or instrument to which the Company or any Subsidiary is bound or by which the Company or any Subsidiary or any of their respective properties may be bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to the Company or any Subsidiary or (iii) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to the Company or any Subsidiary.
- (b) The execution, delivery and performance by Holdings of this Agreement will not (i) contravene, result in any breach of, or constitute a default under, or result in the creation of any Lien in respect of any property of Holdings under, any indenture, mortgage, deed of trust, loan, note purchase or credit agreement, lease, corporate charter or by-laws, or any agreement or instrument to which Holdings is bound or by which Holdings or any of its properties may be bound or affected, (ii) conflict with or result in a breach of any of the terms, conditions or provisions of any order, judgment, decree, or ruling of any court, arbitrator or Governmental Authority applicable to Holdings, or (iii) violate any provision of any statute or other rule or regulation of any Governmental Authority applicable to Holdings.

5.07 Governmental Authorizations, etc.

- (a) No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by the Company of this Agreement, the Letter of Credit Linked Notes or the Term Notes.
- (b) No consent, approval or authorization of, or registration, filing or declaration with, any Governmental Authority is required in connection with the execution, delivery or performance by Holdings of this Agreement.

5.08 Litigation; Observance of Agreements, Statutes and Orders.

- (a) Except as disclosed in Schedule 5.08 or in the Disclosure Documents, there are no actions, suits or proceedings pending or, to the knowledge of the Company, threatened against or affecting the Company or any Subsidiary or any property of the Company or any Subsidiary in any court or before any arbitrator of any kind or before or by any Governmental Authority that, individually or in the aggregate, could reasonably be expected, after offset for claims against insurers, to have a Material Adverse Effect.
- (b) Except as disclosed in Schedule 5.08 or in the Disclosure Documents, neither the Company nor any Subsidiary is in default under any term of any agreement or instrument to which it is a party or by which it is bound, or any order, judgment, decree or ruling of any court,

arbitrator or Governmental Authority or is in violation of any applicable law, ordinance, rule or regulation (including without limitation Environmental Laws) of any Governmental Authority, which default or violation, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

- **5.09 Taxes**. The Company and its Subsidiaries have filed all tax returns that are required to have been filed in any jurisdiction, and have paid all taxes shown to be due and payable on such returns and all other taxes and assessments levied upon them or their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent, except for any taxes and assessments (i) the amount of which is not individually or in the aggregate Material or (ii) the amount, applicability or validity of which is currently being contested in good faith by appropriate proceedings and with respect to which the Company or a Subsidiary, as the case may be, has established adequate reserves in accordance with GAAP. The Company knows of no basis for any other tax or assessment that could reasonably be expected to have a Material Adverse Effect. The charges, accruals and reserves on the books of the Company and its Subsidiaries in respect of federal, state or other taxes for all fiscal periods are adequate in accordance with GAAP. The federal income tax liabilities of the Company and its Subsidiaries have been determined by the IRS and paid for all fiscal years up to and including the fiscal year ended December 31, 1987.
- **5.10 Title to Property; Leases.** The Company and its Restricted Subsidiaries have good and sufficient title to their respective properties that the Company and the Subsidiaries own or purport to own that individually or in the aggregate are Material, including all such properties reflected in the most recent audited balance sheet referred to in Section 5.05 or purported to have been acquired by the Company or any Restricted Subsidiary after said date (except as sold or otherwise disposed of in the ordinary course of business), in each case free and clear of Liens prohibited by this Agreement. All leases that individually or in the aggregate are Material are valid and subsisting and are in full force and effect in all material respects.
 - **5.11** Licenses, Permits, etc. Except as disclosed in Schedule 5.11,
- (a) the Company and its Restricted Subsidiaries own or possess all licenses, permits, franchises, authorizations, patents, copyrights, service marks, trademarks and trade names, or rights thereto, that individually or in the aggregate are Material, without known material conflict with the rights of others;
- (b) to the best knowledge of the Company, no product of the Company or any of its Restricted Subsidiaries infringes in any Material respect any license, permit, franchise, authorization, patent, copyright, service mark, trademark, trade name or other right owned by any other Person; and
- (c) to the best knowledge of the Company, there is no Material violation by any Person of any right of the Company or any of its Restricted Subsidiaries with respect to any patent, copyright, service mark, trademark, trade name or other right owned or used by the Company or any of its Restricted Subsidiaries.
 - 5.12 Compliance with ERISA.

- (a) The Company and each ERISA Affiliate have operated and administered each Plan in compliance with all applicable laws except for such instances of noncompliance as have not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Company nor any ERISA Affiliate has incurred any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans (as defined in Section 3 of ERISA), and no event, transaction or condition has occurred or exists that could reasonably be expected to result in the incurrence of any such liability by the Company or any ERISA Affiliate, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate, in either case pursuant to Title I or IV of ERISA or to such penalty or excise tax provisions or to Section 401(a)(29) or 412 of the Code, other than such liabilities or Liens as would not be individually or in the aggregate Material.
- (b) The present value of the aggregate accrued benefits under each of the Plans that are subject to Title IV of ERISA (other than Multiemployer Plans), determined as of the valuation date for the most recently ended plan year on the basis of the actuarial assumptions specified for funding purposes in such Plan's most recent actuarial valuation report, did not exceed the aggregate current value of the assets of such Plan allocable to such benefits (i) in the case of any single Plan, by an amount which, if said amount were immediately contributed to the Plans, could have a Material Adverse Effect, and (ii) in the case of all said Plans, by an amount which, if said amount were immediately contributed to the Plans, could have a Material Adverse Effect.
- (c) The Company and its ERISA Affiliates have not incurred withdrawal liabilities (and are not subject to contingent withdrawal liabilities) under Section 4201 or 4204 of ERISA in respect of Multiemployer Plans that individually or in the aggregate are Material.
- (d) The expected obligation for postretirement benefits other than pensions (determined as of the last day of the Company's most recently ended fiscal year in accordance with Financial Accounting Standards Board Statement No. 106, without regard to liabilities attributable to continuation coverage mandated by Section 4980B of the Code and after taking into account any governmental reimbursements) of the Company and its Subsidiaries is not Material.
- (e) The execution and delivery of this Agreement and the issuance and sale of the Letter of Credit Linked Notes and the Term Notes will not involve any transaction that is subject to the prohibitions of Section 406 of ERISA or in connection with which a tax could be imposed pursuant to Section 4975(c)(1)(A)-(D) of the Code. The representations by the Company in the first sentence of this Section 5.12(e) are made in reliance upon and subject to the accuracy of each Lender's representations in Section 10.10(b) as to the sources of the funds used to pay the purchase price of its Participation under this Agreement.
- **5.13 Private Offering by Company**. Neither the Company nor anyone acting on its behalf has offered the Letter of Credit Linked Notes or any similar Securities for sale to, or solicited any offer to buy any of the same from, or otherwise approached or negotiated in respect thereof with, any Person other than the Lenders and not more than seventeen (17) other Institutional Investors, each of which has been offered the Letter of Credit Linked Notes

and Term Notes at a private sale for investment. Neither the Company nor anyone acting on its behalf has taken, or will take, any action that would subject the issuance or sale of the Letter of Credit Linked Notes or Term Notes to the registration requirements of Section 5 of the Securities Act.

- **5.14 Use of Proceeds; Margin Regulations**. The Company will apply the proceeds from the Advances and the Term Loans hereunder for general corporate purposes of the Company and its Subsidiaries. No part of the proceeds from the Credit Extensions hereunder will be used, directly or indirectly, for the purpose of buying or carrying any margin stock within the meaning of Regulation U of the FRB (12 CFR 221), or for the purpose of buying or carrying or trading in any Securities under such circumstances as to involve the Company in a violation of Regulation X of said Board (12 CFR 224) or to involve any broker or dealer in a violation of Regulation T of the FRB (12 CFR 220). Margin stock does not constitute more than 5.0% of the value of the consolidated assets of the Company and its Subsidiaries and the Company does not have any present intention that margin stock will constitute more than 5.0% of the value of such assets. As used in this Section, the terms "margin stock" and "purpose of buying or carrying" shall have the meanings assigned to them in said Regulation U.
- **5.15 Existing Indebtedness**. Except as described therein, Schedule 5.15 sets forth a complete and correct list of all outstanding Indebtedness of the Company and its Restricted Subsidiaries as of March 31, 2003, specifying whether such Indebtedness is secured or unsecured. From March 31, 2003 to the Closing Date, there has been no material change in the amounts, interest rates, sinking funds, installment payments or maturities of such Indebtedness of the Company or its Restricted Subsidiaries. As of the Closing Date, neither the Company nor any Restricted Subsidiary is in default and no waiver of default is currently in effect, in the payment of any principal or interest on any such Indebtedness of the Company or such Restricted Subsidiary, and no event or condition exists with respect to any such Indebtedness of the Company or any Restricted Subsidiary, that would permit (or that with notice or the lapse of time, or both, would permit) one or more Persons to cause such Indebtedness to become due and payable before its stated maturity or before its regularly scheduled dates of payment.
- **5.16 Foreign Assets Control Regulations, etc.** Neither any Credit Extension to the Company hereunder nor its use of the proceeds thereof will violate the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto. Without limiting the foregoing, neither the Company nor any of its Subsidiaries or its Affiliates (a) is or will become a Person whose property or interests in property are blocked pursuant to Section 1 of Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)) or (b) engages or will engage in any dealings or transactions, or be otherwise associated, with any such Person. The Company and its Subsidiaries and its Affiliates are in compliance, in all Material respects, with the Uniting And Strengthening America By Providing Appropriate Tools Required To Intercept And Obstruct Terrorism (USA Patriot Act of 2001). No part of the proceeds from any Credit Extension hereunder will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or

anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

- **5.17 Status under Certain Statutes**. Neither the Company nor any Restricted Subsidiary is subject to regulation under the Investment Company Act of 1940, as amended, the Public Utility Holding Company Act of 1935, as amended, or, except as disclosed in Schedule 5.17, the Federal Power Act, as amended.
- **5.18** Environmental Matters. Except as disclosed in the Disclosure Documents, neither the Company nor any Restricted Subsidiary has knowledge of any Material claim or has received any notice of any Material claim, and no proceeding has been instituted raising any Material claim against the Company or any of its Restricted Subsidiaries or any of their respective real properties now or formerly owned, leased or operated by any of them or other assets, alleging any damage to the environment or violation of any Environmental Laws, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect. Except as otherwise disclosed to each Lender in writing,
- (a) neither the Company nor any Restricted Subsidiary has knowledge of any facts which would give rise to any claim, public or private, of violation of Environmental Laws or damage to the environment emanating from, occurring on or in any way related to real properties now or formerly owned, leased or operated by any of them or to other assets or their use, except, in each case, such as could not reasonably be expected to result in a Material Adverse Effect;
- (b) neither the Company nor any of its Restricted Subsidiaries has stored any Hazardous Materials on real properties now or formerly owned, leased or operated by any of them in a manner contrary to any Environmental Laws and has not disposed of any Hazardous Materials in a manner contrary to any Environmental Laws, in each case in any manner that could reasonably be expected to result in a Material Adverse Effect; and
- (c) all buildings on all real properties now owned, leased or operated by the Company or any of its Restricted Subsidiaries are in compliance with applicable Environmental Laws, except where failure to comply could not reasonably be expected to result in a Material Adverse Effect.
- **5.19** Compliance With Issuance Limits. Both before and after giving effect to any Credit Extension, all Credits issued and outstanding hereunder are in full compliance with the Issuance Limits.
- **5.20 Tax Shelter Regulations**. The Company does not intend to treat the Advances, Term Loans and Credits and related transactions as being a "reportable transaction" (within the meaning of Treasury Regulation Section 1.6011-4).

ARTICLE VI AFFIRMATIVE COVENANTS

So long as any Lender shall have any Credit-Linked Deposit, Letter of Credit Linked

Note or Term Note hereunder or any Credit hereunder shall remain outstanding:

- **6.01 Financial and Business Information**. The Company shall deliver to the Administrative Agent, the Issuer and to each of the Lenders, in form and detail satisfactory to the Administrative Agent, the Issuer and the Required Credit-Linked Lenders:
- (a) Quarterly Statements within 60 days after the end of each quarterly fiscal period in each fiscal year of the Company (other than the last quarterly fiscal period of each such fiscal year), duplicate copies of,
 - (i) a consolidated balance sheet of the Company and its Subsidiaries as at the end of such quarter, and
- (ii) consolidated statements of income, changes in shareholders' equity and cash flows of the Company and its Subsidiaries, for such quarter and (in the case of the second and third quarters) for the portion of the fiscal year ending with such quarter,

setting forth in each case in comparative form the figures for the corresponding periods in the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP applicable to quarterly financial statements generally, and certified by a Senior Financial Officer as fairly presenting, in all material respects, the financial position of the companies being reported on and their results of operations and cash flows, subject to changes resulting from year-end adjustments, provided that delivery within the time period specified above of copies of the Company's Quarterly Report on Form 10-Q prepared in compliance with the requirements therefor and filed with the SEC, if applicable, shall be deemed to satisfy the requirements of this Section 6.01(a) so long as such Quarterly Report on Form 10-Q includes the financial statements identified in clauses (i) and (ii) above;

- (b) Annual Statements within 100 days after the end of each fiscal year of the Company, duplicate copies of,
 - (i) a consolidated balance sheet of the Company and its Subsidiaries, as at the end of such year, and
 - (ii) consolidated statements of income, shareholder's equity and cash flows of the Company and its Subsidiaries, for such year,

setting forth in each case in comparative form the figures for the previous fiscal year, all in reasonable detail, prepared in accordance with GAAP, and accompanied by an opinion thereon of independent certified public accountants of recognized national standing, which opinion shall state that such financial statements present fairly, in all material respects, the financial position of the companies being reported upon and their results of operations and cash flows and have been prepared in conformity with GAAP, and that the examination of such accountants in connection with such financial statements has been made in accordance with generally accepted auditing standards, and that such audit provides a reasonable basis for such opinion in the circumstances; provided that the delivery within the time period specified above of the Company's Annual Report on Form 10-K for such fiscal year (together with the Company's annual report to

shareholders, if any, prepared pursuant to Rule 14a-3 under the Exchange Act) prepared in accordance with the requirements therefor and filed with the SEC, if applicable, shall be deemed to satisfy the requirements of this Section 6.01(b) so long as such Annual Report on Form 10-K includes the financial statements identified in clauses (i) and (ii) above;

- (c) SEC and Other Reports promptly, following the filing or mailing thereof, copies of all material of a financial nature filed with the SEC or sent to the Company's and its Subsidiaries' stockholders generally;
- (d) Outstanding Credits not later than five (5) Business Days after the end of each quarter, the information with respect to the outstanding Credits as is required under Section 2.01(j);
 - (e) [intentionally omitted]
- (f) Notice of Default or Event of Default promptly, and in any event within five (5) Business Days after a Responsible Officer becoming aware of the existence of any Default or Event of Default, a written notice specifying the nature and period of existence thereof and what action the Company is taking or proposes to take with respect thereto;
- (g) ERISA Matters promptly, and in any event within five (5) Business Days after a Responsible Officer becoming aware of any of the following, a written notice setting forth the nature thereof and the action, if any, that the Company or an ERISA Affiliate proposes to take with respect thereto:
- (i) with respect to any Plan, any reportable event, as defined in Section 4043(b) of ERISA and the regulations thereunder, for which notice thereof has not been waived pursuant to such regulations as in effect on the date hereof; or
- (ii) the taking by the PBGC of steps to institute, or the threatening by the PBGC of the institution of, proceedings under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Plan, or the receipt by the Company or any ERISA Affiliate of a notice from a Multiemployer Plan that such action has been taken by the PBGC with respect to such Multiemployer Plan; or
- (iii) any event, transaction or condition that would result in the incurrence of any liability by the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, or in the imposition of any Lien on any of the rights, properties or assets of the Company or any ERISA Affiliate pursuant to Title I or IV of ERISA or such penalty or excise tax provisions, if such liability or Lien, taken together with any other such liabilities or Liens then existing, could reasonably be expected to have a Material Adverse Effect;
- (h) Notices from Governmental Authority promptly, and in any event within 30 days of receipt thereof, copies of any notice to the Company or any Subsidiary from any federal or state Governmental Authority relating to any order, ruling, statute or other law or regulation that the Company reasonably believes could have a Material Adverse Effect;

- (i) Additional Reporting Requirement in the event that Unrestricted Subsidiaries account for more than 10% of the Consolidated Total Assets of the Company and its Subsidiaries, or more than 10% of the consolidated revenue of the Company and its Subsidiaries, in each case as of the last day of the fiscal quarter or fiscal year for which the Company is delivering financial information pursuant to Sections 6.01(a) and (b), then each such set of financial information delivered pursuant to Sections 6.01(a) and (b) shall be accompanied by unaudited financial statements for all Unrestricted Subsidiaries of the Company taken as a group, together with consolidating statements reflecting eliminations or adjustments required to reconcile such group statements to the consolidated financial statements of the Company and its Subsidiaries;
- (j) Requested Information with reasonable promptness, such other data and information relating to the business, operations, affairs, financial condition, assets or properties of the Company or any its Subsidiaries or relating to the ability of the Company to perform its obligations hereunder, under the Letter of Credit Linked Notes and under the Term Notes or the ability of Holdings to perform its obligations hereunder as from time to time may be reasonably requested by any Lender, including without limitation and subject to Section 10.07, such information as is required by SEC Rule 144A under the Securities Act to be delivered to a prospective transferee of the Letter of Credit Linked Notes or Term Notes.
- **6.02 Officer's Certificate**. Each set of financial statements delivered to a Lender pursuant to Section 6.01(a) or Section 6.01(b) hereof shall be accompanied by a certificate of a Senior Financial Officer setting forth:
- (a) Covenant Compliance the information (including reasonably detailed calculations) required in order to establish whether the Company was in compliance with the requirements of Sections 7.02(c)(iv), 7.03(m), 7.04 and 7.05, during the quarterly or annual period covered by the statements then being furnished (including with respect to each such Section, where applicable, the calculations of the maximum or minimum amount, ratio or percentage, as the case may be, permissible under the terms of such Sections, and the calculation of the amount, ratio or percentage then in existence); and
- (b) Event of Default a statement that such officer has reviewed the relevant terms hereof and has made, or caused to be made, under his or her supervision, a review of the transactions and conditions of the Company and its Subsidiaries from the beginning of the quarterly or annual period covered by the statements then being furnished to the date of the certificate and that such review shall not have disclosed the existence during such period of any condition or event that constitutes a Default or an Event of Default or, if any such condition or event existed or exists (including, without limitation, any such event or condition resulting from the failure of the Company or any Subsidiary to comply with any Environmental Law), specifying the nature and period of existence thereof and what action the Company shall have taken or proposes to take with respect thereto.
 - **6.03 Inspection**. The Company shall permit the representatives of each Lender:
- (a) No Default if no Default or Event of Default then exists, at the expense of such Lender, to visit the principal executive office of the Company and to discuss the affairs, finances

and accounts of the Company and its Restricted Subsidiaries with the Company's officers, all at such reasonable times during normal business hours and as often as may be reasonably requested in writing; and

- (b) Default if a Default or Event of Default then exists, at the expense of the Company and upon reasonable prior notice, to visit the principal executive office of the Company, to examine all relevant books of account, records, reports and other papers relating to the Company's and Holdings' performance of their respective obligations under this Agreement, the Letter of Credit Linked Notes and the Term Notes and to make copies and abstracts therefrom, and to discuss the Company's and its Restricted Subsidiaries' affairs, finances and accounts with the Company's officers and independent public accountants (and by this provision the Company authorizes said accountants to discuss the affairs, finances and accounts of the Company and its Restricted Subsidiaries), all at such reasonable times during normal business hours and as often as may be reasonably requested.
- **6.04 Compliance with Laws**. The Company will, and will cause each of its Subsidiaries to, comply with all laws, ordinances or governmental rules or regulations to which each of them is subject, including, without limitation, Environmental Laws, and will obtain and maintain in effect all licenses, certificates, permits, franchises and other governmental authorizations necessary to the ownership of their respective properties or to the conduct of their respective businesses, in each case to the extent necessary to ensure that non-compliance with such laws, ordinances or governmental rules or regulations or failures to obtain or maintain in effect such licenses, certificates, permits, franchises and other governmental authorizations could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **6.05 Insurance**. The Company will, and will cause each of its Restricted Subsidiaries to, maintain, with financially sound and reputable insurers, insurance with respect to their respective properties and businesses against such risks, of such types, on such terms and in such amounts as is customary in the case of entities of established reputations engaged in the same or a similar business and similarly situated.
- **6.06 Maintenance of Properties**. The Company will, and will cause each of its Restricted Subsidiaries to, maintain and keep, or cause to be maintained and kept, their respective properties in good repair, working order and condition (other than ordinary wear and tear), so that the business carried on in connection therewith may be properly conducted at all times, provided that this Section shall not prevent the Company or any Restricted Subsidiary from discontinuing the operation and the maintenance of any of its properties or failing to carry on any such business if the Company has concluded that such discontinuance or failure could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **6.07 Payment of Taxes and Claims.** The Company will, and will cause each of its Subsidiaries to, file all income tax and similar tax returns required to be filed in any jurisdiction and to pay and discharge all taxes shown to be due and payable on such returns and pay all other taxes, assessments, governmental charges, or levies imposed on them or any of their properties, assets, income or franchises, to the extent such taxes and assessments have become due and payable and before they have become delinquent and all claims for which sums have become due and payable that have or might become a Lien on properties or assets of the Company or any

Subsidiary, provided that neither the Company nor any Subsidiary need file any such returns or pay any such tax or assessment or claims (i) if the amount, applicability or validity thereof is contested by the Company or such Subsidiary on a timely basis in good faith and in appropriate proceedings, diligently conducted, and the Company or a Subsidiary has established adequate reserves therefor in accordance with GAAP on the books of the Company or such Subsidiary, or (ii) to the extent that the failure to file such returns and the nonpayment of all such taxes, assessments and claims, either individually or in the aggregate, could not reasonably be expected to have a Material Adverse Effect.

- **6.08 Corporate Existence, Etc.** Subject to Section 7.02, the Company will at all times preserve and keep in full force and effect its corporate existence, and the Company will at all times preserve and keep in full force and effect the corporate existence of each of its Restricted Subsidiaries (unless merged into the Company or a Restricted Subsidiary) and all rights and franchises of the Company and its Restricted Subsidiaries, unless, in the good faith judgment of the Company, the termination of or failure to preserve and keep in full force and effect such corporate existence, right or franchise could not, individually or in the aggregate, have a Material Adverse Effect.
- **6.09 Books and Records**. The Company will, and will cause each Restricted Subsidiary to, (a) maintain proper books of record and account, in which full, true and correct entries in conformity with GAAP consistently applied shall be made of all financial transactions and matters involving the assets and business of the Company or such Restricted Subsidiary, as the case may be; and (b) maintain such books of record and account in material conformity with all applicable requirements of any Governmental Authority having regulatory jurisdiction over the Company or such Restricted Subsidiary, as the case may be.
- **6.10 Purpose of Credits**. Each Credit shall be used directly or indirectly to cover (a) a default in the performance of any non-financial or commercial obligations of the Company or any Subsidiary under specific contracts, including a Credit issued in favor of a bank or other surety who in connection therewith issues a guarantee or similar undertaking, performance bond, surety bond or other similar instrument that covers a default in the performance of any non-financial or commercial obligations under specific contracts or the payment of any financial contractual obligation, or (b) the payment of any financial contractual obligation of the Company or any Subsidiary.

ARTICLE VII NEGATIVE COVENANTS

So long as any Lender shall have any Credit-Linked Deposit, any Letter of Credit Linked Note or any Term Note hereunder or any Credit shall remain outstanding:

7.01 Transactions with Affiliates. The Company will not, and will not permit any Restricted Subsidiary to, enter into directly or indirectly any Material transaction or Material group of related transactions (including without limitation the purchase, lease, sale or exchange of properties of any kind or the rendering of any service) with any Affiliate (other than the Company or another Restricted Subsidiary), except in the ordinary course and pursuant to the reasonable requirements of the Company's or such Restricted Subsidiary's business and upon

fair and reasonable terms no less favorable to the Company or such Restricted Subsidiary than would be obtainable in a comparable arm's-length transaction with a Person not an Affiliate.

7.02 Merger, Consolidation, Sale of Assets, etc.

- (a) The Company will not, and will not permit any Restricted Subsidiary to, consolidate with or merge with any other Person unless immediately after giving effect to any consolidation or merger no Default or Event of Default would exist and:
- (i) in the case of a consolidation or merger of a Restricted Subsidiary, either (x) the surviving or continuing corporation is the Company or another Restricted Subsidiary or (y) such consolidation or merger, if considered as the sale of the assets of such Restricted Subsidiary to such other Person, would be permitted by Section 7.02(c); and
- (ii) in the case of a consolidation or merger of the Company, the successor corporation which results from such consolidation or merger (the "surviving corporation"), if not the Company, shall have executed and delivered to the Issuer, the Administrative Agent and each Lender in writing its assumption of the due and punctual payment of all Credit Obligations, and the due and punctual performance and observation of all covenants in this Agreement, the Letter of Credit Linked Notes and the Term Notes, to be performed or observed by the Company hereunder and under the Credit Documents and shall furnish to the Issuer, the Administrative Agent and such Lenders a customary opinion of counsel solely to the effect that the instrument of assumption has been duly authorized, executed and delivered and constitutes the legal, valid and binding contract and agreement of the surviving corporation enforceable in accordance with its terms, except as enforcement of such terms may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the enforcement of creditors' rights generally and by general equitable principles.
- (b) The Company will not sell, lease (as lessor) or otherwise transfer all or substantially all of its assets in a single transaction or series of transactions to any Person unless:
 - (i) immediately after giving effect to such transaction, no Default or Event of Default shall have occurred and be continuing; and
- (ii) the successor corporation to which all or substantially all of the Company's assets have been sold, leased or transferred (the "successor corporation") shall have executed and delivered to the Issuer, the Administrative Agent and each Lender in writing its assumption of the due and punctual payment of all Credit Obligations, and the due and punctual performance and observation of all covenants in this Agreement, the Letter of Credit Linked Notes and the Term Notes, to be performed or observed by the Company hereunder and under the Credit Documents and shall furnish to the Issuer, the Administrative Agent and such Lenders a customary opinion of counsel solely to the effect that the instrument of assumption has been duly authorized, executed and delivered and constitutes the legal, valid and binding contract and agreement of the successor corporation enforceable in accordance with its terms, except as enforcement of such terms may be limited by bankruptcy, insolvency, reorganization, moratorium and similar laws affecting the enforcement of creditors' rights generally and by general equitable principles.

No such conveyance, transfer or lease of all or substantially all of the assets of the Company shall have the effect of releasing the Company or any successor corporation that shall theretofore have become such in the manner prescribed in this Section 7.02 from its liability under this Agreement or the Credit Documents.

- (c) The Company will not, and will not permit any Restricted Subsidiary to, sell, lease (as lessor), transfer, abandon or otherwise dispose of (collectively, a "Disposition") any assets in one or a series of transactions (including, without limitation, in any sale-leaseback transaction), other than those assets consisting of accounts receivable sold through any Permitted Receivables Transaction, to any Person other than:
- (i) Dispositions in the ordinary course of business of inventory or of equipment, fixtures, supplies or materials no longer required in the business of the Company or any of its Restricted Subsidiaries or that is obsolete;
 - (ii) Dispositions by the Company to a Restricted Subsidiary or by a Restricted Subsidiary to the Company or to another Restricted Subsidiary;
 - (iii) the sale of assets for cash or other property if all of the following conditions are met:
 - (A) such assets (valued at book value) do not constitute a "substantial part" of the assets of the Company and its Restricted Subsidiaries;
 - (B) in the opinion of a Responsible Officer of the Company, the sale is for fair value and is in the best interests of the Company; and
 - (C) immediately after the consummation of the transaction and after giving effect thereto, no Default or Event of Default would exist; or

(iv) the sale of assets meeting the conditions set forth in clauses (B) and (C) of clause (iii) above, as long as the cash proceeds from such sale in excess of a substantial part of the assets of the Company and its Restricted Subsidiaries (net of expenses of sale and an amount reasonably estimated by the Company to represent incremental income taxes due as a result of such sale) are (i) applied within 365 days of the date of receipt to the acquisition of productive assets useful and intended to be used in the operation or expansion of the business of the Company or its Restricted Subsidiaries, and/or (ii) used to repay any Indebtedness of the Company or its Restricted Subsidiaries (other than (x) Indebtedness that is in any manner subordinated in right of payment in any respect to Indebtedness arising or incurred under this Agreement, (y) Indebtedness owing to the Company, any of its Subsidiaries or any Affiliate, and (z) Indebtedness in respect of any revolving credit or similar credit facility providing the Company or any of its Restricted Subsidiaries with the right to obtain loans or other extensions of credit from time to time, except to the extent that in connection with such payment of Indebtedness the availability of credit under such credit facility is permanently reduced by an amount not less than the amount of such proceeds applied to the payment of such Indebtedness).

- (d) For purposes of Section 7.02(c), a sale of assets will be deemed to involve a "substantial part" of the assets of the Company and its Restricted Subsidiaries if the book value of such assets, together with all other assets sold, including mergers effected pursuant to Section 7.02(a)(i)(y) and the issuance, sale, transfer or disposition of stock or other Securities of Restricted Subsidiaries (except to the Company or another Restricted Subsidiary) effected pursuant to Section 7.02(e), during the same fiscal year (except (1) those assets sold pursuant to clauses (i) and (ii) of Section 7.02(c), and (2) those assets consisting of receivables sold through any Permitted Receivables Transaction permitted under Section 7.03(i)) equals at least 15.00% of the Consolidated Total Assets of the Company and its Restricted Subsidiaries determined as of the end of the immediately preceding fiscal year.
- (e) The Company will not permit any Restricted Subsidiary to issue shares of stock or other Securities to any Person other than the Company or another Restricted Subsidiary. The Company will not, and will not permit any Restricted Subsidiary to, issue, sell, transfer or otherwise dispose of its interest in any stock or any other Securities (or any options or warrants to purchase stock or other Securities exchangeable for or convertible into stock or other Securities) of any Restricted Subsidiary (except to the Company or another Restricted Subsidiary) unless such issuance, sale, transfer or disposition (valued at book value) would be permitted under Section 7.02(c).
- **7.03** Liens. The Company will not, and will not permit any of its Restricted Subsidiaries to, directly or indirectly create, incur, assume or permit to exist (upon the happening of a contingency or otherwise) any Lien on or with respect to any of its properties or assets, whether now owned or hereafter acquired, except for the following (which are collectively referred to as "Permitted Liens"):
- (a) Mechanics', materialmen's, carriers' or other like liens arising in the ordinary course of business (including construction of facilities) in respect of obligations that are not due or which are being contested in good faith;
- (b) Liens arising by reason of deposits with, or the giving of any form of security to, any governmental agency or any body created or approved by law or governmental regulation, which is required by law or governmental regulation as a condition to the transaction of any business or the exercise of any privilege, franchise or license (including, without limitation, any Lien arising by reason of one or more letters of credit in connection with any international waste management contract to be performed by the Company or any of its Subsidiaries or their respective Affiliates);
- (c) Liens for taxes, assessments or governmental charges or levies not yet delinquent or Liens for taxes, assessments or governmental charges or levies already delinquent but the validity of which is being contested in good faith;
- (d) Liens (including judgment liens) arising in connection with legal proceedings so long as such proceedings are being contested in good faith and, in the case of judgment liens, execution thereon is stayed;

- (e) Landlords' liens on fixtures located on premises leased by the Company or any Restricted Subsidiary in the ordinary course of business;
- (f) [intentionally omitted]
- (g) Liens arising by reason of deposits to qualify the Company or any Restricted Subsidiary to conduct business, to maintain self-insurance, or to obtain the benefit of, or comply with, Laws;
- (h) Liens that secure any Indebtedness of a Restricted Subsidiary owing to the Company or another Restricted Subsidiary or by the Company to a Restricted Subsidiary;
- (i) Liens created or deemed to be created under Permitted Receivables Transactions at any time provided such Liens do not extend to any property or assets other than the trade receivables sold pursuant to such Permitted Receivables Transactions, interests in the goods or products (including returned goods and products), if any, relating to the sales giving rise to such trade receivables; any security interests or other Liens and property subject thereto (other than on any leases or related lease payment rights or receivables between the Company and any of its Restricted Subsidiaries, as lessors or sublessors) from time to time purporting to secure the payment by the obligors describing the collateral securing such trade receivables (together with any financing statements signed by such obligors describing the collateral securing such trade receivables) pursuant to such Permitted Receivables Transactions;
 - (j) Liens existing at Closing and reflected on Schedule 7.03 hereto;
- (k) (i) Liens upon property or assets which are created prior to or contemporaneously with, or within 360 days after, (1) in the case of the acquisition of such property or assets, the completion of such acquisition; and (2) in the case of the construction, development or improvement of such property or assets, the later to occur of the completion of such construction, development or improvement or the commencement of operation or use of the property or assets, which Lien secures or provides for the payment, financing or refinancing, directly or indirectly, of all or any part of the acquisition cost of such property or assets or the cost of construction, development or improvement thereof; or (ii) any Lien upon property or assets existing at the time of the acquisition thereof, which Lien secures obligations assumed by the Company or any Restricted Subsidiary; or (iii) any conditional sales agreement or other title retention agreement with respect to any property or assets acquired by the Company or any Restricted Subsidiary; or (iv) any Lien existing on the property or assets or shares of stock of a corporation or firm at the time such corporation or firm is merged into or consolidated with the Company or any Restricted Subsidiary or at the time such corporation becomes a Restricted Subsidiary; or (v) any Lien existing on the property, assets or shares of stock of any successor which shall have become the Company in accordance with the provisions set forth in Section 7.02; provided, in each case, that any such Lien described in the foregoing clauses (ii), (iii), (iv) or (v) does not attach to or affect property or assets owned by the Company or any Restricted Subsidiary prior to the event referred to in such clauses; provided further (for avoidance of doubt) that any such Lien described in the foregoing clauses (ii), (iii) or (iii) may be created in favor of any Governmental Authority.

- (l) Any extension, renewal or refunding (or successive extensions, renewals or refundings) in whole or in part of any Indebtedness secured by any Lien referred to in the foregoing clauses, inclusive, provided that the principal amount of the Indebtedness secured is not increased or the maturity thereof reduced and the Lien securing such Indebtedness shall be limited to the property or assets that, immediately prior to such extension, renewal or refunding, secured such Indebtedness and additions to such property or assets; and
- (m) Liens securing Priority Debt, provided that Priority Debt shall not at any time exceed 15.00% of Consolidated Tangible Assets as of the then most recently ended fiscal quarter or fiscal year of the Company for which financial information is reportable pursuant to Sections 6.01(a) and (b).
- **7.04 Limitation on Consolidated Debt.** The Company will not permit at any time the ratio of (a) Consolidated Debt to (b) Adjusted Consolidated EBITDA for the most recently ended period of four consecutive fiscal quarters of the Company to be greater than 3.75 to 1.00.
- **7.05 Limitation on Priority Debt**. The Company will not permit Priority Debt at any time to exceed 15.00% of Consolidated Tangible Assets as of the end of the most recently ended fiscal quarter or fiscal year of the Company for which financial information is reportable pursuant to Sections 6.01(a) and (b).
- **7.06 Nature of Business**. The Company will not, and will not permit any Restricted Subsidiary to, engage in any business if, as a result, the general nature of the business of the Company and its Restricted Subsidiaries, taken as a whole, which would then be engaged in by the Company and its Restricted Subsidiaries would be substantially changed from the nature of the business engaged in by the Company and its Restricted Subsidiaries, taken as a whole, on the Closing Date.
- **7.07 Designation of Unrestricted Subsidiaries.** The Company may designate any newly-formed or newly-acquired Subsidiary as a Restricted Subsidiary or an Unrestricted Subsidiary; provided that if the Company shall fail to so designate by the end of the fiscal quarter in which such Subsidiary is formed or acquired, such Subsidiary shall be deemed to be a Restricted Subsidiary. In addition, the Company may designate any Restricted Subsidiary as an Unrestricted Subsidiary and any Unrestricted Subsidiary as a Restricted Subsidiary; provided that, (a) if such Subsidiary initially is a Restricted Subsidiary, then such Restricted Subsidiary may be subsequently redesignated as a Restricted Subsidiary may be subsequently designated as a Restricted Subsidiary may be subsequently designated as a Restricted Subsidiary may be subsequently redesignated as an Unrestricted Subsidiary, but no further changes in designation may be made; (c) immediately before and after designation of a Restricted Subsidiary as an Unrestricted Subsidiary or of an Unrestricted Subsidiary as a Restricted Subsidiary there exists no Default or Event of Default; and (d) Holdings may not be designated as an Unrestricted Subsidiary.

7.08 Margin Stock. The Company shall not use the Credits, the Advances or the Term Loans, whether directly or indirectly, and whether immediately, incidentally or ultimately, to purchase or carry margin stock with in the meaning of Regulation U of the FRB.

ARTICLE VIII EVENTS OF DEFAULT AND REMEDIES

- **8.01 Events of Default**. Any of the following shall constitute an Event of Default:
- (a) Non-Payment of Drawing or Principal Amount. The Company fails to pay when and as required to be paid herein or in any other Credit Document (i) within six Business Days after the applicable Reimbursement Date of any drawing under any Credit, the amount of such drawing (unless within such six Business Day period a Request for Term Loans has been received pursuant to Section 2.03(b), and such Term Loans are made in accordance with the terms and conditions of Sections 2.03, 4.02 and 4.03 within 15 Business Days of the receipt by the Administrative Agent of such Request for Term Loans), or (ii) on the Maturity Date, the aggregate principal amount of any Term Loan or Advances outstanding; or
- (b) Non-Payment of Interest and Other Amounts. The Company defaults in the payment of any interest on any Letter of Credit Linked Note or Term Note, any Facility Fee or Fronting Fee due hereunder, or other amount payable hereunder or under any other Credit Document for more than five (5) Business Days after the same becomes due and payable; or
- (c) Specific Covenants. The Company defaults in the performance of or compliance with (i) Section 7.05, which default is not remedied within ten (10) Business Days after a Responsible Officer obtaining actual knowledge of such default, or (ii) Sections 7.02, 7.03, or 7.04; or
- (d) Other Defaults. The Company defaults in the performance of or compliance with any term contained herein (other than those referred to in paragraphs (a), (b) and (c) of this Section 8.01) and such default is not remedied within thirty (30) days after a Responsible Officer obtaining actual knowledge of such default; or
- (e) Representations and Warranties. Any representation or warranty made in writing by or on behalf of the Company or Holdings or by any officer of the Company or Holdings in this Agreement or in any writing furnished in connection with the transactions contemplated hereby proves to have been false or incorrect in any material respect on the date as of which made; or
- (f) Cross-Defaults. (i) A default in the payment of any principal of or premium or make-whole amount or interest on any Material Debt has occurred and continues beyond any period of grace provided with respect thereto, or (ii) any other default in the performance of or compliance with any term of any evidence of any Material Debt or of any mortgage, indenture or other agreement relating thereto exists, and as a consequence of such default such Material Debt has become, or has been declared (or one or more Persons are entitled to declare such Material Debt to be), due and payable before its stated maturity or before its regularly scheduled dates of payment, or (iii) as a consequence of the occurrence or continuation of any event, (x) the

Company or any Restricted Subsidiary has become obligated to purchase or repay any Material Debt before its regular maturity or before its regularly scheduled dates of payment, or (y) one or more Persons have the right to require the Company or any Restricted Subsidiary so to purchase or repay any Material Debt, or (iv) an event or condition shall occur which results in an "Event of Default" under the Ten Year Agreement or the Five Year Agreement; or

- (g) Failure to Pay Debts; Voluntary Proceedings. The Company, any Restricted Subsidiary or Holdings (i) is generally not paying, or admits in writing its inability to pay, its debts as they become due, (ii) files, or consents by answer or otherwise to the filing against it of, a petition for relief or reorganization or arrangement or any other petition in bankruptcy, for liquidation or to take advantage of any bankruptcy, insolvency, reorganization, moratorium or other similar law of any jurisdiction, (iii) makes an assignment for the benefit of its creditors, (iv) consents to the appointment of a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, (v) is adjudicated as insolvent or to be liquidated, or (vi) takes corporate action for the purpose of any of the foregoing; or
- (h) Involuntary Proceedings. A court or governmental authority of competent jurisdiction enters an order appointing, without consent by the Company, any Restricted Subsidiary or Holdings, as applicable, a custodian, receiver, trustee or other officer with similar powers with respect to it or with respect to any substantial part of its property, or constituting an order for relief or approving a petition for relief or reorganization or any other petition in bankruptcy or for liquidation or to take advantage of any bankruptcy or insolvency law of any jurisdiction, or ordering the dissolution, winding-up or liquidation of the Company, any Restricted Subsidiary or Holdings, or any such petition shall be filed against the Company, any Restricted Subsidiary or Holdings and such petition shall not be dismissed within 60 days; or
- (i) Judgments. A final judgment or judgments for the payment of money aggregating in excess of 10.00% of Consolidated Net Worth are rendered against one or more of the Company, any Restricted Subsidiary or Holdings and which judgments are not, within 60 days after entry thereof, bonded, discharged or stayed pending appeal, or are not discharged within 60 days after the expiration of such stay; or
- (j) Invalidity of Holdings Guaranty. If for any reason (other than payment in full of all obligations under this Agreement, the Letter of Credit Linked Notes and the Term Notes) the Holdings Guaranty shall cease to be in full force and effect (other than as expressly provided in Section 11.09 of this Agreement), or the Company or Holdings (or any other Person on behalf of the Company or Holdings) shall contest the validity or enforceability of such Holdings Guaranty, deny that Holdings has any further liability hereunder, or seek to revoke such Holdings Guaranty; or
- (k) ERISA. If (i) any Plan shall fail to satisfy the minimum funding standards of ERISA or the Code for any plan year or part thereof or a waiver of such standards or extension of any amortization period is sought or granted under Section 412 of the Code, (ii) a notice of intent to terminate any Plan shall have been or is reasonably expected to be filed with the PBGC or the PBGC shall have instituted proceedings under ERISA Section 4042 to terminate or appoint a trustee to administer any Plan or the PBGC shall have notified the Company or any

ERISA Affiliate that a Plan may become a subject of any such proceedings, (iii) the aggregate "amount of unfunded benefit liabilities" (within the meaning of Section 4001(a)(18) of ERISA) under all Plans, determined in accordance with Title IV of ERISA, shall exceed 10.00% of Consolidated Net Worth, (iv) the Company or any ERISA Affiliate shall have incurred or is reasonably expected to incur any liability pursuant to Title I or IV of ERISA or the penalty or excise tax provisions of the Code relating to employee benefit plans, (v) the Company or any ERISA Affiliate withdraws from any Multiemployer Plan, or (vi) the Company or any Subsidiary establishes or amends any employee welfare benefit plan that provides post-employment welfare benefits in a manner that would increase the liability of the Company or any Subsidiary thereunder; and any such event or events described in clauses (i) through (vi) above, either individually or together with any other such event or events, could reasonably be expected to have a Material Adverse Effect; or

(l) Change in Control. At any time any Person or group of Persons (within the meaning of Section 13 or 14 of the Securities Exchange Act of 1934, as amended) shall have acquired beneficial ownership (within the meaning of Rule 13d-3 promulgated by the Securities and Exchange Commission under said Act) of 25% or more of the outstanding shares of common voting stock of the Company; or during any period of twelve consecutive calendar months, individuals who were directors of the Company on the first day of such period (together with any new directors whose election by such board or whose nomination for election by the shareholders of the Company was approved by a vote of a majority of the directors still in office who were either directors at the beginning of such period or whose election or nomination for election was previously so approved) shall cease to constitute a majority of the board of directors of the Company.

As used in Section 8.01(k), the terms "employee benefit plan" and "employee welfare benefit plan" shall have the respective meanings assigned to such terms in Section 3 of ERISA.

- **8.02 Remedies Upon Event of Default**. If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Credit-Linked Lenders, take any or all of the following actions, which actions shall be taken simultaneously under this Agreement, the Ten Year Agreement and the Five Year Agreement:
- (a) declare any obligation of the Issuer to make Credit Extensions and the obligations of the Lenders to make Term Loans to be terminated, whereupon such commitments and obligation shall be terminated;
- (b) declare all amounts owing or payable hereunder (including the Prepayment Premium, if any) or under any other Credit Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Company;
- (c) require that the Company Cash Collateralize the Credit Obligations (other than Advances and Term Loans) (in an amount equal to the Outstanding Amount thereof, as such amount may vary from time to time); and

(d) exercise on behalf of itself, the Issuer and the Lenders all rights and remedies available to it, the Issuer and the Lenders under the Credit Documents or applicable law;

provided, however, that upon the occurrence of an Event of Default pursuant to Sections 8.01(g) or (h), any obligation of the Issuer or the Lenders to make Credit Extensions shall automatically terminate, all unpaid amounts as aforesaid shall automatically become due and payable, and the obligation of the Company to Cash Collateralize the Credit Obligations (other than Advances and Term Loans) as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent or any Lender.

Thereafter, at the request of the Required Lenders, the Administrative Agent shall (x) withdraw from the Credit-Linked Deposit Account and distribute to the Lenders an amount equal to the excess of the Total Credit-Linked Deposit over the Outstanding Amount of the Credit Obligations (other than Advances and Term Loans), and (y) subject to, and to the extent of, the Company's compliance with any Cash Collateralization requirements as aforesaid, withdraw from the Credit-Linked Deposit Account and distribute to the Lenders the aggregate amount of any remaining Credit-Linked Deposits to the extent of the amount credited to the Credit-Linked Deposit Account; in each case subject to any notice required pursuant to the Money Market Account Agreement.

8.03 Application of Funds. After the exercise of remedies provided for in Section 8.02 (or after the obligations of the Issuer to make Credit Extensions have been automatically terminated and the Credit Obligations have automatically been required to be Cash Collateralized as set forth in the proviso to Section 8.02), any amounts received by the Administrative Agent, the Issuer or the Lenders on account of the Credit Obligations shall be applied by the Administrative Agent in the following order:

First, to payment of fees, indemnities, expenses and other amounts (including interest thereon, Attorney Costs and amounts payable under Article III) payable to the Administrative Agent in its capacity as such;

Second, to payment of fees, indemnities and other amounts (excluding principal and interest on any Borrowings but including interest, if any, on any such fees, indemnities and other amounts) payable to the Issuer and the Lenders (including Attorney Costs and amounts payable under Article III), ratably among them in proportion to the amounts described in this clause Second payable to them;

Third, to payment of that portion of the Credit Obligations constituting accrued and unpaid interest on the Borrowings, to the Issuer and ratably among the Lenders in proportion to the respective amounts described in this clause Third payable to them;

Fourth, to payment of that portion of the Credit Obligations constituting unpaid principal of the Borrowings, to the Issuer and ratably among the Lenders in proportion to the respective amounts described in this clause Fourth held by them;

Fifth, to the Administrative Agent for the account of the Issuer, to Cash Collateralize that portion of Credit Obligations comprised of the aggregate undrawn amount of Credits; and

Last, the balance, if any, after all of the Credit Obligations have been paid in full, to the Company or as otherwise required by Law.

Subject to Section 2.01(c), amounts used to Cash Collateralize the aggregate undrawn amount of Credits pursuant to clause Fifth above shall be applied to satisfy drawings under such Credits as they occur. If any amount remains on deposit as Cash Collateral after all Credits have either been fully drawn or expired, such remaining amount shall be applied to the other Credit Obligations, if any, in the order set forth above.

ARTICLE IX ADMINISTRATIVE AGENT

- **9.01 Appointment and Authorization of Administrative Agent.** (a) Each of the Lenders and the Issuer hereby irrevocably appoints, designates and authorizes the Administrative Agent to take such action on its behalf under the provisions of this Agreement and each other Credit Document and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement or any other Credit Document, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere herein or in any other Credit Document, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, nor shall the Administrative Agent have or be deemed to have any fiduciary relationship with the Issuer, any Lender or participant, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Credit Document or otherwise exist against the Administrative Agent. Without limiting the generality of the foregoing sentence, the use of the term "agent" herein and in the other Credit Documents with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable Law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only an administrative relationship between independent contracting parties.
- (b) The Issuer shall act on behalf of the Lenders with respect to any Credits issued by it and the Issuer Documents associated therewith, and the Issuer shall have all of the benefits and immunities (i) provided to the Administrative Agent in this Article IX (other than Section 9.07) with respect to any acts taken or omissions suffered by the Issuer in connection with Credits issued by it or proposed to be issued by it and the Issuer Documents pertaining to such Credits as fully as if the term "Administrative Agent" as used in this Article IX and in the definition of "Agent-Related Person" included the Issuer with respect to such acts or omissions, and (ii) as additionally provided herein with respect to the Issuer.
- **9.02 Delegation of Duties.** The Administrative Agent may execute any of its duties under this Agreement or any other Credit Document by or through agents, employees or attorneys-in-fact and shall be entitled to advice of counsel and other consultants or experts concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects in the absence of gross negligence or willful misconduct.

9.03 Liability of Administrative Agent. No Agent-Related Person shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Credit Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct in connection with its duties expressly set forth herein), or (b) be responsible in any manner to any Lender or participant for any recital, statement, representation or warranty made by any Credit Party or any officer thereof, contained herein or in any other Credit Document, or in any certificate, report, statement or other document referred to or provided for in, or received by the Administrative Agent under or in connection with, this Agreement or any other Credit Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Credit Document, or for any failure of any Credit Party or any other party to any Credit Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Lender or participant to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Credit Document, or to inspect the properties, books or records of any Credit Party or any Affiliate thereof.

9.04 Reliance by Administrative Agent. (a) The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, communication, signature, resolution, representation, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, electronic mail message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to any Credit Party), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent shall be fully justified in failing or refusing to take any action under any Credit Document unless it shall first receive such advice or concurrence of the Required Credit-Linked Lenders as it deems appropriate and, if it so requests, it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Credit Document in accordance with a request or consent of the Required Lenders or Required Credit-Linked Lenders, as applicable (or such greater number of Lenders as may be expressly required hereby in any instance) and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Lenders.

(b) For purposes of determining compliance with the conditions specified in Section 4.01, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

9.05 Notice of Default. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to defaults in the payment of principal, interest and fees required to be paid to the Administrative Agent for the account of the Lenders, unless the Administrative Agent shall have received written notice from the Issuer, a Lender or the Company referring to this Agreement, describing

such Default or Event of Default and stating that such notice is a "notice of default." The Administrative Agent will notify the Lenders of its receipt of any such notice. The Administrative Agent shall take such action with respect to such Default or Event of Default as may be directed by the Required Credit-Linked Lenders in accordance with Article VIII; provided, however, that unless and until the Administrative Agent has received any such direction, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable or in the best interest of the Lenders.

9.06 Credit Decision; Disclosure of Information by Administrative Agent. Each Lender and the Issuer acknowledges that no Agent-Related Person has made any representation or warranty to it, and that no act by the Administrative Agent hereafter taken, including any consent to and acceptance of any assignment or review of the affairs of any Credit Party or any Affiliate thereof, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender as to any matter, including whether Agent-Related Persons have disclosed material information in their possession. Each Lender and the Issuer represents to the Administrative Agent that it has, independently and without reliance upon any Agent-Related Person and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of the Credit Parties and their respective Subsidiaries, and made its own decision to enter into this Agreement and to extend credit to the Company and any other Credit Parties hereunder. Each Lender and the Issuer hereby acknowledge that it has not relied on any Agent-Related Person (other than as a recipient of legal opinions of counsel to the Administrative Agent delivered pursuant to Section 4.01(a)(vi)) with respect to any bank or other regulatory Laws relating to the transactions contemplated hereby. Each Lender and the Issuer also represents that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Credit Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of the Company and the other Credit Parties. Except for notices, reports and other documents expressly required to be furnished to the Issuer and the Lenders by the Administrative Agent herein, the Administrative Agent shall not have any duty or responsibility to provide the Issuer or any Lender with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of any of the Credit Parties or any of their respective Affiliates which may come into the possession of any Agent-Related Person.

9.07 Indemnification of Administrative Agent. The Lenders shall indemnify upon demand, in Dollars, the Administrative Agent, in its capacity as Administrative Agent, and the Affiliates (to the extent such Affiliates are acting for the Administrative Agent in such capacity), officers, directors, employees, agents and attorneys-in-fact thereof who are acting for or on behalf of the Administrative Agent (pursuant to the express authorization of, or by express delegation from, the Administrative Agent) in connection with this Agreement (to the extent not reimbursed by or on behalf of any Credit Party and without limiting the obligation of any Credit Party to do so), *pro rata* in accordance with its Pro Rata Share, and hold harmless each such

Person from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Person in any way relating to or arising out of or in connection with (before or after the Closing Date) the execution, delivery, enforcement, performance or administration of any Credit Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby (the "Agent Indemnified Liabilities"), incurred by any such Person in connection with the Administrative Agent's exercise of powers and performance of duties expressly provided for under this Agreement; provided, however, that no Lender shall be liable for the payment to any such Person of any portion of such Agent Indemnified Liabilities resulting from such Person's own gross negligence or willful misconduct; and provided further, however, that no action taken in accordance with the directions of the Required Lenders or Required Credit-Linked Lenders, as applicable, shall be deemed to constitute gross negligence or willful misconduct for purposes of this Section; and provided further, however, that the Agent Indemnified Liability was incurred by or asserted against the Administrative Agent in its capacity as such, or such Affiliate or other Person acting for the Administrative Agent in such capacity. Without limitation of the foregoing, each Lender shall reimburse the Administrative Agent upon demand for such Lender's Pro Rata Share of any costs or out-of-pocket expenses (including reasonable Attorney Costs) incurred by the Administrative Agent (i) in connection with any enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Credit Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Credit Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), or (ii) otherwise, while an Event of Default exists, to the extent that the Administrative Agent is not reimbursed for such expenses by or on behalf of the Company within 30 days after demand therefor; provided in each case that such cost or expense was incurred by the Administrative Agent in its capacity as such; and provided, however, in each case that before making any demand on the Lenders for reimbursement of any such cost or expense, the Administrative Agent shall have first requested the Company to pay such amounts and made a good faith determination that the Company will not honor such request. Notwithstanding the foregoing, no Lender shall be liable for any indirect, special, punitive or consequential damages claimed by any Person under this Section 9.07 (other than such damages claimed by such Person as a result of such damages being claimed against such Person by a third party). The undertaking in this Section shall survive termination of the Credit-Linked Deposits, the payment of all other Credit Obligations and other amounts due hereunder and the resignation of the Administrative Agent. Notwithstanding the foregoing, (a) no Lender shall be obligated to reimburse or indemnify the Administrative Agent, in its capacity as Administrative Agent, and the Affiliates (to the extent such Affiliates are acting for the Administrative Agent in such capacity), officers, directors, employees, agents and attorneys-in-fact thereof, in respect of the transactions contemplated by Section 10.19 resulting from a Deposit Account Interest Non-Payment Event or in respect of any other nonpayment of funds owing to the Lenders, if such other nonpayment results solely from the Administrative Agent's nonperformance of its paying agency duties hereunder (and not from any action or inaction of the Company or any other unrelated Person), and (b) no Lender shall be obligated to reimburse or indemnify BAS.

9.08 Administrative Agent in its Individual Capacity. Bank of America and its Affiliates may make loans to, issue letters of credit and guarantees or other undertakings for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with each of the Credit Parties and their respective Affiliates as though Bank of America were not the Administrative Agent or the Issuer hereunder and without notice to or consent of the Lenders. The Lenders acknowledge that, pursuant to such activities, Bank of America or its Affiliates may receive information regarding any Credit Party or its Affiliates (including information that may be subject to confidentiality obligations in favor of such Credit Party or such Affiliate) and acknowledge that the Administrative Agent shall be under no obligation to provide such information to them.

9.09 Successor Administrative Agent. The Administrative Agent may resign as Administrative Agent upon 30 days' notice to the Lenders; provided that any such resignation by Bank of America shall also constitute its resignation as Issuer under this Agreement; provided, however, the Administrative Agent may not resign if (i) it is the Issuer of any outstanding Credits under this Agreement or (ii) any Credit-Linked Deposits of any Lender exist. If the Administrative Agent resigns under this Agreement, the Required Credit-Linked Lenders shall appoint a successor administrative agent for the Lenders which successor administrative agent shall require the consent of the Company at all times other than during the existence of any Default or Event of Default (which consent of the Company shall not be unreasonably withheld or delayed). If no successor administrative agent is appointed prior to the effective date of the resignation of the Administrative Agent, the Administrative Agent may appoint, after consulting with the Credit-Linked Lenders and the Company, a successor administrative agent from among the Credit-Linked Lenders with the consent of the Company (which consent of the Company shall be required at all times other than during the existence of any Default or Event of Default and shall not be unreasonably withheld or delayed). The successor administrative agent under this Agreement, the Ten Year Agreement and the Five Year Agreement shall at all times be the same Person. Upon the acceptance of its appointment as successor administrative agent shall succeed to all the rights, powers and duties of the retiring Administrative Agent and the term "Administrative Agent" shall mean such successor administrative agent, and the retiring Administrative Agent and the term "Administrative Agent" shall mean such successor administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article IX and Sections 10.04 and 10.05 shall inure to its benefit as to any actions taken or omitted to be taken

ARTICLE X MISCELLANEOUS

10.01 Amendments, Etc. No amendment or waiver of any provision of this Agreement or any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)), and no consent to any departure by the Company or any other Credit Party therefrom (other than a departure of Holdings pursuant to Section 11.09 of this Agreement), shall be effective unless (A) in writing signed by the Required Credit-Linked Lenders and the Company or the applicable Credit Party, as the case may be, and acknowledged by the Administrative

Agent and (B) a corresponding amendment, waiver or consent under (i) the Ten Year Agreement and (ii) the Five Year Agreement is also effective (it being understood that the same terms and conditions with respect to such amendment, waiver or consent shall apply to all Credit-Linked Lenders), and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no such amendment, waiver or consent shall, without the written consent of each Lender with Credit Obligations directly affected (whose consent shall be required for any such amendment, waiver or consent in addition to that of the Required Credit-Linked Lenders):

- (a) waive any condition set forth in Section 4.01;
- (b) extend the Maturity Date or increase the Credit-Linked Deposit of any Lender (or reinstate any obligation to make Credit Extensions terminated pursuant to Section 8.02 in which such Lender participates);
- (c) postpone any date fixed by this Agreement or any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)) for any payment of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b));
- (d) reduce the principal of, or the rate of interest specified herein on, any Borrowing, or (subject to clause (iii) of the proviso below) any fees or other amounts payable hereunder or under any other Credit Document (other than Issuer Documents to the extent permitted by Section 2.01(b)); provided, however, that only the consent of the Required Credit-Linked Lenders shall be necessary to amend the definition of "Default Rate," the amount of any increase in the Applicable Rate pursuant to the definition thereof during the existence of any Event of Default, or the interest rate applicable from time to time to Borrowings under Section 2.04(a)(ii), or to waive any obligation of the Company to pay interest at the Default Rate or at the rate set forth in Section 2.04(a)(ii), or to increase the Applicable Rate during the existence of an Event of Default;
 - (e) change Section 2.08 or Section 8.03 in a manner that would alter the *pro rata* sharing of payments required thereby;
- (f) change any provision of this Section 10.01, or the definition of "Required Credit-Linked Lenders", "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder; or
- (g) release Holdings from any of the Guaranteed Obligations under the Holdings Guaranty (other than as expressly provided in Section 11.09 of this Agreement);

and, provided further, that (i) no amendment, waiver or consent shall, unless in writing and signed by the Issuer in addition to the Lenders required above, modify or waive any provision of Section 2.01(a) (including defined terms used therein) or otherwise affect the rights or duties of the Issuer under this Agreement or any Issuer Document relating to any Credit issued or to be issued by it; (ii) no amendment, waiver or consent shall, unless in writing and signed by the

Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Credit Document; and (iii) the Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed only by the respective parties thereto.

10.02 Notices and Other Communications; Facsimile Copies.

- (a) General. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission). All such written notices shall be mailed, faxed or delivered to the applicable address, facsimile number or (subject to subsection (c) below) electronic mail address, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:
- (i) if to the Company, Holdings, the Administrative Agent or the Issuer, to the address, facsimile number, electronic mail address or telephone number specified for such Person on Schedule 10.02 or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the other parties; and
- (ii) if to any other Lender, to the address, facsimile number, electronic mail address or telephone number specified in its Administrative Questionnaire or to such other address, facsimile number, electronic mail address or telephone number as shall be designated by such party in a notice to the Company, the Administrative Agent and the Issuer.

All such notices and other communications shall be deemed to be given or made upon the earlier to occur of (i) actual receipt by the relevant party hereto and (ii) (A) if delivered by hand or by courier, when signed for by or on behalf of the relevant party hereto; (B) if delivered by mail, four Business Days after deposit in the mails, postage prepaid; (C) if delivered by facsimile, when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of subsection (c) below), when delivered; provided, however, that notices and other communications to the Administrative Agent and the Issuer pursuant to Article II shall not be effective until actually received by any such Person. In no event shall a voicemail message be effective as a notice, communication or confirmation hereunder.

- (b) Effectiveness of Facsimile Documents and Signatures. Credit Documents may be transmitted and/or signed by facsimile. The effectiveness of any such documents and signatures shall, subject to applicable Law, have the same force and effect as manually-signed originals and shall be binding on all Credit Parties, the Administrative Agent, the Issuer and the Lenders. The Administrative Agent may also require that any such documents and signatures be confirmed by a manually-signed original thereof; provided, however, that the failure to request or deliver the same shall not limit the effectiveness of any facsimile document or signature.
- (c) Limited Use of Electronic Mail. Electronic mail and Internet and intranet websites may be used only to distribute routine communications, other than financial statements and other similar information, and to distribute Credit Documents for execution by the parties thereto, and may not be used for any other purpose.

(d) Reliance by Administrative Agent, the Issuer and Lenders. The Administrative Agent, the Issuer and the Lenders shall be entitled to rely and act upon any notices purportedly given by or on behalf of the Company even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Company shall indemnify each Agent-Related Person, each Issuer-Related Person and each Lender from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Company. All telephonic notices to and other communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

10.03 No Waiver; Cumulative Remedies. No failure by any Lender, the Issuer or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein or therein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

10.04 Attorney Costs, Expenses and Taxes. The Company agrees (a) (i) to pay or reimburse each of the Administrative Agent, the Issuer and special counsel for the Lenders for all reasonable costs and expenses incurred in connection with the development, preparation, negotiation and execution of this Agreement and the other Credit Documents (whether or not the transactions contemplated hereby are consummated), and the consummation and administration of the transactions contemplated hereby, including all reasonable Attorney Costs, expenses and disbursements and (ii) to pay or reimburse each of the Administrative Agent, the Issuer and each Lender for all reasonable costs and expenses incurred in connection with any amendment, waiver, consent or other modification of the provisions of this Agreement and the other Credit Documents (whether or not the transactions contemplated thereby are consummated), including without limitation any out-of-pocket costs and expenses incurred in connection with the transactions contemplated by Section 10.19 of this Agreement, and the consummation and administration of the transactions contemplated thereby, including all reasonable Attorney Costs, expenses and disbursements, but excluding any such costs, expenses, attorneys' fees, and disbursements of the Issuer and the Administrative Agent in connection with the transactions contemplated in Section 10.19 resulting from a Deposit Account Interest Non-Payment Event; and (b) to pay or reimburse the Administrative Agent, the Issuer and each Lender for all costs and expenses incurred in connection with the enforcement, attempted enforcement, or preservation of any rights or remedies under this Agreement or the other Credit Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Credit Obligations and during any legal proceeding, including any proceeding under any Debtor Relief Law), including all Attorney Costs. The foregoing costs and expenses shall include all search, filing, recording, title insurance and appraisal charges and fees and taxes related thereto, and other out-of-pocket expenses incurred by the Administrative Agent and the cost of independent public accountants and other outside experts retained by the Administrative Agent, the Issuer or any Lender. All amounts due under this Section 10.04 shall be payable within ten Business Days after demand therefor. The agreements in this Section 10.04 shall

survive the termination of the Credit-Linked Deposits and repayment of all Credit Obligations and all other amounts due hereunder.

10.05 Indemnification by the Company. Whether or not the transactions contemplated hereby are consummated, the Company shall indemnify and hold harmless each Agent-Related Person, each Issuer-Related Person, each Lender and their respective Affiliates, directors, officers, employees, counsel, agents and attorneys-in-fact (collectively, the "Indemnitees") from and against any and all liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements (including reasonable Attorney Costs) of any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against any such Indemnitee in any way relating to or arising out of or in connection with (before or after the Closing Date) (a) the execution, delivery, enforcement, performance or administration of any Credit Document or any other agreement, letter or instrument delivered in connection with the transactions contemplated thereby or the consummation of the transactions contemplated thereby, including without limitation the transactions contemplated by Section 10.19 of this Agreement but excluding any such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses and disbursements of the Issuer and the Administrative Agent in connection with the transactions contemplated in Section 10.19 resulting from a Deposit Account Interest Non-Payment Event. (b) any Credit-Linked Deposit, Credit or the use or proposed use of the proceeds therefrom (including any refusal by the Issuer to honor a demand for payment under a Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Credit), or (c) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory (including any investigation of, preparation for, or defense of any pending or threatened claim, investigation, litigation or proceeding) and regardless of whether any Indemnitee is a party thereto (all the foregoing, collectively, the "Indemnified Liabilities"), in all cases, whether or not caused by or arising, in whole or in part, out of the negligence of the Indemnitee; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such liabilities, obligations, losses, damages, penalties, claims, demands, actions, judgments, suits, costs, expenses or disbursements are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. No Indemnitee shall be liable for any indirect, special, punitive or consequential damages relating to this Agreement or any other Credit Document or arising out of its activities in connection herewith or therewith (whether before or after the Closing Date). Notwithstanding the foregoing, the Company shall not be liable for any indirect, special, punitive or consequential damages claimed by an Indemnitee in connection with the transactions contemplated by Section 10.19 of this Agreement (other than such damages claimed by an Indemnitee as a result of such damages being claimed against such Indemnitee by a third party). All amounts due under this Section 10.05 shall be payable within ten Business Days after demand therefor. The agreements in this Section shall survive the resignation of the Administrative Agent, the replacement of any Lender, the termination of the Credit-Linked Deposits and the repayment, satisfaction or discharge of all the other Credit Obligations.

10.06 Payments Set Aside. To the extent that any payment by or on behalf of the Company is made to the Administrative Agent, the Issuer or any Lender, or the Administrative

Agent, the Issuer or any Lender exercises its right of set-off, and such payment or the proceeds of such set-off or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, the Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such set-off had not occurred, and (b) the Issuer and each Lender severally agree to pay to the Administrative Agent upon demand its applicable share of any amount so recovered from or paid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect, in the applicable currency of such recovery or payment.

10.07 Successors and Assigns. (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Company may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Issuer and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of subsection (b) of this Section 10.07, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section 10.07, (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section 10.07 (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section 10.07 and, to the extent expressly contemplated hereby, the Indemnitees) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) Any Lender may at any time assign to another Lender or to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Credit-Linked Deposit and Credit Obligations at the time owing to it); provided that unless such assignment is to another Lender (i) upon such assignment, the aggregate principal amount of the Credit-Linked Deposit and Credit Obligations held by the assignee, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000; (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Credit Obligations and the Credit-Linked Deposit assigned; and (iii) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption. Subject to acceptance and recording thereof by the Administrative Agent pursuant to subsection (c) of this Section 10.07, from and after the effective date specified in each Assignment and Assumption, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the

case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto but shall continue to be entitled to the benefits of Sections 3.01, 3.02, 3.03, 10.04 and 10.05 with respect to facts and circumstances occurring prior to the effective date of such assignment). Upon request, the Company (at its expense) shall execute and deliver a Letter of Credit Linked Note and one or more Term Notes to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a Participation in such rights and obligations in accordance with subsection (d) of this Section 10.07.

- (c) The Administrative Agent, acting solely for this purpose as an agent of the Company, shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Credit-Linked Deposit of, and amounts of the Credit Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive absent manifest error and the Company, the Administrative Agent, the Issuer and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Company, the Issuer and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (d) Any Lender may at any time, without notice to or the consent of the Administrative Agent, sell participations to any Person (other than a natural person or the Company or any of the Company's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Credit-Linked Deposit and/or such Lender's Term Loans and/or Participations in Credit Obligations owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Company, the Administrative Agent, the Issuer and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to Section 10.01 that directly affects such Participant. Subject to subsection (e) of this Section 10.07, the Company agrees that each Participant shall be entitled to the benefits of Sections 3.01, 3.02 and 3.03 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section 10.07. To the extent permitted by Law, each Participant also shall be entitled to the benefits of Section 2.09 as though it were a Lender.
- (e) A Participant shall not be entitled to receive any greater payment under Section 3.01 or 3.05 than the applicable Lender would have been entitled to receive with respect

to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Company's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of Section 3.01 unless the Company is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Company, to comply with Section 10.15 as though it were a Lender.

- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) **Assignment of Credit-Linked Deposit.** The Credit-Linked Deposit funded by any Lender under Section 2.01(c)(ii) shall not be released in connection with any assignment of its Credit-Linked Deposit, but shall instead be purchased by the relevant assignee and continue to be held for application (if not already applied in accordance with the terms of Section 2.01(c)(ii) or 2.03) pursuant to Section 2.01(c)(ii) in respect of such assignee's obligations under the Credit-Linked Deposit assigned to it.

10.08 Confidentiality. For the purposes of this Section 10.08, "Confidential Information" means information delivered to the Administrative Agent, the Issuer or any Lender by or on behalf of the Company or any Subsidiary in connection with the transactions contemplated by or otherwise pursuant to this Agreement that is proprietary in nature and that was clearly marked or labeled or otherwise adequately identified when received by it as being confidential information of the Company or such Subsidiary, provided that such term does not include information that (a) was publicly known or otherwise known to it prior to the time of such disclosure, (b) subsequently becomes publicly known through no act or omission by it or any Person acting on its behalf, (c) otherwise becomes known to it other than through disclosure by the Company or any Subsidiary or (d) constitutes financial statements delivered to it under Section 6.01 that are otherwise publicly available. Each of the Administrative Agent, the Issuer and the Lenders will maintain the confidentiality of such Confidential Information delivered to or received by it in accordance with procedures adopted by it in good faith to protect confidential information of third parties delivered to it, provided that the Administrative Agent, the Issuer or any Lender may deliver or disclose Confidential Information to (i) its directors, officers, employees, agents, attorneys and Affiliates (to the extent such disclosure reasonably relates to the administration of the Credit Documents by the Administrative Agent, the Issuer or any Lender), (ii) its financial advisors and other professional advisors who agree to hold confidential the Confidential Information substantially in accordance with the terms of this Section 10.08, (iii) any other party to this Agreement, (iv) any Eligible Assignee to which any Lender sells or offers to sell its Credit-Linked Deposit and Credit Obligations or any part thereof or any Participation therein (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 10.08), (v) any Person from which it offers to purchase any security of the Company (if such Person has agreed in writing prior to its receipt of such Confidential Information to be bound by the provisions of this Section 10.08), (vi) any federal or state regulatory authority having jurisdiction over it, (vii) the National Association of Insurance Commissioners or any similar organization, or any nationally

recognized rating agency that requires access to information about its investment portfolio, or (viii) any other Person to which such delivery or disclosure may be necessary or appropriate (w) to effect compliance with any law, rule, regulation or order applicable to it, (x) in response to any subpoena or other legal process, (y) in connection with any litigation to which it is a party or (z) if an Event of Default has occurred and is continuing, to the extent it may reasonably determine such delivery and disclosure to be necessary or appropriate in the enforcement or for the protection of the rights and remedies under the Letter of Credit Linked Notes, the Term Notes and this Agreement. Each Eligible Assignee, by its execution of an Assignment and Assumption, will be deemed to have agreed to be bound by and to be entitled to the benefits of this Section 10.08 as though it were a party to this Agreement. Notwithstanding anything herein to the contrary, "Confidential Information" shall not include, and the Issuer, the Administrative Agent and each Lender may disclose without limitation of any kind, any information with respect to the "tax treatment" and "tax structure" (in each case, within the meaning of Treasury Regulation Section 1.6011-4) of the transactions contemplated hereby and all materials of any kind (including opinions or other tax analyses) that are provided to the Issuer, the Administrative Agent or such Lender relating to such tax treatment and tax structure; provided that with respect to any document or similar item that in either case contains information concerning the tax treatment or tax structure of the transaction as well as other information, this sentence shall only apply to such portions of the document or similar item that relate to the tax treatment or tax structure of the Credits, Advances, Term Loans and transactions contemplated hereby.

10.09 [Intentionally Omitted].

- **10.10 Representations and Warranties of Each Lender**. Each Lender (for itself, and not for any other Lender) represents and warrants to the Issuer and the Company:
- (a) Purchase for Investment. Such Lender is an institutional "accredited investor" within the meaning of subparagraphs (1), (2), (3) or (7) of Rule 501(a) promulgated under the Securities Act. Such Lender represents that it is purchasing its Participation in this Agreement, its Letter of Credit Linked Note and its Term Notes, if any, for its own account or for one or more separate accounts maintained by it or for the account of one or more pension or trust funds and not with a view to the distribution thereof, provided that the disposition of its or their property shall at all times be within its or their control. Such Lender understands that the Letter of Credit Linked Notes and Term Notes have not been registered under the Securities Act and may be resold only if registered pursuant to the provisions of the Securities Act or if an exemption from registration is available, except under circumstances where neither such registration nor such an exemption is required by law, and that the Issuer, the Administrative Agent and the Company are not required to register the Letter of Credit Linked Notes and Term Notes.
- (b) Source of Funds. Such Lender represents that at least one of the following statements is an accurate representation as to each source of funds (a "Source") to be used by it to pay the purchase price of its Participation in this Agreement:
- (i) the Source is an "insurance company general account" (as the term is defined in the United States Department of Labor's Prohibited Transaction Exemption ("PTE") 95-60) in respect of which the reserves and liabilities (as defined by the annual statement for life

insurance companies approved by the National Association of Insurance Commissioners (the "NAIC Annual Statement")) for the general account contract(s) held by or on behalf of any employee benefit plan together with the amount of the reserves and liabilities for the general account contract(s) held by or on behalf of any other employee benefit plans maintained by the same employer (or affiliate thereof as defined in PTE 95-60) or by the same employee organization in the general account do not exceed 10% of the total reserves and liabilities of the general account (exclusive of separate account liabilities) plus surplus as set forth in the NAIC Annual Statement filed with such Lender's state of domicile; or

- (ii) the Source is a separate account that is maintained solely in connection with such Lender's fixed contractual obligations under which the amounts payable, or credited, to any employee benefit plan (or its related trust) that has any interest in such separate account (or to any participant or beneficiary of such plan (including any annuitant)) are not affected in any manner by the investment performance of the separate account; or
- (iii) the Source is either (A) an insurance company pooled separate account, within the meaning of PTE 90-1 or (B) a bank collective investment fund, within the meaning of the PTE 91-38 and, except as disclosed by such Lender to the Issuer and the Company in writing pursuant to this clause (iii), no employee benefit plan or group of plans maintained by the same employer or employee organization beneficially owns more than 10% of all assets allocated to such pooled separate account or collective investment fund; or
- (iv) the Source constitutes assets of an "investment fund" (within the meaning of Part V of PTE 84-14 (the "QPAM Exemption") managed by a "qualified professional asset manager" or "QPAM" (within the meaning of Part V of the QPAM Exemption), no employee benefit plan's assets that are included in such investment fund, when combined with the assets of all other employee benefit plans established or maintained by the same employer or by an affiliate (within the meaning of Section V(c)(1) of the QPAM Exemption) of such employer or by the same employee organization and managed by such QPAM, exceed 20% of the total client assets managed by such QPAM, the conditions of Section I(c) and (g) of the QPAM Exemption are satisfied, neither the QPAM nor a person controlling or controlled by the QPAM (applying the definition of "control" in Section V(e) of the QPAM Exemption) owns a 5% or more interest in the Issuer or the Company and (i) the identity of such QPAM and (ii) the names of all employee benefit plans whose assets are included in such investment fund have been disclosed to the Issuer and the Company in writing pursuant to this clause (iv); or
- (v) the Source constitutes assets of one or more "plans" (within the meaning of Part IV of PTE 96-23 (the "INHAM Exemption")) managed by an "inhouse asset manager" or "INHAM" (within the meaning of Part IV of the INHAM exemption), the conditions of Section I(a), (g) and (h) of the INHAM Exemption are satisfied, neither the INHAM nor a person controlling or controlled by the INHAM (applying the definition of "control" in Section IV(d)(3) of the INHAM Exemption) owns a 5% or more interest in the Issuer or the Company and (A) the identity of such INHAM and (B) the name(s) of the employee benefit plan(s) whose assets constitute the Source have been disclosed to the Issuer and the Company in writing pursuant to this clause (v); or

(vi) the Source is a governmental plan; or

- (vii) the Source is one or more employee benefit plans, or a separate account or trust fund comprised of one or more employee benefit plans, each of which has been identified to the Issuer and the Company in writing pursuant to this clause (vii); or
 - (viii) the Source does not include assets of any employee benefit plan, other than a plan exempt from the coverage of ERISA.

As used in this Section 10.10(b), the terms "employee benefit plan", "governmental plan", "party in interest" and "separate account" shall have the respective meanings assigned to such terms in Section 3 of ERISA.

- (c) Authorization; Enforceability. The execution, delivery and performance of this Agreement has been duly authorized by all necessary corporate action on the part of such Lender. This Agreement has been duly executed and delivered by such Lender and is the legally valid and binding obligation of such Lender, enforceable against such Lender in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles.
- (d) No Withholding. As of the date hereof and as of the date of any assignment under Section 10.07 (after giving effect thereto), no Lender is subject to United States federal withholding taxes on amounts payable pursuant to the terms of this Agreement. Each Lender has delivered to the Administrative Agent all forms required pursuant to Section 10.15(b) to ensure that the Administrative Agent shall not be required to effect any withholding with respect to any amounts payable pursuant to the terms of this Agreement.
- **10.11** Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- **10.12 Integration**. This Agreement, together with the other Credit Documents, comprises the complete and integrated agreement of the parties on the subject matter hereof and thereof and supersedes all prior agreements, written or oral, on such subject matter. In the event of any conflict between the provisions of this Agreement and those of any other Credit Document, the provisions of this Agreement shall control; provided that the inclusion of supplemental rights or remedies in favor of the Administrative Agent, the Issuer or the Lenders in any other Credit Document shall not be deemed a conflict with this Agreement. Each Credit Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.
- 10.13 Survival of Representations and Warranties. All representations and warranties made hereunder and in any other Credit Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent, the Issuer and each Lender, regardless of any investigation made by the Administrative Agent, the Issuer or any Lender or on their behalf and notwithstanding that the Administrative Agent, the Issuer or any Lender may have had notice or

knowledge of any Default or Event of Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Credit Obligation or any other Credit Obligation shall remain unpaid or unsatisfied or any Credit shall remain outstanding.

- **10.14 Severability**. If any provision of this Agreement or the other Credit Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Credit Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 10.15 Tax Forms. (a) (i) Each Lender and the Issuer that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code (a "Foreign Lender") shall deliver to the Administrative Agent, prior to receipt of any payment subject to withholding under the Code (or upon accepting an assignment of an interest herein), two duly signed completed copies of either IRS Form W-8BEN or any successor thereto (relating to such Foreign Lender and entitling it to an exemption from, or reduction of, withholding tax on all payments to be made to such Foreign Lender by the Company pursuant to this Agreement) or IRS Form W-8ECI or any successor thereto (relating to all payments to be made to such Foreign Lender by the Company pursuant to this Agreement), and, if necessary, such other evidence satisfactory to the Company and the Administrative Agent that such Foreign Lender is entitled to an exemption from, or reduction of, U.S. withholding tax, including any exemption pursuant to Section 881(c) of the Code. Thereafter and from time to time, each such Foreign Lender shall (A) promptly submit to the Administrative Agent such additional duly completed and signed copies of one of such forms (or such successor forms as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then current United States laws and regulations to avoid, or such evidence as is satisfactory to the Company and the Administrative Agent of any available exemption from or reduction of, United States withholding taxes in respect of all payments to be made to such Foreign Lender by the Company pursuant to this Agreement, (B) promptly notify the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (C) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Foreign Lender, and as may be reasonably necessary (including the re-designation of its Lending Office) to avoid any
- (ii) Each Foreign Lender, to the extent it does not act or ceases to act for its own account with respect to any portion of any sums paid or payable to such Foreign Lender under any of the Credit Documents (for example, in the case of a typical participation by such Foreign Lender), shall deliver to the Administrative Agent on the date when such Foreign Lender ceases to act for its own account with respect to any portion of any such sums paid or payable, and at such other times as may be necessary in the determination of the Administrative Agent (in the reasonable exercise of its discretion), (A) two duly signed completed copies of the forms or statements required to be provided by such Foreign Lender as set forth above, to establish the

portion of any such sums paid or payable with respect to which such Foreign Lender acts for its own account that is exempt from, or subject to a reduced rate of, U.S. withholding tax, and (B) two duly signed completed copies of IRS Form W-8IMY (or any successor thereto), together with any information such Foreign Lender chooses to transmit with such form, and any other certificate or statement of exemption required under the Code, to establish that such Foreign Lender is not acting for its own account with respect to a portion of any such sums payable to such Foreign Lender.

- (iii) The Company shall not be required to pay any additional amount to any Foreign Lender under Section 3.01 (A) with respect to any Taxes required to be deducted or withheld on the basis of the information, certificates or statements of exemption such Foreign Lender transmits with an IRS Form W-8IMY pursuant to this Section 10.15(a) or (B) if such Foreign Lender shall have failed to satisfy the foregoing provisions of this Section 10.15(a); provided that if such Foreign Lender shall have satisfied the requirement of this Section 10.15(a) on the date such Foreign Lender became a Lender or ceased to act for its own account with respect to any payment under any of the Credit Documents, nothing in this Section 10.15(a) shall relieve the Company of its obligation to pay any amounts pursuant to Section 3.01 in the event that, as a result of any change in any applicable law, treaty or governmental rule, regulation or order, or any change in the interpretation, administration or application thereof, such Foreign Lender is no longer properly entitled to deliver forms, certificates or other evidence at a subsequent date establishing the fact that such Foreign Lender or other Person for the account of which such Foreign Lender receives any sums payable under any of the Credit Documents is not subject to withholding or is subject to withholding at a reduced rate.
- (iv) The Administrative Agent may, without reduction, withhold any Taxes required to be deducted and withheld from any payment under any of the Credit Documents with respect to which the Company is not required to pay additional amounts under this Section 10.15(a) or Section 3.01.
- (b) Upon the request of the Administrative Agent, the Issuer and each Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Administrative Agent two duly signed completed copies of IRS Form W-9. If the Issuer or such Lender fails to deliver such forms, then the Administrative Agent may withhold from any interest payment to the Issuer or such Lender an amount equivalent to the applicable back-up withholding tax imposed by the Code, without reduction.
- (c) If any Governmental Authority asserts that the Administrative Agent did not properly withhold or backup withhold, as the case may be, any tax or other amount from payments made to or for the account of the Issuer or any Lender, the Issuer or such Lender shall indemnify the Administrative Agent therefor, including all penalties and interest, any taxes imposed by any jurisdiction on the amounts payable to the Administrative Agent under this Section, and costs and expenses (including reasonable Attorney Costs) of the Administrative Agent. The obligation of the Issuer and the Lenders under this Section shall survive the termination of the Credit-Linked Deposits, repayment of all Credit Obligations and all other amounts due hereunder and the resignation or replacement of the Administrative Agent.

10.16 GOVERNING Law. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK APPLICABLE TO AGREEMENTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE; PROVIDED THAT THE ADMINISTRATIVE AGENT, THE ISSUER AND EACH LENDER SHALL RETAIN ALL RIGHTS ARISING UNDER FEDERAL LAW.

ANY LEGAL ACTION OR PROCEEDING WITH RESPECT TO THIS AGREEMENT OR ANY OTHER CREDIT DOCUMENT MAY BE BROUGHT IN THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK OR OF THE UNITED STATES FOR SOUTHERN DISTRICT OF SUCH STATE, AND BY EXECUTION AND DELIVERY OF THIS AGREEMENT, THE COMPANY CONSENTS, FOR ITSELF AND IN RESPECT OF ITS PROPERTY, TO THE NON-EXCLUSIVE JURISDICTION OF THOSE COURTS. THE COMPANY IRREVOCABLY WAIVES ANY OBJECTION, INCLUDING ANY OBJECTION TO THE LAYING OF VENUE OR BASED ON THE GROUNDS OF FORUM NON CONVENIENS, WHICH IT MAY NOW OR HEREAFTER HAVE TO THE BRINGING OF ANY ACTION OR PROCEEDING IN SUCH JURISDICTION IN RESPECT OF ANY CREDIT DOCUMENT OR OTHER DOCUMENT RELATED THERETO.

10.17 Waiver of Right to Trial by Jury. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY CREDIT DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY CREDIT DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER FOUNDED IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

10.18 Time of the Essence. Time is of the essence of the Credit Documents.

10.19 Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event. (a) For purposes of this Section 10.19, (i) the term "Credit Rating Downgrade Event" means that the credit ratings established by S&P and Moody's with respect to the senior, unsecured long-term, non-credit enhanced debt securities ("Long-Term Debt") of Bank of America Corporation have been lowered below "A-" by S&P and below "A3" by Moody's or that the Long-Term Debt of Bank of America Corporation has ceased to be rated by S&P and Moody's for a period of more than ten (10) consecutive Business Days, in each case as evidenced by the public announcement thereof; provided that if the credit ratings established for the Long-Term Debt of Bank of America Corporation by either S&P or Moody's are unavailable for any reason, the rating established by the other rating agency shall be used for purposes of this Section 10.19; and (ii) the term "Deposit Account Interest Non-Payment Event" means that an

interest payment in respect of the Credit-Linked Deposits payable pursuant to Section 2.01(j)(i) shall not have been made by Bank of America and distributed by the Administrative Agent to the Lenders pursuant to Section 2.01(j)(i) within five (5) Business Days of when due. The Administrative Agent will promptly notify each Lender (A) after officers within the agency management group (or similar group performing the functions of the present agency management group) of Bank of America obtain knowledge, if a Credit Rating Downgrade Event has occurred, and (B) if a Deposit Interest Non-Payment Event has occurred.

- (b) In the event that a Credit Rating Downgrade Event or a Deposit Account Interest Non-Payment Event has occurred and is continuing, the Required Lenders may, by notice to the Administrative Agent and the Company (each a "Required Notice"), identify the Credit Rating Downgrade Event or Deposit Account Interest Non-Payment Event which then exists, and request that either of the following actions be taken, and upon such request the Company, the Administrative Agent, the Issuer and the Lenders agree to use reasonable commercial efforts to cause such action to be taken:
- (i) A new deposit account (the "New Account") shall be opened by the Administrative Agent at a bank which has a credit rating of at least "A+" from S&P and "A1" from Moody's (the "New Depository Bank"). The New Depository Bank, the Administrative Agent, the Issuer, the Company and the Lenders shall execute such documents as may be reasonably necessary or appropriate so that the Issuer is granted by the Administrative Agent and the Lenders a first priority perfected security interest in the New Account securing the Participations. Upon execution and delivery of such documents (including customary legal opinions), and evidence reasonably satisfactory to the Issuer of perfection with first priority of such security interest, all funds then held in the Credit-Linked Deposit Account shall be transferred in immediately available funds to the New Depository Bank for credit to the New Account. Thereafter, the New Account shall be deemed to be the "Credit-Linked Deposit Account" under this Agreement and the provisions of this Agreement relating to the Credit-Linked Deposit Account shall apply *mutatis mutandis* to the New Account. Except for the transfer of the Credit-Linked Deposit Account as provided above, the rights of the Administrative Agent, on behalf of the Issuer and the Lenders, in and to the Credit-Linked Deposit Account hereunder shall remain unchanged and in full force and effect.

In the event that the interest payable by the New Depository Bank on the funds on deposit in the New Account is less than the interest set forth in Section 2.01(j)(i) of this Agreement with respect to the Total Credit-Linked Deposit, then the Company shall pay such additional amounts at such times as may be required so that the Lenders receive no less than the interest payable under Section 2.01(j)(i), at the times set forth in Section 2.01(j)(i). If a Deposit Account Interest Non-Payment Event has triggered the procedures set forth in this Section 10.19(b), then Bank of America shall (A) indemnify the Company for all additional amounts the Company pays pursuant to the immediately preceding sentence and (B) reimburse the Company and each Lender for all costs and expenses, including Attorney Costs, incurred in connection with such Deposit Account Interest Non-Payment Event.

The parties to this Agreement shall execute and deliver an amendment to this Agreement and such other documents as may be reasonably required to accomplish the

foregoing. Notwithstanding anything to the contrary in this Section 10.19(b)(i), the documentation, terms and conditions utilized to establish and grant a security interest in the New Account and the other documentation (including legal opinions) and actions contemplated herein (including any amendment of this Agreement) shall be reasonably satisfactory in form and substance to the Company, the Issuer, the Administrative Agent and each Lender.

or

(ii) The Administrative Agent and the Lenders shall cause to be established, and beneficially owned by the Lenders, a trust (the "Trust") which shall in turn establish a securities account (the "Securities Account") at a securities intermediary (the "Securities Intermediary") having a credit rating of at least "A+" from S&P and at least "A1" from Moody's. The Securities Intermediary, the Administrative Agent, the Issuer, the Company and the Lenders shall execute such documents as may be reasonably necessary or appropriate so that the Issuer is granted by the Trust, the Administrative Agent and the Lenders, as appropriate, a first priority perfected security interest in the Securities Account securing the Participations. Upon execution and delivery of such documents (including customary legal opinions), and evidence reasonably satisfactory to the Issuer of perfection with first priority of such security interest, all funds then held in the Credit-Linked Deposit Account shall be transferred in immediately available funds to the Securities Intermediary for credit to the Securities Account. Such funds shall be invested in such LIBOR-based short-term debt investments, having a credit rating of at least "A+" from S&P and at least "A1" from Moody's as may be reasonably agreed upon by the Required Lenders, the Administrative Agent, the Issuer and the Company. Thereafter, the Securities Account shall be deemed to be the "Credit Linked Deposit Account" under this Agreement and the provisions of this Agreement relating to the Credit-Linked Deposit Account shall apply *mutatis mutandis* to the Securities Account. Except for the transfer of funds from the Credit-Linked Deposit Account to the Securities Account as provided above, the rights of the Administrative Agent, on behalf of the Issuer and the Lenders, in and to the Securities Account hereunder shall remain unchanged and in full force and effect.

In the event that the interest payable with respect to the investments in the Securities Account is less than the interest set forth in Section 2.01(j)(i) with respect to the Total Credit-Linked Deposit, then the Company shall pay such additional amounts at such times as may be required so that the Lenders receive no less than the interest payable under Section 2.01(j)(i), at the times set forth in Section 2.01(j)(i). If a Deposit Account Interest Non-Payment Event has triggered the procedures set forth in this Section 10.19(b), then Bank of America shall (A) indemnify the Company for all additional amounts the Company pays pursuant to the immediately preceding sentence and (B) reimburse the Company and each Lender for all costs and expenses, including Attorney Costs, incurred in connection with such Deposit Account Interest Non-Payment Event.

The parties to this Agreement shall execute and deliver an amendment to this Agreement and such other documents as may be reasonably required to accomplish the foregoing. Notwithstanding anything to the contrary in this Section 10.19(b)(ii), the documentation, terms and conditions utilized to establish the Trust and Securities Account and to grant a security interest in and to the Security Account and the other documentation (including

legal opinions) and actions contemplated herein (including any amendment of this Agreement) shall be reasonably satisfactory in form and substance to the Company, the Issuer, the Administrative Agent and each Lender.

(iii) The Company hereby agrees to use commercially reasonable efforts from time to time to inform the Administrative Agent, which shall promptly inform the Lenders upon receipt of such information, of the status of the transactions contemplated in Sections 10.19(b)(i) or (ii), as applicable. In the event that neither of the transactions contemplated in subsections (i) or (ii) above, or transactions substantially similar to the transactions in such sections, are consummated in form and substance reasonably satisfactory to the Issuer, the Administrative Agent, the Company and each Lender within 120 days (or such longer period as the Company and the Required Lenders may mutually agree) after the receipt by the Company of a Required Notice, then the Required Lenders may, by notice to the Administrative Agent, terminate the credit facility established under Section 2.01. The Required Lenders may, within 90 days of issuing a Required Notice, provide the Company written notice of their election to retract such Required Notice, but shall have no right to retract the Required Notice after such 90 day period. Upon delivery of a notice of retraction, the actions required by Section 10.19(b)(i) or (ii) shall no longer be required to be taken in connection with the event described in the Required Notice. After delivery of a retraction notice or in the event the credit facility is not terminated at the end of the 120 day period, the Required Lenders may not deliver another Required Notice under this Section 10.19 unless the credit ratings from S&P or Moody's on the Long-Term Debt of Bank of America are further lowered. Upon such termination, all funds then or at any time thereafter held in the Credit-Linked Deposit Account in excess of the sum of the then Outstanding Amount of all Credit Obligations (other than Advances and Term Loans) shall be withdrawn from the Credit-Linked Deposit Account by the Administrative Agent and distributed by the Administrative Agent to each Lender in accordance with its Pro Rata Share, subject to any notice required pursuant to the Money Market Account Agreement; provided that the Company may, pursuant to Section 2.03(b), request that the Lenders make Term Loans as provided in Section 2.03(b)(i); and with respect to all Term Loans (whether then outstanding or thereafter made), the restriction that Term Loans cannot be prepaid until after the second anniversary of the Closing Date set forth in Section 2.03(b)(vi) and the Prepayment Premium, if any, shall continue to apply.

ARTICLE XI HOLDINGS GUARANTY

11.01 Guaranty. For value received and hereby acknowledged and as an inducement to the Lenders and the Issuer to make the Credit Extensions available to the Company, Holdings hereby unconditionally and irrevocably guarantees (a) the full punctual payment when due, whether at stated maturity, by acceleration or otherwise, of all Credit Obligations of the Company now or hereafter existing whether for principal, interest, fees, expenses or otherwise, and (b) the strict performance and observance by the Company of all agreements, warranties and covenants applicable to the Company in the Credit Documents and (c) the obligations of the Company under the Credit Documents (such Credit Obligations collectively being hereafter referred to as the "Guaranteed Obligations"). The foregoing guaranty is a guaranty of payment

and not of collection. For purposes of this Article XI, the obligations of Holdings under this Article XI are referred to as the "Holdings Guaranty."

- 11.02 Guaranty Absolute. Holdings guarantees that the Guaranteed Obligations will be paid strictly in accordance with the terms hereof, regardless of any law, regulation or order now or hereafter in effect in any jurisdiction affecting any of such terms or the rights of any Lender, the Issuer or the Administrative Agent with respect thereto. The liability of Holdings under the Holdings Guaranty with regard to the Guaranteed Obligations shall be absolute and unconditional irrespective of
- (a) any change in the time, manner or place of payment of, or in any other term of, all or any of its Guaranteed Obligations or any other amendment or waiver of or any consent to departure from this Agreement or any other Credit Document (with regard to such Guaranteed Obligations);
 - (b) any release or amendment or waiver of or consent to departure from any other guaranty for all or any of its Guaranteed Obligations;
 - (c) any change in ownership of the Company;
 - (d) any acceptance of any partial payment(s) from the Company or Holdings; or
- (e) any other circumstance similar or dissimilar to the foregoing which might otherwise constitute a defense available to, or a discharge of, the Company in respect of its Credit Obligations under any Credit Document.

The Holdings Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any Guaranteed Obligation is rescinded or must otherwise be returned by the Lenders, the Issuer or the Administrative Agent upon the insolvency, bankruptcy or reorganization of the Company or otherwise, all as though such payment had not been made.

11.03 Effectiveness; Enforcement. The Holdings Guaranty shall be effective and shall be deemed to be made with respect to each Credit Extension as of the time it is made, issued or extended. No invalidity, irregularity or unenforceability by reason of any bankruptcy or similar law, or any law or order of any government or agency thereof purporting to reduce, amend or otherwise affect any liability of the Company, and no defect in or insufficiency or want of powers of the Company or irregular or improperly recorded exercise thereof, shall impair, affect, be a defense to or claim against such guaranty. The Holding Guaranty is a continuing guaranty and shall (a) survive any termination of this Agreement, and (b) remain in full force and effect until payment in full of, and performance of, all Guaranteed Obligations and all other amounts payable under this Agreement. The Holdings Guaranty is made for the benefit of the Administrative Agent, the Issuer and the Lenders and their successors and assigns, and may be enforced from time to time as often as occasion therefor may arise and without requirement on the part of the Administrative Agent, the Issuer or the Lenders first to exercise any rights against the Company, or to resort to any other source or means of obtaining payment of any of the said obligations or to elect any other remedy.

11.04 Waiver. Except as otherwise specifically provided in any of the Credit Documents, Holdings hereby waives promptness, diligence, protest, notice of protest, all suretyship defenses, notice of acceptance and any other notice with respect to any of its Guaranteed Obligations and the Holdings Guaranty and any requirement that the Lenders, the Issuer or the Administrative Agent protect, secure, perfect any security interest or Lien or any property subject thereto or exhaust any right or take any action against the Company or any other Person. Holdings also irrevocably waives, to the fullest extent permitted by law, all defenses which at any time may be available to it in respect of its Guaranteed Obligations by virtue of any statute of limitations, valuation, stay, moratorium law or other similar law now or hereafter in effect.

11.05 Expenses. Holdings hereby promises to reimburse (a) the Administrative Agent, the Issuer and the Lenders for all reasonable out-of-pocket fees and disbursements (including all reasonable Attorneys Costs), incurred or expended in connection with the preparation, filing or recording, or interpretation of the Holdings Guaranty, the other Credit Documents to which Holdings is a party, or any amendment, modification, approval, consent or waiver hereof or thereof, and (b) the Administrative Agent, the Issuer and the Lenders and their respective affiliates for all reasonable out-of-pocket fees and disbursements (including reasonable Attorney Costs), incurred or expended in connection with the enforcement of its Guaranteed Obligations (whether or not legal proceedings are instituted). Holdings will pay any taxes (including any interest and penalties in respect thereof) other than the Lenders' taxes based on overall income or profits, payable on or with respect to the transactions contemplated by the Holdings Guaranty, Holdings hereby agreeing jointly and severally to indemnify each Lender with respect thereto.

11.06 Concerning Joint and Several Liability of Holdings.

- (a) Holdings hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the Company, with respect to the payment and performance of all of its Guaranteed Obligations (including, without limitation, any Guaranteed Obligations arising under this Article XI), it being the intention of the parties hereto that all such Guaranteed Obligations shall be the joint and several Guaranteed Obligations of Holdings and the Company without preferences or distinction among them.
- (b) If and to the extent that the Company shall fail to make any payment with respect to any of its Credit Obligations as and when due or to perform any of its Guaranteed Obligations in accordance with the terns thereof, then in each such event Holdings will make such payment with respect to, or perform, such Guaranteed Obligation.
- (c) The Guaranteed Obligations of Holdings under the provisions of this Article XI constitute full recourse obligations of Holdings enforceable against Holdings to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of this Agreement or any other circumstance whatsoever.
- (d) Except as otherwise expressly provided in this Agreement, Holdings hereby waives notice of acceptance of its joint and several liability, notice of any Credit Extensions made or issued under this Agreement, notice of any action at any time taken or omitted by the Administrative Agent, the Issuer or the Lenders under or in respect of any of the Guaranteed

Obligations, and, generally, to the extent permitted by applicable law, all demands, notices and other formalities of every kind in connection with this Agreement. Holdings hereby assents to, and waives notice of, any extension or postponement of the time for the payment of any of the Guaranteed Obligations, the acceptance of any payment of any of the Guaranteed Obligations, the acceptance of any partial payment thereon, any waiver, consent or other action or acquiescence by the Administrative Agent, the Issuer or the Lenders at any time or times in respect of any Default or Event of Default by the Company or Holdings in the performance or satisfaction of any term, covenant, condition or provision of this Agreement or any other Credit Document, any and all other indulgences whatsoever by the Administrative Agent, the Issuer or the Lenders in respect of any of the Guaranteed Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Guaranteed Obligations or the addition, substitution or release, in whole or in part, of the Company or Holdings. Without limiting the generality of the foregoing, Holdings assents to any other action or delay in acting or failure to act on the part of the Lenders, the Issuer or the Administrative Agent with respect to the failure by the Company or Holdings to comply with its respective Credit Obligations or Guaranteed Obligations, including, without limitation, any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with applicable laws or regulations thereunder, which might, but for the provisions of this Article XI, afford grounds for terminating, discharging or relieving Holdings, in whole or in part, from any of the Guaranteed Obligations under this Article XI, it being the intention of Holdings that, so long as any of the Guaranteed Obligations hereunder remain unsatisfied, the Guaranteed Obligations of Holdings under this Article XI shall not be discharged except by performance and then only to the extent of such performance. The Guaranteed Obligations of Holdings under this Article XI shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, reconstruction or similar proceeding with respect to the Company or Holdings or the Lenders, the Issuer or the Administrative Agent. The joint and several liability of Holdings hereunder shall continue in full force and effect notwithstanding any absorption, merger, consolidation, amalgamation or any other change whatsoever in the name, membership, constitution or place of formation of the Company or Holdings, the Lenders, the Issuer or the Administrative Agent.

- (e) Holdings shall be liable under this Article XI only for the maximum amount of such liabilities that can be incurred under applicable law without rendering this Article XI voidable under applicable law relating to fraudulent conveyance and fraudulent transfer, and not for any greater amount. Accordingly, if any obligation under any provision under this Article XI shall be declared to be invalid or unenforceable in any respect or to any extent, it is the stated intention and agreement of Holdings, the Administrative Agent, the Issuer and the Lenders that any balance of the obligation created by such provision and all other obligations of Holdings under this Article XI to the Lenders, the Issuer or the Administrative Agent shall remain valid and enforceable, and that all sums not in excess of those permitted under applicable law shall remain fully collectible by the Lenders, the Issuer and the Administrative Agent from the Company or Holdings, as the case may be.
- (f) The provisions of this Article XI are made for the benefit of the Administrative Agent, the Issuer and the Lenders and their successors and assigns, and may be enforced in good faith by them from time to time against Holdings as often as occasion therefor may arise and

without requirement on the part of the Administrative Agent, the Issuer or the Lenders first to marshal any of their claims or to exercise any of their rights against the Company or Holdings or to exhaust any remedies available to them against the Company or Holdings or to resort to any other source or means of obtaining payment of any of the obligations hereunder or to elect any other remedy. The provisions of this Article XI shall remain in effect until all of the Guaranteed Obligations shall have been paid in full or otherwise fully satisfied, any remaining Credit-Linked Deposits in the Credit-Linked Deposit Account are withdrawn and distributed to the Lenders in accordance with their respective Pro Rata Shares, and all outstanding Credit shall have expired, matured or otherwise been terminated. If at any time, any payment, or any part thereof, made in respect of any of the Guaranteed Obligations, is rescinded or must otherwise be restored or returned by the Lenders, the Issuer or the Administrative Agent upon the insolvency, bankruptcy or reorganization of the Company or Holdings, or otherwise, the provisions of this Article XI will forthwith be reinstated in effect, as though such payment had not been made.

11.07 Waiver. Until the final payment and performance in full of all of the Credit Obligations, Holdings shall not exercise and Holdings hereby waives any rights Holdings may have against the Company arising as a result of payment by Holdings hereunder, by way of subrogation, reimbursement, restitution, contribution or otherwise, and will not prove any claim in competition with the Administrative Agent, the Issuer or any Lender in respect of any payment hereunder in any bankruptcy, insolvency or reorganization case or proceedings of any nature; Holdings will not claim any setoff, recoupment or counterclaim against the Company in respect of any liability of the Company to Holdings; and Holdings waives any benefit of and any right to participate in any collateral security which may be held by the Administrative Agent, the Issuer or any Lender.

11.08 Subrogation; Subordination. The payment of any amounts due with respect to any indebtedness of the Company for money borrowed or credit received now or hereafter owed to Holdings is hereby subordinated to the prior payment in full of all of the Credit Obligations. Holdings agrees that, after the occurrence of any default in the payment or performance of any of the Credit Obligations, Holdings will not demand, sue for or otherwise attempt to collect any such indebtedness of the Company to Holdings until all of the Credit Obligations shall have been paid in full. If, notwithstanding the foregoing sentence, Holdings shall collect, enforce or receive any amounts in respect of such indebtedness while any Credit Obligations are still outstanding, such amounts shall be collected, enforced and received by Holdings as trustee for the Lenders, the Issuer and the Administrative Agent and be paid over to the Administrative Agent at Default, for the benefit of the Lenders, the Issuer, and the Administrative Agent on account of the Credit Obligations without affecting in any manner the liability of Holdings under the other provisions hereof.

11.09 Release of Guaranty. The Administrative Agent, the Issuer and the Lenders hereby agree that the foregoing Holdings Guaranty shall remain in full force and effect until the occurrence of either (a) the consolidation or merger of Holdings into the Company or its successors, (b) the consolidation or merger of the Company or its successors into Holdings, or (c) the release of Holdings from all other guarantees of the Company's obligations under the Company's existing credit facilities or other senior indebtedness, at which time the foregoing

Holdings Guaranty shall be concurrently released without any further action on the part of the Lenders, the Administrative Agent or the Issuer.

ARTICLE XII ENTIRE AGREEMENT

12.01 Entire Agreement. THIS AGREEMENT AND THE OTHER CREDIT DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

(remainder of page intentionally left blank)

IN WITNESS WHEREOF, the parties hereto have caused this Seven Year Letter of Credit and Term Loan Agreement to be duly executed as of the date first above written.

WASTE MANAGEMENT, INC.

By: /s/ Ronald H. Jones

Name: Ronald H. Jones

Title: Vice President & Treasurer

WASTE MANAGEMENT HOLDINGS, INC.

(for the limited purposes of Article XI)

By: /s/ Ronald H. Jones

Name: Ronald H. Jones

Title: Vice President & Treasurer

By: /s/ Jay Clement

Name: Jay Clement Title: Assistant Treasurer

BANK OF AMERICA, N.A., as

Administrative Agent and Issuer

By:/s/ Brian D. Corum

Name: Brian D. Corum Title: Managing Director

LENDERS:

JOHN HANCOCK LIFE INSURANCE COMPANY

By: /s/ Kathleen E. McDonough

Name: Kathleen E. McDonough Title: Managing Director

METROPOLITAN LIFE INSURANCE COMPANY

By: /s/ A. Dennis White

Name: A. Dennis White

Title: Director

ALLSTATE LIFE INSURANCE COMPANY

By: /s/ Robert B. Bodett

Name: Robert B. Bodett

By: /s/ Jerry D. Zinkula

Name: Jerry D. Zinkula Authorized Signatories

NATIONWIDE LIFE INSURANCE COMPANY

By: /s/ Joseph P. Young

Name: Joseph P. Young Title: Associate Vice President

MASSACHUSETTS MUTUAL LIFE INSURANCE COMPANY

By: David L. Babson & Company Inc. as

Investment Adviser

By: /s/ Mark A. Ahmed

Name: Mark A. Ahmed Title: Managing Director

C.M. LIFE INSURANCE COMPANY

By: David L. Babson & Company Inc. as

Investment Sub-Adviser

By: /s/ Mark A. Ahmed

Name: Mark A. Ahmed Title: Managing Director

GENERAL ELECTRIC CAPITAL ASSURANCE COMPANY

By: GE Asset Management Inc., its investment adviser

By: /s/ John R. Endres

Name: John R. Endres

Title: Vice President – Private Investments

2003 WASTE MANAGEMENT, INC. DIRECTORS' DEFERRED COMPENSATION PLAN (Established January 30, 2003)

ARTICLE I. PURPOSES OF PLAN AND DEFINITIONS

1.1 Purpose

Waste Management, Inc. (the "Company"), hereby establishes the 2003 Waste Management, Inc. Directors' Deferred Compensation Plan for the purpose of providing nonemployee directors ("Directors") of the Company the opportunity to defer all or a portion of their cash compensation, to promote the long-term growth of the Company by increasing the proprietary interest of Directors in the Company, to attract and retain Directors with outstanding competence and ability, to stimulate the active interest of such persons in the development and financial success of the Company and to further the identity of interests of such Directors with those of the Company's stockholders generally.

1.2 Definitions

- (a) Annual Stock Retainer means the portion of the annual retainer paid to Directors in Stock Units, as determined from time to time by the Committee, the Board or another committee thereof. Any portion of an annual retainer paid in cash shall be considered Cash Compensation and may be deferred only by election under § 3.2(a).
- (b) *Beneficiary* means the person or persons designated by the Participant, as provided in § 4.3, to receive any payments otherwise due the Participant under this Plan in the event of the Participant's death.
- (c) **Board of Directors** or **Board** means the Board of Directors of the Company.
- (d) *Cash Compensation* means all of the cash compensation payable to a Participant, for service on the Board and its committees, including the cash portion of any annual retainer, meeting, and other fees, but excluding any expense reimbursements.
- (e) *Code* means the Internal Revenue Code of 1986, as amended.
- (f) Committee means the Nominating & Governance Committee of the Board as is designated by the Board.
- (g) Common Stock means the Company's common stock, \$.01 par value.
- (h) Company means Waste Management, Inc, a Delaware corporation.

- (i) **Determination Date** means the date on which payment of a Participant's deferred compensation is made, as determined in accordance with § 4.1.
- (j) **Director** or **Eligible Director** means each member of the Board of the Company who is not an employee of the Company or any subsidiary.
- (k) Effective Date means January 30, 2003.
- (1) **Election Date** means the date on which the Director makes an election to defer receipt of all or a portion of Cash Compensation pursuant to the terms of the Plan.
- (m) *Election Effective Date* means a date upon which a Participant's deferred compensation is credited to his or her Stock Unit Account pursuant to § 3.3 of this Plan.
- (n) *Fair Market Value* of a share of Common Stock means, as of a particular date, the lowest reported trading price of a share of Common Stock on the New York Stock Exchange, or if no sales occurred on such day, then on the last preceding day on which there were such sales. *Fair Market Value* of a Stock Unit shall be deemed to be equal to the Fair Market Value of one share of Common Stock.
- (o) *Participant* means an Eligible Director of the Company.
- (p) *Plan* means this 2003 Waste Management, Inc. Directors' Deferred Compensation Plan, as amended from time to time.
- (q) **Stock Unit Account** means the bookkeeping account maintained for each Participant to record certain amounts deferred by the Participant in accordance with Article III hereof.
- (r) Stock Unit means a unit equal to one share of Common Stock (as adjusted pursuant to § 3.5).
- (s) Valuation Date means each day on which a sale or sales of the Common Stock on the New York Stock Exchange is reported or a quotation for the Common Stock is available.

ARTICLE II. ADMINISTRATION OF THE PLAN

2.1 Committee

This Plan shall be administered by the Committee.

2.2 Committee's Powers

Subject to the provisions hereof, the Committee shall have full and exclusive power and authority to administer this Plan and to take all actions that are specifically contemplated hereby

or are necessary or appropriate in connection with the administration hereof. The Committee shall also have full and exclusive power to interpret this Plan and to adopt such rules, regulations, and guidelines for carrying out this Plan as it may deem necessary or proper, all of which powers shall be exercised in the best interests of the Company and in keeping with the objectives of this Plan. The Committee may, in its discretion, determine the eligibility of individuals to participate herein, determine the amount of Cash Compensation a Participant may elect to defer, or waive any restriction or other provision of this Plan. The Committee may correct any defect or supply any omission or reconcile any inconsistency in this Plan in the manner and to the extent the Committee deems necessary or desirable to carry it into effect.

2.3 Committee Determinations Conclusive

Any decision of the Committee in the interpretation and administration of this Plan shall lie within its sole and absolute discretion and shall be final, conclusive, and binding on all parties concerned.

2.4 Committee Liability

No member of the Committee or officer or employee of the Company to whom the Committee has delegated authority in accordance with the provisions of § 2.5 of this Plan shall be liable for anything done or omitted to be done by him or her, by any member of the Committee, or by any officer or employee of the Company in connection with the performance of any duties under this Plan, except for his or her own willful misconduct or as expressly provided by statute.

2.5 Delegation of Authority

The Committee may delegate to the Chief Executive Officer or to other senior officers of the Company its duties under this Plan pursuant to such conditions or limitations as the Committee may establish.

2.6 Shares Available under the Plan

Subject to the provisions of § 3.5(b) hereof, no shares of Common Stock shall be issued under this Plan if it shall result in the aggregate number of shares of Common Stock issued under the Plan to exceed 500,000 shares. The Committee may from time to time adopt and observe such procedures concerning the counting of shares against the Plan maximum as it may deem appropriate. The Board and the appropriate officers of the Company shall from time to time take whatever actions are necessary to file any required documents with governmental authorities, stock exchanges and transaction reporting systems to ensure that shares of Common Stock are available for issuance pursuant to the Plan.

ARTICLE III. ACCOUNTS

3.1 Establishment of Accounts

Each Eligible Director as of the Effective Date shall become a Participant as of the Effective Date. The Company shall set up an appropriate record (hereinafter called the "Stock Unit Account") which will from time to time reflect the name of each Participant, the number of Stock Units credited to such Participant pursuant to § 3.2, and the Fair Market Value of that number of Stock Units credited to the Participant determined as of a relevant Election Effective Date.

3.2 Amount of Deferral

- (a) *Cash Deferral.* A Participant may elect to defer receipt of one hundred (100%) percent or fifty (50%) percent of the Cash Compensation payable to the Participant for serving on the Company's Board of Directors, and any committees thereof, for each calendar year for which such deferral is elected. The amount deferred shall be credited to his or her Stock Unit Account in accordance with § 3.3 below.
- (b) Annual Stock Retainer. The Annual Stock Retainer will be credited to a Director's Stock Unit Account in accordance with § 3.3 below.

3.3 Crediting of Deferred Amounts

- (a) **2003 Elections.** One-half of the portion of the Participant's Annual Stock Retainer for 2003 shall be credited to his or her account on February 21, 2003, and the other one-half of the Annual Stock Retainer for 2003, along with, if a Participant elects to defer any portion of his or her annual Cash Compensation for 2003, the Cash Compensation the Participant has elected to defer pursuant to the initial election described in Section 5.1, shall be credited in Stock Units to a Participant's Stock Unit Account on July 15, 2003.
- (b) *Elections after 2003.* After 2003, one-half of the Annual Stock Retainer for the calendar year along with one-half of the Cash Compensation that a Participant elects to defer in such year, shall be credited in Stock Units to a Participant's Stock Unit Account on the fifteenth (15th) day of January and the fifteenth (15th) day of July of such calendar year (the "Election Effective Date") based upon the Fair Market Value of a share of Common Stock on such Election Effective Date.

3.4 Dividends

As of each date that dividends are paid with respect to Common Stock, a Participant who has any outstanding Stock Units credited to his or her Stock Unit Account shall have an additional amount of Stock Units credited to his or her Stock Unit Account equal to the number of Stock Units having a Fair Market Value on the payment date of such dividend equal to the dollar amount of dividends paid per share of Common Stock multiplied by the number of Stock

Units credited to the Participant's Stock Unit Account immediately before payment of such dividend.

3.5 Adjustments

- (a) Exercise of Corporate Powers. The existence of this Plan and any outstanding Stock Units credited hereunder shall not affect in any manner the right or power of the Company or its stockholders to make or authorize any or all adjustments, recapitalizations, reorganizations or other changes in the capital stock of the Company or its business or any merger or consolidation of the Company, or any issue of bonds, debentures, preferred or prior preference stock (whether or not such issue is prior to, on a parity with or junior to the Common Stock) or the dissolution or liquidation of the Company, or any sale or transfer of all or any part of its assets or business, or any other corporate act or proceeding of any kind, whether or not of a character similar to that of the acts or proceedings enumerated above.
- (b) Recapitalizations, Reorganizations and Other Activities. In the event of any subdivision or consolidation of outstanding shares of Common Stock, declaration of a dividend payable in shares of Common Stock or other stock split, then (i) the number of Stock Units and shares of Common Stock available for grant under the Plan and (ii) the appropriate Fair Market Value and other price determinations for such Stock Units shall each be proportionately adjusted by the Board to reflect such transaction. In the event of any other recapitalization or capital reorganization of the Company, any consolidation or merger of the Company with another corporation or entity, the adoption by the Company of any plan of exchange affecting the Common Stock or any distribution to holders of Common Stock of securities or property (other than normal cash dividends or dividends payable in Common Stock), the Board shall make appropriate adjustments to (i) the number of Stock Units and shares of Common Stock available for grant under the Plan and (ii) the appropriate Fair Market Value and other price determinations for such Stock Units to give effect to such transaction; provided that such adjustments shall only be such as are necessary to preserve, without increasing, the value of such Stock Units. In the event of a corporate merger, consolidation, acquisition of property or stock, separation, reorganization or liquidation, the Board shall be authorized to issue or assume units by means of substitution of new units, as appropriate, for previously issued units or an assumption of previously issued units as part of such adjustment.

ARTICLE IV. PAYMENTS

4.1 Period of Deferral

A Participant may elect that payment of the Stock Units credited to the Participant under the Plan be made on any date permitted by the Committee, beginning (i), in the case of Stock Units attributable to Cash Compensation deferred under the Plan, no earlier than the date of

termination of Board service, or (ii), in the case of Stock Units attributable to an Annual Stock Retainer, no earlier than one year following the Participant's date of termination of Board service, in either case as specified on the deferral election form (each such date is hereinafter referred to as the "Determination Date" for the relevant portion of the Participant's Stock Unit Account). If no effective election is made by a Participant, all Stock Units credited to the Participant shall be distributed on the date one year following the Participant's date of termination of Board service, and such date shall be his or her Determination Date. In the event of the death of a Director prior to the date specified on the deferral election form, such Director's date of death shall be his or her Determination Date.

4.2 Form of Payment

Distribution to a Participant of the number of shares of Common Stock represented by the Stock Units credited to the Participant's Stock Unit Account attributable to deferred Cash Compensation or an Annual Stock Retainer, as applicable, shall be made in Common Stock as soon as administratively feasible (but not later than sixty (60) days) following the relevant Determination Date, except in the case of the death of the Participant. In lieu of issuing fractional shares, a cash payment shall be made for any fractional share by multiplying such fraction by the closing sale price for a share of Common Stock on the Determination Date (or if such date is not a day on which exchange trading takes place in Common Stock, the immediately preceding day on which such trading occurred).

4.3 Death Prior to Payment

In the event that a Participant dies prior to payment pursuant to the Plan, any such unpaid amounts, shall be paid to the Participant's designated Beneficiary in Common Stock within sixty (60) days following the Company's notification of the Participant's death. If no Beneficiary has been designated, such payment shall be made to the Participant's estate. A beneficiary designation, or revocation of a prior beneficiary designation, shall be effective only if it is made in writing on a form provided by the Company, signed by the Participant and received by the Committee.

4.4 Payments to Minors and Incompetents

Should the Participant become incompetent or should the Participant designate a Beneficiary who is a minor or incompetent, the Company shall be authorized to make such payment in Common Stock to a parent or guardian of such minor or incompetent, or directly to such minor or incompetent, whichever manner the Committee shall determine in its sole discretion.

ARTICLE V. ELECTING DEFERRALS

5.1 Manner of Electing Deferral

Each election made by a Participant to defer Cash Compensation under the Plan (i) shall take the form of a written document (provided by the Company) signed by the Participant and filed with the Committee, (ii) shall designate the calendar year for which the deferral is made and

the period of deferral, and (iii) cannot be revoked or modified if the proposed revocation or modification applies to amounts deferred with respect to a calendar year which has already commenced at the time such revocation or modification is proposed to be effected; provided, however, that Eligible Directors as of the Effective Date may elect within thirty (30) days after approval of this Plan by the Board of Directors to defer fifty (50%) percent or one hundred (100%) percent of the remainder of the Cash Compensation payable for calendar year 2003.

5.2 Election by a New Director

An election to defer Cash Compensation under the Plan may be made by a new Director of the Company within thirty (30) days after election to the Company's Board of Directors and shall apply to Cash Compensation payable after the date of such election.

ARTICLE VI. MISCELLANEOUS

6.1 Unfunded Plan

Nothing contained herein shall be deemed to create a trust of any kind or create any fiduciary relationship. This Plan shall be unfunded. Any funds invested hereunder shall continue for all purposes to be part of the general funds of the Company. To the extent that a Participant acquires a right to receive payments from the Company under the Plan, such right shall not be greater than the right of any unsecured general creditor of the Company and such right shall be an unsecured claim against the general assets of the Company. Although bookkeeping accounts may be established with respect to Participants who are entitled to rights under this Plan, any such accounts shall be used merely as a bookkeeping convenience. The Company shall not be required to segregate any assets that may at any time be represented by cash, stock or rights thereto, nor shall this Plan be construed as providing for such segregation, nor shall the Company, the Board or the Committee be deemed to be a trustee of any cash, stock or rights thereto granted or to be granted under this Plan. Any liability or obligation of the Company to any Participant with respect to cash, stock or rights thereto under this Plan shall be based solely upon any contractual obligations that may be created by this Plan, and no such liability or obligation of the Company shall be deemed to be secured by any pledge or other encumbrance on any property of the Company. Neither the Company nor the Board nor the Committee shall be required to give any security or bond for the performance of any obligation that may be created by this Plan.

6.2 Title to Funds Remains with Company

Amounts or shares credited to each Participant's Stock Unit Account shall not be specifically set aside or otherwise segregated, but will be combined with corporate assets. Title to such funds or shares will remain with the Company and the Company's only obligation will be to make timely payments to Participants in accordance with the Plan.

6.3 Statement of Account

A statement will be furnished to each Participant not less often than annually and shall reflect the balance of the Participant's Stock Unit Account as of the preceding December 31.

6.4 Assignability

Except as provided in § 4.3, no right to receive payment hereunder shall be transferable or assignable by a Participant except by will or the laws of descent and distribution or pursuant to a qualified domestic relations order as defined by the Code or Title I of the Employee Retirement Income Security Act of 1974, as amended, or the rules thereunder. Any attempted assignment of any benefit under this Plan in violation of this § 6.4 shall be null and void.

6.5 Amendment, Modification, Suspension, or Termination

The Board may amend, modify, suspend, or terminate this Plan for the purpose of meeting or addressing any changes in legal requirements or for any other purpose permitted by law, except that no amendment, modification, or termination shall, without the consent of the Participant, impair the rights of any Participant to the balance in such Participant's Stock Unit Account as of the date of such amendment, modification, or termination. The Board may at any time and from time to time delegate to the Committee any or all of its authority under this § 6.5. Notwithstanding the foregoing, if an amendment would constitute a material revision of the Plan, then to the extent required by applicable law, or deemed necessary by the Board or Committee, such amendment shall be subject to stockholder approval. A "material revision" includes, but is not limited to, a revision that: (i) materially increases the benefits accruing to Participants under this Plan, (ii) materially increases the aggregate number of securities that may be issued under this Plan, (iii) materially modifies the requirements as to eligibility for participation in this Plan, (iv) changes the types of awards available under the Plan, or (v) materially extends the term of the Plan.

6.6 Governing Law

This Plan and all determinations made and actions taken pursuant hereto, to the extent not otherwise governed by mandatory provisions of the Code or the securities laws of the United States, shall be governed by and construed in accordance with the laws of the State of Texas.

6.7 Effective Date

The Plan shall be effective as of the Effective Date.

WASTE MANAGEMENT, INC.

COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES

(In Millions, Except Ratios) (Unaudited)

	Six Months Ended June 30,	
	2003	2002
Income before income taxes, cumulative effect of changes in accounting		
principles and minority interests	\$460	\$574
Fixed charges deducted from income:		
Interest expense	219	234
Implicit interest in rents	34	29
	253	263
Earnings available for fixed charges	\$713	\$837
Interest expense	\$219	\$234
Capitalized interest	10	9
Implicit interest in rents	34	29
Total fixed charges	\$263	\$272
	_	
Ratio of earnings to fixed charges	2.7	3.1

CERTIFICATIONS

- I, A. Maurice Myers, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Waste Management, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a 15(e) and 15d 15(e)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - b. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

By: /s/ A. MAURICE MYERS

A. Maurice Myers

President and Chief Executive Officer

Date: August 5, 2003

- I, David P. Steiner, certify that:
 - 1. I have reviewed this quarterly report on Form 10-Q of Waste Management, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a 15(e) and 15d 15(e)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
 - b. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

By: /s/ DAVID P. STEINER

David P. Steiner

Executive Vice President and

Chief Financial Officer

Date: August 5, 2003

CERTIFICATION PURSUANT TO

18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Waste Management, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2003 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, A. Maurice Myers, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

A. Maurice Myers

President and Chief Executive Officer

August 5, 2003

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Waste Management, Inc. and will be retained by Waste Management, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO

18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Waste Management, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2003 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, David P. Steiner, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

David P. Steiner

Executive Vice President and Chief Financial Officer

August 5, 2003

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Waste Management, Inc. and will be retained by Waste Management, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.